6

7

8

9

10

11

12

# ASSEMBLY AMENDMENT 1, TO ASSEMBLY SUBSTITUTE AMENDMENT 1, TO 2007 SENATE BILL 40

July 10, 2007 - Offered by Representative HUEBSCH.

- 1 At the locations indicated, amend the substitute amendment as follows:
- 2 **1.** Page 1, line 3: delete lines 3 to 11.
  - **2.** Page 2, line 1: delete lines 1 to 7.
- **3.** Page 2, line 7: after that line insert:
- 5 "Section 1e. 5.37 (1) of the statutes is amended to read:

5.37 (1) Voting machines shall give every elector a reasonable opportunity to vote for any person for any office and on any proposition the elector is entitled to vote on, assure privacy to the elector so no one will know how the elector is voting or has voted, preclude the electors from voting for persons or propositions upon which they are not entitled to vote and from voting more than once for the same office or on the same proposition. Voting machines shall be constructed to lock so they cannot be manipulated, tampered with, or show the number of votes registered for any

candidate or proposition while voting is in progress. The machines shall provide a method for electors to vote a straight party ticket, shall permit voting a split ticket and shall record each vote cast.

**SECTION 1f.** 5.64 (1) (ar) 1. a. of the statutes is repealed.

**Section 1g.** 5.64 (1) (ar) 1m. of the statutes is created to read:

5.64 (1) (ar) 1m. When voting for president and vice president, the ballot shall permit an elector to vote only for the candidates on one ticket jointly or to write in the names of persons in both spaces.

**Section 1h.** 5.64 (1) (b) of the statutes is amended to read:

5.64 (1) (b) The names of the candidates for the offices of president and vice president that are certified under s. 8.16 (7) or that are contained in nomination papers filed under s. 8.20 shall appear on the ballot in the form prescribed in s. 7.08 (2) (a). The names of the candidates on the regular party tickets nominated at the primary or replacements appointed under s. 8.35 (2) shall appear in a separate column under the party designation. The columns shall be arranged from left to right according to rank, based on the number of votes received by each party's candidate for president or governor at the last general election beginning with the party that received the most votes. To the right of the columns for parties qualifying under s. 5.62 (1) (b) shall be placed the columns for parties qualifying under s. 5.62 (2) in the same order in which the parties filed petitions with the board. Any column required under par. (e) 2. shall be placed next in order. To the right of the party columns shall be a column for the names of independent candidates for each office, or more than one column if the first column does not provide sufficient space for the names of all such candidates.

**Section 1i.** 5.91 (2) of the statutes is repealed.".

**4.** Page 2, line 7: after that line insert:

**"Section 1i.** 5.66 (1) of the statutes is amended to read:

5.66 (1) For Except as provided in sub. (1m), for local elections, where necessary, municipal clerks shall have sufficient ballots printed or otherwise prepared whenever a voting system does not utilize printed ballots to assure a ballot for all electors or voting machines. For all other elections the municipal clerks shall certify to their county clerk, on the first day of the 2nd month preceding the month in which the primary is held, the approximate number of electors in the municipality. The county clerk shall total these estimates and order a sufficient supply to assure ballots for all electors and voting machines.

**SECTION 1j.** 5.66 (1m) of the statutes is created to read:

5.66 (1m) No 1st class city that prints ballots for distribution to electors may print ballots for any election prior to election day in a number greater than 200 percent of the votes cast in the city at the most recent election that corresponds to the election for which ballots are to be printed. For purposes of this subsection, the most recent spring election corresponds to a current spring election; the most recent presidential election corresponds to a current general election at which the president is to be elected; the most recent gubernatorial election corresponds to a current general election at which the governor is to be elected; and the most recent regular election for an office to be filled at a special election corresponds to a special election called to fill the same office."

**5.** Page 2, line 7: after that line insert:

**"Section 1d.** 5.35 (6) (a) 4a. of the statutes is amended to read:

 $\mathbf{2}$ 

5.35 **(6)** (a) 4a. Instructions prescribed by the board for electors for whom identification is required under s. 6.345 or for whom proof of residence under s. 6.34 is required under s. 6.55 (2).

**SECTION 1k.** 6.29 (1) of the statutes is amended to read:

6.29 (1) No names may be added to a registration list for any election after the close of registration, except as authorized under this section or s. 6.55 (2) or 6.86 (3) (a) 2. Any person whose name is not on the registration list but who is otherwise a qualified elector is entitled to vote at the election upon compliance with this section, if the person complies with all other requirements for voting at the polling place.

**Section 1kb.** 6.33 (1) of the statutes is amended to read:

6.33 (1) The board shall prescribe the format, size, and shape of registration forms. All forms shall be printed on cards and each item of information shall be of uniform font size, as prescribed by the board. The municipal clerk shall supply sufficient form forms to meet voter registration needs. The forms shall be designed to obtain from each applicant information as to name; date; residence location; citizenship; date of birth; age; the number of a valid operator's license issued to the elector under ch. 343 or the last 4 digits of the elector's social security account number; whether the applicant has resided within the ward or election district for at least 10 days; whether the applicant elects to be subject to an identification requirement under s. 6.345 (1); whether the applicant has been convicted of a felony for which he or she has not been pardoned, and if so, whether the applicant is incarcerated, or on parole, probation, or extended supervision; whether the applicant is disqualified on any other ground from voting; and whether the applicant is currently registered to vote at any other location. The form shall include a space for the applicant's signature and the signature of any corroborating elector. The form

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

shall include a space to enter the name of any special registration deputy under s. 6.26 or 6.55 (6) or inspector, municipal clerk, or deputy clerk under s. 6.55 (2) who obtains the form and a space for the deputy, inspector, clerk, or deputy clerk to sign his or her name, affirming that the deputy, inspector, clerk, or deputy clerk has accepted the form. The form shall include a space for entry of the ward and aldermanic district, if any, where the elector resides and any other information required to determine the offices and referenda for which the elector is certified to vote. The form shall also include a space where the clerk may record an indication of whether the form is received by mail, a space where the clerk may record an indication of the type of identifying document submitted by the elector as proof of residence under s. 6.34, whenever required, and a space where the clerk, for any applicant who possesses a valid voting identification card issued to the person under s. 6.47 (3), may record the identification serial number appearing on the voting identification card. Each county clerk shall obtain sufficient registration forms for completion by an elector who desires to register to vote at the office of the county clerk under s. 6.28 (4).

**Section 1kc.** 6.345 of the statutes is created to read:

**6.345 Optional identification requirement.** (1) Any elector may, at the time of registration, elect to be required to present identification whenever the elector votes at an election.

(2) Any elector who is registered to vote may, by appearing in person at the office of the municipal clerk or board of election commissioners of the municipality where the elector resides, elect to be required to present identification whenever the elector votes at an election. The municipal clerk or board of election commissioners shall require an elector who makes a request under this subsection to present

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

19

20

21

22

23

- identification and sign a request on a form prescribed by the government accountability board.
- (3) Except as provided in ss. 6.79 (7) and 6.87 (4) (b), if an elector has elected to be required to present identification under this section, the elector may not vote in an election unless the elector presents identification.
- (4) An elector who files an election under sub. (1) or (2) may, by appearing personally at the office of the municipal clerk or board of election commissioners of the municipality where the elector resides, file a written revocation of the election on a form prescribed by the government accountability board.
- (5) An election or revocation under this section becomes effective on the 31st day before the next election occurring in the municipality after the election is filed, or if filed later than the 31st day before the next election, on the 31st day before the 2nd succeeding election occurring in the municipality after the election is filed.
  - (6) In this section, "identification" means any of the following:
  - (a) A valid operator's license issued to the elector under ch. 343.
  - (b) A valid identification card issued to the elector under s. 343.50.
- 17 (c) A valid, current identification card issued to the elector by a U.S. uniformed service.
  - **Section 1kd.** 6.36 (2) (d) of the statutes is created to read:
  - 6.36 (2) (d) The list shall contain, next to the name of each elector, an indication of whether the elector has elected to be required to provide identification under s. 6.345.".
    - **6.** Page 2, line 7: after that line insert:
  - "Section 1p. 6.79 (2) (a) and (d) of the statutes are amended to read:

6.79 (2) (a) Unless information on the poll list is entered electronically, the				
municipal clerk shall supply the inspectors with 2 copies of the most current official				
registration list or lists prepared under s. 6.36 (2) (a) for use as poll lists at the polling				
place. Except as provided in sub. subs. (6) and (7), each person elector, before				
receiving a serial number, shall state his or her full name and address and, if the poll				
list indicates that the elector has elected to be subject to an identification				
requirement under s. 6.345, present to the officials a valid operator's license issued				
to the elector under ch. 343, a valid, current identification card issued to the elector				
by a U.S. uniformed service, or a valid identification card issued to the elector under				
s. 343.50. The officials shall verify that the name and address provided stated by the				
person elector are the same as the person's elector's name and address on the poll list.				
(d) If the poll list, indicates that proof of residence under s. 6.34 is required and				
any document provided by the elector under par. (a) does not constitute proof of				
residence under s. 6.34, the officials shall require the elector to provide proof of				
residence. If proof of residence is provided, the officials shall verify that the name				
and address on the document submitted as proof of residence provided is the same				
as the name and address shown on the registration list. If proof of residence is				
required and not provided, or if the elector does not present a license or identification				
card under par. (a), whenever required, the officials shall offer the opportunity for				
the elector to vote under s. 6.97.				
<b>SECTION 1pb.</b> 6.79 (3) (title) of the statutes is amended to read:				

**Section 1pc.** 6.79 (3) of the statutes is renumbered 6.79 (3) (a).

**Section 1pd.** 6.79(3)(b) of the statutes is created to read:

LICENSE, OR IDENTIFICATION CARD.

6.79 (3) (b) If the poll list indicates that the elector has elected to be subject to an identification requirement under s. 6.345 and identification under sub. (2) is not provided by the elector, or if the name or any photograph appearing on the document that is provided cannot be verified by the officials, the elector shall not be permitted to vote, except as authorized under sub. (7), but if the elector is entitled to cast a provisional ballot under s. 6.97, the officials shall offer the opportunity for the elector to vote under s. 6.97.

**Section 1pe.** 6.79 (7) of the statutes is created to read:

6.79 (7) LICENSE SURRENDER. If an elector is required to provide identification under s. 6.345 and the elector receives a citation or notice of intent to revoke or suspend an operator's license from a law enforcement officer in any jurisdiction that is dated within 60 days of the date of an election and the elector is required to surrender his or her operator's license issued to the elector under ch. 343 at the time the citation or notice is issued, the elector may present an original copy of the citation or notice in lieu of an operator's license under ch. 343. In such case, the elector shall cast his or her ballot under s. 6.965.

**Section 1pf.** 6.82 (1) (a) of the statutes is amended to read:

6.82 (1) (a) When any inspectors are informed that an elector is at the entrance to the polling place who as a result of disability is unable to enter the polling place, they shall permit the elector to be assisted in marking a ballot by any individual selected by the elector, except the elector's employer or an agent of that employer or an officer or agent of a labor organization which represents the elector. The If the elector is required to provide identification under s. 6.345, except as authorized in s. 6.79 (7), the individual selected by the elector shall present to the inspectors a valid operator's license issued to the elector under ch. 343, a valid, current identification

 $\mathbf{2}$ 

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

card issued to the elector by a U.S. uniformed service, or a valid identification card issued to the elector under s. 343.50 and, if the license or identification card does not constitute proof of residence under s. 6.34, shall also provide proof of residence under s. 6.34 for the assisted elector, whenever required, and all other information necessary for the elector to obtain a ballot under s. 6.79 (2). The inspectors shall issue a ballot to the individual selected by the elector and shall accompany the individual to the polling place entrance where the assistance is to be given. If the ballot is a paper ballot, the assisting individual shall fold the ballot after the ballot is marked by the assisting individual. The assisting individual shall then immediately take the ballot into the polling place and give the ballot to an inspector. The inspector shall distinctly announce that he or she has "a ballot offered by .... (stating person's name), an elector who, as a result of disability, is unable to enter the polling place without assistance". The inspector shall then ask, "Does anyone object to the reception of this ballot?" If no objection is made, the inspectors shall record the elector's name under s. 6.79 and deposit the ballot in the ballot box, and shall make a notation on the poll list: "Ballot received at poll entrance".".

### **7.** Page 2, line 7: after that line insert:

""Section 1pg.  $6.86\ (1)\ (ac)$  of the statutes is amended to read:

6.86 (1) (ac) Any elector qualifying under par. (a) may make written application to the municipal clerk for an official ballot by means of facsimile transmission or electronic mail. Any application under this paragraph shall contain a copy of the applicant's original signature. An elector requesting a ballot under this paragraph shall return with the voted ballot a copy of the request bearing an original signature of the elector as provided in s. 6.87 (4) (a).

**Section 1ph.** 6.86 (1) (ar) of the statutes is amended to read:

6.86 (1) (ar) Except as authorized in s. 6.875 (6), the municipal clerk shall not issue an absentee ballot unless the clerk receives a written application therefor from a qualified elector of the municipality. The clerk shall retain each absentee ballot application until destruction is authorized under s. 7.23 (1). Except as authorized in s. 6.79 (7), if an elector is required to provide identification under s. 6.345 and the elector applies for an absentee ballot in person at the clerk's office, the clerk shall not issue the elector an absentee ballot unless the elector presents a valid operator's license issued to the elector under ch. 343, a valid, current identification card issued to the elector by a U.S. uniformed service, or a valid identification card issued to the elector under s. 343.50. The clerk shall make a copy of the document presented by the elector and shall enclose the copy in the certificate envelope.

**Section 1pi.** 6.86 (3) (a) 1. of the statutes is amended to read:

6.86 (3) (a) 1. Any elector who is registered and who is hospitalized, may apply for and obtain an official ballot by agent. The agent may apply for and obtain a ballot for the hospitalized absent elector by presenting a form prescribed by the board and containing the required information supplied by the hospitalized elector and signed by that elector and any other elector residing in the same municipality as the hospitalized elector, corroborating the information contained therein. The corroborating elector shall state on the form his or her full name and address. Except as authorized in s. 6.87 (4) (b), if the elector is required to provide identification under s. 6.345, the agent shall present the identification required under sub. (1) (ar). The clerk shall make a copy of the document presented by the agent and shall enclose the copy in the certificate envelope.

**SECTION 1pj.** 6.869 of the statutes is amended to read:

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

**6.869 Uniform instructions.** The board shall prescribe uniform instructions for absentee voters. The instructions shall include information concerning the procedure for correcting errors in marking a ballot and obtaining a replacement for a spoiled ballot. The procedure shall, to the extent possible, respect the privacy of each elector and preserve the confidentiality of each elector's vote. For electors who are subject to an identification requirement under s. 6.345, the instructions shall include information regarding the identification that is required under s. 6.86 (1) (ar) or the copy of the identification that is required under s. 6.87 (4).

**Section 1pk.** 6.87 (3) (d) of the statutes is amended to read:

6.87 (3) (d) A municipal clerk may, if the clerk is reliably informed by an absent elector of a facsimile transmission number or electronic mail address where the elector can receive an absentee ballot, transmit a facsimile or electronic copy of the absent elector's ballot to that elector in lieu of mailing under this subsection if, in the judgment of the clerk, the time required to send the ballot through the mail may not be sufficient to enable return of the ballot by the time provided under sub. (6). An elector may receive an absentee ballot under this subsection only if the elector has filed a valid application for the ballot under s. 6.86 (1). If the clerk transmits an absentee ballot under this paragraph, the clerk shall also transmit a facsimile or electronic copy of the text of the material that appears on the certificate envelope prescribed in sub. (2), together with instructions prescribed by the board. The instructions shall require the absent elector to make and subscribe to the certification as required under sub. (4) (a) and to enclose the absentee ballot in a separate envelope contained within a larger envelope, that shall include the completed certificate. The elector shall then affix sufficient postage unless the absentee ballot qualifies for mailing free of postage under federal free postage laws

 $\mathbf{2}$ 

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

and shall mail the absentee ballot to the municipal clerk. Except as authorized in s. 6.97 (2), an absentee ballot received under this paragraph shall not be counted unless it is cast in the manner prescribed in this paragraph and in accordance with the instructions provided by the board.

**SECTION 1pL.** 6.87 (4) of the statutes is renumbered 6.87 (4) (a) and amended to read:

6.87 (4) (a) Except as otherwise provided in s. 6.875, the elector voting absentee shall make and subscribe to the certification before one witness who is an adult U.S. citizen. The absent elector, in the presence of the witness, shall mark the ballot in a manner that will not disclose how the elector's vote is cast. The elector shall then, still in the presence of the witness, fold the ballots so each is separate and so that the elector conceals the markings thereon and deposit them in the proper envelope. If a consolidated ballot under s. 5.655 is used, the elector shall fold the ballot so that the elector conceals the markings thereon and deposit the ballot in the proper envelope. If proof of residence is required Except as authorized in par. (b) and notwithstanding s. 343.43 (1) (f), if the elector is required to provide identification under s. 6.345, the elector shall enclose a copy of the identification in the envelope. If proof of residence under s. 6.34 is required and the document enclosed by the elector under this paragraph does not constitute proof of residence under s. 6.34, the elector shall also enclose proof of residence under s. 6.34 in the envelope. Proof of residence is required if the elector is not a military elector or an overseas elector, as defined in s. 6.34 (1), and the elector registered by mail and has not voted in an election in this state. If the elector requested a ballot by means of facsimile transmission or electronic mail under s. 6.86 (1) (ac), the elector shall enclose in the envelope a copy of the request which bears an original signature of the elector. The

elector may receive assistance under sub. (5). The return envelope shall then be sealed. The witness may not be a candidate. The envelope shall be mailed by the elector, or delivered in person, to the municipal clerk issuing the ballot or ballots. If the envelope is mailed from a location outside the United States, the elector shall affix sufficient postage unless the ballot qualifies for delivery free of postage under federal law. Failure to return an unused ballot in a primary does not invalidate the ballot on which the elector's votes are cast. Return of more than one marked ballot in a primary or return of a ballot prepared under s. 5.655 or a ballot used with an electronic voting system in a primary which is marked for candidates of more than one party invalidates all votes cast by the elector for candidates in the primary.

**Section 1pm.** 6.87 (4) (b) of the statutes is created to read:

6.87 (4) (b) If the absentee elector is required to provide identification under s. 6.345 and the elector has received a citation or notice of intent to revoke or suspend an operator's license from a law enforcement officer in any jurisdiction that is dated within 60 days of the date of the election and the elector is required to surrender his or her operator's license issued to the elector under ch. 343 at the time the citation or notice is issued, the elector may enclose a copy of the citation or notice in lieu of an operator's license under ch. 343 if the elector is voting by mail, or may present an original copy of the citation or notice in lieu of an operator's license under ch. 343 if the elector is voting at the office of the municipal clerk.

**Section 1pn.** 6.875 (6) (c) 1. of the statutes is amended to read:

6.875 (6) (c) 1. Upon their visit to the home or facility under par. (a), the deputies shall personally offer each elector who has filed a proper application for an absentee ballot the opportunity to cast his or her absentee ballot. If an elector is present who has not filed a proper application for an absentee ballot, the 2 deputies

may accept an application from the elector and shall issue a ballot to the elector if the elector is qualified, the elector presents identification, whenever required under s. 6.345, and the application is proper. The deputies shall each witness the certification and may, upon request of the elector, assist the elector in marking the elector's ballot. All voting shall be conducted in the presence of the deputies. Upon request of the elector, a relative of the elector who is present in the room may assist the elector in marking the elector's ballot. No individual other than a deputy may witness the certification and no individual other than a deputy or relative of an elector may render voting assistance to the elector.

**Section 1po.** 6.88 (3) (a) of the statutes is amended to read:

under s. 7.52, at any time between the opening and closing of the polls on election day, the inspectors shall, in the same room where votes are being cast, in such a manner that members of the public can hear and see the procedures, open the carrier envelope only, and announce the name of the absent elector or the identification serial number of the absent elector if the elector has a confidential listing under s. 6.47 (2). When the inspectors find that the certification has been properly executed, the applicant is a qualified elector of the ward or election district, and the applicant has not voted in the election, they shall enter an indication on the poll list next to the applicant's name indicating an absentee ballot is cast by the elector. They shall then open the envelope containing the ballot in a manner so as not to deface or destroy the certification thereon. The inspectors shall take out the ballot without unfolding it or permitting it to be unfolded or examined. Unless the ballot is cast under s. 6.95, the inspectors shall verify that the ballot has been endorsed by the issuing clerk. If the poll list indicates that proof of residence under s. 6.34 is required and no proof

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

of residence is enclosed or the name or address on the document that is provided is not the same as the name and address shown on the poll list, or if the elector is required to provide identification under s. 6.345 and no copy of the identification is enclosed or the name on the document that is provided cannot be verified by the inspectors, the inspectors shall proceed as provided under s. 6.97 (2). The inspectors shall then deposit the ballot into the proper ballot box and enter the absent elector's name or voting number after his or her name on the poll list in the same manner as if the elector had been present and voted in person.

**Section 1pp.** 6.965 of the statutes is created to read:

6.965 Voting procedure for electors presenting citation or notice in **lieu of license.** Whenever any elector is allowed to vote at a polling place under s. 6.79 (7) by presenting a citation or notice of intent to revoke or suspend an operator's license in lieu of an operator's license issued to the elector under ch. 343, the inspectors shall, before giving the elector a ballot, write on the back of the ballot the serial number of the elector corresponding to the number kept at the election on the poll list or other list maintained under s. 6.79 and the notation "s. 6.965." If voting machines are used in the municipality where the elector is voting, the elector's vote may be received only upon an absentee ballot furnished by the municipal clerk which shall have the notation "s. 6.965" written on the back of the ballot by the inspectors before the ballot is given to the elector. If the municipal clerk receives an absentee ballot from an elector who presents a citation or notice, or copy thereof, under s. 6.87 (4) (b), the clerk shall enter a notation on the certificate envelope "Ballot under s. 6.965, stats." Upon receiving the envelope, the inspectors shall open and write on the back of the ballot the serial number of the elector corresponding to the number kept at the election on the poll list or other list maintained under s. 6.79 and the

 $\mathbf{2}$ 

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

notation "s. 6.965." The inspectors shall indicate on the poll list or other list maintained under s. 6.79 the fact that the elector is voting by using a citation or notice in lieu of a license. The inspectors shall then deposit the ballot. The ballot shall then be counted under s. 5.85, or under s. 7.51 or 7.52.

**Section 1pq.** 6.97 (title) of the statutes is amended to read:

## 6.97 (title) Voting procedure for individuals not providing required proof of residence or identification.

**Section 1pr.** 6.97 (1) and (2) of the statutes are amended to read:

6.97 (1) Whenever any individual who is required to provide proof of residence under s. 6.34 in order to be permitted to vote appears to vote at a polling place and cannot provide the required proof of residence, the inspectors shall offer the opportunity for the individual to vote under this section. Whenever any individual who is required to provide identification under s. 6.345 appears to vote at a polling place and does not present identification, the inspectors or the municipal clerk shall similarly offer the opportunity for the individual to vote under this section. If the individual wishes to vote, the inspectors shall provide the elector with an envelope marked "Ballot under s. 6.97, stats." on which the serial number of the elector is entered and shall require the individual to execute on the envelope a written affirmation stating that the individual is a qualified elector of the ward or election district where he or she offers to vote and is eligible to vote in the election. The inspectors shall, before giving the elector a ballot, write on the back of the ballot the serial number of the individual corresponding to the number kept at the election on the poll list or other list maintained under s. 6.79 and the notation "s. 6.97". If voting machines are used in the municipality where the individual is voting, the individual's vote may be received only upon an absentee ballot furnished by the

municipal clerk which shall have the corresponding number from the poll list or other list maintained under s. 6.79 and the notation "s. 6.97" written on the back of the ballot by the inspectors before the ballot is given to the elector. When receiving the individual's ballot, the inspectors shall provide the individual with written voting information prescribed by the board under s. 7.08 (8). The inspectors shall indicate on the list the fact that the individual is required to provide proof of residence or identification under s. 6.345 but did not do so. The inspectors shall notify the individual that he or she may provide proof of residence or identification to the municipal clerk or executive director of the municipal board of election commissioners. The inspectors shall also promptly notify the municipal clerk or executive director of the name, address, and serial number of the individual. The inspectors shall then place the ballot inside the envelope and place the envelope in a separate carrier envelope.

(2) Whenever any individual who votes by absentee ballot is required to provide proof of residence in order to be permitted to vote and does not provide the required proof of residence under s. 6.34, the inspectors shall treat the ballot as a provisional ballot under this section. Whenever any individual who is required to provide identification under s. 6.345, other than an individual who is exempted under s. 6.87 (4), votes by absentee ballot and does not enclose a copy of the identification specified in s. 6.86 (1) (ar), the inspectors shall similarly treat the ballot as a provisional ballot under this section. Upon removing the ballot from the envelope, the inspectors shall write on the back of the absentee ballot the serial number of the individual corresponding to the number kept at the election on the poll list or other list maintained under s. 6.79 and the notation "s. 6.97". The inspectors shall indicate on the list the fact that the individual is required to provide proof of

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

residence or a copy of the identification specified in s. 6.86 (1) (ar) but did not do so. The inspectors shall promptly notify the municipal clerk or executive director of the municipal board of election commissioners of the name, address, and serial number of the individual. The inspectors shall then place the ballot inside an envelope on which the name and serial number of the elector is entered and shall place the envelope in a separate carrier envelope.

**SECTION 1ps.** 6.97 (3) of the statutes is renumbered 6.97 (3) (b) and amended to read:

6.97 (3) (b) Whenever the municipal clerk or executive director of the municipal board of election commissioners is informed by the inspectors that a ballot has been cast under this section, the clerk or executive director shall promptly provide written notice to the board of canvassers of each municipality, special purpose district, and county that is responsible for canvassing the election of the number of ballots cast under this section in each ward or election district. The municipal clerk or executive director then shall determine whether each individual voting under this section is qualified to vote in the ward or election district where the individual's ballot is cast. If the elector is required to provide a license or identification card or copy thereof under s. 6.79 (2) or 6.87 (4) and fails to do so, the elector bears the burden of correcting the omission by providing the license or identification card or copy thereof at the polling place before the closing hour or at the office of the municipal clerk or board of election commissioners no later than 4 p.m. on the day after the election. The municipal clerk or executive director shall make a record of the procedure used to determine the validity of each ballot cast under this section. If, prior to 4 p.m. on the day after the election, the municipal clerk or executive director determines that the individual is qualified to vote in the ward or election district where the individual's

ballot is cast, the municipal clerk or executive director shall notify the board of canvassers for each municipality, special purpose district and county that is responsible for canvassing the election of that fact.

**SECTION 1pt.** 6.97 (3) (a) and (c) of the statutes are created to read:

6.97 (3) (a) Whenever an elector who votes by provisional ballot under sub. (1) or (2) because the elector does not provide a license or identification card or copy thereof under s. 6.79 (2) or 6.87 (4) later appears at the polling place where the ballot is cast before the closing hour and provides the license or identification card, the inspectors shall remove the elector's ballot from the separate carrier envelope, shall note on the poll list that the elector's provisional ballot is withdrawn, and shall deposit the elector's ballot in the ballot box. If the inspectors have notified the municipal clerk or executive director of the board of election commissioners that the elector's ballot was cast under this section, the inspectors shall notify the clerk or executive director that the elector's provisional ballot is withdrawn.

(c) A ballot cast under this section by an elector for whom a valid license or identification card or copy thereof is required under s. 6.79 (2) or 6.87 (4) shall not be counted unless the municipal clerk or executive director of the board of election commissioners provides timely notification that the elector has provided a valid license or identification card or copy thereof under this section.

**Section 1pu.** 6.97 (4) of the statutes is amended to read:

6.97 (4) Whenever a board of canvassers receives timely notification from the municipal clerk or executive director of the board of election commissioners under sub. (3) (b) that an individual who has voted under this section is qualified to vote in the ward or election district where the individual's ballot is cast, the board of canvassers shall promptly reconvene and, if the ballot cast by the individual is

otherwise valid, shall count the ballot and adjust the statements, certifications and determinations accordingly. If the municipal clerk or executive director transmits returns of the election to the county clerk or board of election commissioners, the municipal clerk or executive director shall transmit to the county clerk or board of election commissioners a copy of the amended returns together with all additional ballots counted by each board of canvassers.

**Section 1pw.** 7.08 (1) (c) of the statutes is amended to read:

7.08 (1) (c) Prescribe forms required by ss. 6.24 (3) and (4), 6.30 (4), 6.33 (1), 6.345, 6.40 (1) (a), 6.47 (1) (am) 2. and (3), 6.55 (2), and 6.86 (2) to (3). All such forms shall contain a statement of the penalty applicable to false or fraudulent registration or voting through use of the form. Forms are not required to be furnished by the board.

**SECTION 1px.** 7.08 (8) (title) of the statutes is amended to read:

7.08 (8) (title) Electors voting without proof of residence <u>or identification</u> or pursuant to court order.".

**8.** Page 2, line 7: after that line insert:

"Section 3b. 7.50(2)(a) of the statutes is repealed.

**Section 3c.** 10.02 (3) (b) 1. of the statutes is amended to read:

10.02 (3) (b) 1. If an elector wishes to vote for all candidates nominated by any party, the elector shall make a cross (7) or depress the lever or button next to the party designation shown at the top of the ballot. Unless a name has been erased or crossed out, another name written in, a cross made next to the name of a candidate for the same office in another column or a sticker applied, a cross next to a party designation at the top of the column is a vote for all the party's candidates listed in the column.

13

14

15

16

17

18

19

20

21

22

23

If an elector does not wish to vote for all the candidates nominated by one party, the 1  $\mathbf{2}$ The elector shall make a cross (X) next to or separately depress the levers or buttons 3 next to each candidate's name for whom he or she intends to vote, or shall insert or 4 write in the name of a candidate. 5 **Section 3cm.** 10.02 (3) (c) of the statutes is amended to read: 6 10.02 (3) (c) In presidential elections, unless the elector wishes to vote for all 7 candidates nominated by any party, the elector shall make a cross (x) next to or 8 depress the button or lever next to the set of candidates for president and vice 9 president for whom he or she intends to vote. A vote for candidates for president and 10 vice president is a vote for the presidential electors of those candidates.". **9.** Page 2. line 7: after that line insert: 11

""Section 2m. 7.15 (2) (b) of the statutes is amended to read:

7.15 (2) (b) Cities over 500,000 population may prepare their own official and sample ballots, subject to s. 5.66 (1m). Official ballots not utilized as absentee ballots shall be printed so they are ready at least 2 days before the election.

**Section 2n.** 7.15 (2) (c) of the statutes is amended to read:

7.15 (2) (c) With Subject to s. 5.66 (1m) and with the consent of the county clerk, municipalities may prepare their own ballots whenever voting machines or electronic voting systems are used in elections where candidates for both local offices and national, state or county offices appear on the ballot. This paragraph does not apply to cities under par. (b).".

**10.** Page 2, line 7: after that line insert:

**"Section 3bm.** 7.52 (3) (a) of the statutes is amended to read:

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

7.52 (3) (a) The board of absentee ballot canvassers shall first open the carrier envelope only, and, in such a manner that a member of the public, if he or she desired, could hear, announce the name of the absent elector or the identification serial number of the absent elector if the elector has a confidential listing under s. 6.47 (2). When the board of absentee ballot canvassers finds that the certification has been properly executed and the applicant is a qualified elector of the ward or election district, the board of absentee ballot canvassers shall enter an indication on the poll list next to the applicant's name indicating an absentee ballot is cast by the elector. The board of absentee ballot canvassers shall then open the envelope containing the ballot in a manner so as not to deface or destroy the certification thereon. The board of absentee ballot canvassers shall take out the ballot without unfolding it or permitting it to be unfolded or examined. Unless the ballot is cast under s. 6.95, the board of absentee ballot canvassers shall verify that the ballot has been endorsed by the issuing clerk. If the poll list indicates that proof of residence is required and no proof of residence is enclosed or the name or address on the document that is provided is not the same as the name and address shown on the poll list, or if the elector is required to provide a copy of identification under s. 6.87 (4) and no copy of the identification is enclosed or the name on the document cannot be verified by the canvassers, the board of absentee ballot canvassers shall proceed as provided under s, 6.97 (2). The board of absentee ballot canvassers shall mark the poll list number of each elector who casts an absentee ballot on the back of the elector's ballot. The board of absentee ballot canvassers shall then deposit the ballot into the proper ballot box and enter the absent elector's name or poll list number after his or her name on the poll list.

**Section 3bp.** 10.02 (3) (form) (a) of the statutes is amended to read:

 $\mathbf{2}$ 

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

10.02 (3) (form) (a) Upon entering the polling place and before being permitted to vote, an elector shall state his or her name and address. If an elector is not registered to vote, an elector may register to vote at the polling place serving his or her residence if the elector provides proof of residence or the elector's registration is verified by another elector of the same municipality where the elector resides. If an elector has elected to be subject to an identification requirement under s. 6.345 (1), and any proof of residence provided by the elector does not constitute identification, the elector shall also present a valid Wisconsin operator's license, a valid, current identification card issued to the elector by a U.S. uniformed service, or a valid Wisconsin identification card unless the elector is exempted from this requirement. Where ballots are distributed to electors, the initials of 2 inspectors must appear on the ballot. Upon being permitted to vote, the elector shall retire alone to a voting booth or machine and cast his or her ballot, except that an elector who is a parent or guardian may be accompanied by the elector's minor child or minor ward. An election official may inform the elector of the proper manner for casting a vote, but the official may not in any manner advise or indicate a particular voting choice.".

### **11.** Page 2, line 7: after that line insert:

**"Section 3c.** 9.20 (4) of the statutes is amended to read:

9.20 (4) The Except as provided in sub. (9), the common council or village board shall, without alteration, either pass the ordinance or resolution within 30 days following the date of the clerk's final certificate, or submit it to the electors at the next spring or general election, if the election is more than 6 weeks after the date of the council's or board's action on the petition or the expiration of the 30-day period, whichever first occurs. If there are 6 weeks or less before the election, the ordinance

or resolution shall be voted on at the next election thereafter. The council or board by a three-fourths vote of the members-elect may order a special election for the purpose of voting on the ordinance or resolution at any time prior to the next election, but not more than one special election for direct legislation may be ordered in any 6-month period.

**SECTION 3ce.** 9.20 (9) of the statutes is created to read:

- 9.20 (9) A common council or village board is not required to act under sub. (4) if any of the following applies:
- (a) The proposed ordinance or resolution does not substantially relate to any city or village governmental function or responsibility.
- (b) The proposed ordinance or resolution is primarily ceremonial or aspirational.".
  - **12.** Page 3, line 5: after that line insert:

"Section 3u. 13.093 (2) (b) of the statutes is amended to read:

13.093 (2) (b) Executive budget bills introduced under s. 16.47 (1) are exempt from the fiscal estimate requirement under par. (a) but shall, if they contain a provision affecting a public retirement fund er, providing a tax exemption, or imposing a mandate, as defined in s. 13.59 (1) (b), be analyzed as to those provisions by the respective joint survey committee or the joint committee on state mandates. If such a bill imposes a mandate, as defined in s. 13.59 (1) (b), the bill shall be simultaneously referred to the joint committee on state mandates and the joint committee on finance. If such a bill contains a provision providing a tax exemption, the bill shall be simultaneously referred to the joint survey committee on tax exemptions and the joint committee on finance. The report of the joint survey

- 1 committee on tax exemptions shall be prepared within 60 days of introduction for 2 bills introduced under s. 16.47 (1).".
  - **13.** Page 3, line 5: after that line insert:
- 4 "Section 3u. 11.386 of the statutes is created to read:

#### 11.386 Use of moneys derived from employment for political purposes.

- (1) In this section, "employer" includes the state and every local governmental unit, as defined in s. 16.97 (7).
- (2) No employer or labor organization may increase the salary of an officer or employee, or give an emolument to an officer, employee, or other person, with the intention that the increase in salary, or the emolument, or a part of it, be used to make a contribution or disbursement.
- (3) No employer or labor organization may discriminate against an officer or employee with respect to any term or condition of employment for failing to make a contribution; failing to support or oppose a candidate, proposition, political party, or committee; or supporting or opposing a candidate, proposition, political party, or committee.
- (4) No employer or other person who is responsible for the disbursement of moneys in payment of wages or salaries may withhold any portion of an employee's wages or salary for the purpose of making a contribution to a committee or for use as a contribution to a committee except upon the written request of the employee. Any such request shall be made on a form prescribed by the board informing the employee of the prohibition under sub. (3). The request is valid for 12 months from the date on which it is made by the employee unless the employer and employee agree to an earlier termination date.

(5) Each person who withholds moneys under sub. (4) shall maintain open for public inspection for a period of no less than 3 years from the date on which a withholding occurs, during normal business hours, documents and books of accounts which shall include a copy of each employee's request for withholding, the amounts and dates on which moneys are withheld under the request, and the amounts and dates on which moneys are transferred to any committee by the person. Each such person shall deliver or transmit copies of such information to the board upon its request.

**Section 3vb.** 11.387 of the statutes is created to read:

11.387 Use of deductions for payments to labor organizations. (1) In this section:

- (a) "All-union agreement" has the meaning given under s. 111.02 (1).
- (b) "Fair-share agreement" has the meaning given under s. 111.70 (1) (f) or 111.81 (9).
  - (2) No labor organization may use moneys derived from an all-union agreement or a fair-share agreement that are paid by an individual who is not a member of the organization for the purpose of making a contribution or disbursement, unless authorized by the individual. Any authorization shall be made in the manner provided under s. 11.386 (4).".
- **14.** Page 3, line 7: delete the material beginning with that line and ending with page 4, line 23.
  - **15.** Page 4, line 23: after that line insert:

23 "Section 6d. 13.121 (4) of the statutes is renumbered 13.121 (4) (a) and 24 amended to read:

 $\mathbf{2}$ 

13.121 (4) (a) For the purpose of premium determinations under s. 40.05 (4) and (5) each member of the legislature shall accrue sick leave at a rate equivalent to a percentage of time worked recommended for such positions by the director of the office of state employment relations and approved by the joint committee on employment relations in the same manner as compensation for such positions is determined under s. 20.923, except as provided in par. (b). This percentage of time worked shall be applied to the sick leave accrual rate established under s. 230.35 (2). The approved percentage shall be incorporated into the compensation plan under s. 230.12 (1).

**Section 6h.** 13.121 (4) (b) of the statutes is created to read:

13.121 (4) (b) No member of the legislature may receive sick leave as a member of the legislature during any term of office that begins after the effective date of this paragraph .... [revisor inserts date].".

**16.** Page 5, line 6: after that line insert:

"Section 7m. 13.48 (10) (a) of the statutes is amended to read:

13.48 (10) (a) No state board, agency, officer, department, commission or body corporate may enter into a contract for the construction, reconstruction, remodeling of or addition to any building, structure, or facility, in connection with any building project which involves a cost in excess of \$150,000 without completion of final plans and arrangement for supervision of construction and prior approval by the building commission. The building commission may not approve a contract for the construction, reconstruction, renovation or remodeling of or an addition to a state building as defined in s. 44.51 (2) unless it determines that s. 44.57 has been complied with or does not apply. This section applies to the department of transportation only

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

22

23

24

in respect to buildings, structures and facilities to be used for administrative or operating functions, including buildings, land and equipment to be used for the motor vehicle emission inspection and maintenance program under s. 110.20.".

**17.** Page 6, line 25: after that line insert:

"Section 9nh. 13.48 (32r) of the statutes is repealed.".

**18.** Page 6, line 25: after that line insert:

**"Section 9og.** 13.489 (5) (b) of the statutes is amended to read:

13.489 (5) (b) All project information included in any report required under this subsection par. (a) shall be reported on both a cumulative basis from the inception of the project and on an updated basis for the period since the department's last report under this subsection.

**Section 9oh.** 13.489 (5) (c) of the statutes is created to read:

13.489 (5) (c) With the report submitted under par. (a), by February 1 of each year, the department of transportation shall include a current project schedule for all projects enumerated under s. 84.013 (3) or approved under s. 84.013 (6), showing the annual funding required until completion for each project."

- **19.** Page 7, line 7: after "ss." insert "13.58 (5) (b) 5...".
- **20.** Page 7, line 8: delete the material beginning with "(Lg)" and ending with "(14)" and substitute "(Lg), and 16.973 (10) to (15)".
  - **21.** Page 7, line 15: after that line insert:
- 21 "Section 9rg. 13.58 (5) (b) 5. of the statutes is created to read:

13.58 (5) (b) 5. Review any executive branch information technology project identified in a report submitted to the committee by the department of administration under s. 16.973 (15) to determine whether the project should be

 $\mathbf{2}$ 

continued or implemented.	The committee may	forward any	recommendations
regarding the project to the	governor and to the leg	gislature und	er s. 13.172 (2).".

- **22.** Page 7, line 15: after that line insert:
- **"Section 9t.** 13.59 of the statutes is created to read:
  - 13.59 Joint committee on state mandates. (1) Definitions. In this section:
- (a) "Local governmental unit" has the meaning given in s. 19.42 (7u).
- (b) "Mandate," except in sub. (6), means a statutory provision placing a requirement on a local governmental unit and, in sub. (6), has the meaning given in s. 227.112 (1) (b). The term does not include any statutory provision that relates to employment discrimination or the compensation, benefits, leave, collective bargaining rights, or conditions of employment of employees or retirees of a local governmental unit or prevailing wages under s. 66.0903.
- (2) CREATION. There is created a joint committee on state mandates, consisting of 3 majority party and 2 minority party senators and 3 majority party and 2 minority party representatives to the assembly appointed as are the members of standing committees in their respective houses. At least one member of the majority party of each house who is appointed to the joint committee on state mandates must also be a member of the joint committee on finance.
- (3) COMMITTEE PROCEDURES. (a) The committee shall meet at the call of its cochairpersons.
- (b) All actions of the committee require the approval of a majority of all of the members.
  - (4) POWERS AND DUTIES. (a) The committee shall:

- 1. Provide the legislature with a report under sub. (5) concerning each bill that would impose a mandate.
- 2. On a regular basis, review existing mandates and evaluate their desirability as a matter of public policy, cost–effectiveness, and financial responsibility.
  - (b) The committee may make investigations and hold hearings.
- (5) Report. (a) Upon the introduction in either house of the legislature of a bill that would impose a mandate, the bill shall at once be referred to the joint committee on state mandates by the presiding officer, unless the presiding officer determines that the mandate has a minimal fiscal effect on local governmental units. The bill may not be considered further by either house or any other committee until the earlier of 30 days after referral or the time at which the joint committee on state mandates submits a written report, to the chief clerk of the house in which the bill is introduced, doing all of the following:
- 1. Describing the fiscal effect on state government and on local governmental units of the mandate contained in the bill.
  - 2. Identifying the objective of the mandate contained in the bill.
- 3. Determining whether it is possible to achieve the objective without imposing a mandate.
- 4. Determining whether the mandate contained in the bill is state-imposed or is addressing a requirement imposed by the federal government.
- 5. Explaining the effect of the mandate contained in the bill on the revenues and expenditures of state government and local governmental units, including an explanation as to whether unrestricted or restricted state aid, grants, or tax benefits are currently being provided or potentially available under existing law to meet the costs of the mandate.

- 6. Identifying whether the mandate contained in the bill has a recurring or nonrecurring impact.
  - 7. Identifying any method of reimbursement for any costs of the mandate contained in the bill or any method of waiver or appeal of the requirements contained in the mandate.
  - 8. Addressing whether it is appropriate to consider an expiration date for the mandate contained in the bill.
  - 9. Addressing the desirability of the mandate contained in the bill as a matter of public policy.
  - 10. Providing any other information that the committee considers to be appropriate.
  - (b) If the committee's report concludes that the bill has a negative uncompensated fiscal effect on local governmental units, and that the mandate contained in the bill is wholly state-imposed, the committee shall prepare and offer an amendment to the bill that increases the amount of the appropriation under s. 20.505 (1) (bm) or such other appropriation designated in the amendment and creates any other provisions required by s. 13.591 (2).
  - (c) The report under par. (a) shall be reproduced as an appendix to the bill and attached to it as are amendments. The reproduction shall be in lieu of inclusion in the daily journal of the house in which the bill is introduced.
  - (6) IDENTIFICATION OF MANDATES. (a) The legislative fiscal bureau shall identify all mandates, other than mandates that have a minimal fiscal effect, existing on the effective date of this paragraph .... [revisor inserts date], and submit that information to the joint committee on state mandates by January 1, 2009.

(b) The committee shall introduce one or more bills amending the statutes in each house of the legislature repealing all mandates that are wholly state-imposed and that have a negative uncompensated fiscal effect on local governmental units.

**Section 9v.** 13.591 of the statutes is created to read:

- **13.591 Funding of state-imposed mandates. (1)** DEFINITION. In this section, "mandate" has the meaning set forth in s. 13.59 (1) (b).
- (2) APPROPRIATION INCREASE. The legislature may not enact a bill on or after the effective date of this subsection .... [revisor inserts date], that contains a mandate unless the bill has had a public hearing before the joint committee on state mandates or contains an appropriation to provide for reimbursement under s. 16.59 for the current fiscal biennium, and requires that an appropriation be provided in all subsequent fiscal years in which the mandate is imposed, by the applicable amount specified in the report prepared under s. 13.59 (5), to provide for reimbursement under s. 16.59.
- (3) Enforcement prohibited if unfunded. If a bill that contains a mandate is enacted after the effective date of this subsection .... [revisor inserts date], is not in compliance with sub. (2), or if the legislature does not provide an appropriation as required by sub. (2) for the mandate, the mandate contained in the enacted bill may not be enforced until the required appropriation is provided.
- (4) Existing unfunded mandates. If the joint committee on state mandates determines that a law enacted or rule promulgated on or before the effective date of this subsection .... [revisor inserts date], contains a mandate that is wholly state-imposed and that has a negative uncompensated fiscal effect on local governmental units, the committee shall introduce a bill in each house of the

- legislature repealing the law or making the rule ineffective unless the committee determines that the uncompensated fiscal effect is minimal.".
- **23.** Page 7, line 15: after that line insert:
- 4 "Section 9rg. 13.58 (5) (b) 5. of the statutes is created to read:
  - 13.58 (5) (b) 5. Review any University of Wisconsin System, institution, or college campus information technology project identified in a report submitted to the committee by the Board of Regents under s. 36.59 (7) to determine whether the project should be continued or implemented. The committee may forward any recommendations regarding the project to the governor and to the legislature under s. 13.172 (2)."
- **24.** Page 7, line 22: delete lines 22 to 25.
- **25.** Page 8, line 1: delete lines 1 to 25.
- **26.** Page 9, line 1: delete lines 1 to 25.
- **27.** Page 10, line 15: delete lines 15 to 24.
- 28. Page 15, line 17: delete "family long-term" and substitute "family".
- **29.** Page 16, line 1: delete lines 1 to 8.
- **30.** Page 16, line 22: after that line insert:
- 18 "Section 20m. 14.037 of the statutes is created to read:
  - 14.037 Legislative approval of Indian gaming on lands taken into trust after October 17, 1988. The governor may not concur with a determination of the U.S. secretary of the interior, under 25 USC 2719 (b) (1) (A), that a gaming establishment proposed to be located on lands acquired by the U.S. secretary of the interior in trust for the benefit of an Indian tribe after October 17, 1988, would be in

5

6

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

- the best interest of the Indian tribe and its members and would not be detrimental to the surrounding community, unless the legislature first concurs in the determination by joint resolution.".
  - **31.** Page 16, line 23: delete lines 23 to 25.
  - **32.** Page 17, line 1: delete lines 1 to 3.
  - **33.** Page 17, line 13: after that line insert:
- 7 "Section 24. 15.01 (2) of the statutes is amended to read:
  - 15.01 (2) "Commission" means a 3-member governing body in charge of a department or independent agency or of a division or other subunit within a department, except for the Wisconsin waterways commission which shall consist of 5 members and the parole commission which shall consist of 8 members. A Wisconsin group created for participation in a continuing interstate body, or the interstate body itself, shall be known as a "commission", but is not a commission for purposes of s. 15.06. The parole commission created under s. 15.145 (1) shall be known as a "commission", but is not a commission for purposes of s. 15.06. The sentencing commission created under s. 15.105 (27) shall be known as a "commission" but is not a commission for purposes of s. 15.06 (1) to (4m), (7), and (9).".
  - **34.** Page 17, line 19: delete the material beginning with "office" and ending with "and the" on line 22 and substitute "office of justice assistance in the department of administration and the".
  - **35.** Page 18, line 5: delete lines 5 to 7 and substitute "division shall be headed by an "administration,". The office of justice assistance in the department of administration and the office of credit unions in the".
    - **36.** Page 18, line 9: delete "administration," and substitute "administration".

8

11

12

13

14

15

16

17

18

19

20

1	<b>37.</b> Page 18, line 10: delete the material beginning with 0"the" and ending
2	with "administration," on line 11.

- **38.** Page 18, line 12: after that line insert:
- 4 "Section 26g. 15.04 (1) (n) of the statutes is created to read:
- 15.04 (1) (n) Programmatic goals for grants and loans. Establish clear and measurable goals for each grant or loan program administered by the department or independent agency, including all of the following:
  - 1. Specific goals for the program that are tied to statutory policy objectives.
- 9 2. At minimum, one quantifiable benchmark for each program goal described in subd. 1.
  - 3. A requirement that each contract with a recipient of a grant or loan under the program specify the frequency and format of reports to be submitted to the department or independent agency and the performance measures to be included in the reports.
  - 4. A method for evaluating the projected or anticipated results of the grant or loan program with actual outcomes as determined by evaluating the information described in subds. 1. to 3.
  - 5. A requirement that the department or independent agency annually and independently verify, from a sample of grants and loans, the accuracy of the information reported under subd. 3.".
  - **39.** Page 18, line 20: delete lines 20 to 22.
- **40.** Page 19, line 13: delete lines 13 to 16 and substitute:
- 23 "Section 33m. 15.105 (19) of the statutes is repealed.".
- **41.** Page 19, line 18: delete lines 18 to 22.

- 1 **42.** Page 19, line 23: delete lines 23 to 25.
- 2 **43.** Page 20, line 1: delete lines 1 and 2.
- 3 **44.** Page 20, line 10: delete lines 10 to 24.
- 4 **45.** Page 21, line 1: delete lines 1 to 25.
- 5 **46.** Page 22, line 1: delete lines 1 to 25.
- 6 **47.** Page 23, line 1: delete lines 1 to 20.
- 7 **48.** Page 23, line 20: after that line insert:
- 8 "Section 67g. 15.255 (1) (a) (intro.) of the statutes is amended to read:
- 9 15.255 (1) (a) (intro.) The board shall be composed of  $\frac{15}{14}$  members as follows:
- 10 **Section 67m.** 15.255 (1) (a) 7. of the statutes is repealed.".
- 11 **49.** Page 23, line 21: delete the material beginning with that line and ending with page 24, line 9.
- 13 **50.** Page 24, line 9: after that line insert:

- **"Section 68d.** 15.497 (2) (a) of the statutes is amended to read:
- 15 15.497 (2) (a) One representative each of the state departments of the 16 American Legion, the Disabled American Veterans, the Veterans of Foreign Wars, the Marine Corps League, the Navy Club of the U.S.A., the Veterans of World War II 17 (AMVETS), the Veterans of World War I of the U.S.A., Inc., the American 18 19 Ex-prisoners of War, the Vietnam Veterans Against the War, Inc., the Vietnam 20 Veterans of America, Inc., the Catholic War Veterans of the U.S.A., the Jewish War 21Veterans of the U.S.A., the Polish Legion of American Veterans, the National 22Association for Black Veterans, Inc., the Army and Navy Union of the United States

of America, the Wisconsin Association of Concerned Veteran Organizations, the

- 1 United Women Veterans, Inc., the U.S. Submarine Veterans of World War II, the
- 2 Federation of Minority Veterans, Inc., the Wisconsin Vietnam Veterans, Inc., and the
- 3 Military Order of the Purple Heart.
- **SECTION 68e.** 15.497 (2) (e) of the statutes is created to read:
- 5 15.497 (2) (e) One representative of the Wisconsin Council of the Military
- 6 Officers Association of America.
- **Section 68f.** 15.497 (2) (f) of the statutes is created to read:
- 8 15.497 (2) (f) One representative of The Retired Enlisted Association.".
- **51.** Page 25, line 6: delete "235, and" and substitute "235, and".
- 10 **52.** Page 25, line 6: after that line insert:
- "Section **69m.** 16.003 (2) of the statutes is amended to read:
- 12 16.003 (2) STAFF. Except as provided in ss. 16.548, 16.57, 978.03 (1), (1m) and
- 13 (2), 978.04 and 978.05 (8) (b), the secretary shall appoint the staff necessary for
- performing the duties of the department. All staff shall be appointed under the
- classified service except as otherwise provided by law.".
- **53.** Page 26, line 22: delete "235, or" and substitute "235, or".
- 17 **54.** Page 27, line 3: delete lines 3 to 10.
- 18 **55.** Page 27, line 20: delete lines 20 to 25.
- 19 **56.** Page 28, line 1: delete lines 1 to 23.
- 20 **57.** Page 29, line 1: delete lines 1 to 8.
- 21 **58.** Page 29, line 9: delete lines 9 to 15.
- 22 **59.** Page 29, line 18: after that line insert:
- 23 "Section 80t. 16.415 (4) of the statutes is created to read:

- 1 16.415 (4) As soon as practicable after the effective date of this subsection ....
  2 [revisor inserts date], the secretary shall convert the state payroll system from a
  3 biweekly pay period to a monthly pay period.".
  - **60.** Page 29, line 21: delete "235, or" and substitute "235, or".
- **61.** Page 29, line 21: after that line insert:
  - **"Section 81h.** 16.42 (1) (intro.) of the statutes is amended to read:
    - 16.42 (1) (intro.) All agencies, other than the legislature and the courts, no later than September 15 of each even-numbered year, in the form and content prescribed by the department, but subject to the requirements of sub. (3), shall prepare and forward to the department and to the legislative fiscal bureau the following program and financial information:".
  - **62.** Page 29, line 22: delete that line.
- **63.** Page 29, line 22: after that line insert:
- **"Section 82h.** 16.42 (3) of the statutes is created to read:
  - 16.42 (3) (a) In this subsection, "zero-based budgeting" means the compilation of a budget in which each component is justified on the basis of cost, need, and relation to the statutory responsibilities of the state agency for which the budget is made.
  - (b) 1. Except as provided in subd. 2., beginning with the 2009–11 fiscal biennium, the department shall require 20 percent of agencies under sub. (1) to submit their biennial budget requests prepared using the principles of zero-based budgeting for each of its activities, units, and programs. In each fiscal biennium thereafter, the department shall require a different 20 percent of the agencies to submit their biennial budget requests in this form.

- 2. The department shall require each agency to submit its biennial budget request prepared using the principles of zero-based budgeting for each of its activities, units, and programs at least once during any 5 consecutive fiscal biennia.".
  - **64.** Page 29, line 23: delete that line and substitute:

**"Section 83g.** 16.423 (5) of the statutes is created to read:

16.423 (5) Notwithstanding sub. (4), a state agency may use any format for its report. If a state agency uses a format for a report that is different from the format that the state agency used for one or more previous reports, the state agency shall reissue all previous reports in the new format.

**SECTION 83L.** 16.43 of the statutes is amended to read:

16.43 Budget compiled. The secretary shall compile and submit to the governor or the governor-elect and to each person elected to serve in the legislature during the next biennium, not later than November 20 of each even-numbered year, a compilation giving all of the data required by s. 16.46 to be included in the state budget report, except the recommendations of the governor and the explanation thereof. The secretary shall not include in the compilation any provision for the development or implementation of an information technology development project for an executive branch agency that is not consistent with the strategic plan of the agency, as approved under s. 16.976. The secretary may distribute the budget compilation in printed or optical disk format. As part of the budget compilation, the secretary shall also submit the base budget review reports under s. 16.423 for the current fiscal biennium and the most recent statements submitted under s. 16.42 (1) (b) for each state agency."

**65.** Page 29, line 24: after that line insert:

 $\mathbf{2}$ 

"Section 85d. 16.467 of the statutes is created to read:

16.467 Preparation of biennial budget bill or bills according to generally accepted accounting principles. For the fiscal biennium that occurs after the fiscal year in which the department determines, in any audited financial report prepared by the department, that the state does not have a budget deficit according to generally accepted accounting principles, as adopted by the governmental accounting standards board or its successor bodies, and each fiscal biennium thereafter, the biennial budget bill or bills shall be prepared according to generally accepted accounting principles.

**Section 85h.** 16.47 (1) of the statutes is amended to read:

16.47 (1) Except as provided in s. 16.529 (2) and subject to ss. 25.40 (3) (c) and 655.27 (6), the executive budget bill or bills shall incorporate the governor's recommendations for appropriations for the succeeding biennium. The appropriation method shown in the bill or bills shall in no way affect the amount of detail or manner of presentation which may be requested by the joint committee on finance. Appropriation requests may be divided into 3 allotments: personal services, other operating expenses and capital outlay or such other meaningful classifications as may be approved by the joint committee on finance. No executive budget bill or bills may increase the state's budget deficit, if any, according to generally accepted accounting principles, as reported in any audited financial report prepared by the department."

**66.** Page 29, line 24: after that line insert:

**"Section 85c.** 16.50 (1) (a) of the statutes is amended to read:

16.50 (1) (a) Each department except the legislature and the courts shall prepare and submit to the secretary an estimate of the amount of money which it proposes to expend, encumber or distribute under any appropriation in ch. 20. The department of administration shall prepare and submit estimates for expenditures from appropriations under ss. 20.855, 20.865, 20.866 and 20.867. The secretary may waive the submission of estimates of other than administrative expenditures from such funds as he or she determines, but the secretary shall not waive submission of estimates for the appropriations under s. 20.285 (1) (im) and (n) nor for expenditure of any amount designated as a refund of an expenditure under s. 20.001 (5). Estimates shall be prepared in such form, at such times and for such time periods as the secretary requires. Revised Except as provided in par. (c), revised and supplemental estimates may be presented at any time under rules promulgated by the secretary.

**Section 85e.** 16.50 (1) (c) of the statutes is created to read:

16.50 (1) (c) 1. The department may not approve any revised or supplemental estimate submitted by the department of transportation under par. (a) for any appropriation of federal funds under s. 20.395 unless the department of transportation has submitted a request to revise or supplement the estimate to the joint committee on finance and the request is approved under subd. 2. or the department of transportation has submitted a plan including the revised or supplemental estimate to the joint committee on finance under s. 84.03 (2) (b) 1. and the plan is approved under s. 84.03 (2) (c).

2. If the department of transportation submits a request under subd. 1. and the cochairpersons of the joint committee on finance do not notify the department of transportation within 14 working days after the date of the submittal that the

committee has scheduled a meeting for the purpose of reviewing the request, the request is approved. If, within 14 working days after the date of the submittal, the cochairpersons of the committee notify the department of transportation that the committee has scheduled a meeting for the purpose of reviewing the request, the department of administration may not revise or supplement any estimate specified in the request until it is approved by the committee, as submitted or as modified.

**SECTION 85f.** 16.50 (6) of the statutes is amended to read:

- 16.50 (6) PROPORTIONAL SPENDING. If the secretary determines that expenditures of general purpose or segregated fund revenues are utilized to match revenues received under s. 16.54 or 20.001 (2) (b) for the purposes of combined program expenditure, the secretary may require that disbursements of the general purpose revenue and corresponding segregated revenue be in direct proportion to the amount of program revenue or corresponding segregated revenue which is available or appropriated in ch. 20 or as condition of a grant or contract. If Subject to sub. (1) (c), if the secretary makes such a determination, the agency shall incorporate the necessary adjustments into the expenditure plans provided for in sub. (1).".
- **67.** Page 30, line 12: delete the material beginning with that line and ending with page 34, line 2.
  - **68.** Page 34, line 9: after that line insert:
  - **"Section 89h.** 16.53 (1) (d) 1. of the statutes is amended to read:
- 16.53 (1) (d) 1. The secretary, with the approval of the joint committee on employment relations, shall fix the time and frequency for payment of salaries due elective and appointive officers and employees of the state. As determined under this subdivision, the salaries shall be paid either monthly, semimonthly or for each

- 2 -week period, except that as soon as practicable after the effective date of this subdivision .... [revisor inserts date], the salaries shall be paid monthly.".
- **69.** Page 34, line 20: delete the material beginning with that line and ending with page 35, line 11.
- **70.** Page 35, line 19: delete lines 19 to 25.
- **71.** Page 36, line 1: delete lines 1 to 8.
- **72.** Page 36, line 9: delete lines 9 to 24.
- **73.** Page 36, line 25: delete that line.
- **74.** Page 37, line 1: delete lines 1 to 8.
- **75.** Page 37, line 9: delete lines 9 to 25.
- **76.** Page 38, line 1: delete lines 1 and 2.
- **77.** Page 38, line 2: after that line insert:

- "Section 99c. 16.59 of the statutes is created to read:
- **16.59 State funding of mandates. (1)** In this section:
- 15 (a) "Local governmental unit" has the meaning given in s. 19.42 (7u).
- 16 (b) "Mandate" has the meaning given in s. 227.112 (1) (b).
  - (2) From the appropriation under s. 20.505 (1) (bm) or such other appropriations designated by law for such purpose, on the basis of the report prepared under s. 13.59 (5) and other relevant information available to the department, the department shall reimburse local governmental units for their approximate costs not otherwise funded by the state that are attributable to mandates that are subject to the funding requirements under s. 13.591 or 227.112. Reimbursements under this subsection shall be made on an annual basis and shall

- be accompanied by a statement identifying each mandate for which reimbursement
  is made and the amount of reimbursement for each mandate.".
- 3 **78.** Page 38, line 2: after that line insert:
- 4 "Section 99m. 16.548 of the statutes is repealed.".
- **79.** Page 38, line 5: delete "235, or" and substitute "235, or".
- 80. Page 39, line 16: delete the material beginning with that line and ending with page 40, line 2.
- 8 **81.** Page 40, line 18: delete lines 18 to 25.
- 9 **82.** Page 45, line 23: after that line insert:
- 10 "Section 112d. 16.841 of the statutes is repealed.".
- 11 **83.** Page 45, line 23: after that line insert:
- 12 "**Section 112e.** 16.838 (1) (b) of the statutes is amended to read:
- 13 16.838 (1) (b) "Authority" means a body created under subch. II of ch. 114 or ch. 231, 232, 233, 234, <del>235,</del> or 237.".
- 15 **84.** Page 46, line 1: after "PROJECTS." insert "(a)".
- 16 **85.** Page 46, line 6: after that line insert:

18

19

20

21

22

- "(b) The department shall measure and verify each energy conservation construction project funded under this subsection in accordance with the performance measurement and verification guidelines adopted by the federal Energy Management Program.
- (c) The department shall, to the extent feasible, use the procedures under s. 16.858 to carry out energy conservation construction projects funded under this subsection. In any contract entered into by the department under s. 16.858 that is

- funded under this subsection, the contract shall set forth the minimum savings in
- 2 energy usage that will be realized by the state from construction of the project and
- 3 the contractor shall guarantee that the savings will be realized.".
- 4 **86.** Page 46, line 14: after that line insert:
- 5 "**Section 112w.** 16.848 (2) (b) of the statutes is repealed.".
- **87.** Page 48, line 23: delete "235, or" and substitute "235, or".
- 7 **88.** Page 48, line 24: delete the material beginning with that line and ending with page 50, line 2.
- 9 **89.** Page 50, line 3: delete lines 3 to 7.

17

18

19

20

21

- 10 **90.** Page 50, line 7: after that line insert:
- "Section 118e. 16.964 (title) of the statutes is repealed.
- 12 **Section 118g.** 16.964 (1) (intro.) of the statutes is repealed.
- 13 **SECTION 118i.** 16.964 (1) (a) of the statutes is renumbered 165.25 (12).
- **SECTION 118k.** 16.964 (1) (b) of the statutes is renumbered 165.25 (13) and amended to read:
  - 165.25 (13) Juvenile Justice improvement plan on behalf of the governor. The plan shall be submitted to the joint committee on finance in accordance with s. 16.54 and to the appropriate standing committees of each house of the legislature as determined by the presiding officer of each house. The plan shall be updated periodically and shall be based on an analysis of the state's juvenile justice needs and problems.
- 23 **Section 118m.** 16.964 (1) (c) of the statutes is renumbered 165.25 (14).
- **SECTION 1180.** 16.964 (1) (d) of the statutes is renumbered 165.25 (15).

1	<b>SECTION 118q.</b> 16.964 (1) (e) of the statutes is renumbered 165.25 (16).".
2	<b>91.</b> Page 50, line 8: delete that line and substitute:
3	"Section 119c. 16.964 (1) (f) of the statutes is renumbered 165.25 (17) and
4	amended to read:
5	165.25 (17) Maintain a statistical analysis center to serve as a clearing house
6	of justice system data and information and conduct justice system research and data
7	analysis <del>under this section</del> .".
8	<b>92.</b> Page 50, line 9: before that line insert:
9	"Section 119e. 16.964 (1) (g) of the statutes is renumbered 165.845 (1) (a) and
10	amended to read:
11	165.845 (1) (a) Collect information concerning the number and nature of
12	offenses known to have been committed in this state and such other information as
13	may be useful in the study of crime and the administration of justice. The office
14	department of justice may determine any other information to be obtained regarding
15	crime and justice system statistics. The information shall include data requested by
16	the federal bureau of investigation under its system of uniform crime reports for the
17	United States.
18	<b>Section 119g.</b> 16.964 (1) (h) of the statutes is renumbered 165.845 (1) (b) and
19	amended to read:
20	165.845 (1) (b) Furnish all reporting officials with forms or instructions or both
21	that specify the nature of the information required under par. $(g)$ $(a)$ , the time it is
22	to be forwarded, the method of classifying and any other matters that facilitate

**Section 119i.** 16.964 (1) (i) of the statutes is repealed.

collection and compilation.

23

**SECTION 119k.** 16.964 (2) of the statutes is renumbered 165.845 (2) and amended to read:

165.845 **(2)** All persons in charge of law enforcement agencies and other criminal and juvenile justice system agencies shall supply the office department of justice with the information described in sub. (1) (g) (a) on the basis of the forms or instructions or both to be supplied by the office department under sub. (1) (g) (a).".

**93.** Page 50, line 9: delete lines 9 to 11 and substitute:

"Section 120g. 16.964 (3) of the statutes is repealed.".

**94.** Page 50, line 11: after that line insert:

"Section 120m. 16.964 (5) of the statutes is renumbered 165.986, and 165.986 (1), (2), (3) (intro.), (4) and (6), as renumbered, are amended to read:

165.986 (1) The office department shall provide grants from the appropriation under s. 20.505 (6) (c) 20.455 (2) (cr) to cities to employ additional uniformed law enforcement officers whose primary duty is beat patrolling. A city is eligible for a grant under this subsection section in fiscal year 1994–95 if the city has a population of 25,000 or more. A city may receive a grant for a calendar year if the city applies for a grant before September 1 of the preceding calendar year. Grants shall be awarded to the 10 eligible cities submitting an application for a grant that have the highest rates of violent crime index offenses in the most recent full calendar year for which data is available under the uniform crime reporting system of the federal bureau of investigation.

(2) A city applying to the <u>office department</u> for a grant under this <u>subsection</u> section shall include a proposed plan of expenditure of the grant moneys. The grant moneys that a city receives under this <u>subsection</u> may be used for salary and

fringe benefits only. Except as provided in par. (c) sub. (3), the positions for which funding is sought must be created on or after April 21, 1994, and result in a net increase in the number of uniformed law enforcement officers assigned to beat patrol duties.

- (3) (intro.) During the first 6 months of the first year of a grant, a city may, with the approval of the office department, use part of the grant for the payment of salary and fringe benefits for overtime provided by uniformed law enforcement officers whose primary duty is beat patrolling. A city may submit a request to the office department for a 3-month extension of the use of the grant for the payment of overtime costs. To be eligible to use part of the first year's grant for overtime costs, the city shall provide the office department with all of the following:
- (4) The office department shall develop criteria which, notwithstanding s. 227.10 (1), need not be promulgated as rules under ch. 227, for use in determining the amount to grant to cities under this subsection section. The office department may not award an annual grant in excess of \$150,000 to any city. The office department shall review any application and plan submitted under par. (b) sub. (2) to determine if that application and plan meet the requirements of this subsection section. The grant that a city receives under this subsection section may not supplant existing local resources.
- (6) The office department may make grants to additional cities with a population of 25,000 or more after fiscal year 1994–95. Eligibility for grants under this paragraph subsection shall be determined and allocations made as provided in this subsection section.".
  - **95.** Page 50, line 12: delete that line and substitute:

"Section 122d. 16.964 (10) of the statutes is renumbered 165.987, and 165.987 (1) (intro.), (2), and (3) (intro.) and (e), as renumbered, are amended to read:

165.987 (1) (intro.) In this subsection section:

- (2) From the appropriation under s. 20.505 (6) (ke) 20.455 (2) (kb), the office department of justice shall provide grants to law enforcement agencies for the purchase, installation, or maintenance of digital recording equipment for making audio or audio and visual recordings of custodial interrogations or for training personnel to use such equipment. Grants awarded under this subsection section may be used to reimburse law enforcement agencies for expenses incurred or payments made on or after July 7, 2005. Grants awarded under this subsection section may be used to support recording of custodial interrogations of either juveniles or adults and of interrogations related to either misdemeanor or felony offenses. The office department of justice may award more than one grant under this subsection section to a law enforcement agency. The office department of justice shall develop criteria and procedures to administer this subsection section. Notwithstanding s. 227.10 (1), the criteria and procedures need not be promulgated as rules under ch. 227.
- (3) (intro.) A law enforcement agency shall include the following information in an application for a grant under this subsection section:
  - (e) Any other information required by the office department of justice.

**SECTION 122e.** 16.964 (11) of the statutes is renumbered 165.935 and amended to read:

165.935 <u>Services for sexual assault victims.</u> From the appropriation under s. 20.505 (6) 20.455 (5) (gj), the office <u>department of justice</u> shall provide grants to nonprofit organizations that provide services to victims of sexual assault. The <u>office</u> <u>department of justice</u> shall develop criteria and procedures for use in selecting

 $\mathbf{2}$ 

grantees and administering the grant program. Notwithstanding s. 227.10 (1), the criteria and procedures need not be promulgated as rules under ch. 227.".

**96.** Page 50, line 13: delete the material beginning with that line and ending with page 51, line 7, and substitute:

"Section 122x. 16.964 (12) of the statutes is renumbered 165.95, and 165.95 (1) (intro.), (2), (3) (intro.) and (k), (4), (5), (6), (7), (8), (9), (10) and (11), as renumbered, are amended to read:

165.95 (1) (intro.) In this subsection section, "violent offender" means a person to whom one of the following applies:

- (2) The effice department of justice shall make grants to counties to enable them to establish and operate programs, including suspended and deferred prosecution programs and programs based on principles of restorative justice, that provide alternatives to prosecution and incarceration for criminal offenders who abuse alcohol or other drugs. The office department of justice shall make the grants from the appropriations under s. 20.505 (6) (b) and (ku) 20.455 (2) (em) and (kx). The office department of justice shall collaborate with the departments of corrections and health and family services in establishing this grant program.
- (3) (intro.) A county shall be eligible for a grant under par. (b) sub. (2) if all of the following apply:
- (k) The county complies with other eligibility requirements established by the office department of justice to promote the objectives listed in subds. 1. and 2 pars.

  (a) and (b).

 $\mathbf{2}$ 

- (4) In implementing a program that meets the requirements of par. (c) sub. (3), a county department may contract with or award grants to a religious organization under s. 59.54 (27).
- (5) (a) A county that receives a grant under this subsection section shall create an oversight committee to advise the county in administering and evaluating its program. Each committee shall consist of a circuit court judge, the district attorney or his or her designee, the state public defender or his or her designee, a local law enforcement official, a representative of the county, a representative of each other county agency responsible for providing social services, including services relating to child welfare, mental health, and the Wisconsin Works program, representatives of the departments of corrections and health and family services, a representative from private social services agencies, a representative of substance abuse treatment providers, and other members to be determined by the county.
- (b) A county that receives a grant under this subsection section shall comply with state audits and shall submit an annual report to the office department of justice and to the oversight committee created under subd. 1. par. (a) regarding the impact of the program on jail and prison populations and its progress in attaining the goals specified in par. (c) 2. and 6 sub. (3) (b) and (f).
- (6) Two or more counties may jointly apply for and receive a grant under this subsection section. If counties submit a joint application, they shall include with their application a written agreement specifying each county department's role in developing, administering, and evaluating the program. The oversight committee established under par. (e) 1. sub. (5) (a) shall consist of representatives from each county.

- (7) Grants provided under this subsection shall be provided on a calendar year basis beginning on January 1, 2007. If the office department of justice decides to make a grant to a county under this subsection section, the office department of justice shall notify the county of its decision and the amount of the grant no later than September 1 of the year preceding the year for which the grant will be made.
- (8) The office department of justice shall assist a county receiving a grant under this subsection section in obtaining funding from other sources for its program.
- (9) The office department of justice shall inform any county that is applying for a grant under this subsection section whether the county meets the requirements established under par. (c) sub. (3), regardless of whether the county receives a grant.
- (10) The office department of justice shall enter into one or more contracts with another person for the purpose of evaluating the grant program established under this subsection section. The office department of justice shall fund such contracts from moneys appropriated under s. 20.505 (6) (b) and (ku) 20.455 (2) (em) and (kx) with 1 percent of the amount awarded as grants under par. (b) sub. (2).
- (11) By December 31, 2011, the office department of justice, in collaboration with the departments of corrections and health and family services, shall submit a report to the chief clerk of each house of the legislature, for distribution to the appropriate standing committees under s. 13.172 (3), regarding savings that have been generated through the implementation of the grant program. The report shall also include recommendations regarding how the grant program should be structured in the future.".
  - **97.** Page 51, line 8: delete lines 8 to 20.

22

23

24

1	98. Page 56, line 22: after that line insert:
2	"Section 128v. 16.973 (15) of the statutes is created to read:
3	16.973 (15) No later than March 1 and September 1 of each year, submit to the
4	joint committee on information policy and technology a report that documents for
5	each executive branch agency information technology project with an actual or
6	projected cost greater than \$1,000,000 or that the department of administration has
7	identified as a large, high-risk information technology project under sub. (10) (a) all
8	of the following:
9	(a) Original and updated project cost projections.
10	(b) Original and updated completion dates for the project and any stage of the
11	project.
12	(c) An explanation for any variation between the original and updated costs and
13	completion dates under pars. (a) and (b).
14	(d) A copy of any contract entered into by the department for the project and
15	not provided in a previous report.
16	(e) All sources of funding for the project.
17	(f) The amount of any funding provided for the project through a master lease
18	under s. 16.76 (4).
19	(g) Information about the status of the project, including any portion of the
20	project that has been completed.
21	(h) Any other information about the project, or related information technology

99. Page 56, line 24: delete the material beginning with that line and ending with page 57, line 14.

projects, requested by the joint committee on information policy and technology.".

- **100.** Page 57, line 15: delete the material beginning with that line and ending with page 71, line 22.
- **101.** Page 72, line 4: delete "family long-term" and substitute "family".
- **102.** Page 72, line 9: after that line insert:
- 5 "Section 156t. 19.42 (10) (m) of the statutes is repealed.".
- **103.** Page 72, line 9: after that line insert:
- 7 "Section 156m. 19.36 (14) of the statutes is created to read:
- 8 19.36 (14) Utility Security System Plans. (a) In this subsection:
  - 1. "Security system plan" means a plan for the physical or electronic security of facilities, telecommunications systems, or information technology systems owned or operated by a utility, including any information, photograph, audio or visual presentation, schematic diagram, survey, recommendation, consultation, or other communication related to such a plan, and including any threat assessment, vulnerability or capability assessment, or threat response plan or any emergency evacuation plan.
  - 2. "Utility" means a person that generates, transmits, or distributes electricity, transports or distributes natural gas, operates a public water system, or provides telecommunications or sewer service.
  - (b) An authority may withhold access to any record containing a security system plan or a portion of a security system plan if the authority determines that a facility or system that is the subject of the plan is so vital to the state that the incapacity or destruction of the facility or system would have a debilitating impact on the physical or economic security of the state or on public health, safety, or welfare."

- 1 **104.** Page 72, line 10: delete lines 10 to 12 and substitute.
- 2 "Section 157. 19.42 (10) (p) of the statutes is repealed.".
- 3 **105.** Page 72, line 15: after that line insert:
- 4 "Section 158t. 19.42 (13) (L) of the statutes is repealed.".
- 5 **106.** Page 72, line 16: delete lines 16 to 18 and substitute.
- 6 "Section 159. 19.42 (13) (o) of the statutes is repealed.".
- 7 **107.** Page 72, line 19: delete lines 19 to 25.
- 8 **108.** Page 73, line 1: delete lines 1 to 10.
- 9 **109.** Page 73, line 24: delete "family long-term" and substitute "family".
- 10 **110.** Page 74, line 20: delete lines 20 to 25.
- 11 **111.** Page 75, line 1: delete lines 1 to 25.
- 12. Page 76, line 1: delete lines 1 to 6.
- 13 Page 76, line 6: after that line insert:
- **"Section 167c.** 20.002 (11) (d) 3. of the statutes is amended to read:
- 15 20.002 (11) (d) 3. The industrial building construction loan fund under s.
- 16 560.10<u>, 2005 stats</u>.".
- 17 **114.** Page 76, line 17: delete lines 17 to 20 and substitute:
- **Section 170p.** 20.003 (4) (ft) of the statutes is repealed.
- 19 **Section 171p.** 20.003 (4) (fv) of the statutes is repealed.".
- 20 **115.** Page 76, line 22: delete "\$130,000,000" and substitute "\$65,000,000".
- 21 **116.** Page 76, line 24: delete "\$130,000,000" and substitute "\$65,000,000".
- 22 **117.** Page 77, line 2: after that line insert:

"Section 174c. 20.003 (6) of the statutes is created to read:

20.003 (6) RESTRICTION ON AUTHORIZATION OF CERTAIN PUBLIC DEBT. No bill may be enacted by the legislature that would authorize any new public debt, the debt service of which is paid from general purpose revenue, to be contracted in an amount exceeding \$430,000,000 in any fiscal biennium for projects enumerated in the long-range state building program.".

**118.** Page 77, line 2: after that line insert:

**"Section 174c.** 20.003 (4m) of the statutes is amended to read:

20.003 (4m) Required general fund structural balance. No bill may be adopted by the legislature if the bill would cause in any the 2nd fiscal year of a fiscal biennium the amount of moneys designated as "Total Expenditures" in the summary under s. 20.005 (1) for that fiscal year, less any amounts transferred to the budget stabilization fund in that fiscal year, to exceed the sum of the amount of moneys designated as "Taxes" and "Departmental Revenues" in the summary under s. 20.005 (1) for that fiscal year."

- **119.** Page 91, line 9: decrease the dollar amount for fiscal year 2007–08 by \$522,800 and decrease the dollar amount for fiscal year 2008–09 by \$522,800 to reduce funding for the purpose for which the appropriation is made, and adjust the NET APPROPRIATION totals accordingly.
- **120.** Page 91, line 9: increase the dollar amount for fiscal year 2007–08 by \$43,900 and increase the dollar amount for fiscal year 2008–09 by \$43,900 to increase funding for the purpose for which the appropriation is made and to increase the authorized FTE positions for the department of agriculture, trade and consumer

- protection by 0.5 GPR position, and adjust the NET APPROPRIATION totals accordingly.
  - **121.** Page 91, line 10: decrease the dollar amount for fiscal year 2007–08 by \$32,200 and decrease the dollar amount for fiscal year 2008–09 by \$32,200 for the purpose of eliminating 0.5 vacant GPR position, and adjust the NET APPROPRIATION totals accordingly.
  - **122.** Page 91, line 10: decrease the dollar amount for fiscal year 2007–08 by \$108,800 and decrease the dollar amount for fiscal year 2008–09 by \$108,800 to reduce funding for the purpose for which the appropriation is made, and adjust the NET APPROPRIATION totals accordingly.
  - **123.** Page 91, line 11: decrease the dollar amount for fiscal year 2007–08 by \$38,400 and decrease the dollar amount for fiscal year 2008–09 by \$38,400 to reduce funding for the purpose for which the appropriation is made, and adjust the NET APPROPRIATION totals accordingly.
  - **124.** Page 91, line 11: decrease the dollar amount for fiscal year 2007–08 by \$102,300 and decrease the dollar amount for fiscal year 2008–09 by \$102,300 for the purpose of decreasing the authorized positions for the department of agriculture, trade and consumer protection by 1.5 GPR positions and for other costs related to privacy protection, and adjust the NET APPROPRIATION totals accordingly.
  - **125.** Page 91, line 13: increase the dollar amount for fiscal year 2007–08 by \$54,900 and increase the dollar amount for fiscal year 2008–09 by \$54,900 to increase funding for the purpose for which the appropriation is made and to increase the authorized FTE positions for the department of agriculture, trade and consumer protection by 1.05 PR positions.

 $\mathbf{2}$ 

- 126. Page 91, line 19: decrease the dollar amount for fiscal year 2007–08 by \$5,700 and decrease the dollar amount for fiscal year 2008–09 by \$5,700 to decrease funding for the purpose for which the appropriation is made and to decrease the authorized FTE positions for the department of agriculture, trade and consumer protection by 0.15 PR position.
- **127.** Page 91, line 21: decrease the dollar amount for fiscal year 2007–08 by \$13,400 and decrease the dollar amount for fiscal year 2008–09 by \$13,400 to decrease funding for the purpose for which the appropriation is made and to decrease the authorized FTE positions for the department of agriculture, trade and consumer protection by 0.35 PR position.
- **128.** Page 93, line 1: decrease the dollar amount for fiscal year 2007–08 by \$430,400 and decrease the dollar amount for fiscal year 2008–09 by \$430,400 to reduce funding for the purpose for which the appropriation is made.
- **129.** Page 93, line 1: increase the dollar amount for fiscal year 2007–08 by \$76,900 and increase the dollar amount for fiscal year 2008–09 by \$76,900 for the purpose of increasing the authorized FTE positions for the department of agriculture, trade and consumer protection by 1.0 GPR position.
- **130.** Page 93, line 13: decrease the dollar amount for fiscal year 2007–08 by \$146,500 and decrease the dollar amount for fiscal year 2008–09 by \$146,500 to reduce funding for the purpose for which the appropriation is made.
- **131.** Page 94, line 12: decrease the dollar amount for fiscal year 2007–08 by \$100,000 and decrease the dollar amount for fiscal year 2008–09 by \$100,000 to decrease funding for the purpose for which the appropriation is made.
  - **132.** Page 94, line 12: after that line insert:

"(g) 1 Grant for pulp and paper mill PR 2,500,000 -0-". В 2 **133.** Page 94. line 14: after that line insert: 3 "(qm) Grants for soybean crushing 4 facilities 4,000,000 -0-". SEG В **134.** Page 94. line 16: after that line insert: 5 6 "(u) Grant for pulp and paper mill; 7 recycling fund 2,500,000 SEG В -0-". 8 **135.** Page 95, line 1: decrease the dollar amount for fiscal year 2007–08 by 9 \$66,700 and decrease the dollar amount for fiscal year 2008–09 by \$66,700 to reduce 10 funding for the purpose for which the appropriation is made. 11 **136.** Page 95, line 19: decrease the dollar amount for fiscal year 2007–08 by \$4.000,000 and decrease the dollar amount for fiscal year 2008-09 by \$5.000,000 to 12 13 reduce funding for the purpose for which the appropriation is made. 137. Page 96. line 7: decrease the dollar amount for fiscal year 2007-08 by 14 15 \$471,600 and decrease the dollar amount for fiscal year 2008-09 by \$471,600 to 16 reduce funding for the purpose for which the appropriation is made. 17 **138.** Page 96, line 7: decrease the dollar amount for fiscal year 2007–08 by \$76,900 and decrease the dollar amount for fiscal year 2008-09 by \$76,900 for the 18 19 purpose of decreasing the authorized FTE positions for the department of 20 agriculture, trade and consumer protection by 1.0 GPR position. 21 **139.** Page 97, line 7: decrease the dollar amount for fiscal year 2007–08 by 22 \$124,300 and decrease the dollar amount for fiscal year 2008-09 by \$124,300 for the 23purpose of eliminating 1.95 vacant GPR positions.

 $\mathbf{2}$ 

- **140.** Page 97, line 7: decrease the dollar amount for fiscal year 2007–08 by \$415,300 and decrease the dollar amount for fiscal year 2008–09 by \$415,300 to decrease funding for the purpose for which the appropriation is made.
- **141.** Page 97, line 9: decrease the dollar amount for fiscal year 2007–08 by \$3,000 and decrease the dollar amount for fiscal year 2008–09 by \$3,000 to decrease funding for the purposes for which the appropriation is made.
- **142.** Page 97, line 10: decrease the dollar amount for fiscal year 2007–08 by \$32,000 and decrease the dollar amount for fiscal year 2008–09 by \$32,000 to decrease funding for the purposes for which the appropriation is made.
- **143.** Page 97, line 13: decrease the dollar amount for fiscal year 2007–08 by \$1,919,600 and decrease the dollar amount for fiscal year 2008–09 by \$919,600 to decrease funding for the purposes for which the appropriation is made.
- **144.** Page 98, line 5: decrease the dollar amount for fiscal year 2007–08 by \$25,000 and decrease the dollar amount for fiscal year 2008–09 by \$25,000 to decrease funding for the purpose for which the appropriation is made.
- **145.** Page 98, line 12: decrease the dollar amount for fiscal year 2007–08 by \$60,700 and decrease the dollar amount for fiscal year 2008–09 by \$60,700 to decrease funding for the purpose for which the appropriation is made.
- **146.** Page 98, line 16: decrease the dollar amount for fiscal year 2007–08 by \$71,200 and decrease the dollar amount for fiscal year 2008–09 by \$71,200 to decrease funding for the purposes for which the appropriation is made.

- **147.** Page 98, line 20: decrease the dollar amount for fiscal year 2007–08 by \$25,400 and decrease the dollar amount for fiscal year 2008–09 by \$25,400 to decrease funding for the purposes for which the appropriation is made.
  - **148.** Page 100, line 14: decrease the dollar amount for fiscal year 2007–08 by \$112,800 and decrease the dollar amount for fiscal year 2008–09 by \$112,800 to decrease funding for the purposes for which the appropriation is made.
- **149.** Page 101, line 7: decrease the dollar amount for fiscal year 2007–08 by \$616,900 and decrease the dollar amount for fiscal year 2008–09 by \$616,900 to decrease funding for the purposes for which the appropriation is made.
- **150.** Page 101, line 11: decrease the dollar amount for fiscal year 2007–08 by \$19,100 and decrease the dollar amount for fiscal year 2008–09 by \$19,100 for the purpose of eliminating 0.3 vacant GPR position.
- **151.** Page 101, line 11: decrease the dollar amount for fiscal year 2007–08 by \$61,500 and decrease the dollar amount for fiscal year 2008–09 by \$61,500 to decrease funding for the purpose for which the appropriation is made.
- **152.** Page 101, line 13: decrease the dollar amount for fiscal year 2007–08 by \$130,000 and decrease the dollar amount for fiscal year 2008–09 by \$130,000 to decrease funding for the purposes for which the appropriation is made.
- **153.** Page 101, line 16: decrease the dollar amount for fiscal year 2007–08 by \$150,600 and decrease the dollar amount for fiscal year 2008–09 by \$150,600 to decrease funding for the purposes for which the appropriation is made.

- **154.** Page 101, line 18: decrease the dollar amount for fiscal year 2007–08 by \$4,500 and decrease the dollar amount for fiscal year 2008–09 by \$4,500 to decrease funding for the purpose for which the appropriation is made.
  - **155.** Page 102, line 8: delete lines 8 to 10.
- **156.** Page 102, line 18: decrease the dollar amount for fiscal year 2007–08 by \$299,900 to decrease funding for the purpose for which the appropriation is made.
- **157.** Page 102, line 18: decrease the dollar amount for fiscal year 2008–09 by \$2,999,000 to reflect the elimination of the program.
  - **158.** Page 103, line 8: delete lines 8 and 9.
  - **159.** Page 104, line 1: decrease the dollar amount for fiscal year 2007–08 by \$1,000,000 and decrease the dollar amount for fiscal year 2008–09 by \$1,000,000 to reduce funding for the purpose for which the appropriation is made.
  - **160.** Page 104, line 11: decrease the dollar amount for fiscal year 2007–08 by \$142,300 and decrease the dollar amount for fiscal year 2008–09 by \$142,300 to decrease funding for the purpose for which the appropriation is made.
  - **161.** Page 106, line 8: decrease the dollar amount for fiscal year 2007–08 by \$75,900 and decrease the dollar amount for fiscal year 2008–09 by \$91,900 for the purpose of decreasing the authorized FTE positions for the office of the commissioner of insurance by 1.0 PR position for an insurance financial examiner–chief position in the office's bureau of financial analysis and examinations.
  - **162.** Page 106, line 8: decrease the dollar amount for fiscal year 2007–08 by \$102,300 and decrease the dollar amount for fiscal year 2008–09 by \$102,300 for the purpose of decreasing funding related to privacy protection.

1	163. Page 109, line 3: decrease the dollar amount for fiscal year 2007–08 by					
2	\$49,500 and decrease the dollar amount for fiscal year 2008–09 by \$49,500 for the					
3	purpose of reducing funding for the purposes for which the appropriation is mad					
4	<b>164.</b> Page 111, line 10: delete that line.					
5	<b>165.</b> Page 112, line 4: delete lines 4 to 6.					
6	<b>166.</b> Page 113, line 3: decrease the dollar amount for fiscal year 2007–08 by					
7	\$1,653,100 and decrease the dollar amount for fiscal year $2008-09$ by $$1,653,100$					
8	decrease funding for the purpose for which the appropriation is made.					
9	167. Page 113, line 4: decrease the dollar amount for fiscal year 2008–09 by					
10	\$100,000 to decrease funding for the purposes for which the appropriation is made.					
11	168. Page 113, line 6: decrease the dollar amount for fiscal year 2007–08 by					
12	\$125,400 and decrease the dollar amount for fiscal year 2008-09 by \$125,400 to					
13	decrease funding for the purpose for which the appropriation is made.					
14	169. Page 113, line 9: decrease the dollar amount for fiscal year 2007–08 by					
15	\$597,200 and decrease the dollar amount for fiscal year 2008-09 by \$597,200 to					
16	decrease funding for the purpose for which the appropriation is made.					
17	170. Page 114, line 3: decrease the dollar amount for fiscal year 2007–08 by					
18	\$620,900 and decrease the dollar amount for fiscal year 2008-09 by \$1,241,800 to					
19	decrease funding for the purpose for which the appropriation is made.					
20	<b>171.</b> Page 114, line 3: after that line insert:					
21	"(cd) Loan program for physicians in					
22	shortage areas $ ext{GPR}  ext{ A}  ext{ } -0-  ext{ } 450,000$ ".					
23	172. Page 114, line 3: after that line insert:					

1	"(cp) Loan program for veterinarians
2	of food–producing animals GPR A $-0-$ 125,000".
3	173. Page 114, line 19: decrease the dollar amount for fiscal year 2007–08 by
4	\$10,719,400 and decrease the dollar amount for fiscal year $2008-09$ by $$15,719,400$
5	to decrease funding for the purpose for which the appropriation is made.
6	174. Page 114, line 21: decrease the dollar amount for fiscal year 2007-08 by
7	\$417,800 and decrease the dollar amount for fiscal year 2008-09 by \$835,600 to
8	decrease funding for the purpose for which the appropriation is made.
9	175. Page 115, line 2: decrease the dollar amount for fiscal year 2007-08 by
10	\$19,000 and decrease the dollar amount for fiscal year 2008-09 by \$38,000 to
11	decrease funding for the purpose for which the appropriation is made.
12	<b>176.</b> Page 115, line 4: delete that line.
13	<b>177.</b> Page 115, line 7: delete lines 7 and 8.
14	178. Page 115, line 16: decrease the dollar amount for fiscal year 2007-08 by
15	\$10,000 and decrease the dollar amount for fiscal year 2008-09 by \$20,000 to
16	decrease funding for the purpose for which the appropriation is made.
17	179. Page 116, line 1: decrease the dollar amount for fiscal year 2007-08 by
18	\$40,400 and decrease the dollar amount for fiscal year 2008-09 by \$40,400 for the
19	purpose of eliminating 1.0 vacant GPR position.
20	180. Page 117, line 3: decrease the dollar amount for fiscal year 2007-08 by
21	$$298,\!400$ and decrease the dollar amount for fiscal year $2008-09$ by $$298,\!400$ for the

purpose of eliminating 5.06 vacant GPR positions.

1 **181.** Page 119, line 5: decrease the dollar amount for fiscal year 2007-08 by 2 \$38,000 and decrease the dollar amount for fiscal year 2008-09 by \$38,000 for the 3 purpose of eliminating funding for miscellaneous international education activities. **182.** Page 119, line 7: decrease the dollar amount for fiscal year 2007-08 by 4 \$133.100 and decrease the dollar amount for fiscal year 2008-09 by \$133.100 for the 5 6 purpose of eliminating 2.75 vacant GPR positions. **183.** Page 120, line 15: delete lines 15 and 16. 7 **184.** Page 121, line 9: decrease the dollar amount for fiscal year 2007-08 by 8 9 \$19,319,200 and decrease the dollar amount for fiscal year 2008-09 by \$66,075,200 10 for the purpose of paying general school aids. 11 **185.** Page 121, line 10: after that line insert: 12"(am) Teacher merit pay **GPR** Α 250,000 250,000". **186.** Page 121, line 17: delete "consolidation" and substitute "efficiency 13 14 incentive". 15 187. Page 121, line 18: decrease the dollar amount for fiscal year 2007-08 by 16 \$1,055,800 and decrease the dollar amount for fiscal year 2008-09 by \$2,040,900 for 17the purpose of eliminating adjustment to bilingual-bicultural education aids. **188.** Page 122, line 7: decrease the dollar amount for fiscal year 2007-08 by 18 19 \$1.458,100 and decrease the dollar amount for fiscal year 2008-09 by \$1.835,200 for 20 the purpose of eliminating the increase to the school breakfast reimbursement rate. 21 **189.** Page 122, line 10: after that line insert: 22 "(cr) Aid for pupil transportation GPR 34,315,000 A 34,315,000

1	<b>190.</b> Page 122, line 12: decrease th	e dollar	amoun	t for fiscal yea	ar 2008–09 by
2	\$540,000 for the purpose of eliminating	g expan	sion of	the student	achievement
3	guarantee in education program.				
4	<b>191.</b> Page 122, line 12: after that	line ins	ert:		
5	"(cw) Aid for transportation; youth				
6	options program	GPR	A	20,000	20,000
7	(cy) Aid for transportation; open				
8	enrollment	GPR	A	500,000	500,000".
9	<b>192.</b> Page 122, line 13: delete line	s 13 an	d 14.		
10	<b>193.</b> Page 122, line 17: after that	line ins	ert:		
11	"(dn) School safety grants	GPR	A	5,000,000	5,000,000".
12	<b>194.</b> Page 122, line 20: delete that	t line.			
13	<b>195.</b> Page 123, line 3: after that li	ne inse	rt:		
14	"(fr) Racine County Parental Choice				
15	Program	GPR	S	-0-	-0-".
16	<b>196.</b> Page 123, line 8: increase the	e dollar	amount	for fiscal yea	r 2007-08 by
17	\$91,000 and increase the dollar amount	for fisca	ıl year 2	2008-09 by \$9	1,000 for the
18	purpose of providing grants to support gi	fted and	d talente	ed pupils.	
19	<b>197.</b> Page 123, line 8: after that li	ne inse	rt:		
20	"(fz) Grants for science, technology,				
21	engineering, and mathematics				
22	programs	GPR	A	61,500	61,500".
23	<b>198.</b> Page 123, line 17: delete line	s 17 to 2	21.		

- **199.** Page 124, line 2: delete that line.
- **200.** Page 124, line 4: delete "Grants for national" and substitute "National".
- **201.** Page 124, line 5: delete "or master educator".
- **202.** Page 124, line 6: decrease the dollar amount for fiscal year 2007–08 by \$140,600 and decrease the dollar amount for fiscal year 2008–09 by \$199,500 for the purpose of eliminating grants to persons receiving master educator licenses.
- **203.** Page 124, line 6: delete "licensure".

- **204.** Page 124, line 14: decrease the dollar amount for fiscal year 2007–08 by \$108,900 and decrease the dollar amount for fiscal year 2008–09 by \$108,900 for the purpose of eliminating increase to the precollege scholarship program.
- **205.** Page 125, line 3: decrease the dollar amount for fiscal year 2007–08 by \$42,263,200 and decrease the dollar amount for fiscal year 2008–09 by \$45,961,400 for the purpose for which the appropriation is made.
  - **206.** Page 125, line 8: decrease the dollar amount for fiscal year 2007–08 by \$114,700 and decrease the dollar amount for fiscal year 2008–09 by \$114,700 for the purpose for which the appropriation is made.
  - **207.** Page 125, line 10: decrease the dollar amount for fiscal year 2007–08 by \$5,394,400 and decrease the dollar amount for fiscal year 2008–09 by \$5,394,400 for the purposes for which the appropriation is made.
- **208.** Page 125, line 11: decrease the dollar amount for fiscal year 2007–08 by \$659,400 and decrease the dollar amount for fiscal year 2008–09 by \$659,400 for the purpose for which the appropriation is made.

 $\mathbf{2}$ 

3

4

5

6

7

8

9

10

11

12

13

14

15

- **209.** Page 126, line 3: decrease the dollar amount for fiscal year 2007–08 by \$163,000 and decrease the dollar amount for fiscal year 2008–09 by \$163,000 for the purpose for which the appropriation is made.
- **210.** Page 126, line 6: decrease the dollar amount for fiscal year 2007–08 by \$500,000 and decrease the dollar amount for fiscal year 2008–09 by \$500,000 for the purpose for which the appropriation is made.
- **211.** Page 126, line 8: decrease the dollar amount for fiscal year 2007–08 by \$448,800 and decrease the dollar amount for fiscal year 2008–09 by \$448,800 for the purpose for which the appropriation is made.
- **212.** Page 126, line 10: decrease the dollar amount for fiscal year 2007–08 by \$437,200 and decrease the dollar amount for fiscal year 2008–09 by \$437,200 for the purpose for which the appropriation is made.
- **213.** Page 126, line 11: decrease the dollar amount for fiscal year 2007–08 by \$219,300 and decrease the dollar amount for fiscal year 2008–09 by \$219,300 for the purpose for which the appropriation is made.
  - **214.** Page 126, line 13: delete that line.
- 17 **215.** Page 126, line 15: delete that line.
- 216. Page 127, line 6: decrease the dollar amount for fiscal year 2007-08 by \$3,124,500 and decrease the dollar amount for fiscal year 2008-09 by \$17,737,300 for the purposes for which the appropriation is made.
- 21 **217.** Page 128, line 11: delete lines 11 and 12.

- **218.** Page 129, line 5: increase the dollar amount for fiscal year 2007–08 by \$100,000 and increase the dollar amount for fiscal year 2008–09 by \$100,000 for the purpose for which the appropriation is made.
  - **219.** Page 129, line 5: delete "B" and substitute "A".
- 220. Page 129, line 15: decrease the dollar amount for fiscal year 2007–08 by \$2,169,200 and decrease the dollar amount for fiscal year 2008–09 by \$2,222,600 for the purpose for which the appropriation is made.
  - **221.** Page 130, line 3: decrease the dollar amount for fiscal year 2007–08 by \$2,145,200 and decrease the dollar amount for fiscal year 2008–09 by \$2,145,200 for the purpose for which the appropriation is made.
  - **222.** Page 130, line 4: decrease the dollar amount for fiscal year 2007–08 by \$1,780,500 and decrease the dollar amount for fiscal year 2008–09 by \$2,329,400 for the purpose for which the appropriation is made.
    - **223.** Page 130, line 6: decrease the dollar amount for fiscal year 2007–08 by \$376,100 and decrease the dollar amount for fiscal year 2008–09 by \$841,900 for the purpose for which the appropriation is made.
    - **224.** Page 131, line 3: decrease the dollar amount for fiscal year 2007–08 by \$104,700 and decrease the dollar amount for fiscal year 2008–09 by \$104,700 for the purpose of eliminating 1.5 vacant GPR positions.
  - **225.** Page 131, line 3: decrease the dollar amount for fiscal year 2007–08 by \$665,200 and decrease the dollar amount for fiscal year 2008–09 by \$665,200 to decrease funding for the purposes for which the appropriation is made.

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

20

21

22

23

- **226.** Page 131, line 5: decrease the dollar amount for fiscal year 2007–08 by \$506,700 and decrease the dollar amount for fiscal year 2008–09 by \$506,700 to decrease funding for the purpose for which the appropriation is made.
  - **227.** Page 131, line 11: decrease the dollar amount for fiscal year 2007–08 by \$3,552,500 and decrease the dollar amount for fiscal year 2008–09 by \$3,552,500 to decrease funding for the purposes for which the appropriation is made.
  - **228.** Page 131, line 12: decrease the dollar amount for fiscal year 2007–08 by \$1,944,900 and decrease the dollar amount for fiscal year 2008–09 by \$1,944,900 to decrease funding for the purposes for which the appropriation is made.
  - **229.** Page 132, line 7: decrease the dollar amount for fiscal year 2007–08 by \$158,900 and decrease the dollar amount for fiscal year 2008–09 by \$158,900 to decrease funding for the purpose for which the appropriation is made.
  - **230.** Page 132, line 8: decrease the dollar amount for fiscal year 2007–08 by \$1,000,000 and decrease the dollar amount for fiscal year 2008–09 by \$2,000,000 to decrease funding for the purpose for which the appropriation is made.
    - **231.** Page 132, line 10: after that line insert:
- 17 "(fc) Driver education, local assistan-

18 ce GPR A 307,500 307,500

19 (fg) Chauffeur training grants GPR A 191,000 191,000".

- **232.** Page 132, line 11: decrease the dollar amount for fiscal year 2007–08 by \$284,700 and decrease the dollar amount for fiscal year 2008–09 by \$284,700 to decrease funding for the purpose for which the appropriation is made.
- **233.** Page 132, line 11: after that line insert:

1	"(fp) Emergency medical technician –
2	basic training; state operations $GPR$ A $-0 -0-$ ".
3	<b>234.</b> Page 133, line 16: delete lines 16 to 19.
4	<b>235.</b> Page 137, line 17: decrease the dollar amount for fiscal year 2007–08 by
5	$$164,\!800$ and decrease the dollar amount for fiscal year $2008-09$ by $$164,\!800$ for the
6	purpose of eliminating 3.0 vacant GPR positions.
7	<b>236.</b> Page 137, line 17: decrease the dollar amount for fiscal year 2007–08 by
8	\$309,300 and decrease the dollar amount for fiscal year 2008-09 by \$309,300 to
9	decrease funding for the purpose for which the appropriation is made.
10	<b>237.</b> Page 138, line 9: decrease the dollar amount for fiscal year 2007–08 by
11	\$22,300 and decrease the dollar amount for fiscal year 2008-09 by \$22,300 to
12	decrease funding for the purpose for which the appropriation is made.
13	238. Page 139, line 9: after that line insert:
14	"(kr) Control of wild animals SEG-S B 246,100 246,200".
15	<b>239.</b> Page 139, line 16: delete that line.
16	<b>240.</b> Page 140, line 8: decrease the dollar amount for fiscal year 2007–08 by
17	$$104,\!100$ and decrease the dollar amount for fiscal year $2008-09$ by $$80,\!000$ for the
18	purpose of reducing funding for a sustainable all-terrain vehicle trails system in
19	northern state forests.
20	<b>241</b> . Page 140, line 15: decrease the dollar amount for fiscal year 2007–08 by
21	\$18,100 and decrease the dollar amount for fiscal year 2008-09 by \$18,100 for the
22	purpose of reducing funding for limited term employees, utilities, and supplies

relating to the operation of new buildings and campgrounds in southern state forests, and adjust the NET APPROPRIATION totals accordingly.

- **242.** Page 140, line 16: decrease the dollar amount for fiscal year 2007–08 by \$117,200 and decrease the dollar amount for fiscal year 2008–09 by \$331,100 for the purpose of reducing funding for limited term employees, utilities, and supplies relating to the operation of new buildings and campgrounds in state parks, and to decrease the authorized FTE positions for the department of natural resources by 3.0 SEG positions on July 1, 2008, and adjust the NET APPROPRIATION totals accordingly.
- **243.** Page 140, line 16: decrease the dollar amount for fiscal year 2007–08 by \$159,500 and decrease the dollar amount for fiscal year 2008–09 by \$138,500 for the purpose of reducing funding for the operation of Milwaukee Lakeshore State Park, and adjust the NET APPROPRIATION totals accordingly.
  - **244.** Page 141, line 17: after that line insert:
- 15 "(cf) Air management motor vehicle
- emission inspection and mainte-
- nance program, state funds GPR A 46,600 46,600".
- **245.** Page 142, line 9: delete lines 9 to 11.
  - **246.** Page 142, line 25: decrease the dollar amount for fiscal year 2007–08 by \$215,200 and decrease the dollar amount for fiscal year 2008–09 by \$215,300 to decrease funding for the purposes for which the appropriation is made.
    - **247.** Page 143, line 18: decrease the dollar amount for fiscal year 2007–08 by \$85,600 and decrease the dollar amount for fiscal year 2008–09 by \$85,600 for the purpose of eliminating 1.0 vacant GPR position.

<b>248.</b> Page 143, line 18: decrease the dollar amount for fiscal year 2007–08 by
\$6,800 and decrease the dollar amount for fiscal year $2008-09$ by $$6,800$ to decrease
funding for the purpose for which the appropriation is made.
040 5 4444 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5

- **249.** Page 144, line 2: decrease the dollar amount for fiscal year 2007–08 by \$281,800 and decrease the dollar amount for fiscal year 2008–09 by \$282,500 to decrease funding for the purpose for which the appropriation is made and to decrease the authorized FTE positions for the department of natural resources by 2.84 SEG positions.
- **250.** Page 144, line 4: decrease the dollar amount for fiscal year 2007–08 by \$32,400 and decrease the dollar amount for fiscal year 2008–09 by \$32,400 to decrease funding for the purpose for which the appropriation is made and to decrease the authorized FTE positions for the department of natural resources by 0.26 SEG position.
  - **251.** Page 144, line 9: after that line insert:
- 15 "(ad) Law enforcement car kill deer;
- 16 general fund GPR A 514,600 514,600".
- 252. Page 144, line 16: decrease the dollar amount for fiscal year 2007-08 by \$2,600 and decrease the dollar amount for fiscal year 2008-09 by \$2,600 for the purpose of reducing funding for warden radios.
  - **253.** Page 144, line 18: decrease the dollar amount for fiscal year 2007–08 by \$1,200 and decrease the dollar amount for fiscal year 2008–09 by \$1,200 for the purpose of reducing funding for warden radios.

 $\mathbf{2}$ 

- **254.** Page 145, line 3: decrease the dollar amount for fiscal year 2007–08 by \$200 and decrease the dollar amount for fiscal year 2008–09 by \$200 for the purpose of reducing funding for warden radios.
  - **255.** Page 145, line 4: delete lines 4 and 5.
- **256.** Page 145, line 20: decrease the dollar amount for fiscal year 2007–08 by \$340,800 and decrease the dollar amount for fiscal year 2008–09 by \$340,800 for the purpose of eliminating 5.0 vacant GPR positions.
- **257.** Page 145, line 20: decrease the dollar amount for fiscal year 2007–08 by \$138,400 and decrease the dollar amount for fiscal year 2008–09 by \$138,400 to decrease funding for the purposes for which the appropriation is made.
- **258.** Page 146, line 4: decrease the dollar amount for fiscal year 2007–08 by \$104,900 and decrease the dollar amount for fiscal year 2008–09 by \$105,800 to decrease funding for the purpose for which the appropriation is made and to decrease the authorized FTE positions for the department of natural resources by 0.71 SEG position.
- **259.** Page 146, line 4: decrease the dollar amount for fiscal year 2007–08 by \$1,000 and decrease the dollar amount for fiscal year 2008–09 by \$1,000 for the purpose of reducing funding for warden radios.
- **260.** Page 146, line 4: decrease the dollar amount for fiscal year 2007–08 by \$82,500 and decrease the dollar amount for fiscal year 2008–09 by \$82,500 for the purpose of reducing funding for limited term employees for environmental enforcement activities.

- **261.** Page 146, line 6: decrease the dollar amount for fiscal year 2007–08 by \$200 and decrease the dollar amount for fiscal year 2008–09 by \$200 for the purpose of reducing funding for warden radios.
  - **262.** Page 146, line 13: decrease the dollar amount for fiscal year 2007–08 by \$16,600 and decrease the dollar amount for fiscal year 2008–09 by \$16,600 for the purpose of reducing funding for warden radios.
  - **263.** Page 147, line 8: decrease the dollar amount for fiscal year 2007–08 by \$8,100 and decrease the dollar amount for fiscal year 2008–09 by \$8,100 to decrease funding for the purpose for which the appropriation is made.
  - **264.** Page 148, line 18: decrease the dollar amount for fiscal year 2007–08 by \$146,800 and decrease the dollar amount for fiscal year 2008–09 by \$146,800 for the purpose of eliminating 2.0 vacant GPR positions, and adjust the NET APPROPRIATION totals accordingly.
  - **265.** Page 148, line 18: decrease the dollar amount for fiscal year 2007–08 by \$141,500 and decrease the dollar amount for fiscal year 2008–09 by \$141,500 to decrease funding for the purpose for which the appropriation is made, and adjust the NET APPROPRIATION totals accordingly.
  - **266.** Page 148, line 20: decrease the dollar amount for fiscal year 2007–08 by \$6,500 and decrease the dollar amount for fiscal year 2008–09 by \$6,500 for the purpose of eliminating 0.1 vacant GPR position, and adjust the NET APPROPRIATION totals accordingly.
  - **267.** Page 148, line 20: decrease the dollar amount for fiscal year 2007–08 by \$62,300 and decrease the dollar amount for fiscal year 2008–09 by \$62,300 to

decrease funding for the purpose for which the appropriation is made, and adjust the NET APPROPRIATION totals accordingly.

**268.** Page 148, line 21: decrease the dollar amount for fiscal year 2007–08 by \$24,100 and decrease the dollar amount for fiscal year 2008–09 by \$24,100 for the purpose of eliminating 0.5 vacant GPR position, and adjust the NET APPROPRIATION totals accordingly.

**269.** Page 148, line 21: decrease the dollar amount for fiscal year 2007–08 by \$193,800 and decrease the dollar amount for fiscal year 2008–09 by \$193,800 to decrease funding for the purpose for which the appropriation is made, and adjust the NET APPROPRIATION totals accordingly.

**270.** Page 148, line 22: decrease the dollar amount for fiscal year 2007–08 by \$294,600 and decrease the dollar amount for fiscal year 2008–09 by \$294,600 to decrease funding for the purpose for which the appropriation is made, and adjust the NET APPROPRIATION totals accordingly.

**271.** Page 149, line 13: decrease the dollar amount for fiscal year 2007–08 by \$106,100 and decrease the dollar amount for fiscal year 2008–09 by \$106,600 to decrease funding for the purpose for which the appropriation is made and to decrease the authorized FTE positions for the department of natural resources by 0.90 SEG position, and adjust the NET APPROPRIATION totals accordingly.

**272.** Page 149, line 14: decrease the dollar amount for fiscal year 2007–08 by \$204,100 and decrease the dollar amount for fiscal year 2008–09 by \$205,000 to decrease funding for the purpose for which the appropriation is made and to decrease the authorized FTE positions for the department of natural resources by 1.72 SEG positions, and adjust the NET APPROPRIATION totals accordingly.

22

- **273.** Page 150, line 16: delete that line. 1 **274.** Page 151, line 13: after that line insert: 2 3 "(az) Resource aids — interpretive 4 25,000 SEG -0-". center Α 5 **275.** Page 152. line 11: delete lines 11 and 12. **276.** Page 152, line 16: decrease the dollar amount for fiscal year 2007–08 by 6 7 \$1,500,000 and decrease the dollar amount for fiscal year 2008-09 by \$2,500,000 for 8 the purpose for which the appropriation is made. 9 **277.** Page 152, line 25: increase the dollar amount for fiscal year 2007–08 by 10 \$10,000 for the purpose of providing grants to counties and municipalities 11 participating in the lightweight utility vehicle pilot program. 12 **278.** Page 153, line 18: decrease the dollar amount for fiscal year 2007–08 by 13 \$80,000 for the purpose of decreasing the amount of law enforcement aids to 14 counties. 15 **279.** Page 154, line 12: increase the dollar amount for fiscal year 2007–08 by 16 \$1,500,000 and increase the dollar amount for fiscal year 2008-09 by \$2,500,000 for 17 the purpose for which the appropriation is made. **280.** Page 155, line 17: decrease the dollar amount for fiscal year 2007–08 by 18 \$25,900 and decrease the dollar amount for fiscal year 2008-09 by \$25,900 to 19 20 decrease funding for the purpose for which the appropriation is made.
  - **281.** Page 156, line 12: decrease the dollar amount for fiscal year 2007–08 by \$149,800 and decrease the dollar amount for fiscal year 2008–09 by \$149,800 to decrease funding for the purpose for which the appropriation is made.

 $\mathbf{2}$ 

3

4

5

9

10

11

12

13

14

15

16

17

18

19

20

21

22

- **282.** Page 156, line 14: decrease the dollar amount for fiscal year 2007–08 by \$44,100 and decrease the dollar amount for fiscal year 2008–09 by \$44,100 to decrease funding for the purpose for which the appropriation is made.
  - **283.** Page 159, line 18: after that line insert:
- "(mc) Resource maintenance and
- 6 development state park, forest
- 7 and riverway roads

- GPR
- $\mathbf{C}$
- 321,400

321,400".

- 8 **284.** Page 159, line 23: delete lines 23 to 25.
  - **285.** Page 160, line 4: decrease the dollar amount for fiscal year 2007–08 by \$31,900 and decrease the dollar amount for fiscal year 2008–09 by \$31,900 for the purpose of reducing funding for rental costs for department of natural resources facilities.
  - **286.** Page 160, line 6: decrease the dollar amount for fiscal year 2007–08 by \$65,000 and decrease the dollar amount for fiscal year 2008–09 by \$65,000 for the purpose of eliminating 1.0 vacant GPR position.
  - **287.** Page 160, line 6: increase the dollar amount for fiscal year 2007–08 by \$434,500 and increase the dollar amount for fiscal year 2008–09 by \$434,500 to increase funding for the purpose for which the appropriation is made.
  - **288.** Page 160, line 14: decrease the dollar amount for fiscal year 2007–08 by \$56,500 and decrease the dollar amount for fiscal year 2008–09 by \$56,500 for the purpose of reducing funding for rental costs for department of natural resources facilities.

- **289.** Page 160, line 19: decrease the dollar amount for fiscal year 2007–08 by \$140,900 and decrease the dollar amount for fiscal year 2008–09 by \$140,900 for the purpose of reducing funding for rental costs for department of natural resources facilities.
- **290.** Page 161, line 2: decrease the dollar amount for fiscal year 2007–08 by \$93,900 and decrease the dollar amount for fiscal year 2008–09 by \$96,100 to decrease funding for the purpose for which the appropriation is made.
- **291.** Page 161, line 2: decrease the dollar amount for fiscal year 2007–08 by \$85,400 and decrease the dollar amount for fiscal year 2008–09 by \$85,400 for the purpose of reducing funding for rental costs for department of natural resources facilities.
- **292.** Page 162, line 9: decrease the dollar amount for fiscal year 2007–08 by \$34,200 and decrease the dollar amount for fiscal year 2008–09 by \$34,200 for the purpose of eliminating 0.4 vacant GPR position.
- **293.** Page 162, line 9: increase the dollar amount for fiscal year 2007–08 by \$89,500 and increase the dollar amount for fiscal year 2008–09 by \$89,500 to increase funding for the purposes for which the appropriation is made.
- **294.** Page 163, line 2: decrease the dollar amount for fiscal year 2007–08 by \$70,200 and decrease the dollar amount for fiscal year 2008–09 by \$70,300 to decrease funding for the purpose for which the appropriation is made and to decrease the authorized FTE positions for the department of natural resources by 0.37 SEG position.

1	<b>295.</b> Page 164, line 11: decrease the dollar amount for fiscal year 2007–08 by
2	\$8,800 and decrease the dollar amount for fiscal year 2008-09 by \$8,800 for the
3	purpose of eliminating 0.2 vacant GPR position.
4	<b>296.</b> Page 170, line 7: decrease the dollar amount for fiscal year 2007–08 by
5	\$122,700 and decrease the dollar amount for fiscal year 2008-09 by \$126,000 to
6	decrease funding for the purpose for which the appropriation is made.
7	<b>297.</b> Page 170, line 11: decrease the dollar amount for fiscal year 2007–08 by
8	\$1,000,000 for the purpose of reducing funding for preliminary engineering for the
9	Kenosha-Racine-Milwaukee commuter rail extension project.
10	<b>298.</b> Page 170, line 20: delete lines 20 and 21.
11	<b>299.</b> Page 173, line 1: delete lines 1 to 6.
12	<b>300.</b> Page 173, line 20: after that line insert:
13	"(ov) Bicycle and pedestrian facilities,
14	local funds SEG-L C -00-
15	(ox) Bicycle and pedestrian facilities,
16	federal funds SEG-F C $-0 -0-$ ".
17	<b>301.</b> Page 174, line 11: increase the dollar amount for fiscal year 2007-08 by
18	\$19,011,100 and increase the dollar amount for fiscal year $2008-09$ by $$20,668,400$
19	for the purpose of increasing funding for the major highway development program.
20	<b>302.</b> Page 175, line 2: increase the dollar amount for fiscal year 2008–09 by
21	\$238,300 for the purpose of increasing funding for state highway rehabilitation.
22	<b>303.</b> Page 175, line 2: increase the dollar amount for fiscal year 2007-08 by

11,000,000 for the purposes for which the appropriation is made.

- **304.** Page 176, line 2: decrease the dollar amount for fiscal year 2007–08 by \$232,000 for the purpose of reducing funding for printing state highway maps.
  - **305.** Page 178, line 10: increase the dollar amount for fiscal year 2007–08 by \$249,700 and increase the dollar amount for fiscal year 2008–09 by \$249,700 for the purpose of issuing 2 vehicle registration plates for most vehicles.
    - **306.** Page 179, line 3: increase the dollar amount for fiscal year 2007–08 by \$75,000 and increase the dollar amount for fiscal year 2008–09 by \$75,000 for the purposes for which the appropriation is made.
      - **307.** Page 180, line 11: delete lines 11 to 13.
  - **308.** Page 182, line 3: decrease the dollar amount for fiscal year 2007–08 by \$845,800 and decrease the dollar amount for fiscal year 2008–09 by \$845,800 for the purpose of eliminating 10.7 vacant GPR positions.
    - **309.** Page 182, line 3: decrease the dollar amount for fiscal year 2007–08 by \$1,549,300 and decrease the dollar amount for fiscal year 2008–09 by \$1,549,300 for the purpose of decreasing the authorized FTE positions for the department of corrections by 39.0 GPR security positions in each fiscal year.
    - **310.** Page 182, line 3: decrease the dollar amount for fiscal year 2007–08 by \$8,914,200 and decrease the dollar amount for fiscal year 2008–09 by \$8,984,000 for the purpose of decreasing the authorized FTE positions for the department of corrections by 50.0 GPR positions in each fiscal year.
  - **311.** Page 182, line 3: decrease the dollar amount for fiscal year 2007–08 by \$1,565,000 and decrease the dollar amount for fiscal year 2008–09 by \$1,896,500 for the purpose of decreasing funding for the earned release program and decreasing the

- authorized FTE positions for the department of corrections by 31.5 GPR positions in each fiscal year.
- **312.** Page 182, line 3: decrease the dollar amount for fiscal year 2007–08 by \$835,200 and decrease the dollar amount for fiscal year 2008–09 by \$835,200 for the purpose of decreasing funding for inmate wages for education and program assignments.
- **313.** Page 182, line 3: decrease the dollar amount for fiscal year 2007–08 by \$635,700 and decrease the dollar amount for fiscal year 2008–09 by \$635,700 for the purpose of decreasing funding for inmate wages for inmates who are involuntarily unassigned.
- **314.** Page 182, line 3: decrease the dollar amount for fiscal year 2007–08 by \$67,700 and decrease the dollar amount for fiscal year 2008–09 by \$67,700 for the purpose of decreasing funding for postage for inmates.
- **315.** Page 182, line 3: decrease the dollar amount for fiscal year 2007–08 by \$3,631,100 and decrease the dollar amount for fiscal year 2008–09 by \$3,631,100 to decrease the authorized FTE positions for the department of corrections by 40.0 GPR unit manager positions.
- **316.** Page 182, line 3: decrease the dollar amount for fiscal year 2007–08 by \$159,600 and decrease the dollar amount for fiscal year 2008–09 by \$159,600 for the purpose of decreasing funding for an executive assistant salary and fringe benefits.
- **317.** Page 182, line 3: decrease the dollar amount for fiscal year 2007–08 by \$12,600,000 and decrease the dollar amount for fiscal year 2008–09 by \$12,600,000 for the purpose of decreasing administrative funding.

- **318.** Page 182, line 7: increase the dollar amount for fiscal year 2007–08 by \$1,657,400 and increase the dollar amount for fiscal year 2008–09 by \$11,570,300 for the purpose of increasing funding for the earned release program.
  - **319.** Page 182, line 7: increase the dollar amount for fiscal year 2007–08 by \$1,224,200 and increase the dollar amount for fiscal year 2008–09 by \$3,474,800 for the purpose of community alternatives to revocation.
  - **320.** Page 182, line 8: decrease the dollar amount for fiscal year 2007–08 by \$800,300 and decrease the dollar amount for fiscal year 2008–09 by \$800,300 for the purpose of eliminating 13.13 vacant GPR positions.
  - **321.** Page 182, line 8: decrease the dollar amount for fiscal year 2007–08 by \$39,500 and decrease the dollar amount for fiscal year 2008–09 by \$39,500 for the purpose of decreasing full funding for community corrections positions.
  - **322.** Page 182, line 17: decrease the dollar amount for fiscal year 2007–08 by \$1,745,500 and decrease the dollar amount for fiscal year 2008–09 by \$4,040,500 for the purpose of community alternatives to revocation.
  - **323.** Page 184, line 9: decrease the dollar amount for fiscal year 2007–08 by \$2,600 and decrease the dollar amount for fiscal year 2008–09 by \$1,200 for the purpose of which the appropriation is made.
- **324.** Page 184, line 11: decrease the dollar amount for fiscal year 2007–08 by \$148,800 and decrease the dollar amount for fiscal year 2008–09 by \$148,800 for the purpose of eliminating 2.0 vacant GPR positions.

- **325.** Page 185, line 1: decrease the dollar amount for fiscal year 2007–08 by \$5,000,000 and decrease the dollar amount for fiscal year 2008–09 by \$5,000,000 for the purposes for which the appropriation is made.
- **326.** Page 185, line 2: after that line insert:
- 5 "(d) Youth diversion GPR A 380,000 380,000".
- **327.** Page 185, line 12: after that line insert:
- 7 "(k) Youth diversion administration PR-S A 24,800 24,800
- 8 (kj) Youth diversion program PR-S A 794,900 794,900".
- **328.** Page 185, line 19: delete lines 19 and 20.
  - **329.** Page 186, line 8: decrease the dollar amount for fiscal year 2007–08 by \$137,800 and decrease the dollar amount for fiscal year 2008–09 by \$170,000 for the purpose of decreasing the authorized FTE positions for the board on aging and long–term care by 3.2 GPR positions to recruit, train, and supervise volunteers.
  - **330.** Page 186, line 8: decrease the dollar amount for fiscal year 2007–08 by \$20,600 and decrease the dollar amount for fiscal year 2008–09 by \$25,600 for the purpose of decreasing the authorized FTE positions for the board on aging and long-term care by 0.5 GPR positions to provide information and advocacy services to enrollees in the Family Care Program who are over 60 years of age.
  - **331.** Page 186, line 10: decrease the dollar amount for fiscal year 2007–08 by \$35,600 and decrease the dollar amount for fiscal year 2008–09 by \$42,500 for the purpose of decreasing the authorized FTE positions for the board on aging and long-term care by 0.8 PR position to recruit, train, and supervise volunteers.

- **332.** Page 186, line 10: decrease the dollar amount for fiscal year 2007–08 by \$20,600 and decrease the dollar amount for fiscal year 2008–09 by \$25,500 for the purpose of decreasing the authorized FTE positions for the board on aging and long-term care by 0.5 PR position to provide information and advocacy services to enrollees in the Family Care Program who are over 60 years of age.
- **333.** Page 187, line 3: decrease the dollar amount for fiscal year 2007–08 by \$650,400 and decrease the dollar amount for fiscal year 2008–09 by \$789,700 to decrease funding for the purposes for which the appropriation is made and to decrease the authorized FTE positions for the child abuse and neglect prevention board by 1.0 GPR position for the purpose of supporting activities for the prevention of maltreatment of children.
- **334.** Page 188, line 23: decrease the dollar amount for fiscal year 2007–08 by \$27,000 and decrease the dollar amount for fiscal year 2008–09 by \$27,000 to decrease funding for the purpose for which the appropriation is made and to decrease the authorized FTE positions for the department of health and family services by 0.18 SEG position.
- **335.** Page 189, line 2: decrease the dollar amount for fiscal year 2007–08 by \$105,500 and decrease the dollar amount for fiscal year 2008–09 by \$105,500 for the purpose of eliminating 1.9 vacant GPR positions.
- **336.** Page 189, line 7: decrease the dollar amount for fiscal year 2007–08 by \$262,500 and decrease the dollar amount for fiscal year 2008–09 by \$345,500 for the purpose of decreasing funding to expand treatment services.

- **337.** Page 189, line 9: decrease the dollar amount for fiscal year 2007–08 by \$930,000 and decrease the dollar amount for fiscal year 2008–09 by \$930,000 for the purpose of eliminating 8.0 vacant GPR positions.
- **338.** Page 189, line 16: increase the dollar amount for fiscal year 2007–08 by \$544,400 to increase funding for the assessments on licensed beds of intermediate care facilities for the mentally retarded.
- **339.** Page 190, line 8: increase the dollar amount for fiscal year 2008–09 by \$5,758,100 to increase funding for the purpose for which the appropriation is made.
- **340.** Page 190, line 10: increase the dollar amount for fiscal year 2008–09 by \$597,200 to increase funding for the purpose for which the appropriation is made.
- **341.** Page 190, line 10: decrease the dollar amount for fiscal year 2007–08 by \$50,000 and decrease the dollar amount for fiscal year 2008–09 by \$50,000 for the purpose of decreasing funding for the Foster Youth Independence Center in the city of Milwaukee.
- **342.** Page 190, line 12: increase the dollar amount for fiscal year 2008–09 by \$7,150,800 to increase funding for the purpose for which the appropriation is made.
- **343.** Page 190, line 12: decrease the dollar amount for fiscal year 2007–08 by \$706,600 and decrease the dollar amount for fiscal year 2008–09 by \$567,100 to decrease funding for domestic abuse grants.
- **344.** Page 190, line 15: increase the dollar amount for fiscal year 2008–09 by \$60,000 to increase funding for the purpose for which the appropriation is made.
- **345.** Page 190, line 17: increase the dollar amount for fiscal year 2008–09 by \$12,418,700 to increase funding for the purpose for which the appropriation is made.

- **346.** Page 190, line 17: decrease the dollar amount for fiscal year 2007–08 by \$164,900 and decrease the dollar amount for fiscal year 2008–09 by \$164,900 for the purpose of eliminating 2.87 vacant GPR positions.
- **347.** Page 190, line 17: increase the dollar amount for fiscal year 2007–08 by \$75,900 and increase the dollar amount for fiscal year 2008–09 by \$75,900 to increase funding for the purpose for which the appropriation is made.
- **348.** Page 190, line 19: increase the dollar amount for fiscal year 2008–09 by \$42,248,900 to increase funding for the purpose for which the appropriation is made.
- **349.** Page 190, line 19: decrease the dollar amount for fiscal year 2007–08 by \$75,900 and decrease the dollar amount for fiscal year 2008–09 by \$75,900 to decrease funding for the purpose for which the appropriation is made.
- **350.** Page 190, line 19: decrease the dollar amount for fiscal year 2007–08 by \$155,900 and decrease the dollar amount for fiscal year 2008–09 by \$650,200 for the purpose of decreasing funding for monthly payments for foster care for children.
- **351.** Page 191, line 2: increase the dollar amount for fiscal year 2008–09 by \$1,117,200 to increase funding for the purposes for which the appropriation is made.
- **352.** Page 191, line 4: increase the dollar amount for fiscal year 2008–09 by \$50,408,800 to increase funding for the purposes for which the appropriation is made.
- **353.** Page 191, line 4: decrease the dollar amount for fiscal year 2007–08 by \$30,100 and decrease the dollar amount for fiscal year 2008–09 by \$62,400 for the purpose of decreasing funding for monthly payments for foster care for children.

- **354.** Page 191, line 7: increase the dollar amount for fiscal year 2008–09 by \$287,900 to increase funding for the purposes for which the appropriation is made.
  - **355.** Page 191, line 7: decrease the dollar amount for fiscal year 2007–08 by \$109,700 and decrease the dollar amount for fiscal year 2008–09 by \$116,600 to decrease funding for the purposes for which the appropriation is made.
  - **356.** Page 191, line 9: increase the dollar amount for fiscal year 2008–09 by \$1,959,500 to increase funding for the purposes for which the appropriation is made.
  - **357.** Page 191, line 11: delete that line.
  - **358.** Page 191, line 21: increase the dollar amount for fiscal year 2008–09 by \$90,300 to increase funding for the purposes for which the appropriation is made.
  - **359.** Page 191, line 22: increase the dollar amount for fiscal year 2008–09 by \$958,400 to increase funding for the purpose for which the appropriation is made.
  - **360.** Page 192, line 3: increase the dollar amount for fiscal year 2008–09 by \$21,878,300 to increase funding for the purpose for which the appropriation is made.
  - **361.** Page 192, line 5: increase the dollar amount for fiscal year 2008–09 by \$1,464,000 to increase funding for the purpose for which the appropriation is made.
  - **362.** Page 192, line 8: increase the dollar amount for fiscal year 2008–09 by \$21,991,100 to increase funding for the purpose for which the appropriation is made.
  - **363.** Page 193, line 9: decrease the dollar amount for fiscal year 2007–08 by \$28,200 and decrease the dollar amount for fiscal year 2008–09 by \$28,200 for the purpose of eliminating 0.44 vacant GPR position.
  - **364.** Page 193, line 9: decrease the dollar amount for fiscal year 2007-08 by \$457,700 and decrease the dollar amount for fiscal year 2008-09 by \$457,700 to

- increase the authorized FTE positions for the department of health and family services by 10.5 GPR positions on July 1, 2007, to support the client assistance for reemployment and economic support system.
  - **365.** Page 193, line 11: decrease the dollar amount for fiscal year 2007–08 by \$13,800,000 and decrease the dollar amount for fiscal year 2008–09 by \$13,800,000 for the purposes for which the appropriation is made.
- **366.** Page 193, line 11: decrease the dollar amount for fiscal year 2008–09 by \$4,340,400 for the purpose of transportation services.
  - **367.** Page 193, line 11: increase the dollar amount for fiscal year 2007–08 by \$3,150,000 and decrease the dollar amount for fiscal year 2008–09 by \$3,794,600 for the purpose of eliminating the assistance for childless adults demonstration project.
  - **368.** Page 193, line 11: increase the dollar amount for fiscal year 2007–08 by \$362,900,000 and increase the dollar amount for fiscal year 2008–09 by \$370,902,800 for the purposes for which the appropriation is made.
    - **369.** Page 193, line 11: decrease the dollar amount for fiscal year 2007–08 by \$50,000,000 and decrease the dollar amount for fiscal year 2008–09 by \$50,000,000 for the purposes for which the appropriation is made.
    - **370.** Page 193, line 11: increase the dollar amount for fiscal year 2007–08 by \$57,909,700 and increase the dollar amount for fiscal year 2008–09 by \$61,750,200 for the purposes for which the appropriation is made.
    - **371.** Page 193, line 11: decrease the dollar amount for fiscal year 2007–08 by \$746,800 and decrease the dollar amount for fiscal year 2008–09 by \$1,493,500 for the purpose of not requiring Medical Assistance recipients who are eligible for, but

 $\mathbf{2}$ 

- not enrolled in, Medicare Part B to enroll in Medicare Part B and for providing benefits to those recipients.
  - **372.** Page 193, line 11: decrease the dollar amount for fiscal year 2007–08 by \$700,700 and decrease the dollar amount for fiscal year 2008–09 by \$5,031,800 for the purpose of maintaining current law for the Family Care Program.
  - **373.** Page 193, line 11: increase the dollar amount for fiscal year 2007–08 by \$2,315,200 and decrease the dollar amount for fiscal year 2008–09 by \$3,225,800 to reflect the elimination of the BadgerCare Plus Medical Assistance program.
  - **374.** Page 193, line 11: decrease the dollar amount for fiscal year 2007–08 by \$6,877,600 and decrease the dollar amount for fiscal year 2008–09 by \$14,627,000 to reflect savings from administrative efficiencies.
  - **375.** Page 193, line 11: increase the dollar amount for fiscal year 2008–09 by \$16,268,800 for the purpose of capping the nursing home assessment at \$75 per bed per month, for eliminating any nursing home reimbursement rate increase in fiscal year 2007–08, and for increasing the nursing home reimbursement rate by 5 percent for fiscal year 2008–09.
  - **376.** Page 193, line 11: increase the dollar amount for fiscal year 2008–09 by \$11,000,000 for the purpose of increasing reimbursement rates to hospitals for providing services to Medical Assistance and BadgerCare recipients.
  - **377.** Page 193, line 11: increase the dollar amount for fiscal year 2008–09 by \$3,000,000 for the purpose of increasing the reimbursement rate to dentists for services provided to Medical Assistance and BadgerCare recipients.

- **378.** Page 193, line 11: decrease the dollar amount for fiscal year 2007–08 by \$5,814,200 and decrease the dollar amount for fiscal year 2008–09 by \$17,240,700 for the purpose of eliminating rate increases to noninstitutional providers for services provided to Medical Assistance and BadgerCare recipients.
- **379.** Page 193, line 11: decrease the dollar amount for fiscal year 2007–08 by \$1,750,000 and decrease the dollar amount for fiscal year 2008–09 by \$1,750,000 for the purpose of eliminating a pilot project to increase the reimbursement rate to dentists for care provided to Medical Assistance and BadgerCare recipients in Brown, La Crosse, and Racine counties.
- **380.** Page 193, line 11: increase the dollar amount for fiscal year 2008-09 by \$375,400 to delete expansion of the medical assistance family planning demonstration project.
- **381.** Page 193, line 14: increase the dollar amount for fiscal year 2007–08 by \$26,200 and increase the dollar amount for fiscal year 2008–09 by \$26,200 for the purpose of a transportation services broker program.
- **382.** Page 193, line 14: decrease the dollar amount for fiscal year 2007–08 by \$193,900 and increase the dollar amount for fiscal year 2008–09 by \$1,104,500 to reflect the elimination of the BadgerCare Plus Medical Assistance program.
- **383.** Page 193, line 14: decrease the dollar amount for fiscal year 2007–08 by \$306,000 and decrease the dollar amount for fiscal year 2008–09 by \$1,174,200 to reflect savings from administrative efficiencies.

- **384.** Page 193, line 14: decrease the dollar amount for fiscal year 2007–08 by \$516,000 and decrease the dollar amount for fiscal year 2008–09 by \$1,240,000 to decrease funding for a vending contract for food stamp program benefits.
  - **385.** Page 193, line 14: decrease the dollar amount for fiscal year 2008–09 by \$487,000 to decrease funding for Medical Assistance claims processing.
    - **386.** Page 194, line 1: increase the dollar amount for fiscal year 2008-09 by \$850,600 for the purpose of eliminating the assistance for childless adults demonstration project.
- **387.** Page 194, line 3: after that line insert:
- 10 "(gp) Medical assistance; hospital
- 11 assessments PR C 1,500,000 1,500,000".
- **388.** Page 194, line 4: delete "or medical".
- **389.** Page 194, line 23: delete lines 23 and 24.
  - **390.** Page 195, line 3: decrease the dollar amount for fiscal year 2007–08 by \$4,284,700 and decrease the dollar amount for fiscal year 2008–09 by \$19,675,400 to reflect the elimination of the BadgerCare Plus Medical Assistance program.
  - **391.** Page 196, line 5: increase the dollar amount for fiscal year 2007–08 by \$13,800,000 and increase the dollar amount for fiscal year 2008–09 by \$13,800,000 for the purposes for which the appropriation is made.
  - **392.** Page 196, line 5: increase the dollar amount for fiscal year 2007–08 by \$50,000,000 and increase the dollar amount for fiscal year 2008–09 by \$50,000,000 for the purposes for which the appropriation is made.

1	<b>393.</b> Page 196, line 5: decrease the dollar amount for fiscal year 2007–08 by					
2	\$7,528,100 and decrease the dollar amount for fiscal year 2008-09 by \$14,318,800					
3	for the purpose of capping the nursing home assessment at \$75 per bed per mon					
4	for eliminating any nursing home reimbursement rate increase in fiscal year					
5	2007-08, and for increasing the nursing home reimbursement rate by 5 percent					
6	fiscal year 2008–09.					
7	<b>394.</b> Page 196, line 12: delete lines 12 to 15.					
8	<b>395.</b> Page 196, line 16: delete lines 16 and 17.					
9	<b>396.</b> Page 196, line 18: delete lines 18 to 23.					
10	<b>397.</b> Page 197, line 6: increase the dollar amount for fiscal year 2008–09 by					
11	\$995,700 to increase funding for the purpose for which the appropriation is made.					
12	<b>398.</b> Page 197, line 9: decrease the dollar amount for fiscal year 2007–08 by					
13	\$400,000 and decrease the dollar amount for fiscal year 2008-09 by \$876,600 to					
14	decrease funding for the Mike Johnson life care and early intervention services					
15	grants.					
16	<b>399.</b> Page 197, line 10: increase the dollar amount for fiscal year 2007–08 by					
17	\$37,500 and increase the dollar amount for fiscal year 2008-09 by \$37,500 for the					
18	purpose for which the appropriation is made.					
19	<b>400.</b> Page 197, line 13: after that line insert:					
20	"(ch) Emergency medical services;					
21	aids GPR A 2,200,000 2,200,000".					

**401.** Page 197, line 13: after that line insert:

- 1 "(cf) Colposcopies and other services GPR A 37,500".
- **402.** Page 197, line 17: increase the dollar amount for fiscal year 2008–09 by \$320,000 to increase funding for the purpose for which the appropriation is made.
- **403.** Page 198, line 9: increase the dollar amount for fiscal year 2008–09 by \$179,300 to increase funding for the purpose for which the appropriation is made.
- 6 **404.** Page 198, line 10: delete lines 10 and 11.
- 7 **405.** Page 198, line 15: decrease the dollar amount for fiscal year 2008–09 by \$3,000,000 for the purpose for which the appropriation is made.
  - **406.** Page 198, line 18: increase the dollar amount for fiscal year 2007–08 by \$2,500,000 and increase the dollar amount for fiscal year 2008–09 by \$2,500,000 for the purpose for which the appropriation is made.
  - **407.** Page 199, line 5: delete lines 5 and 6.
- 13 **408.** Page 199, line 7: delete that line.

10

11

12

14

15

16

17

18

19

20

21

- **409.** Page 199, line 10: increase the dollar amount for fiscal year 2008–09 by \$15,000 to increase funding for the operations of the council on developmental disabilities.
  - **410.** Page 199, line 10: decrease the dollar amount for fiscal year 2007–08 by \$89,800 and decrease the dollar amount for fiscal year 2008–09 by \$89,800 for the purpose of eliminating 1.63 vacant GPR positions.
- **411.** Page 199, line 10: decrease the dollar amount for fiscal year 2007–08 by \$167,600 and decrease the dollar amount for fiscal year 2008–09 by \$275,500 for the purpose of maintaining current law for the Family Care Program.

- **412.** Page 201, line 3: increase the dollar amount for fiscal year 2008–09 by \$28,959,400 to increase funding for services for children and families, for court-attached intake services, and for shelter care, foster care, treatment foster care, and subsidized guardianship care for children.
- **413.** Page 201, line 3: increase the dollar amount for fiscal year 2007–08 by \$7,278,100 and decrease the dollar amount for fiscal year 2008–09 by \$6,310,700 for the purpose of maintaining current law for the Family Care Program.
- **414.** Page 201, line 3: decrease the dollar amount for fiscal year 2007–08 by \$240,000 and decrease the dollar amount for fiscal year 2008–09 by \$477,100 for the purpose of decreasing funding for monthly payments for foster care for children.
- **415.** Page 201, line 4: decrease the dollar amount for fiscal year 2007–08 by \$106,400 and decrease the dollar amount for fiscal year 2008–09 by \$106,400 to decrease funding for the female offender reintegration program.
- **416.** Page 201, line 4: decrease the dollar amount for fiscal year 2007–08 by \$167,000 and decrease the dollar amount for fiscal year 2008–09 by \$167,000 to decrease funding for the quality home care program.
- **417.** Page 201, line 4: decrease the dollar amount for fiscal year 2007–08 by \$250,000 and decrease the dollar amount for fiscal year 2008–09 by \$250,000 to decrease funding for a grant in support of family-centered managed care for children with long-term support needs.
- **418.** Page 202, line 7: delete lines 7 to 8.
  - **419.** Page 203, line 16: increase the dollar amount for fiscal year 2008–09 by \$349,300 to increase funding for the purpose for which the appropriation is made.

- **420.** Page 203, line 16: decrease the dollar amount for fiscal year 2007–08 by \$143,000 and decrease the dollar amount for fiscal year 2008–09 by \$143,000 for the purpose of eliminating 2.04 vacant GPR positions.
  - **421.** Page 203, line 16: decrease the dollar amount for fiscal year 2007–08 by \$43,300 to decrease funding for supplies and services.
  - **422.** Page 203, line 16: decrease the dollar amount for fiscal year 2007–08 by \$227,300 and decrease the dollar amount for fiscal year 2008–09 by \$302,200 to increase the authorized FTE positions for the department of health and family services by 23.0 GPR positions on July 1, 2007, for the performance of information and technology services.
  - **423.** Page 203, line 16: decrease the dollar amount for fiscal year 2007–08 by \$114,100 and decrease the dollar amount for fiscal year 2008–09 by \$114,100 to decrease the authorized FTE positions for the department of health and family services by 1.5 GPR positions on July 1, 2007, for the performance of strategic finance analysis.
  - **424.** Page 203, line 19: increase the dollar amount for fiscal year 2008–09 by \$132,200 to increase funding for the purpose for which the appropriation is made.
  - **425.** Page 205, line 1: delete lines 1 to 24.
  - **426.** Page 206, line 1: delete lines 1 to 24.
- **427.** Page 207, line 1: delete lines 1 to 20.
- **428.** Page 208, line 1: delete lines 1 to 22.
- **429.** Page 209, line 1: delete lines 1 to 23.
- **430.** Page 210, line 1: delete lines 1 to 18.

3

4

5

6

11

18

19

20

21

22

- 1 **431.** Page 211, line 1: delete lines 1 to 21.
  - **432.** Page 212, line 8: decrease the dollar amount for fiscal year 2007–08 by \$87,600 and decrease the dollar amount for fiscal year 2008–09 by \$116,800 to decrease funding for the youth apprenticeship program and to decrease the authorized FTE positions for the department of workforce development by 2.0 GPR positions for the purpose of administering that program.
- 7 **433.** Page 213, line 3: delete lines 3 and 4.
- 434. Page 213, line 5: decrease the dollar amount for fiscal year 2007-08 by \$325,000 and decrease the dollar amount for fiscal year 2008-09 by \$1,100,000 to decrease funding for the purpose for which the appropriation is made.
  - **435.** Page 213, line 9: after that line insert:
- 12 "(fg) Employment transit aids, state
- 13 funds GPR A 550,100 550,100".
- **436.** Page 213, line 10: delete lines 10 to 12.
- 15 **437.** Page 215, line 14: delete lines 14 and 15.
- 16 **438.** Page 216, line 5: decrease the dollar amount for fiscal year 2007–08 by \$150,000 to reduce funding for the purposes for which the appropriation is made.
  - **439.** Page 216, line 5: increase the dollar amount for fiscal year 2008–09 by \$5,072,300 to increase funding for the purposes for which the appropriation is made.
  - **440.** Page 216, line 7: increase the dollar amount for fiscal year 2008–09 by \$28,849,400 to increase funding for the purpose for which the appropriation is made.
  - **441.** Page 216, line 9: increase the dollar amount for fiscal year 2008–09 by \$237,500 to increase funding for the purpose for which the appropriation is made.

 $\mathbf{2}$ 

- **442.** Page 216, line 11: increase the dollar amount for fiscal year 2008–09 by \$120,971,700 to increase funding for the purposes for which the appropriation is made.
  - **443.** Page 216, line 12: delete that line.
- **444.** Page 217, line 4: increase the dollar amount for fiscal year 2007–08 by \$8,317,600 and increase the dollar amount for fiscal year 2008–09 by \$8,317,600 for the purposes of administering the food stamp employment and training program, as a result of not transferring administration of the program to the department of health and family services, and increasing the authorized FTE positions for the department of workforce development by 1.71 PR-S positions for administration of the program.
- **445.** Page 217, line 9: decrease the dollar amount for fiscal year 2007–08 by \$487,900 and decrease the dollar amount for fiscal year 2008–09 by \$521,000 to reduce funding for the purposes for which the appropriation is made.
- **446.** Page 217, line 9: increase the dollar amount for fiscal year 2008–09 by \$18,743,100 to increase funding for the purpose for which the appropriation is made.
- **447.** Page 217, line 10: increase the dollar amount for fiscal year 2007–08 by \$487,900 and increase the dollar amount for fiscal year 2008–09 by \$521,000 to increase funding for the purposes for which the appropriation is made.
- **448.** Page 217, line 10: increase the dollar amount for fiscal year 2008–09 by \$365,197,900 to increase funding for the purposes for which the appropriation is made.

1	<b>449.</b> Page 218, line 6: increase the dollar amount for fiscal year 2008–09 by
2	\$9,232,000 to increase funding for the purpose for which the appropriation is made.
3	450. Page 218, line 9: decrease the dollar amount for fiscal year 2007–08 by
4	\$26,400 and decrease the dollar amount for fiscal year 2008-09 by \$26,400 for the
5	purpose of eliminating 0.43 vacant GPR position.
6	<b>451.</b> Page 220, line 8: increase the dollar amount for fiscal year 2007–08 by
7	\$220,900 and increase the dollar amount for fiscal year $2008-09$ by $$220,900$ for the
8	purpose of transferring duties of the office of justice assistance to the department of
9	justice.
10	${f 452.}$ Page 220, line 8: decrease the dollar amount for fiscal year 2007–08 by
11	$$40,\!200$ and decrease the dollar amount for fiscal year $2008-09$ by $$40,\!200$ for the
12	purpose of eliminating 1.0 vacant GPR position.
13	<b>453.</b> Page 220, line 14: after that line insert:
14	"(cr) Law enforcement officer supple-
15	ment grants $GPR   A   1,000,000   1,000,000$ ".
16	<b>454.</b> Page 220, line 18: decrease the dollar amount for fiscal year 2007–08 by
L7	\$250,000 and decrease the dollar amount for fiscal year $2008-09$ by $$250,000$ for the
18	purpose of reducing grants to community policing.
19	<b>455.</b> Page 220, line 18: after that line insert:
20	"(em) Grants for substance abuse
21	treatment programs for criminal
22	offenders; general purpose reve-
23	nue GPR A -00-".

1	<b>456.</b> Page 221, line 19: after that	line inse	ert:		
2	"(kb) Grants for digital recording of				
3	custodial interrogations	PR-S	A	750,000	750,000".
4	<b>457.</b> Page 222, line 14: after that	line inse	rt:		
5	"(kx) Grants for substance abuse				
6	treatment programs for criminal				
7	offenders; program revenue	PR	$\mathbf{C}$	755,000	755,000
8	(ky) Law enforcement programs				
9	administration	PR-S	A	179,700	179,700".
10	<b>458.</b> Page 222, line 17: increase the	he dollar	amount	for fiscal year	2007-08 by
11	\$2,846,300 and increase the dollar amoun	nt for fisc	al year 2	2008–09 by \$2	2,428,300 for
12	the purpose of transferring duties of the o	office of ju	stice ass	istance to the	department
13	of justice.				
14	<b>459.</b> Page 222, line 18: increase the	he dollar	amount	for fiscal year	2007-08 by
15	\$18,904,900 and increase the dollar amo	unt for fi	scal year	r 2008-09 by	\$18,904,900
16	for the purpose of transferring duties	of the of	ffice of	justice assist	ance to the
17	department of justice.				
18	<b>460.</b> Page 224, line 3: after that l	ine inser	t:		
19	"(gj) Grants for victims of sexual				
20	assault; child pornography sur-				
21	charge	PR	$\mathbf{C}$	-0-	-0-".
22	<b>461.</b> Page 224, line 5: delete that	line.			

County by 0.1 GPR district attorney.

**462.** Page 225, line 3: decrease the dollar amount for fiscal year 2007–08 by 1  $\mathbf{2}$ \$180,600 and decrease the dollar amount for fiscal year 2008-09 by \$180,600 for the 3 purpose of eliminating 4.24 vacant GPR positions. 4 **463.** Page 226, line 5: after that line insert: 5 "(dd) Regional emergency response 6 teams **GPR** A 1,400,000 1,400,000 7 (dp) Emergency response equipment GPR 468,000 468,000 A 8 (dr)Emergency response supplement GPR  $\mathbf{C}$ -0--0-9 Emergency response training 64,900 64,900". (dt) GPR В **464.** Page 226, line 7: after that line insert: 10 19,000". 11 "(f) Civil air patrol aids GPR A 19,000 **465.** Page 226, line 19: after that line insert: 12 13 "(mb) Federal aid. homeland security PR-F  $\mathbf{C}$ 36,729,400 36,584,300 ". 14 15 **466.** Page 227, line 10: decrease the dollar amount for fiscal year 2007–08 by \$700 and decrease the dollar amount for fiscal year 2008-09 by \$700 to decrease 16 17 funding for the purpose for which the appropriation is made. **467.** Page 227, line 11: delete lines 11 to 16. 18 **468.** Page 228, line 9: increase the dollar amount for fiscal year 2007–08 by 19 20 \$12,300 and increase the dollar amount for fiscal year 2008-09 by \$12,500 for the 21purpose of increasing the authorized FTE district attorney positions in Vernon

 $\mathbf{2}$ 

- **469.** Page 228, line 10: increase the dollar amount for fiscal year 2007–08 by \$16,700 and increase the dollar amount for fiscal year 2008–09 by \$16,700 for the purpose of increasing the authorized FTE positions for the department of administration by 0.25 PR assistant district attorney position for Chippewa County beginning July 1, 2007.
- **470.** Page 228, line 10: increase the dollar amount for fiscal year 2007–08 by \$32,400 and increase the dollar amount for fiscal year 2008–09 by \$64,800 for the purpose of increasing the authorized FTE positions for the department of administration by 1.0 PR assistant district attorney position for St. Croix County beginning January 1, 2008.
- **471.** Page 231, line 21: increase the dollar amount for fiscal year 2007–08 by \$35,000 for the purpose of funding a study under 2007 Wisconsin Act .... (this act), section 9153 (3g).
- **472.** Page 237, line 3: decrease the dollar amount for fiscal year 2007–08 by \$80,700 and decrease the dollar amount for fiscal year 2008–09 by \$80,700 for the purpose of eliminating 1.0 vacant GPR position.
- **473.** Page 237, line 3: decrease the dollar amount for fiscal year 2007–08 by \$230,400 and decrease the dollar amount for fiscal year 2008–09 by \$230,400 to decrease the authorized FTE positions for the department by 2.0 GPR positions associated with the federal-state relations office.
  - **474.** Page 237, line 13: delete that line.

- **475.** Page 237, line 15: decrease the dollar amount for fiscal year 2007–08 by \$109,500 and decrease the dollar amount for fiscal year 2008–09 by \$109,500 for the purpose for which the appropriation is made.
- **476.** Page 239, line 1: decrease the dollar amount for fiscal year 2007–08 by \$120,100 and decrease the dollar amount for fiscal year 2008–09 by \$120,100 for the purpose of decreasing the authorized FTE positions for the department of administration by 1.0 PR information technology manager position on loan to the office of justice assistance.
- **477.** Page 239, line 3: decrease the dollar amount for fiscal year 2007–08 by \$70,900 and decrease the dollar amount for fiscal year 2008–09 by \$70,900 for the purpose of decreasing the authorized FTE positions for the department by 1.0 PR records management position.
  - **478.** Page 242, line 3: delete lines 3 and 4.
- **479.** Page 245, line 12: delete the material beginning with that line and ending with page 246, line 15.
  - **480.** Page 250, line 3: increase the dollar amount for fiscal year 2007–08 by \$50,000 and increase the dollar amount for fiscal year 2008–09 by \$100,000 to increase funding for the purpose for which the appropriation is made.
  - **481.** Page 253, line 7: decrease the dollar amount for fiscal year 2007–08 by \$55,400 and decrease the dollar amount for fiscal year 2008–09 by \$72,600 for the purpose of the labor management cooperation program and for eliminating 1.0 4-year project position.

 $\mathbf{2}$ 

- **482.** Page 253, line 12: decrease the dollar amount for fiscal year 2007–08 by \$62,500 and decrease the dollar amount for fiscal year 2008–09 by \$62,500 for the purpose of eliminating 1.0 vacant GPR position.
- **483.** Page 253, line 14: decrease the dollar amount for fiscal year 2007–08 by \$900 and decrease the dollar amount for fiscal year 2008–09 by \$900 for the purpose of eliminating 0.03 vacant GPR position.
- **484.** Page 253, line 16: decrease the dollar amount for fiscal year 2008–09 by \$182,100 for the purpose of maintaining current law regarding eligibility for representation by the state public defender.
- **485.** Page 253, line 18: decrease the dollar amount for fiscal year 2007–08 by \$18,800 and decrease the dollar amount for fiscal year 2008–09 by \$18,800 for the purpose of eliminating 0.47 vacant GPR position.
- **486.** Page 254, line 13: decrease the dollar amount for fiscal year 2007–08 by \$209,600 and decrease the dollar amount for fiscal year 2008–09 by \$209,600 for the purpose of eliminating 3.65 vacant GPR positions.
- **487.** Page 256, line 3: decrease the dollar amount for fiscal year 2007–08 by \$200,500 and decrease the dollar amount for fiscal year 2008–09 by \$187,300 for the purpose of reducing the department of revenue's authorized positions by 3.0 FTE auditor positions for the oil company assessment.
- **488.** Page 256, line 5: decrease the dollar amount for fiscal year 2007–08 by \$22,300 and decrease the dollar amount for fiscal year 2008–09 by \$22,300 for the purpose of eliminating 0.6 vacant GPR position.

- **489.** Page 256, line 5: decrease the dollar amount for fiscal year 2007-08 by \$113,100 and decrease the dollar amount for fiscal year 2008-09 by \$113,100 for the purpose of reducing the department of revenue's authorized positions by 1.0 FTE position.
- 5 **490.** Page 256, line 6: delete lines 6 and 7.
- 491. Page 261, line 6: decrease the dollar amount for fiscal year 2007–08 by \$134,400 and decrease the dollar amount for fiscal year 2008–09 by \$125,500 for the purpose for which the appropriation is made.
- 9 **492.** Page 262, line 2: delete that line.
- 10 **493.** Page 267, line 7: delete lines 7 and 8.
- 11 **494.** Page 268, line 1: delete lines 1 and 2.
- 12 **495.** Page 268, line 4: delete "Municipal" and substitute "County and municipal".
- 14 **496.** Page 268, line 5: delete that line.
- 15 **497.** Page 268, line 9: delete that line.
- 16 **498.** Page 268, line 16: after that line insert:
- 17 "(bn) Dairy manufacturing facility
- investment credit GPR A 600,000 700,000".
- 19 **499.** Page 269, line 6: delete "Enterprise" and substitute "Rural enterprise".
- 500. Page 269, line 20: increase the dollar amount for fiscal year 2007-08 by \$23,429,900 and increase the dollar amount for fiscal year 2008-09 by \$34,935,800 to increase funding for the purposes for which the appropriation is made.

- 1 **501.** Page 270, line 2: delete "and first".
- 2 **502.** Page 270, line 3: delete "dollar credit".
- 503. Page 271, line 1: decrease the dollar amount for fiscal year 2007-08 by \$5,000,000 and decrease the dollar amount for fiscal year 2008-09 by \$5,000,000 for the purpose of making payments for municipal services.
- 6 **504.** Page 273, line 7: after that line insert:
- 7 "(f) Supplemental title fee matching GPR S -0-
- 8 "

16

17

18

19

- 9 **505.** Page 273, line 10: delete lines 10 and 11.
- 10 **506.** Page 273, line 14: delete that line.
- 507. Page 280, line 7: decrease the dollar amount for fiscal year 2008–09 by \$500,000 for the purpose of eliminating funding for the office of justice assistance to provide a pre-sentencing assessment grant to the county that has the highest violent crime rate.
  - **508.** Page 280, line 7: decrease the dollar amount for fiscal year 2007–08 by \$3,452,600 and decrease the dollar amount for fiscal year 2008–09 by \$4,547,400 for the purposes for which the appropriation is made.
  - **509.** Page 280, line 7: decrease the dollar amount for fiscal year 2007–08 by \$50,000 and decrease the dollar amount for fiscal year 2008–09 by \$700,000 for the purpose of reducing general purpose revenue for general program supplementation.
- 21 **510.** Page 286, line 3: delete lines 3 to 11.
- 22 **511.** Page 287, line 11: after that line insert:
- 23 "Section 183r. 20.115 (4) (c) of the statutes is amended to read:

1	20.115 (4) (c) Agricultural investment aids. Biennially, the amounts in the
2	schedule for agricultural research and development grants under s. $93.46\ (2)$ and $(3)$
3	and sustainable agriculture grants under s. 93.47.".
4	<b>512.</b> Page 287, line 12: after that line insert:
5	"Section 187c. 20.115 (4) (g) of the statutes is created to read:
6	20.115 (4) (g) Grant for pulp and paper mill. Biennially, the amounts in the
7	schedule for the grant to a pulp and paper mill under 2007 Wisconsin Act (this act),
8	section 9103 (3w). All moneys received from the Wisconsin Housing and Economic
9	Development Authority under s. 234.165 (4) shall be credited to this appropriation
10	account.
11	<b>Section 187d.</b> 20.115 (4) (u) of the statutes is created to read:
12	20.115 (4) (u) Grant for pulp and paper mill; recycling fund. Biennially, from
13	the recycling fund, the amounts in the schedule for the grant to a pulp and paper mill
14	under 2007 Wisconsin Act (this act), section 9103 (3w).".
15	<b>513.</b> Page 287, line 12: after that line insert:
16	"Section 185t. 20.115 (4) (qm) of the statutes is created to read:
17	20.115 (4) (qm) Grants for soybean crushing facilities. Biennially, from the
18	recycling fund, the amounts in the schedule for grants for soybean crushing facilities
19	under 2007 Wisconsin Act (this act), section 9103 (4u).".
20	<b>514.</b> Page 287, line 12: after that line insert:
21	"Section 185p. 20.115 (4) (r) of the statutes is amended to read:
22	20.115 (4) (r) Agricultural investment aids, agrichemical management fund.
23	Biennially, from the agrichemical management fund, the amounts in the schedule

9

10

11

20

21

22

- for agricultural research and development grants under s. 93.46 (2) and (3) and sustainable agriculture grants under s. 93.47.".
- 3 **515.** Page 287, line 13: delete lines 13 to 20.
- 516. Page 287, line 24: delete the material beginning with that line and ending with page 288, line 13.
- 6 **517.** Page 289, line 20: after that line insert:
- 7 "Section 195t. 20.143 (1) (fm) of the statutes is amended to read:
  - 20.143 (1) (fm) *Minority business projects; grants and loans*. Biennially, the amounts in the schedule for grants under ss. 560.038, 560.039, s. 560.82 and 560.837, grants and loans under s. 560.83, the grant under 1993 Wisconsin Act 110, section 3, and the loans under 1997 Wisconsin Act 9, section 3.".
- 518. Page 290, line 12: after "9110 (4)," insert "and the loans under 2007
   Wisconsin Act .... (this act), section 9108 (5f),".
- 14 **519.** Page 290, line 15: after that line insert:
- **"Section 198fb.** 20.143 (1) (ig) of the statutes is amended to read:
- 20.143 (1) (ig) Gaming economic development and diversification; repayments.

  Biennially, the amounts in the schedule for grants and loans under ss. 560.137 (2)

  and s. 560.138. All moneys received in repayment of loans under ss. s. 560.137 (2),

  2005 stats., and s. 560.138 shall be credited to this appropriation account.
  - **SECTION 198fd.** 20.143 (1) (im) of the statutes is amended to read:
  - 20.143 (1) (im) *Minority business projects; repayments*. All moneys received in repayment of grants or loans under s. 560.83 and loans under 1997 Wisconsin Act 9, section 3, to be used for grants and loans under ss. 560.82, and 560.83, and 560.837,

the grant under 2001 Wisconsin Act 16, section 9110 (7g), and the loans under 1997 Wisconsin Act 9, section 3.

**Section 198fe.** 20.143 (1) (kg) of the statutes is amended to read:

20.143 (1) (kg) American Indian economic liaison and gaming grants specialist and program marketing. The amounts in the schedule for the American Indian economic liaison program under s. 560.87, other than for grants under s. 560.87 (6), for the salary and fringe benefits of, and related supplies and services for, the gaming grants specialist for the programs program under ss. 560.137 and s. 560.138, and for marketing the programs program under ss. 560.137 and s. 560.138. From this appropriation, the department may expend in each fiscal year no more than \$100,000 for marketing the programs program under ss. 560.137 and s. 560.138. All moneys transferred from the appropriation account under s. 20.505 (8) (hm) 6g. shall be credited to this appropriation account. Notwithstanding s. 20.001 (3) (a), the unencumbered balance on June 30 of each year shall revert to the appropriation account under s. 20.505 (8) (hm).

**Section 198ff.** 20.143 (1) (kj) of the statutes is amended to read:

20.143 (1) (kj) Gaming economic development and diversification; grants and loans. Biennially, the amounts in the schedule for grants and loans under ss. 560.137 and s. 560.138, for the grants under s. 560.139 (1) (a) and (2), and for the grants under 2001 Wisconsin Act 16, section 9110 (2k), (11pk), and (11zx). Of the amounts in the schedule, \$500,000 shall be allocated in each fiscal year for the grants under s. 560.137 (3m). All moneys transferred from the appropriation account under s. 20.505 (8) (hm) 6j. shall be credited to this appropriation account. Notwithstanding s. 20.001 (3) (b), the unencumbered balance on June 30 of each odd-numbered year shall revert to the appropriation account under s. 20.505 (8) (hm)."

- 1 **520.** Page 290, line 16: after that line insert:
- "Section 198p. 20.143(1)(x) of the statutes is amended to read:
- 3 20.143 (1) (x) Industrial building construction loan fund. All moneys received
- 4 in the industrial building construction loan fund, for the purpose of s. 560.10, 2005
- 5 stats.".
- 6 **521.** Page 291, line 1: after "(3)" insert "(a)".
- 7 **522.** Page 291, line 5: delete lines 5 to 13.
- 8 **523.** Page 291, line 13: after that line insert:
- 9 "Section 206d. 20.143 (3) (de) of the statutes is repealed.".
- 10 **524.** Page 291, line 17: delete "<u>, for</u>" and substitute "<u>and for</u>".
- 11 **525.** Page 291, line 18: delete the material beginning with "<u>, and</u>" and ending
- 12 with "<u>(km)</u>" on line 20.
- 13 **526.** Page 292, line 5: delete lines 5 to 9.
- 14 **527.** Page 292, line 10: delete the material beginning with that line and
- ending with page 293, line 5.
- 16 **528.** Page 294, line 15: delete the material beginning with that line and
- ending with page 295, line 6.
- 18 **529.** Page 295, line 15: delete the material beginning with that line and
- ending with page 296, line 7.
- 20 **530.** Page 296, line 8: delete lines 8 to 12.
- 21 **531.** Page 296, line 12: after that line insert:
- 22 "Section 221pk. 20.215 (1) (k) of the statutes is repealed.
- **Section 221pm.** 20.215 (1) (ka) of the statutes is repealed.".

kindergarten to 12.".

1	<b>532.</b> Page 296, line 12: after that line insert:
2	"Section 221r. 20.225 (1) (a) of the statutes is amended to read:
3	20.225 (1) (a) General program operations. The amounts in the schedule to
4	carry out its functions other than programming under ss. 39.11 and 39.13. To the
5	greatest extent practicable, the educational communications board shall expend
6	those amounts to carry out functions that support educational programming for
7	grades kindergarten to 12.".
8	<b>533.</b> Page 296, line 18: after "board." insert "To the greatest extent
9	practicable, the educational communications board shall expend those amounts for
10	energy costs incurred in providing educational programming for grades
11	kindergarten to 12.".
12	<b>534.</b> Page 296, line 19: delete the material beginning with that line and
13	ending with page 297, line 10.
14	<b>535.</b> Page 297, line 10: after that line insert:
15	"Section 222b. 20.225 (1) (d) of the statutes is amended to read:
16	20.225 (1) (d) Milwaukee area technical college. The amounts in the schedule
17	to contract with Milwaukee area technical college under s. 39.11 (18). To the greatest
18	extent practicable, the educational communications board shall expend those
19	amounts to contract for educational programming for grades kindergarten to 12.
20	<b>Section 222f.</b> 20.225 (1) (f) of the statutes is amended to read:
21	20.225 (1) (f) Programming. The amounts in the schedule for programming
22	under s. 39.11. To the greatest extent practicable, the educational communications
23	board shall expend those amounts for educational programming for grades

- 1 **536.** Page 297, line 10: after that line insert:
- 2 "Section 223h. 20.235 (1) (cd) of the statutes is created to read:
- 3 20.235 (1) (cd) Loan program for physicians in shortage areas. The amounts
- 4 in the schedule for the loan program for physicians in shortage areas under s.
- 5 39.387.".
- 6 **537.** Page 297, line 10: after that line insert:
- 7 "Section 223j. 20.235 (1) (cp) of the statutes is created to read:
- 8 20.235 (1) (cp) Loan program for veterinarians of food-producing animals. The
- 9 amounts in the schedule for the loan program for veterinarians of food-producing
- 10 animals under s. 39.397.".
- 11 **538.** Page 297, line 13: delete "\$50,000,000" and substitute "\$39,280,600".
- **539.** Page 297, line 14: delete "\$39,280,600 \$55,000,000" and substitute
- 13 "\$39,280,600".
- **540.** Page 297, line 18: delete lines 18 to 20.
- 15 **541.** Page 297, line 21: delete lines 21 to 25.
- 16 **542.** Page 298, line 11: delete the material beginning with that line and
- ending with page 299, line 6.
- 18 **543.** Page 299, line 24: delete "and" and substitute "and".
- 19 **544.** Page 300, line 1: delete the material beginning with ", and" and ending
- 20 with "(a)" on line 3.
- 21 **545.** Page 300, line 4: delete lines 4 to 9.
- 22 **546.** Page 300, line 20: delete the material beginning with that line and
- ending with page 301, line 2.

- 1 **547.** Page 301, line 4: delete lines 4 to 8.
- 2 **548.** Page 301, line 12: after that line insert:
- 3 "Section 236n. 20.255 (2) (am) of the statutes is created to read:
- 4 20.255 (2) (am) *Teacher merit pay*. The amounts in the schedule for teacher merit pay under s. 115.28 (44).".
- 6 **549.** Page 301, line 14: delete "consolidation" and substitute "efficiency incentive".
- 8 **550.** Page 301, line 17: after that line insert:
- 9 **"Section 237e.** 20.255 (2) (cf) of the statutes is amended to read:
- 10 20.255 (2) (cf) Alternative education grants. The amounts in the schedule for
- alternative education grants under s. 115.366 and for payments to the Second
- 12 Chance Partnership under 2005 2007 Wisconsin Act 25 .... (this act), section 9137
- 13 (3q) (3L).".
- **551.** Page 301, line 18: delete lines 18 to 25.
- 15 **552.** Page 302, line 1: delete lines 1 to 10.
- **553.** Page 302, line 11: delete lines 11 to 14.
- 17 **554.** Page 302, line 14: after that line insert:
- 18 **"Section 241t.** 20.255 (2) (dn) of the statutes is created to read:
- 19 20.255 (2) (dn) School safety grants. The amounts in the schedule for grants
- to school districts under s. 115.28 (46m).".
- 21 **555.** Page 302, line 15: delete lines 15 to 17.
- 22 **556.** Page 302, line 17: after that line insert:
- 23 "Section 242p. 20.255 (2) (fr) of the statutes is created to read:

 $\mathbf{2}$ 

3

19

20

21

22

- 20.255 **(2)** (fr) *Racine County Parental Choice Program.* A sum sufficient to make the payments to private schools under s. 118.53 (4) and (4m).".
  - **557.** Page 302, line 17: after that line insert:
- 4 "Section 243. 20.255 (2) (fz) of the statutes is created to read:
- 5 20.255 (2) (fz) Grants for science, technology, engineering, and mathematics 6 programs. The amounts in the schedule for grants to school districts for science, 7 technology, engineering, and mathematics programs under s. 115.28 (46).".
- 8 **558.** Page 302, line 18: delete lines 18 to 22.
- 9 **559.** Page 302, line 23: delete the material beginning with that line and ending with page 303, line 2.
- 11 **560.** Page 303, line 25: delete the material beginning with that line and ending with page 304, line 23.
- 13 **561.** Page 304, line 24: delete the material beginning with that line and ending with page 305, line 2.
- 15 **562.** Page 305, line 2: after that line insert:
- **"Section 251g.** 20.285 (1) (ft) of the statutes is repealed.".
- 17 **563.** Page 305, line 25: delete the material beginning with that line and ending with page 306, line 11.
  - **564.** Page 306, line 16: delete the material beginning with "and" and ending with "2009," on line 17 and substitute "but not including any moneys received from the sale of state property during the period beginning on the effective date of this paragraph .... [revisor inserts date], and ending on June 30, 2009,".
    - **565.** Page 307, line 3: delete lines 3 to 24.

15

16

17

- 1 **566.** Page 308, line 8: delete the material beginning with that line and ending 2 with page 310, line 2. 3
  - **567.** Page 310, line 2: after that line insert:
- 4 "Section 260c. 20.285 (1) (kn) of the statutes is repealed.".
- **568.** Page 310, line 4: delete "Biennially, from the agricultural" and 5 6 substitute "From the".
- **569.** 7 Page 310, line 5: delete "chemical cleanup" and substitute "environmental". 8
  - **570.** Page 310, line 7: delete lines 7 and 8.
- **571.** Page 310, line 8: after that line insert: 10
- 11 "Section 261p. 20.285 (2) (e) (title) of the statutes is amended to read:
- 12 20.285 (2) (e) (title) Use of state funds for entertainment certain purposes.
- 13 **Section 261r.** 20.285 (2) (e) of the statutes is renumbered 20.285 (2) (e) 1.
- 14 **Section 261t.** 20.285 (2) (e) 2. of the statutes is created to read:
  - 20.285 (2) (e) 2. Beginning in fiscal year 2008–09, no general purpose revenues appropriated under this section may be used to fund the Havens Center located within the sociology department at the University of Wisconsin-Madison or to fund the School for Workers under the University of Wisconsin-Extension.
- 19 **Section 261v.** 20.285 (2) (e) 3. of the statutes is created to read:
- 20 20.285 (2) (e) 3. Beginning in fiscal year 2010-2011, no general purpose 21revenues appropriated under this section may be used to fund the law school of the 22 University of Wisconsin.".
- 23 **572.** Page 310, line 8: after that line insert:
- **"Section 261t.** 20.285 (4) (dd) of the statutes is amended to read: 24

 $\mathbf{2}$ 

20.285 (4) (dd) Lawton minority undergraduate grants program. A sum sufficient equal to \$5,218,300 \$5,531,400 in the 2005–06 2007–08 fiscal year and \$5,531,400 \$5,333,900 in the 2006–07 2008–09 fiscal year, and in subsequent fiscal years a sum sufficient equal to the amount calculated under s. 36.34 (1) (c), for the Lawton minority undergraduate grant program under s. 36.34 (1).".

- **573.** Page 310, line 9: delete lines 9 to 18.
- **574.** Page 310, line 19: delete lines 19 to 24.
- **575.** Page 311, line 1: delete lines 1 to 11.
- **576.** Page 311, line 19: delete the material beginning with that line and ending with page 313, line 25.
- **577.** Page 313, line 25: after that line insert:
- 12 "Section 271k. 20.320 (3) of the statutes is repealed.".
- **578.** Page 314, line 7: after that line insert:
- **"Section 273d.** 20.370 (1) (fe) of the statutes is amended to read:

20.370 (1) (fe) Endangered resources — general fund. From the general fund, a sum sufficient in fiscal year 1993–94 and in each fiscal year thereafter that equals the sum of the amount certified in that fiscal year under s. 71.10 (5) (h) 3. for the previous fiscal year and the amounts received under par. (gr) in that fiscal year for the purposes of the endangered resources program, as defined in s. 71.10 (5) (a) 2. The amount appropriated under this subdivision may not exceed \$500,000 in a fiscal year, except that the amount appropriated under this subdivision in fiscal year years 2005–06, 2006–07, 2007–08, and 2008–09 may not exceed \$364,000 and the amount appropriated under this subdivision in fiscal year 2006–07 may not exceed \$364,000 in each of those fiscal years."

- 1 **579.** Page 314, line 12: after that line insert:
- 2 "Section 275g. 20.370 (1) (kr) of the statutes is created to read:
- 20.370 (1) (kr) Control of wild animals. Biennially, the amounts in the schedule for removal activities of the department under s. 29.885. All moneys transferred from the appropriation under sub. (5) (fq) shall be credited to this appropriation.
- **SECTION 275m.** 20.370 (1) (Ls) of the statutes is repealed.".
- 8 **580.** Page 314, line 13: delete lines 13 to 18.
- 9 **581.** Page 315, line 1: delete lines 1 to 6.
- 10 **582.** Page 315, line 16: delete lines 16 to 20.
- 11 **583.** Page 315, line 23: after that line insert:
- "Section 281qj. 20.370 (5) (az) of the statutes is created to read:
- 13 20.370 (5) (az) Resource aids—interpretive center. The amounts in the schedule 14 for a grant to the Florence Wild Rivers Interpretive Center.".
- SECTION 281qr. 20.370 (5) (az) of the statutes, as created by 2007 Wisconsin

  Act .... (this act), is repealed.".
- 17 **584.** Page 316, line 6: delete lines 6 to 9.
- 18 **585.** Page 317, line 3: after that line insert:
- **"Section 282fc.** 20.370 (5) (cu) of the statutes is amended to read:
- 20 20.370 **(5)** (cu) Recreation aids all-terrain vehicle project aids. As a continuing appropriation, the amounts in the schedule from moneys received from all-terrain vehicle fees under s. 23.33 (2) (c) to (e) and (2j) to provide aid to towns, villages, cities, counties, and federal agencies for nonstate all-terrain vehicle

6

7

8

9

10

11

12

13

- projects <u>and to provide grants under s. 23.33 (11m) (g) to counties and municipalities</u>

  participating in the lightweight utility vehicle pilot program.".
- 3 **586.** Page 317, line 3: after that line insert:
- 4 "Section 282fg. 20.370 (5) (fq) of the statutes is amended to read:
  - 20.370 (5) (fq) Wildlife damage claims and abatement. All moneys received under ss. 29.181, 29.559 (1r), and 29.563 (13) and not appropriated under par. pars. (fr) and (fs) and subs. sub. (1) (hs) and (Ls) and (5) (fs) to provide state aid for the wildlife damage abatement program under s. 29.889 (5) (c) and the wildlife damage claim program under s. 29.889 (7) (d), for county administration costs under s. 29.889 (2) (d), and for payments under s. 29.89. On June 30 of each fiscal year, the unencumbered balance of this appropriation or the amount equal to the amount in the schedule under sub. (1) (kr) for that fiscal year, whichever is less, shall be transferred from this appropriation to the appropriation under sub. (1) (kr)."
    - **587.** Page 317, line 9: after "Biennially," insert "from the conservation fund,".
- 15 **588.** Page 317, line 15: delete the material beginning with that line and ending with page 320, line 4.
- **589.** Page 320, line 9: after "(ti)," insert "and".
- 18 **590.** Page 320, line 11: delete the material beginning with ", and" and ending with "(a)" on line 12.
- 20 **591.** Page 320, line 13: delete the material beginning with that line and ending with page 323, line 17.
- **592.** Page 323, line 18: delete lines 18 to 25.
- 23 **593.** Page 324, line 1: delete lines 1 to 5.

- 1 **594.** Page 324, line 25: after that line insert:
- 2 "Section 306g. 20.395 (2) (dq) of the statutes is amended to read:
- 20.395 (2) (dq) Aeronautics assistance, state funds. As a continuing appropriation, the amounts in the schedule for the state's share of airport projects under ss. 114.34 and 114.35; for developing air marking and other air navigational facilities; for administration of the powers and duties of the secretary of transportation under s. 114.31; for costs associated with aeronautical activities under s. 114.31, except for the program under s. 114.31 (3) (b); and for the administration of other aeronautical activities, except aircraft registration under s.
- 10 114.20, authorized by law.
- 11 **Section 306h.** 20.395 (2) (ds) of the statutes is repealed.".
- 12 **595.** Page 325, line 1: delete lines 1 to 5.
- 13 **596.** Page 325, line 9: delete "and".
- **597.** Page 325, line 10: after "9148 (3)" insert ", and for the grant under 2007
- 15 Wisconsin Act .... (this act), section 9148 (9z)".
- 16 **598.** Page 325, line 10: after that line insert:
- 17 **"Section 307c.** 20.395 (2) (jq), (jv) and (jx) of the statutes are repealed.
- **SECTION 307e.** 20.395 (2) (nx) of the statutes is amended to read:
- 20.395 (2) (nx) Transportation enhancement activities, federal funds. All moneys received from the federal government for purposes of transportation enhancement activities under s. 85.026 and for grants under s. 85.024, for such purposes.
- 23 **Section 307g.** 20.395 (2) (ov) of the statutes is created to read:

20.395 **(2)** (ov) *Bicycle and pedestrian facilities, local funds*. All moneys received from any local unit of government for purposes of the bicycle and pedestrian facilities program under s. 85.024, for such purposes.

**Section 307i.** 20.395 (2) (ox) of the statutes is created to read:

20.395 (2) (ox) Bicycle and pedestrian facilities, federal funds. All moneys received from the federal government for purposes of the bicycle and pedestrian facilities program under s. 85.024, for such purposes.".

**599.** Page 325, line 18: after that line insert:

**"Section 309d.** 20.395 (3) (cq) of the statutes is amended to read:

appropriation, the amounts in the schedule for improvement of existing state trunk and connecting highways; for improvement of bridges on state trunk or connecting highways and other bridges for which improvement is a state responsibility, for necessary approach work for such bridges and for replacement of such bridges with at-grade crossing improvements; for the construction and rehabilitation of the national system of interstate and defense highways and bridges and related appurtenances; for special maintenance activities under s. 84.04 on roadside improvements; for bridges under s. 84.10; for the bridge project under s. 84.115; for payment to a local unit of government for a jurisdictional transfer under s. 84.02 (8); for the disadvantaged business demonstration and training program under s. 84.076; for the transfers required under 1999 Wisconsin Act 9, section 9250 (1), and 2003 Wisconsin Act 33, section 9153 (4q); and for the purposes described under 1999 Wisconsin Act 9, section 9150 (8g), and 2001 Wisconsin Act 16, section 9152 (4e), and 2007 Wisconsin Act .... (this act), section 9148 (9i) (b). This paragraph does not apply

highways.".

- to any southeast Wisconsin freeway rehabilitation projects under s. 84.014, or to the installation, replacement, rehabilitation, or maintenance of highway signs, traffic control signals, highway lighting, pavement markings, or intelligent transportation systems, unless incidental to the improvement of existing state trunk and connecting
- **600.** Page 325, line 19: delete the material beginning with that line and ending with page 326, line 4.
- **601.** Page 326, line 8: delete ", less any amount appropriated under par. (bq),".
- **602.** Page 326, line 12: delete "and to" and substitute "and to".
- **603.** Page 326, line 15: delete the material beginning with "<u>, and</u>" and ending with "<u>(a)</u>" on line 16.
- **604.** Page 326, line 17: delete the material beginning with that line and ending with page 327, line 19.
- 605. Page 328, line 1: delete the material beginning with ", and" and ending with "(a)" on line 2.
- **606.** Page 328, line 3: delete lines 3 to 9.
- **607.** Page 328, line 10: delete lines 10 to 25.
- **608.** Page 329, line 19: delete the material beginning with that line and ending with page 330, line 3.
- **609.** Page 330, line 4: delete lines 4 to 7.
- **610.** Page 330, line 8: delete lines 8 to 13.
- **611.** Page 334, line 13: delete that line.
- **612.** Page 334, line 14: delete lines 14 to 19.

- **613.** Page 334, line 20: delete lines 20 to 23.
- **614.** Page 334, line 24: delete the material beginning with that line and ending with page 335, line 11.
- **615.** Page 335, line 22: delete lines 22 and 23.
- **616.** Page 336, line 1: delete lines 1 to 11.
- **617.** Page 336, line 13: delete lines 13 to 21.
- **618.** Page 337, line 3: delete lines 3 to 8.
- **619.** Page 337, line 16: delete lines 16 to 25.
- **620.** Page 338, line 1: delete lines 1 to 5.
- **621.** Page 338, line 6: delete that line.
- **622.** Page 338, line 7: delete lines 7 to 19.
- **623.** Page 338, line 20: delete lines 20 to 24.
- **624.** Page 339, line 1: delete lines 1 to 17.
- **625.** Page 339, line 18: delete lines 18 to 24.
- **626.** Page 339, line 21: delete lines 21 to 25.
- **627.** Page 340, line 1: delete lines 1 to 24.
- **628.** Page 341, line 1: delete lines 1 to 25.
- **629.** Page 342, line 1: delete lines 1 to 20.
- **630.** Page 343, line 5: delete that line and substitute "s. 46.284 (5),".
- **631.** Page 343, line 6: delete "49.685,".

- **632.** Page 343, line 16: delete the material beginning with that line and ending with page 344, line 12.
- **633.** Page 344, line 17: delete "<u>a portion of</u>".
- 634. Page 344, line 18: delete lines 18 to 21 and substitute "contract costs for the Medical Assistance program under s. 49.45, the food stamp program under s. 49.79, and the Badger Care health care program under s. 49.665, other than payments to counties and tribal governing bodies".
- **635.** Page 345, line 7: delete lines 7 to 25.
- **636.** Page 346, line 1: delete lines 1 to 7.
- **637.** Page 346, line 8: delete lines 8 to 16.
- **638.** Page 346, line 17: delete that line.
- **639.** Page 346, line 18: delete lines 18 to 25.
- **640.** Page 347, line 14: delete the material beginning with that line and ending with page 348, line 8.
- **641.** Page 348, line 17: delete the material beginning with that line and ending with page 349, line 2.
- **642.** Page 349, line 3: delete lines 3 to 6.
- **643.** Page 349, line 7: delete lines 7 to 16.
- **644.** Page 349, line 17: delete lines 17 to 20.
- **645.** Page 350, line 6: after that line insert:
- 21 "Section 402r. 20.435 (5) (cf) of the statutes is created to read:

- 1 20.435 (5) (cf) Colposcopies and other services. The amounts in the schedule 2 for aids under s. 146.68.".
- **646.** Page 350, line 7: delete lines 7 to 14.
- **647.** Page 350, line 15: delete lines 15 to 24.
- **648.** Page 350, line 25: delete the material beginning with that line and ending with page 351, line 5.
- **649.** Page 351, line 13: delete lines 13 to 16.
- **650.** Page 351, line 17: delete lines 17 to 22.
- **651.** Page 352, line 11: delete lines 11 to 25.
- **652.** Page 353, line 1: delete lines 1 to 24.
- **653.** Page 353, line 25: delete the material beginning with that line and ending with page 354, line 16.
- **654.** Page 355, line 1: delete lines 1 to 7.
- **655.** Page 356, line 3: delete lines 3 to 9.
- **656.** Page 356, line 10: delete lines 10 to 23.
- **657.** Page 356, line 24: delete that line.
- **658.** Page 357, line 1: delete lines 1 to 10.
- **659.** Page 357, line 11: delete lines 11 to 22.
- **660.** Page 357, line 23: delete lines 23 to 25.
- **661.** Page 358, line 1: delete lines 1 to 25.
- **662.** Page 359, line 1: delete lines 1 to 7.
- **663.** Page 359, line 8: delete lines 8 to 25.

- **664.** Page 360, line 1: delete lines 1 to 24.
- **665.** Page 361, line 1: delete lines 1 to 25.
- **666.** Page 362, line 1: delete lines 1 to 25.
- **667.** Page 363, line 1: delete lines 1 to 23.
- **668.** Page 364, line 1: delete lines 1 to 25.
- **669.** Page 365, line 1: delete lines 1 to 23.
- **670.** Page 366, line 1: delete lines 1 to 25.
- **671.** Page 367, line 1: delete lines 1 to 8.
- **672.** Page 367, line 9: delete lines 9 to 13.
- **673.** Page 367, line 14: delete lines 14 to 22.
- **674.** Page 368, line 20: delete lines 20 and 21.
- **675.** Page 369, line 1: delete lines 1 and 2.
- **676.** Page 369, line 5: delete lines 5 to 7.
- **677.** Page 369, line 8: delete lines 8 to 10.
- **678.** Page 369, line 17: delete lines 17 to 19.
- **679.** Page 370, line 7: delete lines 7 to 10.
- **680.** Page 371, line 7: delete lines 7 to 9.
- **681.** Page 371, line 10: delete lines 10 to 16.
- **682.** Page 371, line 17: delete lines 17 to 25.
- **683.** Page 372, line 1: delete lines 1 to 25.
- **684.** Page 373, line 1: delete lines 1 to 25.

- 1 **685.** Page 374, line 1: delete lines 1 to 14.
- 2 **686.** Page 374, line 15: delete lines 15 to 25.
- 3 **687.** Page 375, line 8: delete lines 8 to 10.
- 4 **688.** Page 375, line 11: delete that line and substitute:
- **SECTION 448m.** 20.455 (2) (i) 13. of the statutes is amended to read:
- 6 20.455 **(2)** (i) 13. The amount transferred to s. 20.505 (6) (k) par. (ky) shall be 7 the amount in the schedule under s. 20.505 (6) (k) par. (ky).".
- 8 **689.** Page 375, line 12: delete that line and substitute:
- 9 "Section 489d. 20.455 (2) (i) 16. of the statutes is amended to read:
- 10 20.455 **(2)** (i) 16. The amount transferred to s. 20.505 (6) (kc) par. (kb) shall be

  11 the amount in the schedule under s. 20.505 (6) (kc) par. (kb).".
- 12 **690.** Page 375, line 23: after that line insert:
- 13 "**Section 492r.** 20.455 (2) (m) of the statutes is amended to read:
- 20.455 (2) (m) *Federal aid*, *state operations*. All moneys received as federal aid, as authorized by the governor under s. 16.54, for state operations <u>and for state</u> agency operations for justice assistance.".
  - **691.** Page 375, line 24: after that line insert:

- 18 **"Section 493m.** 20.455 (2) (n) of the statutes is amended to read:
- 20.455 (2) (n) Federal aid, local assistance. All moneys received as federal aid, as authorized by the governor under s. 16.54, for local assistance and for project grants to improve the administration of criminal justice.".
- 22 **692.** Page 376, line 7: delete lines 7 to 23.
- 23 **693.** Page 376, line 24: delete that line.

- **694.** Page 377, line 1: delete lines 1 and 2.
- **695.** Page 377, line 23: delete the material beginning with that line and ending with page 378, line 3.
- **696.** Page 378, line 11: delete lines 11 to 25.
- **697.** Page 379, line 1: delete lines 1 to 13.
- **698.** Page 379, line 23: delete the material beginning with that line and ending with page 380, line 3.
- **699.** Page 380, line 13: delete lines 13 to 23.
- **700.** Page 381, line 24: delete the material beginning with that line and ending with page 382, line 14.
- **701.** Page 382, line 21: after that line insert:
- 12 "**Section 516d.** 20.505 (1) (bm) of the statutes is created to read:
- 20.505 (1) (bm) *State funding of mandates*. As a continuing appropriation, the amounts in the schedule to reimburse local governmental units as provided under s. 16.59.".
- **702.** Page 382, line 22: delete lines 22 to 25.
- **703.** Page 382, line 25: after that line insert:

- **"Section 517p.** 20.505 (1) (kp) of the statutes is amended to read:
  - 20.505 (1) (kp) *Interagency assistance; justice information systems*. The amounts in the schedule for the development and operation of automated justice information systems under s. 16.971 (9). All moneys transferred from the appropriation account under sub. (6) (m) s. 20.455 (2) (m) shall be credited to this appropriation account.".

**704.** Page 383, line 4: delete lines 4 to 10.

**705.** Page 383, line 12: delete the material beginning with that line and ending with page 384, line 20.

**706.** Page 384, line 21: delete lines 21 to 25.

**707.** Page 385, line 1: delete lines 1 and 2.

**708.** Page 386, line 6: delete lines 6 to 19.

**709.** Page 387, line 10: after that line insert:

"Section 534s. 20.505 (5) (ka) of the statutes, as affected by 2007 Wisconsin Act .... (this act), is amended to read:

20.505 (5) (ka) Facility operations and maintenance; police and protection functions. The amounts in the schedule for the purpose of financing the costs of operation of state-owned or operated facilities that are not funded from other appropriations, including custodial and maintenance services; minor projects; utilities, fuel, heat and air conditioning; assessments levied by the department under s. 16.847 (3) for debt service costs and energy cost savings generated at departmental facilities; and costs incurred under ss. 16.858 and 16.895 by or on behalf of the department; and supplementing the costs of operation of child care facilities for children of state employees under s. 16.841; and for police and protection functions under s. 16.84 (2) and (3). All moneys received from state agencies for the operation of such facilities, parking rental fees established under s. 16.843 (2) (bm) and miscellaneous other sources, all moneys received from assessments under s. 16.895, all moneys received for the performance of gaming protection functions under s. 16.84 (3), and all moneys transferred from the appropriation account under s. 20.865 (2) (e) for this purpose shall be credited to this appropriation account.".

**710.** Page 387, line 11: delete lines 11 to 19. 1 2 **711.** Page 388, line 7: delete the material beginning with ", and to" and ending with "(a)" on line 8. 3 4 **712.** Page 388, line 18: after that line insert: **"Section 535r.** 20.505 (6) (a) of the statutes is repealed.". 5 6 **713.** Page 388, line 19: delete lines 19 to 24 and substitute: 7 "Section 536g. 20.505 (6) (b) of the statutes is renumbered 20.455 (2) (em) and 8 amended to read: 9 20.455 (2) (em) Alternatives to prosecution and incarceration for persons who 10 use alcohol or other drugs Grants for substance abuse treatment programs for criminal offenders; general purpose revenue. The amounts in the schedule for 11 12 making grants to counties under s. 16.964 (12) (b) 165.95 (2) and entering into 13 contracts under s. 16.964 (12) (j) 165.95 (10).". 14 **714.** Page 388, line 24: after that line insert: 15 "Section 536i. 20.505 (6) (c) of the statutes is renumbered 20.455 (2) (cr) and 16 amended to read: 17 20.455 (2) (cr) Law enforcement officer supplement grants. The amounts in the 18 schedule to provide grants for uniformed law enforcement officers under s. 16.964 (5) 19 <u>165.986</u>. 20 **Section 538m.** 20.505 (6) (gj) of the statutes is renumbered 20.455 (5) (gj) and 21 amended to read: 22 20.455 (5) (gi) Grants for victims of sexual assault; child pornography 23 surcharge. All moneys received as part C of any child pornography surcharge

imposed under s. 973.042 for the grant program under s. 16.964 (11) 165.935.

1	<b>Section 538t.</b> 20.505 (6) (i) of the statutes is repealed.".
2	715. Page 388, line 25: delete the material beginning with that line and
3	ending with page 389, line 5, and substitute:
4	"Section 539b. 20.505 (6) (k) of the statutes is renumbered 20.455 (2) (ky)."
5	<b>716.</b> Page 389, line 6: delete that line and substitute:
6	"Section 540b. 20.505 (6) (kc) of the statutes is renumbered 20.455 (2) (kb) and
7	amended to read:
8	20.455 (2) (kb) Grants for digital recording of custodial interrogations. The
9	amounts in the schedule for grants to law enforcement agencies under s. $16.964$ (10)
10	165.987 for equipment or training used to digitally record custodial interrogations.
11	All moneys transferred from the appropriation account under s. 20.455 (2) (i) 16.
12	shall be credited to this appropriation account.".
13	<b>717.</b> Page 389, line 6: after that line insert:
14	"Section 540e. 20.505 (6) (km) of the statutes is repealed.
15	<b>Section 540g.</b> 20.505 (6) (ku) of the statutes is renumbered 20.455 (2) (kx) and
16	amended to read:
17	20.455 (2) (kx) Grants for substance abuse treatment programs for criminal
18	offenders; program revenue. All moneys received under s. 961.41 (5) (c) 2. or 973.043
19	for the purpose of making grants to counties under s. $\frac{16.964}{(12)}$ (b) $\frac{165.95}{(2)}$ and
20	entering into contracts under s. $\frac{16.964}{(12)}$ (j) $\frac{165.95}{(10)}$ .
21	<b>Section 540i.</b> 20.505 (6) (m) of the statutes is repealed.
22	<b>Section 540k.</b> 20.505 (6) (mb) of the statutes is renumbered 20.465 (3) (mb).
23	<b>Section 540m.</b> 20.505 (6) (p) of the statutes is repealed.".

718. Page 389, line 10: after that line insert:

- 1 "Section 542g. 20.505 (8) (hm) 11a. of the statutes is repealed.".
- 2 **719.** Page 390, line 9: delete lines 9 to 16.
- **720.** Page 390, line 17: delete the material beginning with that line and ending with page 391, line 4.
- 5 **721.** Page 391, line 23: delete the material beginning with that line and ending with on page 392, line 2.
- 7 **722.** Page 392, line 9: delete lines 9 to 22.
- 8 **723.** Page 393, line 3: delete lines 3 to 5.
- 9 **724.** Page 393, line 14: delete lines 14 to 19.
- 10 **725.** Page 394, line 11: delete lines 11 to 16.
- 726. Page 394, line 17: delete the material beginning with that line and ending with page 395, line 10.
- 13 **727.** Page 395, line 10: after that line insert:
- **"Section 568m.** 20.835 (2) (co) (title) of the statutes is amended to read:
- 15 20.835 (2) (co) (title) Enterprise Rural enterprise zone jobs credit.".
- 728. Page 395, line 10: after that line insert:
- 17 "Section **568h.** 20.835 (2) (bn) of the statutes is created to read:
- 18 20.835 (2) (bn) Dairy manufacturing facility investment credit. The amounts
- in the schedule to make the payments under ss. 71.07 (3p), 71.28 (3p), and 71.47
- 20 (3p).".
- 21 **729.** Page 395, line 11: delete lines 11 to 16.
- **730.** Page 395, line 17: delete lines 17 to 19.
- 23 **731.** Page 395, line 20: delete lines 20 to 25.

9

10

11

12

19

20

21

22

23

- 1 **732.** Page 396, line 1: delete that line.
- 2 **733.** Page 396, line 1: after that line insert:
- 3 "Section **572d.** 20.855 (4) (fd) of the statutes is created to read:
- 20.855 (4) (fd) Motor vehicle sales and use taxes; transportation fund. A sum sufficient to pay the amounts determined under s. 77.68 into the transportation fund.".
  - **734.** Page 396, line 1: after that line insert:
- 8 "Section 572b. 20.855 (4) (f) of the statutes is amended to read:
  - 20.855 (4) (f) Supplemental title fee matching. From the general fund, a sum sufficient equal to the amount of supplemental title fees collected under s. 342.14 (3m), as determined under s. 85.037, plus \$3,750,000 in each fiscal year, to be transferred to the environmental fund on October 1 annually.".
- 13 **735.** Page 396, line 2: delete lines 2 to 7.
- **736.** Page 396, line 8: delete lines 8 to 12.
- 737. Page 396, line 13: delete the material beginning with that line and ending with page 397, line 5.
- 17 **738.** Page 397, line 5: after that line insert:
- **"Section 581c.** 20.865 (2) (am) of the statutes is amended to read:
  - 20.865 **(2)** (am) *Space management and child care*. The amounts in the schedule to finance the unbudgeted costs of remodeling, moving, additional rental costs, and move-related vacant space costs resulting from relocations of state agencies directed by the department of administration, and the unbudgeted costs of assessments for child care facilities under s. 16.841 (4) incurred by state agencies.
    - **Section 581e.** 20.865 (2) (gm) of the statutes is amended to read:

2

3

4

5

6

7

8

9

10

11

12

13

14

20.865 (2) (gm) Space management and child care; program revenues. From the appropriate program revenue and program revenue-service accounts, a sum sufficient to finance the unbudgeted costs of remodeling, moving, additional rental costs, and move-related vacant space costs resulting from relocations of state agencies directed by the department of administration, and the unbudgeted costs of assessments for child care facilities under s. 16.841 (4) incurred by state agencies.

**SECTION 581m.** 20.865 (2) (gm) of the statutes is amended to read:

- 20.865 (2) (qm) Space management and child care; segregated revenues. From the appropriate segregated funds, a sum sufficient to finance the unbudgeted costs of remodeling, moving, additional rental costs, and move-related vacant space costs resulting from relocations of state agencies directed by the department of administration, and the unbudgeted costs of assessments for child care facilities under s. 16.841 (4) incurred by state agencies.".
- **739.** Page 397, line 12: delete "and" and substitute "and".
- **740.** Page 397, line 13: delete "and (bq),".
- **741.** Page 397, line 16: delete "(bt)," and substitute "(bt),".
- 742. Page 397, line 16: delete the material beginning with "and," and ending with "any" on line 18 and substitute "and interest on".
- 19 **743.** Page 398, line 1: decrease the underscored dollar amount by \$78,743,000.
- 21 **744.** Page 398, line 1: decrease the underscored dollar amount by \$17,500,000.
- **745.** Page 398, line 9: decrease the underscored dollar amount by \$5,000,000.

- **746.** Page 398, line 9: decrease the underscored dollar amount by \$2,950,000.
- **747.** Page 398, line 12: delete lines 12 to 23 and substitute:
  - **"Section 584b.** 20.866 (2) (ta) of the statutes is amended to read:
    - 20.866 (2) (ta) Natural resources; Warren Knowles-Gaylord Nelson stewardship 2000 program. From the capital improvement fund a sum sufficient for the Warren Knowles-Gaylord Nelson stewardship 2000 program under s. 23.0917. The state may contract public debt in an amount not to exceed \$572,000,000 \$717,000,000 for this program. Except as provided in s. 23.0917 (4g) (b), (4m) (k), (5) and (5m), the amounts obligated, as defined in s. 23.0917 (1) (e), under this paragraph may not exceed \$46,000,000 in fiscal year 2000-01, may not exceed \$46,000,000 in fiscal year 2001-02, and may not exceed \$60,000,000 in each fiscal year beginning with fiscal year 2009-10 2006-07, and may not exceed \$25,000,000 in each fiscal year beginning with fiscal year 2007-08 and ending with fiscal year 2019-20.".
      - **748.** Page 401, line 19: after that line insert:
  - "Section **591s.** 20.866 (2) (tz) of the statutes is amended to read:
    - 20.866 **(2)** (tz) Natural resources; Warren Knowles-Gaylord Nelson stewardship program. From the capital improvement fund a sum sufficient for the purposes specified in s. 23.0915 (1). The state may contract public debt in an amount not to exceed \$231,000,000 \$228,950,000 for this purpose.".
    - **749.** Page 401, line 20: delete the material beginning with that line and ending with page 402, line 2.
- **750.** Page 404, line 8: decrease the underscored dollar amount by \$65,000,000.

- 1 **751.** Page 404, line 16: after that line insert:
- 2 "Section **596jj.** 20.866 (2) (zbq) of the statutes is repealed.".
- **752.** Page 406, line 15: increase the underscored dollar amount by \$4,300,000.
- 5 **753.** Page 406, line 17: delete the material beginning with that line and ending with page 409, line 15, and substitute:
- 7 **"Section 606d.** 20.867 (3) (bt) of the statutes is repealed.".
- 8 **754.** Page 409, line 22: delete "and" and substitute "and".
- 9 **755.** Page 409, line 25: delete the material beginning with "<u>, and</u>" and ending with "<u>(a)</u>" on page 410, line 1.
- 11 **756.** Page 410, line 10: delete lines 10 to 25.
- 12 **757.** Page 411, line 1: delete lines 1 to 13.
- **758.** Page 411, line 14: delete lines 14 to 16 and substitute.
- **"Section 616.** 20.923 (4) (b) 7. of the statutes is repealed.".
- 15 **759.** Page 411, line 16: after that line insert:
- **"Section 616m.** 20.923 (4) (c) 1. of the statutes is repealed.".
- **760.** Page 411, line 17: delete lines 17 to 19.
- **761.** Page 411, line 21: delete lines 21 and 22.
- 19 **762.** Page 411, line 23: delete lines 23 and 24.
- 20 **763.** Page 411, line 25: delete the material beginning with that line and ending with page 412, line 4.
- 22 **764.** Page 412, line 4: after that line insert:

- 1 "Section **627m.** 20.923 (6) (ah) of the statutes is repealed.".
- 2 **765.** Page 412, line 5: delete lines 5 to 7.
- **766.** Page 412, line 10: delete lines 10 to 12 and substitute.
- 4 "Section **629.** 20.923 (6) (hr) of the statutes is repealed.".
- 5 **767.** Page 412, line 13: delete lines 13 to 16.
- 6 **768.** Page 412, line 17: delete lines 17 to 24.
- 7 **769.** Page 413, line 1: delete lines 1 to 3.
- 8 **770.** Page 413, line 4: delete lines 4 to 11.
- 9 **771.** Page 422, line 3: after that line insert:
- "Section 635q. 21.49 (2m) of the statutes is repealed.".
- 11 **772.** Page 422, line 4: delete the material beginning with that line and ending with page 423, line 2.
- 13 **773.** Page 423, line 15: after that line insert:

23

provided in sub. (2c):

- **"Section 638s.** 23.0915 (1) (intro.) of the statutes is amended to read:
- 23.0915 (1) Designated amounts. (intro.) The legislature intends that the department will expend the following designated amounts under the Warren Knowles-Gaylord Nelson stewardship program from the appropriation under s. 20.866 (2) (tz) for the following purposes in each fiscal year, the expenditures beginning with fiscal year 1990–91 and ending in fiscal year 1999–2000, except as
- **SECTION 638v.** 23.0915 (2c) of the statutes is repealed and recreated to read:
- 22 23.0915 (2c) COMMITMENTS FOR EXPENDITURE. (a) In this subsection, "commit

for expenditure" means to encumber, set aside, or otherwise commit.

2

3

4

5

6

7

8

9

16

17

18

19

20

21

22

23

(b) No moneys may be committed for expenditure from the appropriation under
s. $20.866(2)(tz)$ for a purpose under sub. $(1)(a)$ to $(n)$ beginning on the effective date
of this paragraph [revisor inserts date]. Any moneys that have been committed
for expenditure from the appropriation under s. $20.866\ (2)\ (tz)$ for a purpose under
$sub.\ (1)\ (a)\ to\ (n)\ before\ the\ effective\ date\ of\ this\ paragraph\\ [revisor\ inserts\ date],$
shall be provided from the appropriation under s. $20.866\ (2)\ (ta)$ and shall be treated
as moneys obligated from the subprogram under s. 23.0917 that encompasses that
purpose.".

- **774.** Page 423, line 15: after that line insert:
- **"Section 638r.** 23.0917 (2) (a) 3m. of the statutes is created to read:
- 11 23.0917 (2) (a) 3m. A subprogram for recreational boating aids.".
- 12 **775.** Page 423, line 15: after that line insert:
- "Section 638r. 23.0916 of the statutes is created to read:
- 23.0916 Stewardship land access. (1) Definitions. In this section:
- 15 (a) "Land" has the meaning given in s. 23.0917 (1) (d).
  - (b) "Stewardship land" means land that is acquired in whole or in part with funding from the stewardship program under s. 23.0917.
    - (2) REQUIREMENT OF ACCESS; NONDEPARTMENT LAND. (a) Except as provided in par. (b), any person receiving a grant on or after the effective date of this paragraph .... [revisor inserts date], that will be used to acquire stewardship land shall permit public access to the stewardship land for hunting, fishing, trapping, hiking, and cross-country skiing.
    - (b) If the land is acquired in fee simple or if the acquisition is an easement or other conveyance that does not consist of or include or otherwise involve an

- acquisition of development rights, the person may prohibit public access for hunting, fishing, trapping, hiking, or cross-country skiing, or a combination thereof, if the owner and the department jointly determine that it is necessary to do so in order to protect public safety.
- (3) REQUIREMENT OF ACCESS; DEPARTMENT LAND. (a) Except as provided in pars. (b) and (c) and ss. 29.089, 29.091, 29.301 (1) (b), and 29.621 (4), the department shall permit public access for hunting, fishing, trapping, hiking, and cross-country skiing by others on stewardship land that is acquired by the department on or after the effective date of this paragraph .... [revisor inserts date].
- (b) If the land is acquired in fee simple or if the acquisition is an easement or other conveyance that does not consist of or include or otherwise involve an acquisition of development rights, the department may prohibit public access for hunting, fishing, trapping, hiking, or cross-country skiing, or a combination thereof, if the department determines that it is necessary to do so in order to protect public safety.
- (c) The department may prohibit public access for hunting, fishing, or trapping, or any combination thereof, on stewardship land that is located in fish or game refuges.
- (4) REVIEW. The natural resources board, by rule, shall develop a process for the review of determinations made under subs. (2) (b) and (3) (b).
- (5) Reporting requirement. The department shall prepare an annual report that identifies all stewardship land that has been acquired during each fiscal year and upon which public access for hunting, fishing, trapping, hiking, or cross-county skiing is prohibited. For each acquisition, the report shall specify for which of these outdoor activities public access is prohibited and shall include the reason for the

18

20

21

22

23

- 1 prohibition. The department shall submit the report to the joint committee on  $\mathbf{2}$ finance and to the appropriate standing committees of the legislature in the manner 3 provided under s. 13.172 (3). The department shall submit the report no later than 4 November 15 for the preceding fiscal year and shall submit the first report no later than November 15, 2008.". 5 **776.** Page 424, line 4: substitute "2007-08" for "2010-11". 6
- **777.** Page 424, line 6: substitute "\$4,000,000" for "\$14,500,000". 7
- 8 **778.** Page 424, line 9: delete lines 9 to 11 and substitute:
- 9 "Section 642c. 23.0917 (3) (dm) 2. of the statutes is amended to read:
- 10 23.0917 (3) (dm) 2. For each fiscal year beginning with 2002-03 and ending 11 with fiscal year 2009-10 2006-07, \$45,000,000.
- 12 **Section 642d.** 23.0917 (3) (dm) 3. of the statutes is created to read:
- 13 23.0917 (3) (dm) 3. For fiscal year 2007-08, \$16,000,000.
- 14 **Section 642e.** 23.0917 (3) (dm) 4. of the statutes is created to read:
- 15 23.0917 (3) (dm) 4. For each fiscal year beginning with fiscal year 2008–09 and ending with fiscal year 2019-20, \$15,000,000.". 16
  - **779.** Page 424, line 18: delete the material beginning with that line and ending with page 425, line 2, and substitute:
- 19 **"Section 644d.** 23.0917 (4) (d) 1. of the statutes is amended to read:
  - 23.0917 (4) (d) 1. The department may obligate not more than \$11,500,000 in fiscal year 2000-01 and not more than \$11,500,000 in fiscal year 2001-02 under the subprogram except as provided in sub. (5). For each fiscal year beginning with 2002–03 and ending with fiscal year 2009–10 2006–07, the department may obligate not more than \$15,000,000 under the subprogram except as provided in sub. (5). For

each fiscal year beginning with fiscal year 2007-08 and ending with fiscal year 2019-20, the department may obligate not more than \$7,500,000 under the subprogram except as provided in sub. (5).".

**780.** Page 425, line 3: delete lines 3 to 10 and substitute:

**"Section 645b.** 23.0917 (4) (d) 2. of the statutes is amended to read:

23.0917 (4) (d) 2. The Beginning with fiscal year 2000–01 and ending with fiscal year 2006–07, the department may obligate not more than \$8,000,000 in each fiscal year for local assistance.

**SECTION 646b.** 23.0917 (4) (d) 2n. of the statutes is created to read:

23.0917 (4) (d) 2n. Beginning with fiscal year 2007–08 and ending with fiscal year 2019–20, the department may obligate not more than \$4,000,000 in each fiscal year for local assistance.

**SECTION 646m.** 23.0917 (4j) of the statutes is created to read:

23.0917 **(4j)** Recreational boating aids. (a) In this subsection "local governmental unit" means a city, village, town, or county, a lake sanitary district, as defined in s. 30.50 (4q), a public inland lake protection and rehabilitation district organized under ch. 33, or any other local governmental unit, as defined in s. 66.0131 (1) (a), that is established for the purpose of lake management.

(b) For fiscal year 2007–08, the department may not obligate more than \$1,500,000 for cost–sharing with local governmental units for recreational boating projects under s. 30.92. For each fiscal year beginning with fiscal year 2008–09 and ending with fiscal year 2019–20, the department may not obligate more than \$2,500,000 for cost–sharing with local governmental units for recreational boating projects under s. 30.92.".

**781.** Page 425, line 10: after that line insert:

**"Section 646r.** 23.0917 (5m) (a) of the statutes is amended to read:

23.0917 (5m) (a) Beginning in fiscal year 1999–2000, the department, subject to the approval of the governor and the joint committee on finance under sub. (6) (6m), may obligate under the subprogram for land acquisition any amount not in excess of the total bonding authority for that subprogram for the acquisition of land.

**Section 646t.** 23.0917 (6m) of the statutes is created to read:

23.0917 (6m) Review by joint committee on finance. (a) In addition to obtaining any necessary approval of the building commission under s. 13.48 or 13.488, the department may not obligate from the appropriation under s. 20.866 (2) (ta) for a given project or activity any moneys unless it first notifies the joint committee on finance in writing of the proposal. The committee may schedule a meeting to review the department's proposal only if at least 3 members of the committee object to the proposal. If the cochairpersons of the committee do not notify the department within 14 working days after the date of the department's notification that the committee has scheduled a meeting to review the proposal, the department may obligate the moneys. If, within 14 working days after the date of the notification by the department, the cochairpersons of the committee notify the department that the committee has scheduled a meeting to review the proposal, the department may obligate the moneys only upon approval of the committee unless par. (b) applies.

(b) If the committee does not hold the meeting to review the proposal within 60 days after the cochairpersons notify the department that a meeting has been scheduled, the department may obligate the moneys.

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

23

24

sub. (5m).

1	(c) The procedures under pars. (a) and (b) apply only to an amount for a project
2	or activity that exceeds \$500,000, except as provided in pars. (d) and (dm).
3	(d) The procedures under pars. (a) and (b) apply to any land acquisition under

- (dm) The procedures under pars. (a) and (b) apply to an amount for a project or activity that is less than or equal to \$500,000 if all of the following apply:
- 1. The project or activity is so closely related to one or more other department projects or activities for which the department has proposed to obligate or has obligated moneys under s. 20.866 (2) (ta) that the projects or activities, if combined, would constitute a larger project or activity that exceeds \$500,000.
- 2. The project or activity was separated from a larger project or activity by the department primarily to avoid the procedures under pars. (a) and (b).
- (e) This subsection does not apply to land acquired by the department under s. 24.59 (1).".
  - **782.** Page 425, line 17: after that line insert:
  - **"Section 647g.** 23.0917 (8) (e) of the statutes is created to read:
- 23.0917 (8) (e) Beginning with fiscal year 2007–08, the department may not obligate from the appropriation under s. 20.866 (2) (ta) more than 10 percent of the available bonding authority in a fiscal year for the acquisition of parcels of lands that are less than 10 acres in size.".
  - **783.** Page 426, line 5: delete lines 5 to 9 and substitute:
- 22 "**Section 648b.** 23.0917 (12) of the statutes is amended to read:
  - 23.0917 (12) EXPENDITURES AFTER JUNE 30, 2010 2020. If the remaining bonding authority for a subprogram under sub. (3) or, (4), or (4j) on June 30, 2010 2020, is an

6

7

8

9

10

11

12

13

14

15

16

17

18

19

- amount greater than zero, the department may expend any portion of this remaining bonding authority for that subprogram in one or more subsequent fiscal years.".
- 784. Page 426, line 10: delete the material beginning with that line and ending with page 427, line 9.
  - **785.** Page 427, line 14: delete lines 14 to 19 and substitute:
  - "(2) (a) Beginning with fiscal year 2010–11 and ending with fiscal year 2019–20, the department shall establish a grant program under which the department may award a grant to a county for any of the following:
    - 1. Acquisition of land for a county forest under s. 28.11.
  - 2. Acquisition of land for a project that promotes nature-based outdoor recreation or conservation and for which the department is requesting the county's assistance.
  - (b) Grants under this section shall be awarded from the appropriation under s. 20.866 (2) (ta), and, for purposes of s. 23.0917, shall be treated as moneys obligated from the subprogram under s. 23.0917 (3).".
  - **786.** Page 427, line 22: delete lines 22 to 25 and substitute:
  - "(4) A county may not convert the land, or any rights in the land, acquired with grant moneys awarded under sub. (2) (a) 2. to a use that is inconsistent with the type of nature-based outdoor recreation or conservation activity for which the grant was awarded unless the natural resources board approves the conversion."
- 21 **787.** Page 428, line 1: delete lines 1 to 16.
- **788.** Page 430, line 11: after that line insert:
- 23 "Section 658t. 23.197 (14) of the statutes is created to read:

23.197 (14) Antigo; trail development. From the appropriation under s. 20.866 (2) (ta), the department shall provide funding in an amount not to exceed \$600,000 to the city of Antigo for property development related to the ice age trail and the Springbrook trail located within the city. The funding authorized under this subsection shall be in a manner that, for every \$1 expended by the city of Antigo for the property development, the department shall provide \$1. For purposes of s. 23.0917, moneys provided from the appropriation under s. 20.866 (2) (ta) shall be treated as moneys obligated from either or both of the subprograms under s. 23.0917 (3) and (4).".

**789.** Page 434, line 5: after that line insert:

"Section 666m. 23.33 (11m) of the statutes is created to read:

23.33 (11m) LIGHTWEIGHT UTILITY VEHICLES PILOT PROGRAM. (a) In this subsection:

- 1. "Golf cart" means a vehicle whose speed attainable in one mile does not exceed 20 miles per hour on a paved, level surface, and is designed and intended to convey one or more persons and equipment to play the game of golf in an area designated as a golf course.
- 2. "Lightweight utility vehicle" means an engine-driven device having a gross weight of more than 700 pounds but not more than 1,999 pounds that is designed to travel on 4 or more low-pressure tires, is equipped with a cargo area, and is used primarily off a highway. "Lightweight utility vehicle" does not include golf carts or low-speed vehicles.

 $\mathbf{2}$ 

- 3. "Low pressure tire" means a tire that is designed to be mounted on a rim with a maximum diameter of 14 inches and to be inflated with an operating pressure not to exceed 20 pounds per square inch as recommended by the manufacturer.
- 4. "Low-speed vehicle" means a low-speed vehicle, as defined in 49 CFR 571.3, that satisfies the equipment standards under 49 CFR 571.500 and that was originally manufactured to meet the applicable equipment standards under 49 CFR 571.500. "Low-speed vehicle" does not include a golf cart.
  - 5. "Municipality" means a city, village, or town.
- (b) The department of natural resources, in consultation with the department of transportation, shall administer a pilot program to investigate the effects of using lightweight utility vehicles on trails and roadways that are used and authorized to be used by all-terrain vehicles, to evaluate whether it is feasible and appropriate to expand the allowable use of lightweight utility vehicles.
- (c) The counties of Florence, Forest, Sawyer, Marinette, Langlade, Lincoln, Oneida, and Washburn, and the municipalities within those counties, are eligible to participate in the pilot program, and the governing body of each county or municipality may elect to participate in the pilot program by adopting a resolution to that effect. The governing body of each county or municipality may withdraw from the pilot program prior to the end of the pilot program under par. (h) by adopting a resolution to that effect.
- (d) The counties and municipalities in the pilot program may designate any of the following:
- 1. All-terrain vehicle routes and trails within their respective jurisdictions that may be used by operators of lightweight utility vehicles.

 $\mathbf{2}$ 

- 2. All-terrain vehicle routes and trails within their respective jurisdictions upon which lightweight utility vehicle use is prohibited.
- (e) For the purposes of all of the following, a lightweight utility vehicle that is operated as authorized under this subsection is considered an all-terrain vehicle:
- 1. Sections 345.11 (1r), 346.02 (11), 349.02, 885.235 (1g) and (1k), 895.049, and 901.053.
  - 2. Subsections (3), (3g), (4), (4c) to (4x), (6), (7), (10), (12), and (13).
  - 3. Local ordinances enacted by a county or municipality under sub. (11).
  - (f) In addition to the provisions under par. (e), the operation of a lightweight utility vehicle as authorized under the pilot program is subject to all of the following:
  - 1. The operator of a lightweight utility vehicle must possess a valid motor vehicle operator's license.
  - 2. Any trail fees imposed on all-terrain vehicle use by a county or municipality also apply to operation of a lightweight utility vehicle.
  - (g) The department of natural resources, in consultation with the department of transportation and with the counties and municipalities participating in the pilot program, shall evaluate the effect of using lightweight utility vehicles on roadways and on all-terrain vehicle routes and trails upon conclusion of the pilot program. The department may make grants from the appropriation under s. 20.370 (5) (cu) to each participating county and municipality, for the purpose of assisting the department of natural resources in the evaluation. The department of natural resources shall make grants in such a manner that the total amount of grants for a given county, including the grants to municipalities located wholly or partially in that county, does not exceed \$2,000. The department of natural resources shall report the results of its evaluation to the legislature under s. 13.172 (2) no later than January 1, 2010.

11

12

13

14

15

16

17

18

19

21

22

23

- 1 (h) The pilot program under this subsection does not apply after September 30, 2009.".
- 3 **790.** Page 436, line 17: delete lines 17 and 18.
- 4 **791.** Page 436, line 19: delete lines 19 and 20.
- 5 **792.** Page 436, line 20: after that line insert:
- 6 "Section 678d. 25.17 (1) (hm) of the statutes is amended to read:
- 7 25.17 (1) (hm) Industrial building construction loan fund (s. 560.10, 2005 stats.);".
  - **793.** Page 437, line 3: after that line insert:
- "Section **679h.** 25.17 (70) (intro.) of the statutes is amended to read:
  - 25.17 (70) (intro.) No later than June 30 March 30 of every odd-numbered year, after receiving a report from the department of commerce the information required under s. 560.08 (2) (m), and in consultation with the department of commerce, submit to the governor and to the presiding officer of each house of the legislature a plan for making investments in this state. The purpose of the plan is to encourage the board to make the maximum amount of investments in this state, subject to s. 25.15 and consistent with the statutory purpose of each trust or fund managed by the board. The plan shall discuss potential investments to be made during the first to 5th fiscal years following submittal, and shall include, but not be limited to, the following:
- **Section 679j.** 25.17 (70) (a) of the statutes is amended to read:
  - 25.17 (70) (a) A report from the department of commerce containing the information required under s. 560.08 (2) (m) describing the types of investments in businesses in this state which will have the greatest likelihood of enhancing economic development in this state."

11

12

13

14

15

16

17

18

19

20

21

22

23

24

1 **794.** Page 438, line 6: after that line insert:

2 "Section 686t. 25.40 (1) (a) 3. of the statutes is amended to read:

3 25.40 (1) (a) 3. Revenues collected under ss. 341.09 (2) (d), (2m) (a) 1., (4), and (7), 341.14 (2), (2m), (6) (d), (6m) (a), (6r) (b) 2., (6w), (6y), and (8), 341.145 (3), 341.16 (1) (a) and (b) to (c), (2), and (2m), 341.17 (8), 341.19 (1) (a), 341.25, 341.255 (1), (2) (a), (b), and (c), (4), and (5), 341.26 (1), (2), (2m) (am) and (b), (3), (3m), (4), (5), and (7), 341.264 (1), 341.265 (1), 341.266 (2) (b) and (3), 341.268 (2) (b) and (3), 341.30 (3), 341.305 (3), 341.308 (3), 341.36 (1) and (1m), 341.51 (2), and 342.14, except s. 342.14

9 (1r), that are pledged to any fund created under s. 84.59 (2).".

**795.** Page 438, line 7: delete lines 7 and 8.

**796.** Page 438, line 8: after that line insert:

"Section 687d. 25.40 (1) (bn) of the statutes is created to read:

25.40 (1) (bn) From the amount determined under s. 77.68, all moneys transferred from the appropriation account under s. 20.855 (4) (fd).".

**797.** Page 438, line 8: after that line insert:

"Section 687c. 25.40 (3) (c) of the statutes is created to read:

25.40 (3) (c) No executive biennial budget bill introduced under s. 16.47 (1m) may include any provision that amends or repeals any provision of this subsection or that applies notwithstanding any provision of this subsection.".

**798.** Page 438, line 8: after that line insert:

"Section 688n. 25.43 (3) of the statutes is amended to read:

25.43 (3) Except for the purpose of investment as provided in s. 25.17 (2) (d), the environmental improvement fund may be used only for the purposes authorized under ss. 20.320 (1) (r), (s), (sm), (t), (x) and (y), and (2) (s) and (x) and (3) (q), 20.370

- 1 (4) (mt), (mx) and (nz), (8) (mr) and (9) (mt), (mx) and (ny), 20.505 (1) (v), (x) and (y),
- 2 281.58, 281.59, 281.60, 281.61 and 281.62.".
- 3 **799.** Page 438, line 9: delete lines 9 to 11.
- 4 **800.** Page 438, line 12: delete lines 12 to 15.
- 5 **801.** Page 438, line 15: after that line insert:
- 6 "Section **690v.** 25.47 of the statutes is renumbered 25.47 (1).
- 7 **Section 690x.** 25.47 (2) of the statutes is created to read:
- 8 25.47 (2) (a) Moneys deposited in the petroleum inspection fund that are not appropriated may not be transferred from the petroleum inspection fund to any other fund or to any appropriation account in any other fund.
- 11 (b) Moneys that are deposited in the petroleum inspection fund may not be 12 appropriated for purposes other than purposes for which appropriations from the 13 petroleum inspection fund are made on the effective date of this paragraph .... 14 [revisor inserts date].".
- 15 **802.** Page 438, line 20: delete "family long-term" and substitute "family".
- 16 **803.** Page 439, line 3: delete lines 3 to 6.
- 17 **804.** Page 439, line 12: delete lines 12 to 19.
- 18 **805.** Page 440, line 1: delete "health care quality fund \$50,000,000 in each"

  19 and substitute "medical assistance trust fund \$50,000,000 in each".
- 20 **806.** Page 440, line 5: delete lines 5 to 14.
- 21 **807.** Page 442, line 4: delete that line.
- 808. Page 442, line 19: delete the material beginning with that line and ending with page 443, line 13.

- **809.** Page 443, line 20: delete the material beginning with that line and ending with page 444, line 2.
- **810.** Page 444, line 14: delete lines 14 to 25.
- **811.** Page 445, line 1: delete lines 1 to 23.
- **812.** Page 446, line 9: delete lines 9 to 15.
- **813.** Page 446, line 18: delete lines 18 to 25.
- **814.** Page 447, line 1: delete lines 1 to 24.
- **815.** Page 448, line 2: delete lines 2 to 14.
- **816.** Page 449, line 14: delete lines 14 and 15 and substitute:
- 10 "(2) No person may take shovelnose sturgeon or shovelnose sturgeon eggs
  11 unless the person holds a permit from the department under this section.".
- **817.** Page 449, line 16: delete "holds" and substitute "is required to hold".
- **818.** Page 449, line 17: delete "provided by" and substitute "available from".
- **819.** Page 449, line 22: delete the material beginning with that line and ending with page 451, line 3.
- **820.** Page 451, line 6: delete the material beginning with that line and ending with page 452, line 16.
- **821.** Page 455, line 23: delete that line.
- **822.** Page 456, line 1: delete lines 1 to 11.
- **823.** Page 457, line 8: after that line insert:
- **"Section 725k.** 30.62 (2) (g) 5. of the statutes is created to read:

1	30.62 (2) (g) 5. An airboat while used by a person who is trapping during an
2	open season for trapping.".
3	<b>824.</b> Page 457, line 18: after that line insert:
4	"Section 726ad. 30.77 (3) (e) 1. (intro.) of the statutes is amended to read:
5	30.77 (3) (e) 1. (intro.) A municipality, a public inland lake protection and
6	rehabilitation district or a town sanitary district that has in effect an ordinance
7	under par. (am) may charge <del>boat operators reasonable fees for any a boat operator</del>
8	one or more of the following:
9	<b>Section 726af.</b> 30.77 (3) (e) 1. a. of the statutes is amended to read:
10	30.77 (3) (e) 1. a. Use A reasonable fee for the use of a public boat launching
11	facility that the municipality or lake district owns or operates.
12	Section 726ah. 30.77 (3) (e) 1. am. of the statutes is created to read:
13	30.77 (3) (e) 1. am. A fee that does not exceed more than 20 percent of the
14	amount specified in s. 27.01 $(7)$ $(f)$ 2., for the use of a boat launch facility that the
15	municipality or district owns or operates
16	<b>Section 726aj.</b> 30.77 (3) (e) 1. b. of the statutes is amended to read:
17	30.77 (3) (e) 1. b. The <u>A reasonable fee for the</u> municipality's or district's costs
18	for operating or maintaining a water safety patrol unit, as defined in s. $30.79(1)(b)$
19	2.
20	Section 726aL. 30.77 (3) (e) 1. c. of the statutes is amended to read:
21	30.77 (3) (e) 1. c. The A reasonable fee for the municipality's or district's costs
22	for providing other recreational boating services.
23	Section 726an. 30.77 (3) (e) 1m. of the statutes is created to read:

1	30.77 (3) (e) 1m. Any fee collected under subd. 1. am. shall be used for operating
2	or maintaining a water safety patrol unit, as defined in s. $30.79(1)(b)2.$ ".
3	<b>825.</b> Page 461, line 3: after that line insert:
4	"Section 726u. 35.84 (figure) column A lines 10 and 11 of the statutes are
5	amended to read:
6	35.84 (figure) column A Statutes, Hard Covers, s. 35.18
7	10. Each Senator
8	11. Each Representative to the Assembly
9	Section 726v. 35.84 (figure) column A line 60 of the statutes is amended to
10	read:
11	60. Each member and member-elect of Congress from this state $\dots 1\underline{A}$
12	Section 726w. 35.84 (figure) column B lines 71 to 77 of the statutes are
13	amended to read:
14	35.84 (figure) column B Statutes; Soft Covers; s. 35.18
15	71. Each county board chairperson
16	72. Each county clerk
17	73. Each city clerk
18	74. Each town clerk
19	75. Each village clerk
20	76. Each sheriff
21	77. Each county corporation counsel
22	SECTION 726x. 35.84 (figure) column B lines 79 to 84 of the statutes are
23	amended to read:
24	35.84 (figure) column B Statutes; Soft Covers; s. 35.18

1	79. Each register of deeds
2	80. Each coroner or medical examiner
3	81. Each county treasurer
4	82. Each county surveyor
5	83. Each county human services or social services
6	department director
7	84. Each county veterans' service officer $1\underline{A}$ ".
8	<b>826.</b> Page 461, line 4: delete lines 4 to 17.
9	<b>827.</b> Page 464, line 6: after that line insert:
10	"Section 729k. 36.11 (1) (b) of the statutes is amended to read:
11	36.11 (1) (b) Except as provided in this paragraph, the board may purchase,
12	have custody of, hold, control, possess, lease, grant easements and enjoy any lands,
13	buildings, books, records and all other property of any nature which may be
14	necessary and required for the purposes, objects and uses of the system authorized
15	by law. Any lease is subject to the powers of the University of Wisconsin Hospitals
16	and Clinics Authority under s. 233.03 (13) and the rights of the authority under any
17	lease agreement, as defined in s. 233.01 (6). The board shall not permit a facility that
18	would be privately owned or operated to be constructed on state-owned land without
19	obtaining prior approval of the building commission under s. 13.48 (12). The $\underline{\text{Except}}$
20	where a sale occurs under s. 16.848, the board may sell or dispose of such property
21	as provided by law, or any part thereof when in its judgment it is for the best interests
22	of the system and the state. All purchases and sales of real property shall be subject
23	to the approval of the building commission. The provision of all leases of real

- property to be occupied by the board shall be the responsibility of the department of administration under s. 16.84 (5).
- **Section 729m.** 36.11 (1) (e) of the statutes is amended to read:
  - 36.11 (1) (e) The board, with the approval of the building commission, may sell or lease state-owned residence halls to another state agency or nonstate nonprofit agency for purposes of alternate use. This paragraph does not apply to property that is sold under s. 16.848.".
  - **828.** Page 464, line 7: delete lines 7 to 18.
- **829.** Page 464, line 18: after that line insert:
- "Section 731g. 36.11 (22) (d) of the statutes is amended to read:
  - 36.11 (22) (d) Annually, each institution shall report to the office of justice assistance in the department of administration justice statistics on sexual assaults and on sexual assaults by acquaintances of the victims that occurred on each campus of the institution in the previous year. The office department of justice assistance shall include the statistics in appropriate crime reports published by the office department.".
- **830.** Page 465, line 4: after that line insert:
- 18 "Section 732d. 36.11 (48m) of the statutes is created to read:
  - 36.11 (48m) MONTHLY PAY PERIOD. As soon as practicable after the effective date of this subsection .... [revisor inserts date], the board shall convert the system payroll system from a biweekly pay period to a monthly pay period.".
- **831.** Page 466, line 14: delete lines 14 and 15.
- **832.** Page 466, line 15: after that line insert:
  - **"Section 732ye.** 36.27 (1) (a) of the statutes is amended to read:

36.27 (1) (a) Subject to pars. (am), (ar), (b), and (c), the board may establish for different classes of students differing tuition and fees incidental to enrollment in educational programs or use of facilities in the system. Except as otherwise provided in this section, the board may charge any student who is not exempted by this section a nonresident tuition. The board may establish special rates of tuition and fees for the extension and summer sessions and such other studies or courses of instruction as the board deems advisable.

**SECTION 732ym.** 36.27 (1) (am) (intro.) of the statutes is amended to read:

36.27 (1) (am) (intro.) The Except as provided in par. (ar), the board may not increase academic fees for resident undergraduate students beyond an amount sufficient to fund all of the following:

**Section 732ys.** 36.27 (1) (ar) of the statutes is created to read:

- 36.27 **(1)** (ar) 1. In fiscal years 2007–08, 2008–09, 2009–10, and 2010–11, the board may not do any of the following:
- a. Increase academic fees for undergraduate students by more than 4 percent above the academic fees in effect for the previous fiscal year.
- b. Except as provided in subd. 2., increase the total nonallocable segregated fees for undergraduate students by more than 4 percent above the total nonallocable segregated fees in effect for the previous fiscal year.
- 2. If a majority of the undergraduate students enrolled in an institution who vote in a referendum that is open only to undergraduate students approve an increase in the total nonallocable segregated fees for undergraduate students for one or more fiscal years by an amount greater than the maximum increase allowed in subd. 1. b., the board may increase total nonallocable segregated fees for undergraduate students at the institution for the fiscal year or years by the amount

2

3

4

5

6

7

8

9

10

12

13

14

15

16

17

18

19

20

21

- approved in the referendum. This subdivision applies only if all of the following are satisfied:
  - a. More than 10 percent of the undergraduate students enrolled in the institution at the time of the referendum vote to approve the increase.
    - b. The ballot used in the referendum identifies the amount of the increase and the fiscal year or years to which the increase applies and explains the reasons for the increase.".
  - **833.** Page 466, line 16: delete the material beginning with that line and ending with page 467, line 2.
    - **834.** Page 467, line 8: after that line insert:
- 11 "Section 733mr. 36.27 (3n) (b) 2. of the statutes is amended to read:
  - 36.27 (3n) (b) 2. An Except as provided in subd. 2m., an unremarried surviving spouse of an eligible veteran. The remission under this subdivision applies only during the first 10 years after the veteran died.
    - **Section 733mw.** 36.27 (3n) (b) 2m. of the statutes is created to read:
    - 36.27 (3n) (b) 2m. An unremarried surviving spouse of an eligible veteran who had a child with the eligible veteran. The remission under this subdivision applies only until 10 years after the youngest child that the spouse had with the eligible veteran reaches or would have reached 18 years of age, or during the first 10 years after the veteran died, whichever is longer.".
      - **835.** Page 467, line 16: delete lines 16 to 19.
  - **836.** Page 467, line 20: delete lines 20 to 24 and substitute:
- 23 "Section 735g. 36.27 (3p) (b) of the statutes is renumbered 36.27 (3p) (b) 1. and 24 amended to read:

36.27 (3p) (b) 1. The Except as provided in subd. 2. and par. (bm), the board
shall grant full remission of nonresident tuition, academic fees, and segregated fees
charged for 128 credits or 8 semesters, whichever is longer, less the amount".

- **837.** Page 468, line 2: after "veteran." insert "A student who at any time is granted a remission under par. (bg) is not eligible for a remission under this subdivision.".
  - **838.** Page 468, line 2: after that line insert:
- "Section 735g. 36.27 (3p) (b) 2. of the statutes is created to read:
  - 36.27 (**3p**) (b) 2. The board shall grant a remission under subd. 1. to a person for the lesser of the following, less the number of credits or semesters for which the person received remission of fees under s. 38.24 (8):
  - a. One hundred twenty-eight credits or 8 semesters, whichever is longer.
  - b. Until completion of a sufficient number of credits to be awarded a bachelor's degree in the person's major field of study.
  - **Section 735m.** 36.27 (3p) (bg) of the statutes is created to read:
    - 36.27 (**3p**) (bg) 1. Except as provided in par. (bm), the board shall grant remission of nonresident tuition, academic fees, and segregated fees charged for 48 credits or until completion of a sufficient number of credits to be awarded a graduate degree in the student's field of study, whichever is less, less the amount of any academic fees or segregated fees paid under 10 USC 2107 (c) or 38 USC 3104 (a) (7) (A), to any student enrolled as a graduate student who is a veteran. A student who at any time after January 1, 2008, was granted a remission under par. (b) 1. or s. 38.24 (8) (b) is not eligible for a remission under this paragraph.

- 2. The amount of a remission granted under subd. 1. to a graduate student may not exceed the amount of a remission granted under par. (b) 1. to a resident undergraduate student at the same institution for the same number of credits.".
  - **839.** Page 468, line 8: delete lines 8 to 11.
- **840.** Page 469, line 8: after that line insert:
- **"Section 736tr.** 36.30 of the statutes is amended to read:
  - **36.30 Sick leave.** Leave of absence for persons holding positions under s. 20.923 (4g) and (5), faculty and academic staff personnel with pay, owing to sickness, shall be regulated by rules of the board, except that unused sick leave shall accumulate from year to year. A person who is initially employed in a position specified in this subsection after the effective date of this section .... [revisor inserts date], shall only receive 6 paid sick leave days in any year."
    - **841.** Page 475, line 19: after that line insert:
  - "(7) Reports. No later than March 1 and September 1 of each year, the Board of Regents shall submit to the joint committee on information policy and technology a report that documents for each information technology project within the system with an actual or projected cost greater than \$1,000,000 or that the board has identified as a large, high-risk information technology project under sub. (2) (a) all of the following:
    - (a) Original and updated project cost projections.
  - (b) Original and updated completion dates for the project and any stage of the project.
  - (c) An explanation for any variation between the original and updated costs and completion dates under pars. (a) and (b).

1	(d) A copy of any contract entered into by the board for the project and not
2	provided in a previous report.
3	(e) All sources of funding for the project.
4	(f) The amount of any funding provided for the project through a master lease
5	under s. 16.76 (4).
6	(g) Information about the status of the project, including any portion of the
7	project that has been completed.
8	(h) Any other information about the project, or related information technology
9	projects, requested by the joint committee on information policy and technology.".
10	<b>842.</b> Page 475, line 20: delete lines 20 to 25.
11	<b>843.</b> Page 475, line 25: after that line insert:
12	"Section 737m. 38.14 (4) of the statutes is amended to read:
13	38.14 (4) GIFTS AND GRANTS. The district board may accept gifts, grants and
14	bequests to be used in the execution of its functions and may accept grants to provide
15	fiscal and management services relating to the duties of the former office of justice
16	assistance in the department of administration that are described under s. 16.964,
17	2005 stats., for the office of justice assistance in the department of administration
18	justice or its subsidiaries or, if applicable, its successor agency.".
19	<b>844.</b> Page 475, line 25: after that line insert:
20	"Section 737m. 38.17 of the statutes is created to read:
21	38.17 Levy limit. (1) Definition. In this section:
22	(a) "Debt service" includes debt service on debt issued or reissued to fund or
23	refund outstanding municipal obligations, interest on outstanding municipal
24	obligations, and related issuance costs and redemption premiums.

- (b) "Valuation factor" means a percentage equal to the average percentage change in the statewide equalized valuation due to new construction, less improvements removed, as determined for the January 1 equalized valuations in the 5 years preceding the levy, but not less than zero.
- (2) LIMIT. (a) Except as provided in subs. (3) and (4), no district board may levy in 2007 more than it levied in the previous year increased by the valuation factor.
- (b) Except as provided in subs. (3) and (4), no district board may levy in 2008 or 2009 more than the amount determined as follows:
  - 1. Increase the amount levied in the previous year by the valuation factor.
- 2. If the district board offers a collegiate transfer program, reduce the amount determined under subd. 1. by the difference between the amount generated by the district in that school year under s. 38.24 (1m) (a) and the amount that would have been generated by the district in that school year under s. 38.24 (1m) (a) if the applicable percentage under s. 38.24 (1m) (a) had been 37 percent.
- (3) Adjustments. (a) 1. If a district board transfers to another governmental unit responsibility for providing any service that it provided in the preceding fiscal year, the limit otherwise applicable under sub. (2) in the current fiscal year is decreased by the cost that it would have incurred to provide that service, as determined by the department of revenue.
- 2. If a district board increases the services that it provides by adding responsibility for providing a service transferred to it from another governmental unit that provided the service in the previous fiscal year, the limit otherwise applicable under sub. (2) in the current fiscal year is increased by the cost of that service, as determined by the department of revenue.

 $\mathbf{2}$ 

- (b) 1. If the amount of debt service for a district board in the preceding fiscal year is less than the amount of debt service needed in the current fiscal year, as a result of the district board adopting a resolution before July 1, 2007, authorizing the issuance of debt, the limit otherwise applicable under sub. (2) for the current fiscal year is increased by the difference between the 2 amounts, as determined by the department of revenue.
- 2. The limit otherwise applicable under this section does not apply to amounts levied by a district board for the payment of any general obligation debt service, including debt service on debt issued or reissued to fund or refund outstanding municipal obligations, interest on outstanding municipal obligations, or the payment of related issuance costs or redemption premiums, authorized on or after July 1, 2007, by a referendum and secured by the full faith and credit of the district.
- (4) Referendum. (a) 1. A district board may exceed the levy limit under sub. (2) if it adopts a resolution to that effect and the resolution is approved in a referendum. The resolution shall specify the proposed amount of increase in the levy beyond the amount that is allowed under sub. (2), and shall specify whether the proposed amount of increase is for the next fiscal year only or if it will apply on an ongoing basis.
- 2. Except as provided in subd. 3., the district board may call a special referendum for the purpose of submitting the resolution to the electors of the district for approval or rejection.
- 3. A referendum to exceed the limit under sub. (2) for the 2008 levy shall be held at the spring primary or election or September primary or general election in 2008.

- (b) The district board shall publish type A, B, C, D, and E notices of the referendum under s. 10.01 (2). Section 5.01 (1) applies in the event of failure to comply with the notice requirements of this paragraph.
- (c) The referendum shall be held in accordance with chs. 5 to 12. The district board shall provide the election officials with all necessary election supplies. The form of the ballot shall correspond substantially with the standard form for referendum ballots prescribed by the elections board under ss. 5.64 (2) and 7.08 (1) (a). The question shall be submitted as follows: "Under state law, the percentage increase in the levy of the .... (name of district) for the .... (next) fiscal year is limited to .... percent, resulting in a levy of \$..... Shall the .... (name of district) be allowed to exceed this limit such that the percentage increase for the .... (next) fiscal year will be .... percent, resulting in a levy of \$....?"
- (d) Within 14 days after the referendum, the district board shall certify the results of the referendum to the department of revenue. The limit otherwise applicable to the district under sub. (2) is increased for the next fiscal year by the amount approved by a majority of those voting on the question. If the resolution specifies that the increase is for one year only, the amount of the increase shall be subtracted from the base used to calculate the limit for the 2nd succeeding fiscal year.
- (5) Penalty. The department of revenue shall notify the board of any amount levied by a district board that exceeds the district's limit under this section. The board shall reduce the district's state aid under s. 38.28 in the same fiscal year in which the excess levy occurred by an amount equal to the amount of the excess levy. The amount of the reduction shall lapse to the general fund.".

## **845.** Page 475, line 25: after that line insert:

1	"Section 373g. 38.16 (3) of the statutes is created to read:
2	38.16 (3) Notwithstanding sub. (1), beginning in 2010 and annually thereafter
3	any district board that offers a collegiate transfer program shall reduce its property
4	tax levy by an amount equal to the amount of revenue generated by the district in
5	that school year under s. 38.24 (1m) (a) less the amount of revenue that would have
6	been generated by the district in that school year under s. 38.24 (1m) (a) if the
7	applicable percentage under s. 38.24 (1m) (a) had been 37 percent.
8	SECTION 737r. 38.175 of the statutes is created to read:
9	38.175 Enterprise centers. (1) In the section, "enterprise center" means a
10	revenue - generating operation such as a bookstore, cafeteria, or day care center
11	"Enterprise center" includes a public broadcasting station.
12	(2) Beginning in the 2009–10 school year, a district board may not do any of the
13	following:
14	(a) Use property taxes or state aid to fund the operation of an enterprise center
15	(b) Transfer more than 10 percent of the revenue generated by an enterprise
16	center in any school year to fund the operation of another enterprise center.".
17	<b>846.</b> Page 476, line 1: delete lines 1 to 11.
18	847. Page 476, line 11: after that line insert:
19	"Section 738d. 38.24 (1m) (a) of the statutes is renumbered 38.24 (1m) (a)
20	(intro.) and amended to read:
21	38.24 (1m) (a) Liberal arts collegiate transfer programs. (intro.) Uniform fees
22	based on not less than 31% the following percentage of the statewide average
23	operational costs of liberal arts collegiate transfer programs in district schools-:
24	<b>Section 738g.</b> 38.24 (1m) (a) 1. to 3. of the statutes are created to read:

 $\mathbf{2}$ 

4

5

6

7

8

9

10

11

12

13

14

15

18

20

21

1 38.24 ( <b>1m</b> ) (a) 1. In the 2007–08 school year, 31 percent.
--

- 2. In the 2008–09 school year, 42 percent.
- 3. In the 2009–10 school year and in each school year thereafter, 50 percent.
  - **SECTION 738i.** 38.24 (1m) (b) of the statutes is amended to read:
    - 38.24 (1m) (b) Postsecondary and vocational-adult programs. Uniform fees based on not less than 14% of the combined estimated statewide operational cost of postsecondary, exclusive of collegiate transfer, and vocational-adult programs. The board shall maintain statewide uniformity in the program fees charged for postsecondary and vocational-adult credits. Students 62 years old and over shall be exempted from program fees under this paragraph in vocational-adult programs. Students enrolled in adult high school, including students enrolled under s. 118.15 (1) (cm) 3.7 and adult basic education and English as a 2nd language courses, shall be exempted from program fees under this paragraph. The board shall establish fees under this paragraph as if students exempt from fees under sub. (4) were not exempt.
    - **Section 738k.** 38.24 (1m) (e) of the statutes is created to read:
- 16 38.24 (1m) (e) English as a 2nd language. Uniform fees for English as a 2nd language courses.".
  - **848.** Page 476, line 18: after that line insert:
- 19 **"Section 738mr.** 38.24 (7) (b) 2. of the statutes is amended to read:
  - 38.24 (7) (b) 2. An Except as provided in subd. 2m., an unremarried surviving spouse of an eligible veteran. The remission under this subdivision applies only during the first 10 years after the veteran died.
- **Section 738mw.** 38.24 (7) (b) 2m. of the statutes is created to read:

2

3

4

- 38.24 (7) (b) 2m. An unremarried surviving spouse of an eligible veteran who had a child with the eligible veteran. The remission under this subdivision applies only until 10 years after the youngest child that the spouse had with the eligible veteran reaches or would have reached 18 years of age, or during the first 10 years after the veteran died, whichever is longer.".
- 6 **849.** Page 477, line 1: delete lines 1 to 4.
- 850. Page 477, line 11: after "veteran." insert "A student who at any time is granted a remission under s. 36.27 (3p) (bg) is not eligible for a remission under this paragraph.".
- 10 **851.** Page 477, line 17: delete lines 17 to 20.
- 11 **852.** Page 477, line 21: delete lines 21 to 25.
- 12 **853.** Page 478, line 1: delete lines 1 to 6.
- 13 **854.** Page 478, line 7: delete lines 7 to 24 and substitute:
- **"Section 743n.** 38.41 (2) of the statutes is created to read:
- 15 38.41 (2) The board shall award at least two-thirds of the total amount 16 awarded under this section in any school year to businesses located in this state that 17 have no more than 100 employees.".
- 18 **855.** Page 479, line 1: delete lines 1 to 18.
- 19 **856.** Page 479, line 18: after that line insert:
- 20 "Section 743s. 39.387 of the statutes is created to read:
- 39.387 Loan program for physicians in shortage areas. (1) The board shall establish a loan program to defray the cost of tuition, fees, and expenses for residents of this state who are enrolled in a program at the University of Wisconsin

- School of Medicine and Public Health or at the Medical College of Wisconsin, Inc., leading to the degree of doctor of medicine and who agree to practice medicine for not less than 6 years in a health professional shortage area, as defined in s. 560.183 (1) (aj), in this state.
- (2) The board shall make loans under sub. (1) from the appropriation account under s. 20.235 (1) (cd). The maximum amount of a loan that a person may receive during any fiscal year is \$10,000. The maximum amount that a person may receive under this section is \$50,000. The terms of a loan shall provide that the loan recipient is not required to repay the loan while the recipient is enrolled in a program described in sub. (1).
- (3) After the recipient of a loan under sub. (1) has completed the program described in sub. (1), the board shall forgive 10 percent of the loan's principal and interest after the first full year, 10 percent of the loan's principal and interest after the 2nd full year, 10 percent of the loan's principal and interest after the 3rd full year, 10 percent of the loan's principal and interest after the 4th full year, 20 percent of the loan's principal and interest after the 5th full year, and 20 percent of the loan's principal and interest after the 6th full year that the recipient has been employed full time as a doctor of medicine in an area described in sub. (1). The board may forgive loans on a prorated basis for persons who are employed less than full time.
- (4) The board shall deposit into the general fund as general purpose revenue
   earned all repayments of loans made under sub. (1) and the interest on those loans.
- (5) The board shall promulgate rules to implement and administer this section.".
  - **857.** Page 479, line 18: after that line insert:

"Section **743u.** 39.397 of the statutes is created to read:

39.397 Loan program for veterinarians of food-producing animals. (1) The board shall establish a loan program to defray a portion of the cost of tuition, fees, and expenses for persons who are enrolled at an accredited school of veterinary medicine in this state in a curriculum leading to a doctor of veterinary medicine degree and who agree to engage full time in this state for not less than 6 years in a veterinary medicine practice in which not less than 75 percent of the revenue produced by the loan recipient is derived from providing veterinary medical services to food-producing animals, as defined in s. 453.02 (4m).

- (2) The board shall make loans under sub. (1) from the appropriation account under s. 20.235 (1) (cp). The maximum amount of a loan that a person may receive during any fiscal year is \$12,500. The maximum amount that a person may receive under this section is \$50,000. The terms of a loan shall provide that the loan recipient is not required to repay the loan while the recipient is enrolled in the curriculum described in sub. (1) or during any period of loan deferment authorized by the board under rules promulgated under sub. (5).
- (3) After the recipient of a loan under sub. (1) has completed the curriculum described in sub. (1), the board shall forgive 10 percent of the loan's principal and interest after the first full year, 10 percent of the loan's principal and interest after the 2nd full year, 10 percent of the loan's principal and interest after the 3rd full year, 10 percent of the loan's principal and interest after the 4th full year, 20 percent of the loan's principal and interest after the 5th full year, and 20 percent of the loan's principal and interest after the 6th full year that the recipient has been employed full time in this state in a veterinary medicine practice described in sub. (1). The board may forgive loans on a prorated basis for persons who are employed less than full

time. If a loan recipient ceases employment in this state in a veterinary medicine
practice described in sub. (1) during the time period allowed for forgiveness or
if after the end of that time period there remains any unpaid balance on the loan, the
loan recipient shall repay the unpaid balance of the loan plus interest at a rate
determined by the board by rule promulgated under sub. (5).

- (4) The board shall deposit in the general fund as general purpose revenue earned all repayments of loans made under sub. (1) and the interest on those loans.
- (5) The board shall promulgate rules to implement and administer this section, including all of the following:
- (a) Rules relating to verification that a person has been employed as required under sub. (3).
- (b) Rules providing circumstances under which the board may defer repayment of a loan.
- (c) Rules establishing an interest rate for loans that are not forgiven and must be repaid.".
  - **858.** Page 480, line 9: delete lines 9 to 25.
- **859.** Page 481, line 1: delete lines 1 to 24.
- **860.** Page 482, line 1: delete lines 1 to 10.
- **861.** Page 482, line 10: after that line insert:
- 20 "Section 748t. 39.47 (1) of the statutes is amended to read:
  - 39.47 (1) There is established, to be administered by the board, a Minnesota-Wisconsin student reciprocity agreement, the purpose of which shall be to ensure that neither state shall profit at the expense of the other and that the determination of any amounts owed by either state under the agreement shall be

based on an equitable formula which reflects the educational costs incurred by the 2 states, reflects any differentials in usage by residents of either state of the public institutions of higher education located in the other state, and reflects any differentials in the resident tuition charged at comparable public institutions of higher education of the 2 states. The board, representing this state, shall enter into an agreement meeting the requirements of this section with the designated body representing the state of Minnesota.

**Section 748v.** 39.47 (3) of the statutes is amended to read:

39.47 (3) Annually, each state shall determine the number of students for whom nonresident tuition has been waived under the agreement. Each state shall certify to the other state, in addition to the number of students so determined, the aggregate amount of its reimbursement obligation. The state with the smaller larger reimbursement obligation shall receive from the other state pay as provided in the agreement an amount determined by subtracting the reimbursement obligation of the state receiving the payment with the smaller reimbursement obligation from the reimbursement obligation of the state making the payment with the larger reimbursement obligation. The agreement shall provide a reasonable date for payment of any such sums due and owing to either state, after which date interest may be charged on the amount owed. The methodology for determination of the appropriate interest rate shall be included in the agreement. Any payments received by this state under this subsection shall be deposited in the general fund."

- **862.** Page 482, line 11: delete lines 11 to 25.
- **863.** Page 483, line 1: delete lines 1 to 16.

864.	Page 483, lin	ne 17:	delete	the	material	beginning	with	that	line	and
ending with	page 484, lin	e 25.								

**865.** Page 484, line 25: after that line insert:

**"Section 759r.** 40.02 (54) (g) of the statutes is repealed.".

**866.** Page 485, line 2: after that line insert:

**"Section 762h.** 40.05 (1) (b) 4. of the statutes is created to read:

40.05 (1) (b) 4. A participating employer that is a municipal employer, as defined in s. 111.70 (1) (j), may not pay, on behalf of a participating employee, the first 3 percent of earnings that a participating employee is required to pay as contributions under par. (a) if that employee first becomes a participating employee on or after the effective date of this subdivision .... [revisor inserts date].".

**867.** Page 485, line 2: after that line insert:

**"Section 762d.** 40.03 (1) (e) of the statutes is amended to read:

40.03 (1) (e) Shall approve the contribution rates and actuarial assumptions determined by the actuary under sub. (5) (b) and (c), except that the board shall not approve any employee contribution rate under s. 40.05 (1) (a) 1. to 4. that is less than 5 percent of each payment of earnings.

**SECTION 762h.** 40.05 (1) (b) of the statutes is renumbered 40.05 (1) (b) 2. and amended to read:

40.05 (1) (b) 2. In <u>Subject to subds. 3. and 4., in</u> lieu of employee payment, the employer may pay all or part of the contributions required by par. (a), but all the payments shall be available for benefit purposes to the same extent as required contributions deducted from earnings of the participating employees. Action to assume employee contributions as provided under this paragraph shall be taken at

 $\mathbf{2}$ 

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

the time and in the form determined by the governing body of the participating employer. The state shall pay under this paragraph for employees who are covered by a collective bargaining agreement under subch. V of ch. 111 and for employees whose fringe benefits are determined under s. 230.12 an amount equal to 4% of the earnings paid by the state unless otherwise provided in a collective bargaining agreement under subch. V of ch. 111 or unless otherwise determined under s. 230.12. The University of Wisconsin Hospitals and Clinics Authority shall pay under this paragraph for employees who are covered by a collective bargaining agreement under subch. I of ch. 111 and for employees whose fringe benefits are determined under s. 233.10 an amount equal to 4% of the earnings paid by the authority unless otherwise provided in a collective bargaining agreement under subch. I of ch. 111 or unless otherwise determined under s. 233.10. The state shall pay under this paragraph for employees who are not covered by a collective bargaining agreement under subch. V of ch. 111 and for employees whose fringe benefits are not determined under s. 230.12 an amount equal to 4% of the earnings paid by the state unless a different amount is recommended by the director of the office of state employment relations and approved by the joint committee on employment relations in the manner provided for approval of changes in the compensation plan under s. 230.12 (3). The University of Wisconsin Hospitals and Clinics Authority shall pay under this paragraph for its employees who are not covered by a collective bargaining agreement under subch. I of ch. 111 an amount equal to 4% of the earnings paid by the authority unless a different amount is established by the board of directors of the authority under s. 233.10.

**Section 762ip.** 40.05 (1) (b) 1. of the statutes is created to read:

40.05 (1) (b) 1. In this paragraph, "state" does not include any entity specified in s. 40.02 (54) (b) to (k).

**Section 762k.** 40.05 (1) (b) 3. of the statutes is created to read:

40.05 (1) (b) 3. The state may not pay for its employees who are not covered by a collective bargaining agreement under subch. V of ch. 111 the first 5 percent of earnings that the employees are required to pay as contributions under par. (a). For employees whose fringe benefits are determined under s. 230.12, the state shall pay any remaining contributions under par. (a) in an amount determined under s. 230.12. For employees whose fringe benefits are determined under a state compensation plan other than under s. 230.12, the state shall pay any remaining contributions under par. (a) in an amount recommended by the director of the office of state employment relations and approved by the joint committee on employment relations in the manner provided for approval of changes in the compensation plan under s. 230.12 (3).".

**868.** Page 485, line 2: after that line insert:

**"Section 762s.** 40.05 (4) (a) 1. of the statutes is amended to read:

40.05 (4) (a) 1. For Subject to subd. 4., for health insurance, each insured employee and insured retired employee shall contribute the balance of the required premium amounts after applying required employer contributions, if any.".

**869.** Page 485, line 3: delete lines 3 to 17.

**870.** Page 485, line 17: after that line insert:

"Section 763v. 41.41 (10) (a) 1. of the statutes is renumbered 41.41 (10) (a) 1. a. and amended to read:

 $\mathbf{2}$ 

41.41 (10) (a) 1. a. "Estimated Subject to subd. 1. b., "estimated value", for the year following the year in which the department acquires land within the Kickapoo valley reserve or the board acquires land under sub. (7), means the full value of the land determined by the department of revenue and, for each later year, means the value that was used for calculating the aid payment under this subsection on the land for the prior year increased or decreased to reflect the annual percentage change in the equalized valuation of all real property, excluding improvements, in the taxation district in which the land is located, as determined by comparing the most recent determination of equalized valuation under s. 70.57 for all real property to the next preceding determination of equalized valuation under s. 70.57 for all real property.

**Section 763w.** 41.41 (10) (a) 1. b. of the statutes is created to read:

41.41 (10) (a) 1. b. The "estimated value" of the land in the town of Stark in Vernon County shall include, in 2008, the value of improvements constituting the Kickapoo Valley Reserve Visitor Center and the maintenance buildings associated with the Kickapoo Valley Reserve Visitor Center and, in each later year, the value that was included under this subd. 1. b. in the prior year increased or decreased in the manner described in subd. 1. a.".

**871.** Page 485, line 17: after that line insert:

**"Section 770c.** 40.51 (8) of the statutes is amended to read:

40.51 **(8)** Every health care coverage plan offered by the state under sub. (6) shall comply with ss. 631.89, 631.90, 631.93 (2), 631.95, 632.72 (2), 632.746 (1) to (8) and (10), 632.747, 632.748, 632.83, 632.835, 632.85, 632.853, 632.855, 632.87 (3) to (5), 632.895 (5m) and (8) to (14) (15), and 632.896.

1	SECTION 770d. 40.51 (8m) of the statutes is amended to read:
2	40.51 (8m) Every health care coverage plan offered by the group insurance
3	board under sub. (7) shall comply with ss. 631.95, 632.746 (1) to (8) and (10), 632.747
4	632.748, 632.83, 632.835, 632.85, 632.853, 632.855, and 632.895 (11) to (14) (15).".
5	872. Page 485, line 17: after that line insert:
6	"Section 764s. $40.05$ (4) (a) 4. of the statutes is created to read:
7	40.05 (4) (a) 4. Beginning on January 1, 2008, each insured employee who is
8	employed by the state shall pay 10 percent of the premium cost of health insurance
9	coverage. This subdivision shall not apply to an insured employee whose
10	compensation is established under subch. I of ch. III or s. 233.10 or to an insured
11	employee who is a protective occupation participant.".
12	873. Page 485, line 17: after that line insert:
13	"Section 765gg. $40.05$ (4) (bp) 4. of the statutes is created to read:
14	40.05 (4) (bp) 4. This paragraph applies only to faculty and academic staff hired
15	before the effective date of this subdivision [revisor inserts date].".
16	<b>874.</b> Page 487, line 7: after that line insert:
17	"Section 781vL. 43.15 (2) (a) of the statutes is renumbered 43.15 (2).
18	Section 781vn. 43.15 (2) (b), (c), (d) and (e) of the statutes are repealed.
19	<b>Section 781vp.</b> 43.15 (4) (c) 5. of the statutes is repealed.
20	Section 781vr. 43.15 (4) (e) of the statutes is repealed.
21	<b>Section 781vt.</b> 43.15 (5) of the statutes is amended to read:
22	43.15 (5) Capital costs excluded. For the purpose of determining the amount
23	of financial support required under subs. (2) (b) and sub. (4) (b) 2. and (c) 5., amounts

spent for capital projects shall be excluded.

3

4

5

6

7

20

21

22

23

24

**Section 781vv.** 43.53 (2) (a) of the statutes is amended to read:

43.53 (2) (a) Name one of the participants as the library's fiscal agent, who is responsible for the payroll, benefit administration, insurance, and financial record keeping and auditing for the library. The participant's costs of providing the services under this paragraph count toward the financial support required of the participant under s. 43.15 (2) (b) or (4) (b) 2. or (c) 5.".

- **875.** Page 488, line 6: after that line insert:
- 8 "Section 782e. 44.57 of the statutes is repealed.".
- 9 **876.** Page 489, line 11: delete lines 11 to 20.
- 10 **877.** Page 490, line 4: delete lines 4 to 9.
- 11 **878.** Page 491, line 15: delete the material beginning with that line and ending with page 492, line 9.
- 13 **879.** Page 492, line 14: after that line insert:
- **"Section 791m.** 45.61 (2) (am) of the statutes is created to read:
- 45.61 (2) (am) A person who died while on active duty in the U.S. armed forces or in forces incorporated in the U.S. armed forces.".
- 17 **880.** Page 492, line 19: after that line insert:
- 18 "Section 792c. 45.61 (5) of the statutes is renumbered 45.61 (5) (a) and amended to read:
  - 45.61 **(5)** EXPENSES. (a) Expenses incident to the burial under this section of persons described in sub. (2) (a) and (b) to (e) shall be paid from the estate of the decedent, except that if there is no estate or the estate is insufficient, the expense of burial, or necessary part of the burial, shall be paid from the appropriation under s. 20.485 (1) (gk) for members of veterans homes, and the amount expended for those

expenses shall not exceed the amount established for funeral and burial expenses under s. 49.785 (1) (b).

**Section 792e.** 45.61 (5) (b) of the statutes is created to read:

45.61 (5) (b) Expenses incident to the burial under this section of persons described in sub. (2) (am) shall be paid from the estate of the decedent, except that if there is no estate or the estate is insufficient, the expense of burial, or necessary part of the burial, shall be paid by the relatives who requested the burial.".

- **881.** Page 492, line 20: delete lines 20 to 25.
- **882.** Page 493, line 1: delete lines 1 to 7.
- **883.** Page 493, line 12: delete lines 12 to 24.
- **884.** Page 494, line 1: delete lines 1 to 25.
- **885.** Page 495, line 1: delete lines 1 to 25.
- **886.** Page 496, line 1: delete lines 1 to 25.
- **887.** Page 497, line 1: delete lines 1 to 17.
- **888.** Page 497, line 21: after that line insert:
- 16 "Section 814m. 46.03 (18) (f) of the statutes is amended to read:

46.03 (18) (f) Notwithstanding par. (a), any person who submits to an assessment or <u>airman or driver</u> safety plan under s. 23.33 (13) (e), 30.80 (6) (d), <u>114.09</u> (2) (bm), 343.16 (5) (a), 343.30 (1q), 343.305 (10) or 350.11 (3) (d) shall pay a reasonable fee therefor to the appropriate county department under s. 51.42 or traffic safety school under s. 345.60. A county may allow the person to pay the assessment fee in 1, 2, 3 or 4 equal installments. The fee for the <u>airman or</u> driver safety plan may be reduced or waived if the person is unable to pay the complete fee, but no fee for

- assessment or attendance at a traffic safety school under s. 345.60 may be reduced or waived. Nonpayment of the assessment fee is noncompliance with the court order that required completion of an assessment and <u>airman or</u> driver safety plan. Upon a finding that the person has the ability to pay, nonpayment of the <u>airman or</u> driver safety plan fee is noncompliance with the court order that required completion of an assessment and <u>airman or</u> driver safety plan.".
- 7 **889.** Page 497, line 22: delete lines 22 to 25.
- 8 **890.** Page 498, line 1: delete lines 1 to 24.
- 9 **891.** Page 499, line 1: delete lines 1 to 25.
- 10 **892.** Page 500, line 1: delete lines 1 to 25.
- 11 **893.** Page 501, line 1: delete lines 1 to 25.
- 12 **894.** Page 502, line 1: delete lines 1 to 25.
- 13 **895.** Page 503, line 1: delete lines 1 to 24.
- **896.** Page 504, line 1: delete lines 1 and 2.
- 15 **897.** Page 504, line 14: delete lines 14 to 24.
- 16 **898.** Page 505, line 1: delete lines 1 to 7.
- 17 **899.** Page 505, line 22: delete lines 22 to 25.
- 18 **900.** Page 506, line 1: delete lines 1 to 25.
- 19 **901.** Page 507, line 1: delete lines 1 to 14.
- 902. Page 507, line 15: delete the material beginning with that line and ending with page 508, line 2.
- **903.** Page 508, line 3: delete lines 3 to 7.

- **904.** Page 508, line 8: delete lines 8 to 25.
- **905.** Page 509, line 1: delete lines 1 to 15.
- **906.** Page 509, line 16: delete the material beginning with that line and ending with page 510, line 8.
- **907.** Page 510, line 9: delete lines 9 to 25.
- **908.** Page 511, line 1: delete lines 1 to 21.
- **909.** Page 511, line 22: delete the material beginning with that line and ending with page 512, line 7.
- **910.** Page 512, line 8: delete lines 8 to 17.
- **911.** Page 512, line 18: delete lines 18 to 21.
- **912.** Page 512, line 22: delete lines 22 to 25.
- **913.** Page 513, line 1: delete lines 1 to 25.
- **914.** Page 514, line 1: delete lines 1 to 12.
- 915. Page 514, line 13: delete the material beginning with that line and ending with page 515, line 5.
- **916.** Page 515, line 6: delete lines 6 to 25.
- **917.** Page 516, line 1: delete lines 1 to 25.
- **918.** Page 517, line 1: delete lines 1 to 8.
- **919.** Page 517, line 9: delete lines 9 to 20.
- **920.** Page 517, line 21: delete lines 21 to 25.
- **921.** Page 518, line 1: delete lines 1 to 25.
- **922.** Page 519, line 1: delete lines 1 to 5.

- 1 **923.** Page 519, line 6: delete lines 6 to 23.
- 2 **924.** Page 519, line 24: delete that line.
- 3 **925.** Page 520, line 1: delete lines 1 to 23.
- 4 **926.** Page 521, line 1: delete lines 1 to 25.
- 5 **927.** Page 522, line 1: delete lines 1 to 25.
- 6 **928.** Page 523, line 1: delete lines 1 to 25.
- 7 **929.** Page 524, line 1: delete lines 1 to 24.
- 8 **930.** Page 525, line 1: delete lines 1 to 24.
- 9 **931.** Page 526, line 1: delete lines 1 to 13.
- 10 **932.** Page 526, line 14: delete the material beginning with that line and ending with page 528, line 20.
- 933. Page 528, line 21: delete the material beginning with that line and ending with page 530, line 6.
- **934.** Page 530, line 7: delete lines 7 to 10.
- **935.** Page 530, line 11: delete lines 11 to 20.
- 936. Page 530, line 21: delete the material beginning with that line and ending with page 531, line 2.
- 18 **937.** Page 531, line 3: delete lines 3 to 9.
- 19 **938.** Page 531, line 10: delete that line.
- 20 **939.** Page 531, line 11: delete lines 11 to 13.
- 940. Page 531, line 14: delete the material beginning with that line and ending with page 532, line 3.

- **941.** Page 532, line 4: delete lines 4 to 6.
- **942.** Page 532, line 7: delete lines 7 to 13.
- **943.** Page 532, line 14: delete lines 14 to 16.
- 944. Page 532, line 17: delete the material beginning with that line and ending with page 533, line 3.
- **945.** Page 533, line 4: delete lines 4 to 6.
  - **946.** Page 533, line 7: delete the material beginning with that line and ending with page 537, line 2.
  - **947.** Page 537, line 3: after that line insert:
- **"Section 952g.** 46.281 (1) (j) of the statutes is created to read:
  - 46.281 (1) (j) Contract with a person to provide the advocacy services described under s. 16.009 (2) (p) 1. to 5. to actual or potential recipients of the family care benefit who are under age 60 or to their families or guardians. The department may not contract under this paragraph with a county or with a person who has a contract with the department to provide services under s. 46.283 (3) and (4) as a resource center or to administer the family care benefit as a care management organization. The contract under this paragraph shall include as a goal that the provider of advocacy services provide one advocate for every 2,500 individuals under age 60 who receive the family care benefit. The department shall allocate \$190,000 for the contract under this paragraph in fiscal year 2007-08 and \$525,000 in each subsequent fiscal year."
  - **948.** Page 537, line 4: delete the material beginning with that line and ending with page 544, line 19.

- 1 **949.** Page 544, line 22: delete lines 22 to 25.
- **950.** Page 545, line 1: delete the material beginning with that line and ending
- 3 with page 546, line 8.
- 4 **951.** Page 546, line 9: delete lines 9 to 14.
- 5 **952.** Page 546, line 15: delete the material beginning with that line and ending with page 549, line 2.
- 7 **953.** Page 549, line 17: delete lines 17 to 24.
- 8 **954.** Page 549, line 24: after that line insert:
- 9 "Section 998m. 46.284 (4) (m) of the statutes is created to read:
- 46.284 (4) (m) Issue payment for 95 percent of proper claims for reimbursement for family care benefit services within 30 days of receipt of the claims and pay 100 percent of such claims within 45 days of receipt of the claims."
- 13 **955.** Page 549, line 25: delete the material beginning with that line and ending with page 550, line 7.
- 956. Page 550, line 8: delete the material beginning with that line and ending
  with page 551, line 13.
- 957. Page 552, line 24: delete lines 24 and 25 and substitute "developmental disability, as defined in s. 51.01 (5) (a), or degenerative brain disorder, as defined in s. 55.01 (1v)<sub>52</sub>".
- 20 **958.** Page 553, line 22: delete the material beginning with that line and ending with page 564, line 15.
- 22 **959.** Page 564, line 21: delete lines 21 to 24.
- 23 **960.** Page 565, line 1: delete lines 1 to 14.

- **961.** Page 565, line 15: delete lines 15 to 22.
- **962.** Page 565, line 23: delete lines 23 and 24.
- **963.** Page 566, line 1: delete lines 1 to 7.
- **964.** Page 566, line 8: delete lines 8 to 16.
- **965.** Page 566, line 17: delete lines 17 to 25.
- **966.** Page 567, line 1: delete lines 1 to 25.
- **967.** Page 568, line 1: delete lines 1 to 5.
- **968.** Page 568, line 10: delete lines 10 to 14.
- **969.** Page 568, line 19: delete lines 19 to 24.
- **970.** Page 569, line 1: delete lines 1 to 19.
- 971. Page 569, line 20: delete the material beginning with that line and ending with page 570, line 20.
- **972.** Page 570, line 21: delete lines 21 to 25.
- **973.** Page 571, line 1: delete lines 1 to 24.
- **974.** Page 572, line 1: delete lines 1 to 25.
- **975.** Page 573, line 1: delete lines 1 to 25.
- **976.** Page 574, line 1: delete lines 1 to 6.
- **977.** Page 574, line 7: delete lines 7 to 10.
- **978.** Page 574, line 11: delete lines 11 to 16 and substitute:
- 20 "Section 1115b. 46.48 (11m) of the statutes is repealed.".
- **979.** Page 574, line 17: delete lines 17 to 22.

- **980.** Page 574, line 23: delete lines 23 and 24.
- **981.** Page 575, line 1: delete lines 1 to 17.
- **982.** Page 575, line 18: delete lines 18 to 22.
- **983.** Page 575, line 23: delete lines 23 to 25.
- **984.** Page 576, line 1: delete lines 1 to 7.
- **985.** Page 576, line 8: delete lines 8 to 12.
- **986.** Page 576, line 13: delete lines 13 to 25.
- **987.** Page 577, line 1: delete lines 1 to 8.
- 9 988. Page 577, line 9: delete the material beginning with that line and ending
- on page 578, line 5.
- **989.** Page 578, line 6: delete lines 6 to 25.
- **990.** Page 579, line 1: delete lines 1 to 14.
- **991.** Page 579, line 15: delete that line.
- **992.** Page 579, line 16: delete lines 16 to 25.
- **993.** Page 580, line 1: delete lines 1 to 24.
- **994.** Page 581, line 1: delete lines 1 to 25.
- **995.** Page 582, line 1: delete lines 1 to 25.
- **996.** Page 583, line 1: delete lines 1 to 25.
- **997.** Page 584, line 1: delete lines 1 and 2.
- **998.** Page 584, line 10: delete lines 10 to 25.
- **999.** Page 585, line 1: delete lines 1 to 25.

- **1000.** Page 586, line 1: delete lines 1 to 12.
- **1001.** Page 586, line 20: delete lines 20 to 24.
- **1002.** Page 587, line 1: delete lines 1 to 25.
- **1003.** Page 588, line 1: delete lines 1 to 25.
- **1004.** Page 589, line 1: delete lines 1 to 25.
- **1005.** Page 590, line 1: delete lines 1 to 20.
- **1006.** Page 592, line 22: delete lines 22 to 25.
- **1007.** Page 593, line 1: delete lines 1 to 25.
- **1008.** Page 594, line 1: delete lines 1 and 2.
- **1009.** Page 595, line 13: delete lines 13 to 21.
- **1010.** Page 598, line 16: delete lines 16 to 25.
- **1011.** Page 599, line 1: delete lines 1 to 25.
- **1012.** Page 600, line 1: delete lines 1 to 25.
- **1013.** Page 601, line 1: delete lines 1 to 24.
- **1014.** Page 603, line 21: delete lines 21 to 24.
- **1015.** Page 604, line 1: delete lines 1 to 25.
- **1016.** Page 605, line 1: delete lines 1 to 25.
- **1017.** Page 606, line 1: delete lines 1 to 25.
- **1018.** Page 607, line 1: delete lines 1 to 3.
- **1019.** Page 607, line 4: delete lines 4 to 14.
- **1020.** Page 607, line 15: delete lines 15 to 25.

- **1021.** Page 608, line 1: delete lines 1 to 4.
- **1022.** Page 608, line 5: delete lines 5 to 24.
- **1023.** Page 609, line 1: delete lines 1 to 25.
- **1024.** Page 610, line 1: delete lines 1 to 25.
- **1025.** Page 611, line 1: delete lines 1 to 25.
- **1026.** Page 612, line 1: delete lines 1 to 25.
- **1027.** Page 613, line 1: delete lines 1 to 24.
- **1028.** Page 614, line 1: delete lines 1 to 23.
- **1029.** Page 615, line 1: delete lines 1 to 11.
- **1030.** Page 615, line 12: delete lines 12 to 14.
- **1031.** Page 615, line 15: delete lines 15 to 23.
- **1032.** Page 615, line 24: delete that line.
- **1033.** Page 616, line 1: delete lines 1 and 2.
- **1034.** Page 616, line 3: delete lines 3 to 25.
- **1035.** Page 617, line 1: delete lines 1 to 24.
- **1036.** Page 618, line 1: delete lines 1 to 25.
- **1037.** Page 619, line 1: delete lines 1 to 25.
- **1038.** Page 620, line 1: delete lines 1 to 25.
- **1039.** Page 621, line 1: delete lines 1 to 25.
- **1040.** Page 622, line 1: delete lines 1 to 25.
- **1041.** Page 623, line 1: delete lines 1 to 25.

- **1042.** Page 624, line 1: delete lines 1 to 24.
- **1043.** Page 625, line 1: delete lines 1 to 24.
- **1044.** Page 626, line 1: delete lines 1 to 25.
- **1045.** Page 627, line 1: delete lines 1 to 25.
- **1046.** Page 628, line 1: delete lines 1 to 20.
- **1047.** Page 628, line 25: delete "2008" and substitute "2009".
- **1048.** Page 629, line 2: delete "for a child 15 years of age or over.".
- **1049.** Page 629, line 3: delete lines 3 and 4.
- **1050.** Page 629, line 5: delete "age; and \$452".
- **1051.** Page 629, line 9: delete lines 9 to 24.
- **1052.** Page 630, line 1: delete lines 1 to 25.
- **1053.** Page 631, line 1: delete lines 1 to 23.
- **1054.** Page 632, line 25: delete that line.
- **1055.** Page 633, line 1: delete lines 1 to 24.
- **1056.** Page 634, line 1: delete lines 1 to 24.
- **1057.** Page 635, line 1: delete lines 1 to 24.
- **1058.** Page 636, line 1: delete lines 1 to 17.
- **1059.** Page 636, line 20: delete lines 20 to 25.
- **1060.** Page 637, line 1: delete lines 1 to 6.
- **1061.** Page 637, line 10: delete lines 10 to 20.
- **1062.** Page 640, line 11: delete lines 11 to 25.

- **1063.** Page 641, line 1: delete lines 1 to 25.
- **1064.** Page 642, line 1: delete lines 1 to 24.
- **1065.** Page 643, line 1: delete lines 1 and 2.
- **1066.** Page 643, line 8: delete lines 8 to 24.
- **1067.** Page 644, line 1: delete lines 1 to 8.
- **1068.** Page 645, line 25: delete that line.
- **1069.** Page 646, line 1: delete lines 1 to 10.
- **1070.** Page 647, line 2: after that line insert:
- 9 "Section 1370s. 48.981 (7) (a) 14m. of the statutes is amended to read:
- 10 48.981 (7) (a) 14m. A judge conducting proceedings under s. 968.26 (1).".
- **1071.** Page 647, line 3: delete lines 3 to 25.
- **1072.** Page 648, line 1: delete lines 1 to 25.
- **1073.** Page 649, line 1: delete lines 1 to 24.
- **1074.** Page 650, line 1: delete lines 1 to 6.
- **1075.** Page 650, line 13: delete lines 13 to 24.
- **1076.** Page 650, line 25: delete the material beginning with that line and ending with page 653, line 10, and substitute:
- **Section 1399n.** 49.13 (2) (a) of the statutes is amended to read:
- 49.13 (2) (a) The department shall contract with the department of health and family services as provided under s. 49.79 (10) to administer an employment and training program for recipients under the food stamp program. The department may subcontract with a Wisconsin works Works agency to administer the employment

and training program under this subsection. Except as provided in pars. (b) and (bm), the department may shall require able individuals who are 18 to 60 years of age who are not participants in a Wisconsin works Works employment position to participate in the employment and training program under this subsection.".

- **1077.** Page 654, line 19: delete lines 19 to 25.
- **1078.** Page 655, line 1: delete lines 1 to 21.
- **1079.** Page 655, line 21: after that line insert:
- 8 "Section 1409j. 49.147 (2) (b) of the statutes is renumbered 49.147 (2) (a) 3.
  9 and amended to read:
  - 49.147 (2) (a) 3. *Job search assistance*. A Wisconsin works <u>Works</u> agency shall assist a participant in his or her search for unsubsidized employment. In determining an appropriate placement for a participant, a Wisconsin works agency shall give priority to placement in unsubsidized employment over placements under subs. (3) to (5).
  - **Section 1409m.** 49.147 (2) (bm) of the statutes is created to read:
  - 49.147 (2) (bm) Case management services. 1. In lieu of placing the individual in a Wisconsin Works employment position under subs. (3) to (5), the department may provide case management services, which may include those services specified in s. 49.1475, to an individual who applies for a Wisconsin Works employment position if the department determines all of the following:
    - a. The individual meets the eligibility requirements under s. 49.145 (2) and (3).
  - b. The individual is willing to work and has no barriers to employment.
  - c. The individual is job-ready, based on the individual's employment history or education.

- d. The most appropriate placement for the individual is in unsubsidized employment.
  - 2. In determining an appropriate placement for an applicant, a Wisconsin Works agency shall give priority to placement in unsubsidized employment and providing case management services over placements under subs. (3) to (5).".
  - **1080.** Page 655, line 22: delete the material beginning with that line and ending with page 657, line 18.
- **1081.** Page 657, line 19: delete lines 19 to 24.
- **1082.** Page 658, line 1: delete lines 1 to 8.
- **1083.** Page 658, line 8: after that line insert:
- **"Section 1418c.** 49.152 (1) of the statutes is amended to read:
  - 49.152 (1) Petition for review. Any individual whose application for any component of Wisconsin works Works is not acted upon by the Wisconsin works Works agency with reasonable promptness after the filing of the application, as defined by the department by rule, or is denied in whole or in part, whose benefit is modified or canceled, or who believes that the benefit was calculated incorrectly or that the employment position in which the individual was placed or providing case management services under s. 49.147 (2) (bm) in lieu of placement in a Wisconsin Works employment position is inappropriate, may petition the Wisconsin works Works agency for a review of such action. Review is unavailable if the action by the Wisconsin works Works agency occurred more than 45 days prior to submission of the petition for review.
    - **SECTION 1419c.** 49.152 (3) (a) of the statutes is amended to read:

49.152 (3) (a) If, following review under sub. (2), the Wisconsin works Works agency or the department determines that an individual, whose application for a Wisconsin works Works employment position was denied based on eligibility, was in fact eligible, or that the individual was placed in an inappropriate Wisconsin works Works employment position or inappropriately provided case management services under s. 49.147 (2) (bm) in lieu of placement in a Wisconsin Works employment position, the Wisconsin works Works agency shall place the individual in the first available Wisconsin works Works employment position that is appropriate for that individual, as determined by the Wisconsin works Works agency or the department. An individual who is placed in a Wisconsin works Works employment position under this paragraph is eligible for the benefit for that position under s. 49.148 beginning on the date on which the individual begins participation under s. 49.147.".

**1084.** Page 658, line 20: delete "program, in the amount of at least" and substitute "program.".

**1085.** Page 658, line 21: delete that line.

1086. Page 658, line 22: delete "services," and substitute "services.".

**1087.** Page 658, line 23: delete that line.

**1088.** Page 658, line 25: delete "at least \$4,800,600" and substitute 19 "\$4,440,500".

**1089.** Page 659, line 7: delete "\$1,765,600" and substitute "\$1,524,900".

**1090.** Page 659, line 7: delete "\$1,600,300" and substitute "\$1,524,900".

**1091.** Page 659, line 9: delete lines 9 to 20.

**1092.** Page 659, line 20: after that line insert:

"Section 1422c. 49.155 (1m) (intro.) of the statutes is amended to read:

49.155 (1m) ELIGIBILITY. (intro.) A Wisconsin works Works agency shall determine eligibility for a child care subsidy under this section. Under this section, subject to any waiting list placement under sub. (2), an individual may receive a subsidy for child care for a child who has not attained the age of 13 or, if the child is disabled, who has not attained the age of 19, if the individual meets all of the following conditions:".

- **1093.** Page 659, line 21: delete lines 21 to 23.
- **1094.** Page 659, line 23: after that line insert:

"Section 1426c. 49.155 (1m) (c) 1. (intro.) of the statutes is amended to read: 49.155 (1m) (c) 1. (intro.) Except as provided in subds. 1g., and 1h., 1m., 2., and 3., the gross income of the individual's family is at or below 185% of the poverty line for a family the size of the individual's family or, for an individual who is already receiving a child care subsidy under this section on the effective date of this subdivision .... [revisor inserts date], the gross income of the individual's family is at or below 200% of the poverty line for a family the size of the individual's family. In calculating the gross income of the family, the Wisconsin works Works agency shall include income described under s. 49.145 (3) (b) 1. and 3., except that, in calculating farm and self-employment income, the Wisconsin works Works agency shall include the sum of the following:

**Section 1427c.** 49.155 (1m) (c) 1c. of the statutes is created to read:

49.155 (1m) (c) 1c. Except as provided in subds. 1g. and 1h., for an individual who, on or after the effective date of this subdivision .... [revisor inserts date], applies for a child care subsidy under this section or reapplies for a child care subsidy under

this section after losing eligibility, the gross income of the individual's family when the individual applies or reapplies is at or below 175 percent of the poverty line for a family the size of the individual's family and, after the individual is already receiving a child care subsidy under this section, the gross income of the individual's family is at or below 190 percent of the poverty line for a family the size of the individual's family. The Wisconsin Works agency shall calculate the gross income of the family in the same manner as gross income is calculated under subd. 1.

**SECTION 1428c.** 49.155 (1m) (c) 1m. of the statutes is repealed.

**SECTION 1428e.** 49.155 (1m) (c) 2. of the statutes is repealed.

**SECTION 1428g.** 49.155 (1m) (c) 3. of the statutes is repealed.

**Section 1429c.** 49.155 (2) of the statutes is created to read:

49.155 (2) Waiting List. (a) The department shall implement a prioritized waiting list system for applicants who are otherwise eligible for a child care subsidy under sub. (1m). Under the system, an applicant on the waiting list would not receive a child care subsidy unless the amount of funding allocated for child care subsidies under s. 49.175 (1) (p) was sufficient.

(b) Notwithstanding par. (a), an applicant for a child care subsidy who is eligible under sub. (1m) and who is participating in a work component of Wisconsin Works under s. 49.147 may not be placed on a waiting list.

**SECTION 1430c.** 49.155 (3) (a) of the statutes is amended to read:

49.155 (3) (a) —A—Subject to any waiting list placement under sub. (2), a Wisconsin works Works agency shall refer an individual who has been determined eligible under sub. (1m) to a county department under s. 46.215, 46.22, or 46.23 for child care assistance.".

23

**1095.** Page 659, line 23: after that line insert: 1 2 "Section 1425h. 49.155 (1m) (cm) of the statutes is created to read: 3 49.155 (1m) (cm) The total number of months in which the individual has received a child care subsidy under this section after the effective date of this 4 5 paragraph .... [revisor inserts date], for the child for whom the individual would be 6 receiving the child care subsidy does not exceed 60 months, whether or not 7 consecutive.". 8 **1096.** Page 659, line 24: delete the material beginning with that line and ending with page 661, line 6. 9 **1097.** Page 661, line 6: after that line insert: 10 11 "Section 1433r. 49.159 (3) of the statutes is amended to read: 12 49.159 (3) OTHER CUSTODIAL PARENTS. A custodial parent in a Wisconsin works Works group in which the other custodial parent is a participant in a Wisconsin 13 14 works Works employment position or is receiving case management services under s. 49.147 (2) (bm) is eligible for employment training and job search assistance 15 16 services provided by the Wisconsin works Works agency.". 17 **1098.** Page 661, line 7: delete lines 7 to 20. **1099.** Page 661, line 23: delete "\$44,068,500" and substitute "\$42,980,000". 18 **1100.** Page 661, line 24: delete "\$43,392,200" and substitute "\$42,980,000". 19 **1101.** Page 662, line 8: delete "\$38,471,500" and substitute "\$46,584,100". 20 **1102.** Page 662, line 9: delete "\$38,471,500" and substitute "\$46,584,100". 21

**1103.** Page 662, line 13: delete "\$16,670,100" and substitute "\$16,422,900".

**1104.** Page 662, line 14: delete "\$16,868,500" and substitute "\$16,422,900".

- **1105.** Page 662, line 15: delete lines 15 to 17.
- **1106.** Page 662, line 20: delete "\$345,601,800" and substitute
- 3 "<u>\$325,365,200</u>".
- **1107.** Page 662, line 21: delete "\$355,352,000" and substitute
- 5 "\$322,987,900".
- **1108.** Page 663, line 1: delete "\$1,765,600" and substitute "\$1,524,900".
- **1109.** Page 663, line 1: delete "\$1,600,300" and substitute "\$1,524,900".
- **1110.** Page 663, line 5: delete "\$5,311,000" and substitute "\$5,671,100".
- **1111.** Page 663, line 8: delete "at least \$4,800,600" and substitute 10 "\$4,440,500".
- **1112.** Page 663, line 11: delete lines 11 to 13.
- **1113.** Page 664, line 4: delete lines 4 to 6 and substitute "at home if appropriate services are provided <u>and for ongoing services provided in those counties</u>
- to families with children placed in out-of-home care, \$5,707,200 in each fiscal year.".
- **1114.** Page 664, line 7: delete lines 7 to 14.
- **1115.** Page 664, line 19: delete "\$1,510,500" and substitute "\$1,317,700".
- **1116.** Page 664, line 19: delete "\$1,317,700 <u>\$1,532,100</u>" and substitute 18 "\$1,317,700".
- **1117.** Page 664, line 25: delete "\$16,125,400" and substitute "\$39,555,300".
- **1118.** Page 664, line 25: delete "\$6,664,200" and substitute "\$41,600,000".
- **1119.** Page 665, line 1: delete lines 1 to 24.
- **1120.** Page 666, line 1: delete lines 1 to 25.

- **1121.** Page 667, line 1: delete lines 1 to 17.
- **1122.** Page 668, line 1: delete lines 1 to 21.
- **1123.** Page 669, line 6: delete lines 6 to 16 and substitute:
- "(c) If a county or tribal governing body establishes a program under par. (b),
  the county or tribal governing body shall do all of the following:
- 1. Retain all amounts recovered by the county or tribal governing body as a result of its program during the first 3 months in which it recovers any amounts as a result of its program.
  - 2. Pay to the department all amounts recovered by the county or tribal governing body as a result of its program after the 3rd month in which it recovers any amounts as a result of its program.".
- **1124.** Page 669, line 23: delete lines 23 and 24.
- **1125.** Page 670, line 1: delete lines 1 to 17.
- **1126.** Page 670, line 18: delete the material beginning with that line and ending with page 671, line 19.
- **1127.** Page 671, line 20: delete lines 20 to 25.
- **1128.** Page 672, line 1: delete lines 1 to 25.
- **1129.** Page 673, line 1: delete lines 1 to 5.
- **1130.** Page 673, line 14: delete lines 14 to 21.
- **1131.** Page 673, line 23: delete "section 1473,".
- **1132.** Page 675, line 1: delete lines 1 to 10.
- **1133.** Page 675, line 11: delete lines 11 to 20.

- **1134.** Page 675, line 21: delete lines 21 to 25.
- **1135.** Page 676, line 1: delete lines 1 to 23.
- **1136.** Page 677, line 1: delete lines 1 to 25.
- **1137.** Page 678, line 1: delete lines 1 to 24.
- **1138.** Page 679, line 1: delete lines 1 to 25.
- **1139.** Page 680, line 1: delete lines 1 to 25.
- **1140.** Page 681, line 1: delete lines 1 to 25.
- **1141.** Page 682, line 1: delete lines 1 to 24.
- **1142.** Page 683, line 1: delete lines 1 to 24.
- **1143.** Page 684, line 1: delete lines 1 to 25.
- **1144.** Page 685, line 1: delete lines 1 to 25.
- **1145.** Page 686, line 1: delete lines 1 to 25.
- **1146.** Page 687, line 1: delete lines 1 to 25.
- **1147.** Page 688, line 1: delete lines 1 to 24.
- **1148.** Page 689, line 1: delete lines 1 to 25.
- **1149.** Page 690, line 1: delete lines 1 to 24.
- **1150.** Page 691, line 1: delete lines 1 to 25.
- **1151.** Page 692, line 1: delete lines 1 to 25.
- **1152.** Page 693, line 1: delete lines 1 to 25.
- **1153.** Page 694, line 1: delete lines 1 to 15.
- **1154.** Page 694, line 16: delete lines 16 to 25.

- **1155.** Page 695, line 1: delete lines 1 to 11.
- **1156.** Page 695, line 12: delete lines 12 to 21.
- **1157.** Page 695, line 22: delete the material beginning with that line and ending with page 696, line 2.
- **1158.** Page 696, line 3: delete lines 3 to 19.
- **1159.** Page 696, line 20: delete lines 20 to 23.
- **1160.** Page 696, line 24: delete the material beginning with that line and ending with page 699, line 3.
- **1161.** Page 699, line 3: after that line insert:

- **"Section 1520d.** 49.45 (3) (cm) of the statutes is created to read:
  - 49.45 (3) (cm) 1. Except as provided under subd. 2., the department shall issue payment for at least 95 percent of proper provider claims for reimbursement under the Medical Assistance program within 30 days of receipt of the claims and shall issue payment for 100 percent of such claims within 45 days of receipt of the claims.
  - 2. The department may exceed the claims payment deadlines under subd.1. under any of the following circumstances:
  - a. If a claim is filed under Medicare, as defined in par. (L) 1. b., for payment for a service, the department has up to 6 months after the department or the provider receives notice of the disposition of the Medicare claim to issue payment for the service.
- b. The department may issue payments at any time in accordance with a court order or to comply with a hearing decision or a corrective action taken by the department.

- c. If the department is granted a waiver under 42 CFR 447.45 (e) that exempts the department from federal deadlines for payment of claims, the department may exceed the deadlines under subd.1. to the extent permitted in the waiver.".
- **1162.** Page 699, line 4: delete lines 4 to 12.
- **1163.** Page 699, line 13: delete the material beginning with that line and ending with page 700, line 16.
- **1164.** Page 700, line 17: delete lines 17 to 24.
- **1165.** Page 700, line 25: delete the material beginning with that line and ending with page 701, line 21.
- **1166.** Page 702, line 3: delete the material beginning with that line and ending with page 703, line 10.
- **1167.** Page 703, line 13: delete "shall may" and substitute "shall".
- **1168.** Page 703, line 13: delete "<del>(gp), (o), and (w), and (xd)"</del> and substitute "<del>(gp), (o), and (w)"</del>.
- **1169.** Page 704, line 1: delete lines 1 to 10.
- **1170.** Page 704, line 11: delete lines 11 to 16.
- **1171.** Page 704, line 17: delete the material beginning with that line and ending with page 706, line 9.
- **1172.** Page 706, line 10: delete the material beginning with that line and ending with page 707, line 9.
- **1173.** Page 707, line 10: delete the material beginning with that line and ending with page 708, line 3.
- **1174.** Page 708, line 4: delete that line.

- **1175.** Page 708, line 5: delete lines 5 to 10.
- **1176.** Page 708, line 11: delete lines 11 to 21 and substitute:
- 3 "Section 1549b. 49.45 (24r) of the statutes is amended to read:
  - 49.45 (24r) Family Planning Demonstration Project. The department shall request a waiver from the secretary of the federal department of health and human services to permit the department to conduct a demonstration project to provide family planning services, as defined in s. 253.07 (1) (b), under medical assistance to any woman between the ages of 15 18 and 44 whose family income does not exceed 185% of the poverty line for a family the size of the woman's family. If the waiver is granted and in effect, the department shall implement the waiver no later than July 1, 1998, or on the effective date of the waiver, whichever is later."
- **1177.** Page 708, line 22: delete lines 22 to 25.
- **1178.** Page 710, line 8: delete "work together to develop" and substitute "approve".
- **1179.** Page 710, line 13: delete "developed" and substitute "approved".
- **1180.** Page 711, line 4: delete that line.
- **1181.** Page 711, line 5: delete lines 5 to 9.
- **1182.** Page 711, line 10: delete lines 10 to 17.
- **1183.** Page 712, line 3: delete lines 3 to 13.
- **1184.** Page 712, line 14: delete lines 14 to 25.
- **1185.** Page 713, line 1: delete lines 1 to 5.
- **1186.** Page 713, line 6: delete lines 6 to 11.
- **1187.** Page 721, line 13: delete lines 13 to 16.

- 1 **1188.** Page 721, line 17: delete lines 17 to 23.
- 2 **1189.** Page 722, line 3: after that line insert:
- 3 "Section 1589r. 49.46 (2) (b) 3. of the statutes is renumbered 49.46 (2) (b) 3.
- 4 a. and amended to read:

6

7

8

9

10

- 49.46 (2) (b) 3. a. Transportation by emergency medical vehicle to obtain emergency medical care, transportation by specialized medical vehicle to obtain medical care including the unloaded travel of the specialized medical vehicle necessary to provide that transportation or, if authorized in advance by the county department under s. 46.215 or 46.22, transportation by common carrier or private motor vehicle and, if transportation by other means is contraindicated, to obtain nonemergency medical care.
- **Section 1589s.** 49.46 (2) (b) 3. b. of the statutes is created to read:
- 13 49.46 (2) (b) 3. b. To obtain nonemergency medical services, except as provided 14 in subd. 3. a., appropriate transportation that is provided through an entity with 15 which the department has contracted to manage transportation services for the 16 Medical Assistance program.".
- 17 **1190.** Page 722, line 7: delete "(1) (c) (1d)" and substitute "(1) (c)".
- 18 **1191.** Page 722, line 17: delete the material beginning with that line and ending with page 724, line 6.
- 20 **1192.** Page 724, line 11: delete lines 11 to 20.
- 1193. Page 727, line 11: delete the material beginning with that line and ending with page 750, line 2.
- 23 **1194.** Page 750, line 3: delete lines 3 to 14.

- **1195.** Page 750, line 15: delete lines 15 to 18.
- **1196.** Page 750, line 19: delete lines 19 to 24.
- **1197.** Page 755, line 21: delete lines 21 to 25.
- **1198.** Page 756, line 1: delete lines 1 to 25.
- **1199.** Page 757, line 1: delete lines 1 to 8.
- **1200.** Page 757, line 17: delete lines 17 to 24.
- **1201.** Page 758, line 12: delete the material beginning with that line and ending with page 759, line 2.
- **1202.** Page 759, line 8: delete the material beginning with that line and ending with page 760, line 16.
- **1203.** Page 760, line 16: after that line insert:

- **"Section 1637q.** 49.665 (3) of the statutes is amended to read:
  - 49.665 (3) Administration. Subject to sub. (2) (a) 2., the department shall administer a program to provide the health services and benefits described in s. 49.46 (2) to persons that meet the eligibility requirements specified in sub. (4) and issue payment for such services in accordance with s. 49.45 (3) (cm). The department shall promulgate rules setting forth the application procedures and appeal and grievance procedures. The department may promulgate rules limiting access to the program under this section to defined enrollment periods. The department may also promulgate rules establishing a method by which the department may purchase family coverage offered by the employer of a member of an eligible family or of a member of an eligible child's household, or family or individual coverage offered by the employer of an eligible unborn child's mother or her spouse, under circumstances

- in which the department determines that purchasing that coverage would not be more costly than providing the coverage under this section.".
- **1204.** Page 760, line 20: delete "and (x), and (xd)" and substitute "and (x)".
- **1205.** Page 761, line 3: delete "and (x), and (xd)" and substitute "and (x)".
- **1206.** Page 761, line 15: delete "and (x), and (xd)" and substitute "and (x)".
- **1207.** Page 762, line 23: delete the material beginning with that line and ending with page 763, line 2.
- **1208.** Page 763, line 5: delete "October".
- **1209.** Page 763, line 6: delete "1, 2007" and substitute "January 1, 2008".
- **1210.** Page 764, line 1: delete lines 1 to 7.
- **1211.** Page 764, line 10: delete "and (pg), and (xh)" and substitute "and (pg)".
- **1212.** Page 764, line 12: after "assistance," insert "<u>including payment</u>

  13 deadlines under s. 49.45 (3) (cm),".
- **1213.** Page 765, line 1: delete lines 1 to 9.
- **1214.** Page 765, line 13: delete lines 13 to 25.
- **1215.** Page 766, line 1: delete lines 1 to 21.
- **1216.** Page 766, line 22: delete the material beginning with that line and ending with page 767, line 16.
- **1217.** Page 767, line 16: after that line insert:
- 20 "Section 1661g. 49.785 (1m) (b) of the statutes is amended to read:
- 49.785 (1m) (b) If the total funeral and burial expenses for the recipient exceed \$3,500 \$4,500, the county or applicable tribal governing body or organization

- responsible for burial of the recipient is not required to make a payment for funeral and burial expenses under sub. (1) (b).".
- **1218.** Page 767, line 17: delete lines 17 to 24.
- **1219.** Page 767, line 25: delete that line.
- **1220.** Page 768, line 1: delete lines 1 to 8.
- **1221.** Page 768, line 9: delete lines 9 to 15.
- **1222.** Page 768, line 16: delete lines 16 to 21.
- **1223.** Page 768, line 22: delete the material beginning with that line and ending with page 769, line 2.
- **1224.** Page 769, line 3: delete lines 3 to 13.
- **1225.** Page 770, line 10: delete lines 10 to 17.
- **1226.** Page 770, line 18: delete lines 18 to 25.
- **1227.** Page 771, line 1: delete lines 1 to 4.

- **1228.** Page 771, line 4: after that line insert:
- "Section 1677p. 49.84 (5) of the statutes is renumbered 49.84 (5) (b) and
   amended to read:
  - 49.84 (5) (b) A person applying for Wisconsin works under ss. 49.141 to 49.161, aid to families with dependent children under s. 49.19, medical assistance under subch. IV or food stamp program benefits under 7 USC 2011 to 2029 Notwithstanding any other eligibility requirements specified in this chapter, to the extent permitted under federal law, an applicant for an assistance program under this chapter shall, as a condition of eligibility, provide —a declaration and other verification documentary proof of citizenship or satisfactory immigration status as

required by rule by the department by rule administering the program or as required in 42 USC 1320b-7 (d), and only a person who is a U.S. citizen or an alien lawfully admitted for permanent residence or otherwise permanently residing in the United States under color of law shall be eligible for an assistance program under this chapter.

**Section 1677r.** 49.84 (5) (a) of the statutes is created to read:

49.84 **(5)** (a) In this subsection:

- 1. "Applicant" does not include any person applying for a program specified in sub. (6) (c) 1., 2., or 3. or any person excepted under sub. (6) (c) 1. a. to e. or 2.
- 2. "Assistance program under this chapter" means any program administered by the department of health and family services or the department of workforce development under this chapter under which the department administering the program provides services, benefits, or other assistance, except that "assistance program under this chapter" does not include any program specified in sub. (6) (c) 1., 2., or 3.

**Section 1677w.** 49.84 (5) (c) of the statutes is created to read:

- 49.84 (5) (c) 1. Every application for an assistance program under this chapter shall include a certification clause, which shall be completed by the welfare worker or other person processing the application, certifying that he or she has received from the applicant documentary proof that the applicant is a U.S. citizen or an alien lawfully admitted for permanent residence or otherwise permanently residing in the United States under color of law, and stating the nature of the documentary proof.
- 2. In addition to any other penalty, any person processing an application for an assistance program under this chapter who falsely certifies on the application that he or she received documentary proof under subd. 1. shall be required to pay a

- forfeiture of \$250 for each false certification. Notice of the penalty under this
- 2 subdivision shall be printed on the application directly below the certification clause
- 3 required under subd. 1.".
- **1229.** Page 772, line 17: delete the material beginning with that line and
- 5 ending with page 773, line 3.
- **1230.** Page 773, line 4: delete lines 4 to 24.
- **1231.** Page 774, line 1: delete lines 1 to 25.
- **1232.** Page 775, line 1: delete lines 1 to 24.
- **1233.** Page 776, line 1: delete lines 1 to 25.
- **1234.** Page 777, line 1: delete lines 1 to 25.
- **1235.** Page 778, line 1: delete lines 1 to 25.
- **1236.** Page 779, line 1: delete lines 1 to 25.
- **1237.** Page 780, line 1: delete lines 1 to 25.
- **1238.** Page 781, line 1: delete lines 1 to 11.
- **1239.** Page 782, line 23: delete lines 23 to 25.
- **1240.** Page 783, line 1: delete lines 1 to 25.
- **1241.** Page 784, line 1: delete lines 1 to 25.
- **1242.** Page 785, line 1: delete lines 1 to 25.
- **1243.** Page 786, line 1: delete lines 1 to 25.
- **1244.** Page 787, line 1: delete lines 1 to 25.
- **1245.** Page 788, line 1: delete lines 1 to 25.
- **1246.** Page 789, line 1: delete lines 1 to 24.

- **1247.** Page 790, line 1: delete lines 1 to 25.
- **1248.** Page 791, line 1: delete lines 1 to 25.
- **1249.** Page 792, line 1: delete lines 1 to 25.
- **1250.** Page 793, line 1: delete lines 1 to 25.
- **1251.** Page 794, line 1: delete lines 1 to 25.
- **1252.** Page 795, line 1: delete lines 1 to 25.
- **1253.** Page 796, line 1: delete lines 1 to 25.
- **1254.** Page 797, line 1: delete lines 1 to 9.
- **1255.** Page 797, line 10: delete lines 10 to 14.
- **1256.** Page 797, line 15: delete lines 15 to 25.
- **1257.** Page 798, line 1: delete lines 1 to 25.
- **1258.** Page 799, line 1: delete lines 1 to 25.
- **1259.** Page 800, line 1: delete the material beginning with that line and ending with page 804, line 23.
- **1260.** Page 803, line 23: after that line insert:

- 16 "Section 1772g. 50.035 (2) (d) of the statutes is created to read:
  - 50.035 (2) (d) Notwithstanding par. (a) 1., unless exempted by the department under standards that the department shall specify by rule, a community-based residential facility that is initially licensed after the effective date of this paragraph .... [revisor inserts date], shall use a complete National Fire Protection Association 13, 13R, or 13D automatic sprinkler system, as specified in the most current automatic sprinkler systems handbook of the National Fire Protection Association.

- The sprinkler system shall be equipped with residential sprinkler heads in all bedrooms, apartments, other habitable rooms, and corridors of the community-based residential facility and shall be connected to the fire alarm system
  - **1261.** Page 805, line 10: delete "the general fund, except amounts in excess of \$13,800,000 shall be deposited in" and substitute "the general fund, except amounts in excess of \$13,800,000 shall be deposited in".
- **1262.** Page 805, line 15: delete "\$127" and substitute "\$75".
- **1263.** Page 806, line 5: delete that line.

of the community-based residential facility.".

- **1264.** Page 806, line 6: delete the material beginning with that line and ending with page 807, line 7.
- **1265.** Page 807, line 8: delete that line.
- **1266.** Page 807, line 9: delete lines 9 to 12.
- **1267.** Page 807, line 13: delete lines 13 to 22.
- **1268.** Page 807, line 22: after that line insert:
- **"Section 1810r.** 51.03 (6) of the statutes is created to read:
  - 51.03 (6) The department shall issue a request for proposals to provide pharmacy management services for the Winnebago Mental Health Institute and the Wisconsin Resource Center and may, in response to the request for proposals, prepare an offer to continue to provide these pharmacy management services. The department shall select the offer that meets the request–for–proposal requirements at the lowest cost and shall award to the organization submitting that offer a one–year contract with an option to renew for 3 additional one–year periods.".

- **1269.** Page 807, line 23: delete lines 23 to 25.
- **1270.** Page 808, line 1: delete lines 1 to 5.
- **1271.** Page 808, line 6: delete lines 6 to 21.
- 1272. Page 808, line 22: delete the material beginning with that line and ending with page 809, line 21.
- **1273.** Page 809, line 21: after that line insert:
- 7 "Section 1816s. 51.30 (4) (b) 8g. (intro.) of the statutes is amended to read:
  - 51.30 (4) (b) 8g. (intro.) To health care providers in a related health care entity, or to any person acting under the supervision of such a health care provider who is involved with an individual's care, if necessary for the current treatment of the individual. Information that may be released under this subdivision is limited to the individual's name, address, and date of birth; the name of the individual's mental health treatment provider; the date of mental health service provided; the individual's medications, allergies, and diagnosis; and other relevant demographic medical information necessary for the current treatment of the individual. In this subdivision, "related health care entity" means one of the following:".
- **1274.** Page 809, line 22: delete lines 22 to 25.
- **1275.** Page 810, line 1: delete lines 1 to 13.
- **1276.** Page 810, line 13: after that line insert:
- **"Section 1819m.** 51.42 (3) (ar) 4. b. of the statutes is amended to read:
- 51.42 **(3)** (ar) 4. b. Comprehensive diagnostic and evaluation services, including assessment as specified under ss. <u>114.09 (2) (bm)</u>, 343.30 (1q) and 343.305 (10) and assessments under ss. 48.295 (1) and 938.295 (1).".

- **1277.** Page 810, line 14: delete lines 14 to 25.
- **1278.** Page 811, line 1: delete lines 1 to 15.
- **1279.** Page 811, line 16: delete the material beginning with that line and ending with page 813, line 16.
- **1280.** Page 813, line 17: delete lines 17 to 25.
- **1281.** Page 814, line 1: delete lines 1 to 20.
- **1282.** Page 814, line 21: delete the material beginning with that line and ending with page 818, line 8.
- **1283.** Page 818, line 14: delete the material beginning with that line and ending with page 822, line 10.
- **1284.** Page 822, line 10: after that line insert:

- **"Section 1843p.** 59.10 (1) (b) of the statutes is amended to read:
  - 59.10 (1) (b) *Terms*. The term of office of supervisors is 2 years. A board may determine whether the terms shall be concurrent or staggered. Supervisors shall be elected at the election to be held on the first Tuesday in April next preceding the expiration of their respective terms and shall take office on the 3rd Tuesday in April following their election. If the board determines that supervisors shall serve staggered terms, the board shall, by ordinance, provide for a division of supervisors into 2 classes, one class to be elected for one-half of a full term and the other class for a full term and thereafter the supervisors shall be elected for a full term. The board shall publish the ordinance as a class 1 notice, under ch. 985, or as a notice, as described under s. 59.14 (1m) (b), before publication of the notice of the election at which supervisors are to be elected.

1	<b>Section 1843r.</b> 59.14 (1) of the statutes is amended to read:
2	59.14 (1) Whenever a board enacts an ordinance under this chapter the clerk
3	shall immediately publish it the ordinance either in its entirety, as a class 1 notice
4	under ch. 985, or as a notice, as described under sub. (1m) (b); and the clerk shall
5	procure and distribute copies of the ordinance to the several town clerks, who shall
6	file it in their respective offices.
7	SECTION 1843s. 59.14 (1m) of the statutes is created to read:
8	59.14 (1m) (a) In this subsection, "summary" means a brief, precise, and
9	plain-language description that can be easily understood.
10	(b) A notice of an ordinance that may be published under this subsection shall
11	be published as a class 1 notice under ch. 985 and shall contain at least all of the
12	following:
13	1. The number and title of the ordinance.
14	2. The date of enactment.
15	3. A summary of the subject matter and main points of the ordinance.
16	4. Information as to where the full text of the ordinance may be obtained
17	including the phone number of the county clerk, a street address where the full text
18	of the ordinance may be viewed, and a Web site, if any, at which the ordinance may
19	be accessed.".
20	<b>1285.</b> Page 822, line 11: delete lines 11 to 16.
21	<b>1286.</b> Page 822, line 16: after that line insert:
22	"Section 1844g. 59.25 (3) (j) of the statutes is amended to read:
23	59.25 (3) (j) Retain 10% for 20 percent of fines and forfeitures collected as fees

in for receiving and paying into the state treasury all money received by the treasurer

for the state for fines and forfeitures, except that 50%, retain 30 percent of the fines and forfeitures collected within one year after the effective date of this paragraph .... [revisor inserts date], that were imposed at least 180 days before the effective date of this paragraph .... [revisor inserts date], retain 50 percent of the state forfeitures and fines under chs. 341 to 347, 349, and 351 shall be retained as fees, and retain the other fees for receiving and paying money into the state treasury that are prescribed by law.".

**1287.** Page 822, line 16: after that line insert:

"Section 1844h. 59.27 (2m) of the statutes is created to read:

59.27 (2m) Verify the citizenship of any prisoner committed to any jail under the sheriff's charge if the prisoner is charged with a felony or with a violation of s. 23.33 (4c), 30.681, 346.63, or 350.101. If the prisoner is not a U.S. citizen or is not otherwise authorized to be in the United States under federal law, the sheriff shall notify U.S. Immigration and Customs Enforcement.".

**1288.** Page 822, line 17: delete lines 17 to 25.

**1289.** Page 823, line 1: delete lines 1 to 25.

**1290.** Page 824, line 1: delete lines 1 to 25.

**1291.** Page 825, line 1: delete lines 1 to 7.

**1292.** Page 825, line 7: after that line insert:

"Section 1851s. 59.69 (3) (a) of the statutes is amended to read:

59.69 (3) (a) The county zoning agency may direct the preparation of a county development plan or parts of the plan for the physical development of the unincorporated territory within the county and areas within incorporated jurisdictions whose governing bodies by resolution agree to having their areas

- included in the county's development plan. The plan may be adopted in whole or in part and may be amended by the board and endorsed by the governing bodies of incorporated jurisdictions included in the plan. The county development plan, in whole or in part, in its original form or as amended, is hereafter referred to as the development plan. Beginning on January 1, 2010 2015, if the county engages in any program or action described in s. 66.1001 (3), the development plan shall contain at least all of the elements specified in s. 66.1001 (2).".
- **1293.** Page 825, line 7: after that line insert:
- **"Section 1851c.** 59.58 (6) (cg) 3. of the statutes is created to read:
- 10 59.58 (6) (cg) 3. The authority may not use any revenues received under subd.
- 1. for lobbying activities or to contract for lobbying services.".
- **1294.** Page 825, line 8: delete lines 8 to 25.
- **1295.** Page 826, line 1: delete lines 1 to 24.
- **1296.** Page 827, line 1: delete lines 1 to 3.
- **1297.** Page 827, line 17: delete lines 17 to 24.
- **1298.** Page 828, line 1: delete lines 1 to 25.
- **1299.** Page 829, line 1: delete lines 1 to 12.
- **1300.** Page 829, line 12: after that line insert:
- **"Section 1866b.** 60.77 (5) (c) of the statutes is amended to read:
- 20 60.77 **(5)** (c) Issue rules or orders, which shall be published <u>either in their</u> 21 <u>entirety</u>, as a class 1 notice under ch. 985, or as a notice, as described under sub. (5s) 22 (b).
- **Section 1866c.** 60.77 (5m) of the statutes is amended to read:

1	60.77 (5m) Authority to enact ordinances. The commission may enact and
2	enforce ordinances to implement the powers listed under sub. (5). The ordinances
3	shall be published either in their entirety, as a class 1 notice under ch. 985, or as a
4	notice, as described under sub. (5s) (b).
5	<b>Section 1866d.</b> 60.77 (5s) of the statutes is created to read:
6	60.77 (5s) Requirements for notice. (a) In this subsection, "summary" has
7	the meaning given in s. $59.14 (1m) (a)$ .
8	(b) A notice of an ordinance, rule, or order that may be published under this
9	subsection shall be published as a class 1 notice under ch. 985 and shall contain at
10	least all of the following:
11	1. The number and title of the ordinance, rule, or order.
12	2. The date of enactment.
13	3. A summary of the subject matter and main points of the ordinance, rule, or
14	order.
15	4. Information as to where the full text of the ordinance, rule, or order may be
16	obtained, including the phone number of the commission's secretary, a street address
17	where the full text of the ordinance, rule, or order may be viewed, and a Web site, it
18	any, at which the ordinance, rule, or order may be accessed.
19	<b>Section 1866e.</b> 60.80 (1) (intro.) of the statutes is amended to read:
20	60.80 (1) GENERAL REQUIREMENT. (intro.) The town clerk shall publish either
21	in its entirety, as a class 1 notice under ch. 985, or as a notice, as described under sub
22	(5) (b), or post in at least 3 places in the town likely to give notice to the public, the
23	following, within 30 days after passage or adoption:

**Section 1866f.** 60.80 (5) of the statutes is created to read:

 $\mathbf{2}$ 

- 60.80 (5) REQUIREMENTS FOR NOTICE. (a) In this subsection, "summary" has the meaning given in s. 59.14 (1m) (a).
- (b) A notice of a resolution, motion, ordinance, or action that may be published under this subsection shall be published as a class 1 notice under ch. 985 and shall contain at least all of the following:
  - 1. The number and title of the resolution, motion, ordinance, or action.
  - 2. The date of enactment.
- 3. A summary of the subject matter and main points of the resolution, motion, ordinance, or action.
- 4. Information as to where the full text of the resolution, motion, ordinance, or action may be obtained, including the phone number of the town clerk, a street address where the full text of the resolution, motion, ordinance, or action may be viewed, and a Web site, if any, at which the resolution, motion, ordinance, or action may be accessed."

**Section 1866h.** 61.50 (1) of the statutes is amended to read:

61.50 (1) Publication of Notice of Ordinances. Every contract, conveyance, commission, license or other written instrument shall be executed on the part of the village by the president and clerk, sealed with corporate seal, and in pursuance only of authority therefor from the village board. All ordinances and bylaws shall be signed by the president and countersigned by the clerk; and, if any penalty or forfeiture is thereby imposed, the ordinance or bylaw shall be published either in its entirety, as a class 1 notice, under ch. 985, or as a notice, as described under sub. (3) (b), and shall take effect on the day after its the publication or a later date if expressly prescribed. If there is no newspaper published in the village, the village board may in lieu of newspaper publication have copies of said the ordinances and bylaws posted

1	in at least 3 public places in said village, and proof thereof filed and recorded by the
2	village clerk, and the same shall take effect the day after the proof of posting has been
3	filed and recorded, or at a later date if expressly provided in the ordinance or bylaw.
4	<b>Section 1866i.</b> 61.50 (1m) of the statutes is amended to read:
5	61.50 (1m) Indeterminate publication. When any village ordinance is
6	required by law to be published without express designation therein as to class of
7	notice, it the ordinance shall be published either in its entirety, as a class 1 notice
8	under ch. 985, or as a notice, as described under sub. (3) (b).
9	<b>Section 1866j.</b> 61.50 (3) of the statutes is created to read:
10	61.50 (3) REQUIREMENTS FOR NOTICE. (a) In this subsection, "summary" has the
11	meaning given in s. 59.14 (1m) (a).
12	(b) A notice of an ordinance or bylaw that may be published under this
13	subsection shall be published as a class 1 notice under ch. 985 and shall contain at
14	least all of the following:
15	1. The number and title of the ordinance or bylaw.
16	2. The date of enactment.
17	3. A summary of the subject matter and main points of the ordinance or bylaw.
18	4. Information as to where the full text of the ordinance or bylaw may be
19	obtained, including the phone number of the village clerk, a street address where the
20	full text of the ordinance or bylaw may be viewed, and a Web site, if any, at which the
21	ordinance or bylaw may be accessed.
22	Section 1866u. 62.11 (4) (a) of the statutes is amended to read:
23	62.11 (4) (a) Proceedings of the council shall be published in the newspaper
24	designated under s. 985.06 as a class 1 notice, under ch. 985. The proceedings for the

purpose of publication shall include the substance of every official action taken by

5

8

9

10

11

12

13

14

15

16

17

1	the governing body. Except as provided in this subsection all ordinances every
2	ordinance shall be published either in its entirety, as a class 1 notice, under ch. 985,
3	or as a notice, as described under par. (c) 2., within 15 days of passage, and shall take
4	effect on the day after its the publication or at a later date if expressly prescribed.

**Section 1866v.** 62.11 (4) (c) of the statutes is created to read:

- 62.11 (4) (c) 1. In this paragraph, "summary" has the meaning given in s. 59.14 6 7 (1m)(a).
  - 2. A notice of an ordinance that may be published under this paragraph shall be published as a class 1 notice under ch. 985 and shall contain at least all of the following:
    - a. The number and title of the ordinance.
    - b. The date of enactment.
    - c. A summary of the subject matter and main points of the ordinance.
  - d. Information as to where the full text of the ordinance may be obtained, including the phone number of the city clerk, a street address where the full text of the ordinance may be viewed, and a Web site, if any, at which the ordinance may be accessed.".
- **1301.** Page 829, line 13: delete lines 13 to 25. 18
- **1302.** Page 830, line 1: delete lines 1 to 15. 19
- **1303.** Page 830, line 15: after that line insert: 20
- 21 **"Section 1867s.** 62.23 (3) (b) of the statutes is amended to read:
- 22 62.23 (3) (b) The commission may adopt the master plan as a whole by a single 23resolution, or, as the work of making the whole master plan progresses, may from 24time to time by resolution adopt a part or parts of a master plan. Beginning on

 $\mathbf{2}$ 

January 1, 2010 2015, if the city engages in any program or action described in s. 66.1001 (3), the master plan shall contain at least all of the elements specified in s. 66.1001 (2). The adoption of the plan or any part, amendment, or addition, shall be by resolution carried by the affirmative votes of not less than a majority of all the members of the city plan commission. The resolution shall refer expressly to the elements under s. 66.1001 and other matters intended by the commission to form the whole or any part of the plan, and the action taken shall be recorded on the adopted plan or part of the plan by the identifying signature of the secretary of the commission, and a copy of the plan or part of the plan shall be certified to the common council, and also to the commanding officer, or the officer's designee, of any military base or installation, with at least 200 assigned military personnel or that contains at least 2,000 acres, that is located in or near the city. The purpose and effect of the adoption and certifying of the master plan or part of the plan shall be solely to aid the city plan commission and the council in the performance of their duties.".

- **1304.** Page 830, line 16: delete lines 16 to 25.
- **1305.** Page 831, line 1: delete lines 1 to 24.
- **1306.** Page 832, line 1: delete lines 1 to 12.
- **1307.** Page 832, line 12: after that line insert:
- **"Section 1873d.** 62.23 (7a) (b) of the statutes is amended to read:

62.23 (7a) (b) The governing body may enact, without referring the matter to the plan commission, an interim zoning ordinance to preserve existing zoning or uses in all or part of the extraterritorial zoning jurisdiction while the comprehensive zoning plan is being prepared. Such ordinance may be enacted as is an ordinary ordinance but shall be effective for no longer than 2 years after its enactment, unless

 $\mathbf{2}$ 

extended as provided in this paragraph. Within 15 days of its passage, the governing body of the city shall publish the ordinance in a newspaper having general circulation in the area proposed to be zoned as a class 1 notice, under ch. 985, or as a notice, as described under s. 62.11 (4) (c) 2., and the city clerk shall mail a certified copy of the ordinance to the clerk of the county in which the extraterritorial jurisdiction is located and to the clerk of each town affected by the interim zoning ordinance and shall file a copy of the ordinance with the city plan commission. The governing body of the city may extend the interim zoning ordinance for no longer than one year, upon the recommendation of the joint extraterritorial zoning committee established under par. (c). No other interim zoning ordinance shall be enacted affecting the same area or part thereof until 2 years after the date of the expiration of the interim zoning ordinance or the one year extension thereof. While the interim zoning ordinance is in effect, the governing body of the city may amend the districts and regulations of the ordinance according to the procedure set forth in par. (f)."

**1308.** Page 832, line 13: delete lines 13 to 17.

**1309.** Page 832, line 17: after that line insert:

**"Section 1874c.** 66.0137 (4) of the statutes is amended to read:

66.0137 (4) Self-insured health plans. If a city, including a 1st class city, or a village provides health care benefits under its home rule power, or if a town provides health care benefits, to its officers and employees on a self-insured basis, the self-insured plan shall comply with ss. 49.493 (3) (d), 631.89, 631.90, 631.93 (2), 632.746 (10) (a) 2. and (b) 2., 632.747 (3), 632.85, 632.853, 632.855, 632.87 (4) and (5), 632.895 (9) to (14) (15), 632.896, and 767.25 (4m) (d).".

- **1310.** Page 833, line 22: delete the material beginning with that line and ending with page 834, line 21.
- **1311.** Page 834, line 25: after that line insert:
- 4 "Section 1879e. 66.0602 (1) (at) of the statutes is created to read:
- 5 66.0602 (1) (at) "Municipality" means a city, village or town.".
- **1312.** Page 835, line 2: delete ", in an amount that is at least".
- **1313.** Page 835, line 3: delete "\$500".

- **1314.** Page 835, line 5: delete lines 5 to 13 and substitute:
- 9 "Section 1881b. 66.0602 (1) (d) of the statutes is renumbered 66.0602 (1) (d) 10 (intro.) and amended to read:
  - 66.0602 (1) (d) (intro.) "Valuation factor" means <u>a percentage equal to the</u> greater of either zero percent or the percentage one of the following:
    - 1. For a municipality, the sum of the change in the political subdivision's January 1 equalized value due to new construction less improvements removed between the previous year and the current year, but not less than 2. Except as provided in subs. (3), (4), and (5), no political subdivision may increase its levy in any year by a percentage that exceeds the political subdivision's valuation factor. In determining its levy in any year, a city, village, or town shall subtract any tax increment that is calculated under s. 60.85 (1) (L) or 66.1105 (2) (i) and, if the department of revenue does not certify a value increment for a tax incremental district for the current year due to the district's termination, 50 percent of the value increment of such a district in the municipality, calculated for the previous year; and that sum is divided by the municipality's equalized value for the previous year to which the levy relates.

**SECTION 1881e.** 66.0602 (1) (d) 2. of the statutes is created to read:

66.0602 (1) (d) 2. For a county, a percentage equal to the percentage change in the county's January 1 equalized value due to new construction less improvements removed between the previous year and the current year.".

- **1315.** Page 835, line 16: delete "or 2008" and substitute ", 2008, or 2009".
- **1316.** Page 835, line 18: delete "maximum allowable" and substitute 7 "political subdivision's".
- **1317.** Page 835, line 19: delete "city, village, town".
  - **1318.** Page 835, line 20: delete "or county" and substitute "political subdivision".
    - **1319.** Page 835, line 21: after that line insert:
    - **"Section 1882u.** 66.0602 (3) (d) 2. of the statutes is amended to read:
      - 66.0602 (3) (d) 2. The limit otherwise applicable under this section does not apply to amounts levied by a political subdivision for the payment of any general obligation debt service, including debt service on debt issued or reissued to fund or refund outstanding obligations of the political subdivision, interest on outstanding obligations of the political subdivision, or the payment of related issuance costs or redemption premiums, authorized on or after July 1, 2005, and before July 1, 2007, and secured by the full faith and credit of the political subdivision.
        - **Section 1883e.** 66.0602 (3) (d) 4. of the statutes is created to read:
      - 66.0602 (3) (d) 4. If the amount of a lease payment related to a lease revenue bond for a political subdivision in the preceding year is less than the amount of the lease payment needed in the current year, as a result of the issuance of a lease revenue bond before July 1, 2005, the levy increase limit otherwise applicable under

1 this section to the political subdivision in the current year is increased by the 2 difference between these 2 amounts. 3 **Section 1883g.** 66.0602 (3) (d) 5. of the statutes is created to read: 4 66.0602 (3) (d) 5. The limit otherwise applicable under this section does not 5 apply to amounts levied by a political subdivision for the payment of any general 6 obligation debt service, including debt service on debt issued or reissued to fund or 7 refund outstanding obligations of the political subdivision, interest on outstanding 8 obligations of the political subdivision, or the payment of related issuance costs or 9 redemption premiums, authorized by referendum on or after July 1, 2007, and 10 secured by the full faith and credit of the political subdivision.". **1320.** Page 836, line 13: delete lines 13 to 19. 11 12 **1321.** Page 836, line 19: after that line insert: 13 **"Section 1889e.** 66.0602 (3) (e) 6. of the statutes is created to read: 14 66.0602 (3) (e) 6. The amount that a county levies in that year for a countywide emergency medical system. 15 16 **Section 1889g.** 66.0602 (3) (e) 7. of the statutes is created to read: 17 66.0602 (3) (e) 7. The amount that a village levies in that year for police 18 protection services, but this subdivision applies only to a village's levy for the year 19 immediately after the year in which the village changes from town status and 20 incorporates as a village, and only if the town did not have a police force.". 21 **1322.** Page 837, line 2: after that line insert: 22 **"Section 1891e.** 66.0602 (3) (i) of the statutes is created to read: 23 66.0602 (3) (i) If a political subdivision receives an aid increase under s. 79.04

(4m) in the preceding year, the levy increase limit otherwise applicable under this

section to the political subdivision in the current year is decreased to reflect 50 percent of the amount of that aid increase, as determined by the department of revenue.".

**1323.** Page 838, line 18: delete lines 18 to 22.

1324. Page 839, line 1: delete lines 1 to 9 and substitute "be imposed under sub. (6) if a taxation district clerk or a county clerk, through mistake or inadvertence in preparing or delivering the tax roll, causes a political subdivision's actual levy in the year before the current year to be different from the amount intended by the political subdivision's governing body. The political subdivision shall use the intended levy as its base amount in determining its maximum allowable levy for the current year if the department of revenue determines that the final action taken by the political subdivision, in setting its levy for the year before the current year, was to set the levy at the intended amount and not at the actual amount. If the department makes such a determination it may issue a finding that the political subdivision is not liable for a penalty that would otherwise be imposed under sub. (6).".

**1325.** Page 839, line 15: delete lines 15 to 20.

**1326.** Page 839, line 20: after that line insert:

**"Section 1903m.** 66.1001 (1) (b) of the statutes is amended to read:

66.1001 (1) (b) "Local governmental unit" means a city, village, town, county, or regional planning commission that may adopt, prepare, or amend a comprehensive plan and, in the case of a political subdivision, that has a population of at least 2,500.

**Section 1903n.** 66.1001 (3) (intro.) of the statutes is amended to read:

66.1001 (3) Actions, procedures that must be consistent with comprehensive plans. (intro.) Beginning on January 1, 2010 2015, if a local governmental unit engages in any of the following actions, those actions shall be consistent with that local governmental unit's comprehensive plan:".

**1327.** Page 839, line 20: after that line insert:

**"Section 1903c.** 66.0903 (4) (b) 1. of the statutes is amended to read:

66.0903 (4) (b) 1. The laborer, worker, mechanic, or truck driver is employed to go to the source of mineral aggregate such as sand, gravel, or stone that is to be immediately incorporated into the work, and not stockpiled or further transported by truck, pick up that mineral aggregate, and deliver that mineral aggregate to the site of a project that is subject to this section by depositing and immediately incorporating the material substantially in place into the work at the final location where the material is to be incorporated into the work, directly or through spreaders from the transporting vehicle, without the need to pick up and move the material to that final location.".

**1328.** Page 839, line 21: delete lines 21 to 24.

**1329.** Page 839, line 24: after that line insert:

**SECTION 1904m.** 67.05 (6a) (a) 2. a. of the statutes is amended to read:

67.05 **(6a)** (a) 2. a. Direct the school district clerk to call a special election for the purpose of submitting the resolution to the electors for approval or rejection, or direct that the resolution be submitted at the next regularly scheduled primary or election to be held not earlier than 45 days after the adoption of the resolution. The resolution shall not be effective unless adopted by a majority of the school district electors voting at the referendum.".

 $\mathbf{2}$ 

**1330.** Page 839, line 24: after that line insert:

**SECTION 1912b.** 67.12 (12) (a) of the statutes is amended to read:

67.12 (12) (a) Any municipality may issue promissory notes as evidence of indebtedness for any public purpose, as defined in s. 67.04 (1) (b), including but not limited to paying any general and current municipal expense, and refunding any municipal obligations, including interest on them. Each note, plus interest if any, shall be repaid within 10 years after the original date of the note, except that notes issued under this section for purposes of ss. 119.498, 145.245 (12m), 281.58, 281.59, 281.60 and 281.61, or to raise funds to pay a portion of the capital costs of a metropolitan sewerage district, shall be repaid within 20 years after the original date of the note."

**1331.** Page 839, line 24: after that line insert:

"Section 1911s. 67.12 (1) (b) of the statutes is renumbered 67.12 (1) (b) 1. and amended to read:

67.12 (1) (b) 1. Any municipality may issue municipal obligations in anticipation of receiving proceeds from clean water fund loans or grants for which the municipality has received a notice of financial assistance commitment under s. 281.58 (15), from bonds or notes the municipality has authorized or has covenanted to issue under this chapter or from grants that are committed to the municipality. Any municipal obligation issued under this paragraph subdivision may be refunded one or more times. Such obligation and any refundings thereof shall be repaid within 5 years after the original date of the original obligation.

**Section 1911t.** 67.12 (1) (b) 2. of the statutes is created to read:

 $\mathbf{2}$ 

67.12 (1) (b) 2. Any municipality may issue municipal obligations in anticipation of receiving proceeds from brownfields revolving loan program loans or grants under the program described in s. 292.72 if the municipality has received written notification from the department of natural resources that the department intends to distribute such proceeds to the municipality. The obligation shall be repaid within 10 years after the original date of the obligation, except that the obligation may be refunded one or more times. Any refundings shall be repaid within 20 years after the original date of the original obligation.

**Section 1912.** 67.12 (12) (a) of the statutes is amended to read:

67.12 (12) (a) Any municipality may issue promissory notes as evidence of indebtedness for any public purpose, as defined in s. 67.04 (1) (b), including but not limited to paying any general and current municipal expense, and refunding any municipal obligations, including interest on them. Each note, plus interest if any, shall be repaid within 10 years after the original date of the note, except that notes issued under this section for purposes of ss. 119.498, 145.245 (12m), 281.58, 281.59, 281.60 and, 281.61, and 292.72 or to raise funds to pay a portion of the capital costs of a metropolitan sewerage district, shall be repaid within 20 years after the original date of the note.".

**1332.** Page 839, line 24: after that line insert:

"Section 1905c. 66.1019 (3) (a) of the statutes is renumbered 66.1019 (3) and amended to read:

66.1019 (3) Except as provided in par. (b), any Any ordinance enacted by a county, city, village, or town relating to the construction or inspection of multifamily

- dwellings, as defined in s. 101.971 (2), shall conform to subch. VI of ch. 101 and s.
- 2 101.02 (7m).

- **Section 1905d.** 66.1019 (3) (b) of the statutes is repealed.".
- **1333.** Page 839, line 25: delete that line.
- **1334.** Page 840, line 1: delete lines 1 to 25.
- **1335.** Page 841, line 1: delete lines 1 to 7.
- **1336.** Page 841, line 7: after that line insert:
- **"Section 1918g.** 69.22 (1) (a) of the statutes is amended to read:
- 9 69.22 (1) (a) Except as provided under par. (c), \$7 \$20 for issuing one certified copy of a vital record and \$3 for any additional certified copy of the same vital record issued at the same time.
- **SECTION 1918h.** 69.22 (1) (a) of the statutes, as affected by 2007 Wisconsin Act
  13 .... (this act), is amended to read:
  - 69.22 (1) (a) Except as provided under par. (c), \$20 \$7 for issuing one certified copy of a vital record and \$3 for any additional certified copy of the same vital record issued at the same time.
- **SECTION 1918i.** 69.22 (1) (b) of the statutes is amended to read:
  - 69.22 **(1)** (b) Except as provided under par. (c), \$20 for issuing an uncertified copy of a vital record issued under s. 69.21 (2) (a) or (b), or \$7 for verifying information about the event submitted by a requester without issuance of a copy, \$7, and \$3 for any additional copy of the same vital record issued at the same time.
- **SECTION 1918j.** 69.22 (1) (b) of the statutes, as affected by 2007 Wisconsin Act 23 .... (this act), is amended to read:

69.22 (1) (b) Except as provided under par. (c), \$20 for issuing an uncertified
copy of a vital record issued under s. 69.21 (2) (a) or (b), \$7 or for verifying information
about the event submitted by a requester without issuance of a copy, <u>\$7</u> , and \$3 for
any additional copy of the same vital record issued at the same time.
SECTION 1918k. 69.22 (1) (c) of the statutes is amended to read:
69.22 (1) (c) Twelve Twenty dollars for issuing an uncertified copy of a birth
certificate or a certified copy of a birth certificate, \$7 of which shall be forwarded to
the secretary of administration as provided in sub. (1m) and credited to the
appropriations under s. 20.433 (1) (g) and (h); and \$3 for issuing any additional
certified or uncertified copy of the same birth certificate issued at the same time.
SECTION 1918L. 69.22 (1) (c) of the statutes, as affected by 2007 Wisconsin Act
(this act), is amended to read:
69.22 (1) (c) Twenty Twelve dollars for issuing an uncertified copy of a birth
certificate or a certified copy of a birth certificate, \$7 of which shall be forwarded to
the secretary of administration as provided in sub. (1m) and credited to the
appropriations under s. 20.433 (1) (g) and (h); and \$3 for issuing any additional
certified or uncertified copy of the same birth certificate issued at the same time.
Section 1918m. 69.22 (1) (d) of the statutes is amended to read:
69.22 (1) (d) In addition to other fees under this subchapter, \$10 \$20 for
expedited service in issuing a vital record.
SECTION 1918n. 69.22 (1) (d) of the statutes, as affected by 2007 Wisconsin Act
(this act), is amended to read:
69.22 (1) (d) In addition to other fees under this subchapter, \$20 \$10 for
expedited service in issuing a vital record.

**SECTION 1918p.** 69.22 (1m) of the statutes is amended to read:

69.22 (1m) The Except as provided in sub. (1p), the state registrar and any local
registrar acting under this subchapter shall, for each copy of a birth certificate for
which a fee under sub. $(1)$ $(c)$ is charged that is issued during a calendar quarter,
forward to the secretary of administration for deposit in the appropriations under $\mathbf{s}$ .
$20.433\ (1)\ (g)$ and (h) the amounts specified in sub. (1) (c) by the 15th day of the first
month following the end of the calendar quarter.

**SECTION 1918q.** 69.22 (1m) of the statutes, as affected by 2007 Wisconsin Act .... (this act), is amended to read:

69.22 (1m) Except as provided in sub. (1p), the The state registrar and any local registrar acting under this subchapter shall, for each copy of a birth certificate for which a fee under sub. (1) (c) is charged that is issued during a calendar quarter, forward to the secretary of administration for deposit in the appropriations under s. 20.433 (1) (g) and (h) the amounts specified in sub. (1) (c) by the 15th day of the first month following the end of the calendar quarter.

**Section 1918r.** 69.22 (1p) of the statutes is created to read:

- 69.22 (1p) The state registrar and any local registrar acting under this subchapter shall forward to the secretary of administration for deposit in the appropriation account under s. 20.435 (1) (gm) all of the following:
- (a) For any certified copy of a vital record that is issued before July 1, 2010, for which a fee of \$20 under sub. (1) (a) is charged, \$13.
- (b) For any uncertified copy of a vital record that is issued before July 1, 2010, for which a fee of \$20 under sub. (1) (b) is charged, \$13.
- (c) For any copy of a birth certificate that is issued before July 1, 2010, for which a fee of \$20 under sub. (1) (c) is charged, \$8.
  - (d) For expedited service in issuing a vital record before July 1, 2010, \$10.

amended to read:

SECTION 1918s. 69.22 (1p) of the statutes, as created by 2007 Wisconsin Act .... 1 2 (this act), is repealed.". **1337.** Page 841, line 8: delete lines 8 to 20. 3 **1338.** Page 841, line 21: delete the material beginning with that line and 4 5 ending with page 842, line 18. 1339. Page 842, line 19: delete the material beginning with that line and 6 7 ending with page 843, line 9. **1340.** Page 843, line 10: delete the material beginning with that line and 8 9 ending with page 844, line 14. 10 **1341.** Page 844, line 14: after that line insert: 11 **"Section 1932g.** 70.11 (27m) of the statutes is created to read: 12 70.11 (27m) RESTAURANT KITCHEN EQUIPMENT. (a) In this subsection: 13 1. "Machinery" has the meaning given in sub. (27) (a) 2. 2. "Restaurant" includes pizza delivery establishments, snack bars, beverage 14 15 bars, take-out food shops, mobile food services, including mobile snack stands, mobile canteens, and ice cream vendors, and any other entity that is required to have 16 17 a permit under s. 254.64 (1) (a). 18 Machinery and equipment, including refrigerators and other storage equipment, used primarily in the operation of a restaurant's kitchen to prepare or 19 20 serve food or beverages, regardless of whether the machinery or equipment is attached to real property.". 2122 **1342.** Page 844, line 14: after that line insert: 23 "Section 1932r. 70.11 (21) (a) of the statutes is renumbered 70.11 (21) (am) and

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

70.11 (21) (am) All property purchased or constructed as a waste treatment facility used for the treatment of exclusively and directly to remove, store, or cause a physical or chemical change in industrial wastes, as defined in s. 281.01 (5), waste or air contaminants, as defined in s. 285.01 (1), but not for other wastes, as defined in s. 281.01 (7), for the purpose of abating or eliminating pollution of surface waters. the air, or waters of the state if that property is not used to grow agricultural products for sale and, if the property's owner is taxed under ch. 76, if the property is approved by the department of revenue. For the purposes of this subsection, "industrial waste" also includes wood chips, sawdust, and other wood residue from the paper and wood products manufacturing process that can be used as fuel and would otherwise be considered superfluous, discarded, or fugitive material. The department of natural and department of health and family services shall make resources recommendations upon request to the department of revenue regarding such All property purchased or upon which construction began prior to July 31, 1975, shall be subject to s. 70.11 (21), 1973 stats.

**Section 1932t.** 70.11 (21) (ab) of the statutes is created to read:

70.11 **(21)** (ab) In this subsection:

- 1. "Air contaminants" has the meaning given in s. 285.01 (1).
- 2. "Industrial waste" means waste resulting from any process of industry, trade, or business, or the development of any natural resource, that has no monetary or market value, except as provided in subd. 3. b., and that would otherwise be considered superfluous, discarded, or fugitive material. "Industrial waste" does not include other wastes, as defined in s. 281.01 (7).
- 3. "Used exclusively" means to the exclusion of all other uses except any of the following:

 $\mathbf{2}$ 

- a. For other use not exceeding 5 percent of total use.
- b. To produce heat or steam for a manufacturing process, if the fuel consists of either 95 percent or more industrial waste that would otherwise be considered superfluous, discarded, or fugitive material or 50 percent or more of wood chips, sawdust, or other wood residue from the paper and wood products manufacturing process, if the wood chips, sawdust, or other wood residue would otherwise be considered superfluous, discarded, or fugitive material.".
  - **1343.** Page 844, line 24: delete the material beginning with that line and ending with page 845, line 6.
  - **1344.** Page 845, line 6: after that line insert:
- **"Section 1935e.** 70.119 (10) of the statutes is created to read:
  - 70.119 (10) Beginning with payments to be made in fiscal year 2007–08, the department shall compute payments under this section and assessments under sub. (7) (c) using the sum of the appropriation under s. 20.835 (5) (a) and \$5,000,000. Prior to making the payment under sub. (7) (a), the department shall reduce the payment to the city of Madison by \$5,000,000.".
  - **1345.** Page 856, line 2: delete that line and substitute "109–227, P.L. 109–280, excluding sections 811 and 844 of P.L. 109–280, and P.L. 109–432, excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432, and as".
  - **1346.** Page 856, line 24: delete that line and substitute "109–227, P.L. 109–280, excluding sections 811 and 844 of P.L. 109–280, and P.L. 109–432, excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432. The".

- 1 **1347.** Page 857, line 9: delete "and P.L. 109-280" and substitute "P.L.
- 2 <u>109–280</u>".
- 3 **1348.** Page 857, line 10: after "P.L. 109–280," insert "and P.L. 109–432,
- 4 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 5 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,".
- 6 **1349.** Page 857, line 16: delete "and P.L. 109–280" and substitute "P.L.
- 7 <u>109–280</u>".
- 8 **1350.** Page 857, line 17: after "P.L. 109-280," insert "and P.L. 109-432,
- 9 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 10 <u>303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,</u>".
- 11 **1351.** Page 858, line 10: delete "and P.L. 109–280" and substitute "P.L.
- 12 109–280".
- 13 **1352.** Page 858, line 11: after "109–280," insert "and P.L. 109–432, excluding
- division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304,
- 305, 307, 404, 424, and 425, of P.L. 109-432,".
- 16 **1353.** Page 859, line 7: delete "P.L. 109-227, and" and substitute "P.L.
- 17 109–227,".
- 18 **1354.** Page 859, line 8: after "P.L. 109–280" insert ", and P.L. 109–432,
- 19 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 20 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432".
- 21 **1355.** Page 859, line 14: delete "and P.L. 109–280" and substitute "P.L.
- 22 109–280".

- 1 **1356.** Page 859, line 15: after "P.L. 109–280," insert "and P.L. 109–432,
- 2 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 3 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,".
- 4 **1357.** Page 859, line 17: delete "and P.L. 109–280" and substitute "P.L.
- 5 109–280".
- 6 **1358.** Page 859, line 18: after "P.L. 109–280," insert "and P.L. 109–432,
- 7 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 8 303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432,".
- 9 **1359.** Page 860, line 9: after "P.L. 109–280, and" insert "division A, sections
- 10 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304, 305, 307, 404, 424,
- 11 and 425, of ".
- 12 **1360.** Page 861, line 6: delete "and P.L. 109-280" and substitute "P.L.
- 13 109–280".
- 14 **1361.** Page 861, line 7: after "P.L. 109–280" insert ", and P.L. 109–432,
- excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 16 303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432".
- 17 **1362.** Page 864, line 18: delete lines 18 to 21 and substitute "benefit, one of
- the following amounts of payments or distributions received each year by an
- individual, who is at least 65 years of age before the close of the taxable year to which
- the exemption claim relates, from a qualified retirement plan under the Internal
- 21 Revenue Code or from an individual retirement account established under 26 USC
- 22 408:".
- 23 **1363.** Page 864, line 22: delete lines 22 and 23 and substitute:

- 1 "1. For taxable years beginning after December 31, 2008, and before January
- 2 1, 2010, \$500.
- 3 2. For taxable years beginning after December 31, 2009, and before January
- 4 1, 2011, \$1,000.
- 5 3. For taxable years beginning after December 31, 2010, and before January
- 6 1, 2012, \$2,000.
- 4. For taxable years beginning after December 31, 2011, and before January
- 8 1, 2013, \$3,000.
- 5. For taxable years beginning after December 31, 2012, and before January
- 10 1, 2014, \$4,000.
- 11 6. For taxable years beginning after December 31, 2013, and before January
- 12 1, 2015, \$5,000.
- 7. For taxable years beginning after December 31, 2014, and before January
- 14 1, 2016, \$6,000.
- 8. For taxable years beginning after December 31, 2015, and before January
- 16 1, 2017, \$7,000.
- 9. For taxable years beginning after December 31, 2016, and before January
- 18 1, 2018, \$8,000.
- 19 10. For taxable years beginning after December 31, 2017, and before January
- 20 1, 2019, \$9,000.
- 21 11. For taxable years beginning after December 31, 2018, and before January
- 22 1, 2020, \$10,000.
- 23 12. For taxable years beginning after December 31, 2019, and before January
- 24 1, 2021, \$11,000.

- 1 13. For taxable years beginning after December 31, 2020, and before January
- 2 1, 2022, \$12,000.
- 3 14. For taxable years beginning after December 31, 2021, and before January
- 4 1, 2023, \$13,000.
- 5 15. For taxable years beginning after December 31, 2022, and before January
- 6 1, 2024, \$14,000.
- 7 16. For taxable years beginning after December 31, 2023, and before January
- 8 1, 2025, \$15,000.
- 9 17. For taxable years beginning after December 31, 2024, and before January
- 10 1, 2026, \$16,000.
- 11 18. For taxable years beginning after December 31, 2025, and before January
- 12 1, 2027, \$17,000.
- 19. For taxable years beginning after December 31, 2026, and before January
- 14 1, 2028, \$18,000.
- 15 20. For taxable years beginning after December 31, 2027, and before January
- 16 1, 2029, \$19,000.
- 17 21. For taxable years beginning after December 31, 2028, and before January
- 18 1, 2030, \$20,000.
- 19 22. For taxable years beginning after December 31, 2029, the maximum
- subtraction amount from the previous year increased by a percentage equal to the
- 21 percentage increase in Wisconsin per capita personal income, as determined by the
- department of revenue based on the most recent data available from the federal
- Bureau of Economic Analysis, comparing the increase between the year for which the
- 24 most recent data is available and the previous year.".

- **1364.** Page 865, line 1: delete lines 1 to 8.
- **1365.** Page 865, line 18: after that line insert:
- **"Section 1947m.** 71.05 (1) (c) 8. of the statutes is created to read:
  - 71.05 (1) (c) 8. The Wisconsin Health and Educational Facilities Authority under s. 231.03 (6), on or after the effective date of this subdivision .... [revisor inserts date], if the proceeds from the bonds or notes that are issued are used by a health facility, as defined in s. 231.01 (5), to fund the acquisition of information technology hardware or software."
- **1366.** Page 865, line 21: after "(3g)," insert "(3h), (3i),".
- **1367.** Page 867, line 21: delete "An" and substitute "An Subject to subd. 28.

  11 am., an".
- **1368.** Page 868, line 2: after "2006," insert "and subject to subd. 28. am.,".
- **1369.** Page 868, line 2: after that line insert:
- 14 "Section 1953c. 71.05 (6) (b) 28. am. of the statutes is created to read:

71.05 (6) (b) 28. am. Notwithstanding subd. 28. a., for taxable years beginning after December 31, 2006, the department of revenue and the Board of Regents of the University of Wisconsin System shall continue making the calculation described under subd. 28. a. Notwithstanding subd. 28. a., once this calculation exceeds \$6,000, the deduction for tuition expenses and mandatory student fees, as described in subd. 28. (intro.), shall be based on an amount equal to not more than twice the average amount charged by the Board of Regents of the University of Wisconsin System at 4-year institutions for resident undergraduate academic fees for the most recent fall semester, as determined by the Board of Regents by September 1 of that semester, per student for each year to which the claim relates, and the deduction that

may be claimed under this subd. 28. am. first applies to taxable years beginning on the January 1 after the calculation of the Board of Regents, that must occur by September 1, exceeds \$6,000.

**SECTION 1953d.** 71.05 (6) (b) 28. b. of the statutes is amended to read:

71.05 (6) (b) 28. b. From the amount calculated under subd. 28. a. <u>or am.</u>, if the claimant is single or married and filing as head of household and his or her federal adjusted gross income is more than \$50,000 but not more than \$60,000, subtract the product of the amount calculated under subd. 28. a. <u>or am.</u> and the value of a fraction, the denominator of which is \$10,000 and the numerator of which is the difference between the claimant's federal adjusted gross income and \$50,000.

**Section 1953e.** 71.05 (6) (b) 28. c. of the statutes is amended to read:

71.05 (6) (b) 28. c. From the amount calculated under subd. 28. a. or am., if the claimant is married and filing jointly and the claimant's and his or her spouse's federal adjusted gross income is more than \$80,000 but not more than \$100,000, subtract the product of the amount calculated under subd. 28. a. or am. and the value of a fraction, the denominator of which is \$20,000 and the numerator of which is the difference between the claimant's and his or her spouse's federal adjusted gross income and \$80,000.

**SECTION 1953f.** 71.05 (6) (b) 28. d. of the statutes is amended to read:

71.05 (6) (b) 28. d. From the amount calculated under subd. 28. a. or am., if the claimant is married and filing separately and the claimant's federal adjusted gross income is more than \$40,000 but not more than \$50,000, subtract the product of the amount calculated under subd. 28. a. or am. and the value of a fraction, the denominator of which is \$10,000 and the numerator of which is the difference between the claimant's federal adjusted gross income and \$40,000.

 $\mathbf{2}$ 

**SECTION 1953g.** 71.05 (6) (b) 28. e. of the statutes is amended to read:

71.05 (6) (b) 28. e. For an individual who is a nonresident or part-year resident of this state, multiply the amount calculated under subd. 28. a., am., b., c. or d. by a fraction the numerator of which is the individual's wages, salary, tips, unearned income and net earnings from a trade or business that are taxable by this state and the denominator of which is the individual's total wages, salary, tips, unearned income and net earnings from a trade or business. In this subd. 28. e., for married persons filing separately "wages, salary, tips, unearned income and net earnings from a trade or business" means the separate wages, salary, tips, unearned income and net earnings from a trade or business of each spouse, and for married persons filing jointly "wages, salary, tips, unearned income and net earnings from a trade or business" means the total wages, salary, tips, unearned income and net earnings from a trade or business of both spouses.

**Section 1953h.** 71.05 (6) (b) 28. f. of the statutes is amended to read:

71.05 (6) (b) 28. f. Reduce the amount calculated under subd. 28. a., <u>am.</u>, b., c., d. or e. to the individual's aggregate wages, salary, tips, unearned income and net earnings from a trade or business that are taxable by this state.".

- **1370.** Page 868, line 6: delete "a college".
- **1371.** Page 868, line 7: delete "savings account, as described in s. 14.64" and substitute "a college savings account, as described in s. 14.64 any qualified tuition program, as that term is defined in 26 USC 529 (b) (1)".
  - **1372.** Page 868, line 9: after that line insert:
  - **"Section 1954e.** 71.05 (6) (b) 31. of the statutes is amended to read:

reports.

71.05 (6) (b) 31. Any increase in value of -a college savings account, as described
in s. 14.64 any qualified tuition program, as that term is defined in 26 USC 529 (b)
(1), except that the subtraction under this subdivision may not be claimed by any
individual who has made a nonqualified withdrawal, as described in s. 14.64 (2) (e).
<b>Section 1954g.</b> 71.05 (6) (b) 32. (intro.) of the statutes is amended to read:
71.05 (6) (b) 32. (intro.) An amount paid into a college savings account, as
described in s. 14.64 any qualified tuition program, as that term is defined in 26 USC
529 (b) (1), if the beneficiary of the account is one of the following: the claimant; the
claimant's child and the claimant's dependent who is claimed under section 151 (c)
of the Internal Revenue Code; the claimant's grandchild; the claimant's
great-grandchild; or the claimant's niece or nephew; calculated as follows:".
1373. Page 875, line 3: delete the material beginning with that line and
ending with page 876, line 18.
<b>1374.</b> Page 876, line 18: after that line insert:
"Section 1965d. 71.07 (3h) of the statutes is created to read:
71.07 (3h) Workplace wellness program credit. (a) Definitions. In this
subsection:

1. "Claimant" means a person who files a claim under this subsection.

consisting of a questionnaire; a biometric heath screening to measure vital health

statistics, including blood pressure, cholesterol, glucose, weight, and height; a

formula for estimating health risks; an advice database; and a means to generate

2. "Health risk assessment" means a computer-based health-promotion tool

- 3. "Workplace wellness program" means a health or fitness program, as defined by rule under s. 560.207 (4), that is provided with health risk assessments and includes the following programs or services:
  - a. Smoking cessation.
  - b. Weight management.
- c. Stress management.
- d. Worker injury prevention programs.
  - e. Health screenings.
  - f. Nutrition education.
    - g. Health or fitness incentive programs.
  - (b) *Filing claims*. Subject to the limitations provided in this subsection and s. 560.207, for taxable years beginning after December 31, 2008, a claimant may claim as a credit against the taxes imposed under s. 71.02, up to the amount of those taxes, in each taxable year for 3 years, an amount that is equal to 30 percent of the amount that the claimant paid in the taxable year to provide a workplace wellness program to any of the claimant's employees who are employed in this state, not including any amount paid to acquire, construct, rehabilitate, remodel, or repair real property.
  - (c) *Limitations*. 1. The maximum amount of the credits that may be claimed under this subsection and ss. 71.28 (3h) and 71.47 (3h) in any taxable year is \$2,500,000 for all claimants who employ 50 or fewer employees in the taxable year and \$2,500,000 for all claimants who employ more than 50 employees in the taxable year.
  - 2. Partnerships, limited liability companies, and tax-option corporations may not claim the credit under this subsection, but the eligibility for, and the amount of, the credit are based on their payment of amounts under par. (b). A partnership,

2

3

4

5

6

7

8

12

15

16

17

18

19

20

21

22

- limited liability company, or tax-option corporation shall compute the amount of credit that each of its partners, members, or shareholders may claim and shall provide that information to each of them. Partners, members of limited liability companies, and shareholders of tax-option corporations may claim the credit in proportion to their ownership interests.
- (d) *Administration*. Section 71.28 (4) (e) to (h), as it applies to the credit under s. 71.28 (4), applies to the credit under this subsection.".
  - **1375.** Page 876, line 18: after that line insert:
- 9 "Section 1965h. 71.07 (3i) of the statutes is created to read:
- 10 71.07 (3i) BIODIESEL FUEL PRODUCTION CREDIT. (a) Definitions. In this subsection:
  - 1. "Biodiesel fuel" has the meaning given in s. 168.14 (2m) (a).
- 13 2. "Claimant" means a person who is engaged in the business of producing 14 biodiesel fuel in this state and who files a claim under this subsection.
  - (b) Filing claims. Subject to the limitations provided in this subsection, for taxable years beginning after December 31, 2007, and before January 1, 2011, for a claimant who produces at least 2,500,000 gallons of biodiesel fuel in this state in the taxable year, a claimant may claim as a credit against the tax imposed under s. 71.02, up to the amount of the tax, an amount that is equal to the number of gallons of biodiesel fuel produced by the claimant in this state in the taxable year multiplied by 10 cents.
  - (c) *Limitations*. 1. The maximum amount of the credit that a claimant may claim under this subsection in a taxable year is \$1,000,000.

 $\mathbf{2}$ 

- 2. Partnerships, limited liability companies, and tax-option corporations may not claim the credit under this subsection, but the eligibility for, and the amount of, the credit are based on their biodiesel fuel production, as described under par. (b). A partnership, limited liability company, or tax-option corporation shall compute the amount of credit that each of its partners, members, or shareholders may claim and shall provide that information to each of them. Partners, members of limited liability companies, and shareholders of tax-option corporations may claim the credit in proportion to their ownership interests.
- (d) *Administration*. Section 71.28 (4) (e) to (h) as it applies to the credit under s. 71.28 (4), applies to the credit under this subsection.".
- **1376.** Page 877, line 24: after "subsection" insert "and s. 560.208".
- **1377.** Page 878, line 1: after "71.02" insert "or 71.08".
- **1378.** Page 878, line 9: after that line insert:
- "2m. a. The maximum amount of the credits that may be claimed under this subsection and ss. 71.28 (3p) and 71.47 (3p) in fiscal year 2007–08 is \$600,000, as allocated under s. 560.208.
  - b. The maximum amount of the credits that may be claimed under this subsection and ss. 71.28 (3p) and 71.47 (3p) in fiscal year 2008–09, and in each fiscal year thereafter, is \$700,000, as allocated under s. 560.208.".
- **1379.** Page 878, line 23: after "Administration." insert "1.".
- **1380.** Page 878, line 23: delete "(e) to" and substitute "(e), (g), and".
- **1381.** Page 878, line 24: after that line insert:
- 23 "Section 1966d. 71.07 (3w) (title) of the statutes is amended to read:
- 24 71.07 (3w) (title) Enterprise Rural enterprise zone jobs credit.

- **Section 1966e.** 71.07 (3w) (a) 1. of the statutes is amended to read:
- 71.07 (3w) (a) 1. "Base year" means the taxable year beginning during the calendar year prior to the calendar year in which the <u>rural</u> enterprise zone in which
- 4 the claimant is located takes effect.
- **Section 1966f.** 71.07 (3w) (a) 4. of the statutes is amended to read:
- 6 71.07 (**3w**) (a) 4. "Enterprise Rural enterprise zone" means a zone designated under s. 560.799.".
- 8 **1382.** Page 878, line 24: after that line insert:
- 9 "2. If the allowable amount of the claim under par. (b) exceeds the tax otherwise 10 due under s. 71.02 or 71.08 or no tax is due under s. 71.02 or 71.08, the amount of the 11 claim not used to offset the tax due shall be certified by the department of revenue 12 to the department of administration for payment by check, share draft, or other draft 13 drawn from the appropriation account under s. 20.835 (2) (bn)."
- 14 **1383.** Page 879, line 6: delete "-a- an" and substitute "a <u>rural</u>".
- 15 **1384.** Page 879, line 12: on lines 12 and 15, before "enterprise" insert "rural".
- 16 **1385.** Page 880, line 1: on lines 1 and 3, before "enterprise" insert "<u>rural</u>".
- 17 **1386.** Page 880, line 21: delete "-a- an" and substitute "a <u>rural</u>".
- 18 **1387.** Page 882, line 9: delete lines 9 to 25.
- 19 **1388.** Page 883, line 25: delete "2008" and substitute "2009".
- 20 **1389.** Page 884, line 1: delete "or 71.08".
- 21 **1390.** Page 885, line 3: delete "or 71.08".

 $\mathbf{2}$ 

**1391.** Page 886, line 12: delete "2007" and substitute "2008".

**1392.** Page 887, line 10: after that line insert:

**"Section 1990s.** 71.07 (6e) (a) 2. a. of the statutes is amended to read:

71.07 **(6e)** (a) 2. a. An individual who had served on active duty in the U.S. armed forces or in forces incorporated as part of the U.S. armed forces; who was a resident of this state at the time of entry into that active service, or who had been a resident of this state for any consecutive 5-year period after entry into that active duty service; and who, while a resident of this state, died while on active duty.

**Section 1990sc.** 71.07 (6e) (a) 2. b. of the statutes is amended to read:

71.07 **(6e)** (a) 2. b. An individual who had served on active duty under honorable conditions in the U.S. armed forces or in forces incorporated as part of the U.S. armed forces; who was a resident of this state at the time of entry into that active service; who was at least 65 years of age at the time of his or her death or would have been 65 years of age at the close of the year in which the death occurred or who had been a resident of this state for any consecutive 5-year period after entry into that active duty service; who was a resident of this state at the time of his or her death; and who had either a service-connected disability rating of 100 percent under 38 USC 1114 or 1134 or a 100 percent disability rating based on individual unemployability.

**Section 1990se.** 71.07 (6e) (a) 2. c. of the statutes is amended to read:

71.07 **(6e)** (a) 2. c. An individual who had served in the national guard or a reserve component of the U.S. armed forces; who was a resident of this state at the time of entry into that service, or who had been a resident of this state for any consecutive 5-year period after entry into that service; and who, while a resident of

 $\mathbf{2}$ 

this state, died in the line of duty while on active or inactive duty for training purposes.

**Section 1990sg.** 71.07 (6e) (a) 3. (intro.) of the statutes is amended to read:

71.07 **(6e)** (a) 3. (intro.) "Eligible veteran" means an individual who is at least 65 years of age and who is verified by the department of veterans affairs as meeting all of the following conditions:

**Section 1990si.** 71.07 (6e) (a) 3. b. of the statutes is amended to read:

71.07 **(6e)** (a) 3. b. Was a resident of this state at the time of entry into that active service or had been a resident of this state for any consecutive 5-year period after entry into that service.

**SECTION 1990sk.** 71.07 (6e) (a) 3. d. of the statutes is amended to read:

71.07 **(6e)** (a) 3. d. Has <u>either</u> a service-connected disability rating of 100 percent under 38 USC 1114 or 1134 or a 100 percent disability rating based on <u>individual unemployability</u>.

**SECTION 1990sm.** 71.07 (6e) (a) 3e. of the statutes is created to read:

71.07 (**6e**) (a) 3e. "Individual unemployability" means a condition under which a veteran has a service–connected disability rating of either 60 percent under 38 USC 1114 or 1134 or two or more service–connected disability conditions where one condition has at least a 40 percent scheduler rating and the combined scheduler rating for all conditions is at least 70 percent, and has an administrative adjustment added to his or her service–connected disability, due to individual unemployability, such that the federal Department of Veterans Affairs rates the veteran 100 percent disabled.".

**1393.** Page 887, line 15: after "(3n)," insert "(3p),".

- 1 **1394.** Page 887, line 15: delete "(5i), (5j),".
- 2 **1395.** Page 887, line 21: after that line insert:
- 3 **"Section 1991t.** 71.10 (4) (cs) of the statutes is created to read:
- 4 71.10 (4) (cs) Workplace wellness program credit under s. 71.07 (3h).".
- 5 **1396.** Page 887, line 21: after that line insert:
- **SECTION 1991h.** 71.10 (4) (cn) of the statutes is created to read:
- 7 71.10 (4) (cn) Biodiesel fuel production credit under s. 71.07 (3i).".
- 8 **1397.** Page 887, line 22: delete lines 22 to 24.
- 9 **1398.** Page 888, line 5: after that line insert:
- **"Section 1994h.** 71.10 (4) (i) of the statutes is amended to read:
- 11 71.10 (4) (i) The total of claim of right credit under s. 71.07 (1), farmland
- 12 preservation credit under subch. IX, homestead credit under subch. VIII, farmland
- tax relief credit under s. 71.07 (3m), farmers' drought property tax credit under s.
- 14 71.07 (2fd), dairy manufacturing facility investment credit under s. 71.07 (3p), film
- production services credit under s. 71.07 (5f) (b) 2., veterans and surviving spouses
- property tax credit under s. 71.07 (6e), <u>rural</u> enterprise zone jobs credit under s. 71.07
- 17 (3w), earned income tax credit under s. 71.07 (9e), estimated tax payments under s.
- 18 71.09, and taxes withheld under subch. X.".
- 19 **1399.** Page 888, line 20: after "(3g)," insert "(3h), (3i),".
- 20 **1400.** Page 899, line 24: delete "and P.L. 109–280" and substitute "P.L.
- 21 <u>109–280</u>".

- 1 **1401.** Page 899, line 25: after "P.L. 109-280," insert "and P.L. 109-432,
- 2 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 3 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,".
- 4 **1402.** Page 900, line 23: delete "and P.L. 109–280" and substitute "P.L.
- 5 109-280".
- 6 **1403.** Page 900, line 24: after "P.L. 109-280" insert ", and P.L. 109-432,
- 7 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 8 <u>303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432</u>".
- 9 **1404.** Page 901, line 9: delete that line and substitute "109-227, P.L.
- 10 <u>109-280</u>, excluding sections 811 and 844 of P.L. 109-280, and P.L. 109-432,
- 11 <u>excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,</u>
- 12 <u>303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432, and</u>".
- 13 **1405.** Page 901, line 16: delete that line and substitute "109-227, P.L.
- 14 <u>109-280</u>, excluding sections 811 and 844 of P.L. 109-280, and P.L. 109-432,
- excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 16 <u>303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432, apply for</u>".
- 17 **1406.** Page 902, line 8: delete "P.L. 109–227, and" and substitute "P.L.
- 18 109–227,".
- 19 **1407.** Page 902, line 9: after "of P.L. 109–280," insert "and P.L. 109–432,
- 20 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 21 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,".
- 22 **1408.** Page 903, line 8: delete "and P.L. 109-280" and substitute "P.L.
- 23 109–280".

- 1 **1409.** Page 903, line 9: after "109–280" insert ", and P.L. 109–432, excluding
- division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304,
- 3 305, 307, 404, 424, and 425, of P.L. 109-432".
- 4 **1410.** Page 903, line 14: delete "P.L. 109–227, and" and substitute "P.L.
- 5 109–227,".
- 6 **1411.** Page 903, line 15: after "of P.L. 109–280," insert "and P.L. 109–432,
- 7 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 8 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,".
- 9 **1412.** Page 903, line 17: delete "P.L. 109–227, and" and substitute "P.L.
- 10 109-227,".
- 11 **1413.** Page 903, line 18: after "of P.L. 109–280," insert "and P.L. 109–432,
- 12 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 13 303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432,".
- 14 **1414.** Page 904, line 10: after "P.L. 109–280, and" insert "division A, sections
- $15 \qquad \quad 101,\, 104,\, 109,\, 112,\, 113,\, 116,\, 118,\, 120,\, 123\, (a),\, 209,\, 302,\, 303,\, 304,\, 305,\, 307,\, 404,\, 424,$
- and 425, of ".
- 17 **1415.** Page 905, line 9: delete "and P.L. 109-280" and substitute "P.L.
- 18 109–280".
- 19 **1416.** Page 905, line 10: after "109–280" insert ", and P.L. 109–432, excluding
- division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304,
- 21 305, 307, 404, 424, and 425, of P.L. 109–432".
- 22 **1417.** Page 916, line 9: delete that line and substitute "109–227, P.L.
- 23 <u>109-280</u>, excluding sections 811 and 844 of P.L. 109-280, and P.L. 109-432,

- 1 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 2 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432, and as".
- 3 **1418.** Page 917, line 6: delete that line and substitute "109-227, P.L.
- 4 109-280, excluding sections 811 and 844 of P.L. 109-280, and P.L. 109-432,
- 5 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 6 <u>303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432</u>. The".
- 7 **1419.** Page 917, line 16: delete "and P.L. 109–280" and substitute "P.L.
- 8 <u>109–280</u>".
- 9 **1420.** Page 917, line 17: after "P.L. 109-280," insert "and P.L. 109-432,
- 10 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 11 <u>303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,</u>".
- 12 **1421.** Page 917, line 23: delete "and P.L. 109–280" and substitute "P.L.
- 13 <u>109–280</u>".
- 14 **1422.** Page 917, line 24: after "P.L. 109–280," insert "and P.L. 109–432,
- excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 16 <u>303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,</u>".
- 17 **1423.** Page 918, line 17: delete "and P.L. 109–280" and substitute "P.L.
- 18 109–280".
- 19 **1424.** Page 918, line 18: after "109–280," insert "and P.L. 109–432, excluding
- division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304,
- 21 305, 307, 404, 424, and 425, of P.L. 109-432,".
- 22 **1425.** Page 919, line 14: delete "P.L. 109-227, and" and substitute "P.L.
- 23 109–227,".

- 1 **1426.** Page 919, line 15: after "of P.L. 109–280" insert ", and P.L. 109–432,
- 2 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 3 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432".
- 4 **1427.** Page 919, line 21: delete "and P.L. 109–280" and substitute "P.L.
- 5 109–280".
- 6 **1428.** Page 919, line 22: after "P.L. 109–280," insert "and P.L. 109–432,
- 7 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 8 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,".
- 9 **1429.** Page 919, line 24: delete "and P.L. 109–280" and substitute "P.L.
- 10 109–280".
- 11 **1430.** Page 919, line 25: after "P.L. 109–280," insert "and P.L. 109–432,
- 12 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 13 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,".
- 14 **1431.** Page 920, line 16: after "P.L. 109–280, and" insert "division A, sections
- $15 \qquad \quad 101,\, 104,\, 109,\, 112,\, 113,\, 116,\, 118,\, 120,\, 123\, (a),\, 209,\, 302,\, 303,\, 304,\, 305,\, 307,\, 404,\, 424,$
- and 425, of ".
- 17 **1432.** Page 921, line 13: delete ", and P.L. 109–280" and substitute ", P.L.
- 18 109–280".
- 19 **1433.** Page 921, line 14: after "P.L. 109–280" insert ", and P.L. 109–432,
- 20 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 21 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432".
- 22 **1434.** Page 922, line 23: delete the material beginning with that line and
- ending with page 923, line 2.

- 1 **1435.** Page 923, line 2: after that line insert:
- 2 "Section 2020h. 71.26 (1m) (i) of the statutes is created to read:
- 71.26 (**1m**) (i) Those issued under s. 231.03 (6), on or after the effective date of this paragraph .... [revisor inserts date], if the proceeds from the obligations that are issued are used by a health facility, as defined in s. 231.01 (5), to fund the acquisition of information technology hardware or software.".
- 7 **1436.** Page 923, line 11: after "(3g)," insert "(3h), (3i),".
- 8 **1437.** Page 947, line 8: delete that line and substitute "109–227, P.L. 9 109–280, excluding sections 811 and 844 of P.L. 109–280, and P.L. 109–432, excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 11 <u>303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432,</u> and as".
- 12 **1438.** Page 948, line 5: delete that line and substitute "109–227, P.L. 13 109–280, excluding sections 811 and 844 of P.L. 109–280, and P.L. 109–432, excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432, "net".
- 16 **1439.** Page 948, line 22: delete "<u>and P.L. 109–280</u>" and substitute "<u>P.L.</u> 17 109–280".
- 18 **1440.** Page 948, line 23: after "P.L. 109–280," insert "and P.L. 109–432, excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432,".
- 21 **1441.** Page 949, line 20: delete that line and substitute "<u>109–227, P.L.</u> 22 109–280, excluding sections 811 and 844 of P.L. 109–280, and P.L. 109–432,

- 1 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 2 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432, except".
- 3 **1442.** Page 950, line 16: delete that line and substitute "<u>109–227, P.L.</u>
- 4 109-280, excluding sections 811 and 844 of P.L. 109-280, and P.L. 109-432,
- 5 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 6 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432, and as".
- 7 **1443.** Page 951, line 13: delete that line and substitute "109-227, P.L.
- 8 <u>109-280</u>, excluding sections 811 and 844 of P.L. 109-280, and P.L. 109-432,
- 9 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 10 <u>303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432,</u> applies".
- 11 **1444.** Page 951, line 23: delete that line and substitute "109-227, P.L.
- 12 <u>109-280</u>, excluding sections 811 and 844 of P.L. 109-280, and P.L. 109-432,
- 13 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 14 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432, and".
- 15 **1445.** Page 952, line 5: delete that line and substitute "109-227, P.L.
- 16 <u>109-280</u>, excluding sections 811 and 844 of P.L. 109-280, and P.L. 109-432,
- 17 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 18 <u>303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432, apply for</u>".
- 19 **1446.** Page 952, line 24: delete "and P.L. 109–280" and substitute "P.L.
- 20 109-280".
- 21 **1447.** Page 952, line 25: after "P.L. 109–280," insert "and P.L. 109–432,
- excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 23 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,".

- 1 **1448.** Page 953, line 22: delete "and P.L. 109–280" and substitute "P.L. 109–280".
- 1449. Page 953, line 23: after "109–280," insert "and P.L. 109–432, excluding
  division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304,
  305, 307, 404, 424, and 425, of P.L. 109–432,".
- 1450. Page 954, line 14: delete that line and substitute "109–227, P.L. 109–280, excluding sections 811 and 844 of P.L. 109–280, and P.L. 109–432, excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432, and as".
- 10 **1451.** Page 955, line 11: delete "P.L. 109–227, and" and substitute "P.L. 109–227,".
- 12 **1452.** Page 955, line 12: delete that line and substitute "109–280, excluding sections 811 and 844 of P.L. 109–280, and P.L. 109–432, excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432, except that property that,".
- 1453. Page 956, line 7: delete that line and substitute "109–227, P.L. 109–280, excluding sections 811 and 844 of P.L. 109–280, and P.L. 109–432, excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432, and as".
- 20 **1454.** Page 957, line 4: delete "and P.L. 109–280" and substitute "P.L. 109–280".

- 1 **1455.** Page 957, line 5: after "P.L. 109–280," insert "and P.L. 109–432,
- 2 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 3 303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432,".
- 4 **1456.** Page 957, line 10: delete "P.L. 109–227, and" and substitute "P.L.
- 5 109-227,".
- 6 **1457.** Page 957, line 11: after "of P.L. 109–280," insert "and P.L. 109–432,
- 7 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 8 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,".
- 9 **1458.** Page 957, line 13: delete "P.L. 109–227, and" and substitute "P.L.
- 10 109–227,".
- 11 **1459.** Page 957, line 14: after "of P.L. 109–280," insert "and P.L. 109–432,
- 12 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 13 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,".
- 14 **1460.** Page 958, line 7: after "P.L. 109–280, and" insert "division A, sections
- 15 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304, 305, 307, 404, 424,
- and 425, of ".
- 17 **1461.** Page 959, line 4: delete "and P.L. 109-280" and substitute "P.L.
- 18 109–280".
- 19 **1462.** Page 959, line 5: after "P.L. 109–280," insert "and P.L. 109–432,
- 20 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 21 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,".

- 1 **1463.** Page 959, line 21: after "109–280, and" insert "division A, sections 101,
- 2 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304, 305, 307, 404, 424, and
- 3 425, of ".
- 4 **1464.** Page 960, line 18: delete "and P.L. 109–280" and substitute "P.L.
- 5 109–280".
- 6 **1465.** Page 960, line 19: after "P.L. 109–280," insert "and P.L. 109–432,
- 7 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 8 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,".
- 9 **1466.** Page 961, line 13: after "P.L. 109–280, and" insert "division A, sections
- 10 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304, 305, 307, 404, 424,
- 11 and 425, of ".
- 12 **1467.** Page 962, line 10: delete that line and substitute "109–222, P.L.
- 13 109-227, P.L. 109-280, excluding sections 811 and 844 of P.L. 109-280, and P.L.
- 14 109–432, excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a),
- 15 209, 302, 303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432,".
- 16 **1468.** Page 962, line 19: delete the material beginning with that line and
- ending with page 964, line 9.
- 18 **1469.** Page 964, line 9: after that line insert:
- "Section 2038d. 71.28 (3h) of the statutes is created to read:
- 20 71.28 (3h) Workplace wellness program credit. (a) Definitions. In this
- 21 subsection:
- 1. "Claimant" means a person who files a claim under this subsection.

- 2. "Health risk assessment" means a computer-based health-promotion tool consisting of a questionnaire; a biometric heath screening to measure vital health statistics, including blood pressure, cholesterol, glucose, weight, and height; a formula for estimating health risks; an advice database; and a means to generate reports.
- 3. "Workplace wellness program" means a health or fitness program, as defined by rule under s. 560.207 (4), that is provided with health risk assessments and includes the following programs or services:
  - a. Smoking cessation.
  - b. Weight management.
  - c. Stress management.
- d. Worker injury prevention programs.
  - e. Health screenings.
  - f. Nutrition education.
    - g. Health or fitness incentive programs.
  - (b) *Filing claims*. Subject to the limitations provided in this subsection and s. 560.207, for taxable years beginning after December 31, 2008, a claimant may claim as a credit against the taxes imposed under s. 71.23, up to the amount of those taxes, in each taxable year for 3 years, an amount that is equal to 30 percent of the amount that the claimant paid in the taxable year to provide a workplace wellness program to any of the claimant's employees who are employed in this state, not including any amount paid to acquire, construct, rehabilitate, remodel, or repair real property.
  - (c) *Limitations*. 1. The maximum amount of the credits that may be claimed under this subsection and ss. 71.07 (3h) and 71.47 (3h) in any taxable year is \$2,500,000 for all claimants who employ 50 or fewer employees in the taxable year

4

5

6

7

8

9

10

11

12

13

17

18

19

20

21

22

23

- and \$2,500,000 for all claimants who employ more than 50 employees in the taxable year.
  - 2. Partnerships, limited liability companies, and tax-option corporations may not claim the credit under this subsection, but the eligibility for, and the amount of, the credit are based on their payment of amounts under par. (b). A partnership, limited liability company, or tax-option corporation shall compute the amount of credit that each of its partners, members, or shareholders may claim and shall provide that information to each of them. Partners, members of limited liability companies, and shareholders of tax-option corporations may claim the credit in proportion to their ownership interests.
  - (d) *Administration*. Subsection (4) (e) to (h), as it applies to the credit under sub. (4), applies to the credit under this subsection.".
    - **1470.** Page 964, line 9: after that line insert:
- "Section 2038h. 71.28 (3i) of the statutes is created to read:
- 71.28 (3i) BIODIESEL FUEL PRODUCTION CREDIT. (a) Definitions. In this subsection:
  - 1. "Biodiesel fuel" has the meaning given in s. 168.14 (2m) (a).
  - 2. "Claimant" means a person who is engaged in the business of producing biodiesel fuel in this state and who files a claim under this subsection.
  - (b) *Filing claims*. Subject to the limitations provided in this subsection, for taxable years beginning after December 31, 2007, and before January 1, 2011, for a claimant who produces at least 2,500,000 gallons of biodiesel fuel in this state in the taxable year, a claimant may claim as a credit against the tax imposed under s. 71.23, up to the amount of the tax, an amount that is equal to the number of gallons of

- biodiesel fuel produced by the claimant in this state in the taxable year multiplied
  by 10 cents.
  - (c) *Limitations*. 1. The maximum amount of the credit that a claimant may claim under this subsection in a taxable year is \$1,000,000.
  - 2. Partnerships, limited liability companies, and tax-option corporations may not claim the credit under this subsection, but the eligibility for, and the amount of, the credit are based on their biodiesel fuel production, as described under par. (b). A partnership, limited liability company, or tax-option corporation shall compute the amount of credit that each of its partners, members, or shareholders may claim and shall provide that information to each of them. Partners, members of limited liability companies, and shareholders of tax-option corporations may claim the credit in proportion to their ownership interests.
  - (d) *Administration*. Subsection (4) (e) to (h) as it applies to the credit under sub.(4), applies to the credit under this subsection.".
    - **1471.** Page 965, line 15: after "subsection" insert "and s. 560.208".
  - **1472.** Page 965, line 25: after that line insert:
    - "2m. a. The maximum amount of the credits that may be claimed under this subsection and ss. 71.07 (3p) and 71.47 (3p) in fiscal year 2007–08 is \$600,000, as allocated under s. 560.208.
    - b. The maximum amount of the credits that may be claimed under this subsection and ss. 71.07 (3p) and 71.47 (3p) in fiscal year 2008–09, and in each fiscal year thereafter, is \$700,000, as allocated under s. 560.208.".
  - **1473.** Page 966, line 14: after "Administration." insert "1.".
- **1474.** Page 966, line 14: delete "(e) to" and substitute "(e), (g), and".

2

3

4

5

6

7

**1475.** Page 966, line 15: after that line insert:

"2. If the allowable amount of the claim under par. (b) exceeds the tax otherwise due under s. 71.23 or no tax is due under s. 71.23, the amount of the claim not used to offset the tax due shall be certified by the department of revenue to the department of administration for payment by check, share draft, or other draft drawn from the appropriation account under s. 20.835 (2) (bn)."

- **1476.** Page 966, line 16: before that line insert:
- 8 "Section 2039d. 71.28 (3w) (title) of the statutes is amended to read:
- 9 71.28 (3w) (title) Enterprise Rural enterprise zone jobs credit.
- **Section 2039e.** 71.28 (3w) (a) 1. of the statutes is amended to read:
- 71.28 (3w) (a) 1. "Base year" means the taxable year beginning during the calendar year prior to the calendar year in which the <u>rural</u> enterprise zone in which the claimant is located takes effect.
- **SECTION 2039f.** 71.28 (3w) (a) 4. of the statutes is amended to read:
- 15 71.28 **(3w)** (a) 4. "Enterprise Rural enterprise zone" means a zone designated under s. 560.799.".
- 17 **1477.** Page 966, line 22: delete "<u>a an</u>" and substitute "a <u>rural</u>".
- 18 **1478.** Page 967, line 3: on lines 3 and 6, before "enterprise" insert "rural".
- 19 **1479.** Page 967, line 17: on lines 17 and 19, before "enterprise" insert "<u>rural</u>".
- 20 **1480.** Page 968, line 12: delete "<u>a</u> <u>an</u>" and substitute "a <u>rural</u>".
- 21 **1481.** Page 969, line 7: delete lines 7 to 23.
- 22 **1482.** Page 970, line 24: delete "2008" and substitute "2009".
- 23 **1483.** Page 973, line 10: delete "2007" and substitute "2008".

- 1 **1484.** Page 974, line 10: after that line insert:
- 2 "Section 2060s. 71.30 (3) (cn) of the statutes is created to read:
- 3 71.30 (3) (cn) Biodiesel fuel production credit under s. 71.28 (3i).".
- 4 **1485.** Page 974, line 11: delete lines 11 to 13.
- 5 **1486.** Page 974, line 13: after that line insert:
- 6 "Section 2061m. 71.30 (3) (ds) of the statutes is created to read:
- 7 71.30 (3) (ds) Workplace wellness program credit under s. 71.28 (3h).".
- 8 **1487.** Page 974, line 24: before "enterprise" insert "<u>rural</u>".
- 9 **1488.** Page 974, line 24: after "(2m)," insert "dairy manufacturing facility investment credit under s. 71.28 (3p),".
- 11 **1489.** Page 975, line 3: after "(3g)," insert "(3h), (3i),".
- 12 **1490.** Page 986, line 12: delete that line and substitute "109-227, P.L.
- 13 <u>109-280</u>, excluding sections 811 and 844 of P.L. 109-280, and P.L. 109-432,
- 14 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 15 <u>303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432,</u> and as".
- 16 **1491.** Page 987, line 11: delete that line and substitute "109-227, P.L.
- 17 <u>109-280, excluding sections 811 and 844 of P.L. 109-280, and P.L. 109-432,</u>
- 18 <u>excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,</u>
- 19 <u>303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,</u> except".
- 20 **1492.** Page 987, line 23: delete "and P.L. 109–280" and substitute "P.L.
- 21 109-280".

- 1 **1493.** Page 987, line 24: after "P.L. 109-280," insert "and P.L. 109-432,
- 2 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 3 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,".
- 4 **1494.** Page 988, line 5: delete "and P.L. 109–280" and substitute "P.L.
- 5 <u>109–280</u>".
- 6 **1495.** Page 988, line 6: after "P.L. 109–280," insert "and P.L. 109–432,
- 7 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 8 <u>303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,</u>".
- 9 **1496.** Page 988, line 23: delete "and P.L. 109–280" and substitute "P.L.
- 10 109–280".
- 11 **1497.** Page 988, line 24: after "109–280," insert "and P.L. 109–432, excluding
- division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304,
- 13 305, 307, 404, 424, and 425, of P.L. 109-432,".
- 14 **1498.** Page 989, line 22: delete "and P.L. 109–280" and substitute "P.L.
- 15 109–280".
- 16 **1499.** Page 989, line 23: after "109–280," insert "and P.L. 109–432, excluding
- division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304,
- 18 305, 307, 404, 424, and 425, of P.L. 109–432,".
- 19 **1500.** Page 990, line 6: delete that line and substitute "109-227, P.L.
- 20 109-280, excluding sections 811 and 844 of P.L. 109-280, and P.L. 109-432,
- excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432, and".

- 1 **1501.** Page 990, line 9: delete that line and substitute "109-227, P.L.
- 2 109-280, excluding sections 811 and 844 of P.L. 109-280, and P.L. 109-432,
- 3 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 4 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432, apply for".
- 5 **1502.** Page 990, line 25: after "P.L. 109–280, and" insert "division A, sections
- $6 \hspace{1.5cm} 101, \hspace{0.1cm} 104, \hspace{0.1cm} 109, \hspace{0.1cm} 112, \hspace{0.1cm} 113, \hspace{0.1cm} 116, \hspace{0.1cm} 118, \hspace{0.1cm} 120, \hspace{0.1cm} 123 \hspace{0.1cm} (a), \hspace{0.1cm} 209, \hspace{0.1cm} 302, \hspace{0.1cm} 303, \hspace{0.1cm} 304, \hspace{0.1cm} 305, \hspace{0.1cm} 307, \hspace{0.1cm} 404, \hspace{0.1cm} 424, \hspace{0.1cm} 424,$
- 7 and 425, of ".
- 8 **1503.** Page 991, line 24: delete "and P.L. 109–280" and substitute "P.L.
- 9 109–280".
- 10 **1504.** Page 991, line 25: after "P.L. 109–280," insert "and P.L. 109–432,
- 11 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 12 303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432,".
- 13 **1505.** Page 1003, line 8: delete "and P.L. 109-280" and substitute "P.L.
- 14 109–280".
- 15 **1506.** Page 1003, line 9: after "P.L. 109–280," insert "and P.L. 109–432,
- 16 <u>excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,</u>
- 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,".
- 18 **1507.** Page 1004, line 5: delete "and P.L. 109-280" and substitute "P.L.
- 19 109–280".
- 20 **1508.** Page 1004, line 6: after "P.L. 109–280," insert "and P.L. 109–432,
- 21 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 22 <u>303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432,</u>".

- 1 **1509.** Page 1004, line 17: delete that line and substitute "109-227, P.L.
- 2 109-280, excluding sections 811 and 844 of P.L. 109-280, and P.L. 109-432,
- 3 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 4 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432, and".
- 5 **1510.** Page 1004, line 24: delete that line and substitute "<u>109–227, P.L.</u>
- 6 109-280, excluding sections 811 and 844 of P.L. 109-280, and P.L. 109-432,
- 7 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 8 <u>303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432, apply for</u>".
- 9 **1511.** Page 1005, line 15: delete "P.L. 109–227, and" and substitute "P.L.
- 10 109-227,".
- 11 **1512.** Page 1005, line 16: delete that line and substitute "P.L. 109–280,
- excluding sections 811 and 844 of P.L. 109–280, and P.L. 109–432, excluding division
- 13 A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304, 305, 307,
- 14 404, 424, and 425, of P.L. 109-432, and as indirectly".
- 15 **1513.** Page 1006, line 13: delete "and P.L. 109–280" and substitute "P.L.
- 16 109–280".
- 17 **1514.** Page 1006, line 14: after "109-280," insert "and P.L. 109-432,
- 18 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 19 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432,".
- 20 **1515.** Page 1006, line 21: delete that line and substitute "109–227, P.L.
- 21 109-280, excluding sections 811 and 844 of P.L. 109-280, and P.L. 109-432,
- 22 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 23 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432, and".

- 1 **1516.** Page 1006, line 24: delete that line and substitute "109–227, P.L. 109–280, excluding sections 811 and 844 of P.L. 109–280, and P.L. 109–432,
- 3 excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302,
- 4 303, 304, 305, 307, 404, 424, and 425, of P.L. 109-432, apply for".
- **1517.** Page 1007, line 15: after "P.L. 109–280, and" insert "division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304, 305, 307, 404, 424, and 425, of".
- 8 **1518.** Page 1008, line 12: delete "and P.L. 109–280" and substitute "P.L. 9 109–280".
- 1519. Page 1008, line 13: after "109–280," insert "and P.L. 109–432, excluding division A, sections 101, 104, 109, 112, 113, 116, 118, 120, 123 (a), 209, 302, 303, 304, 305, 307, 404, 424, and 425, of P.L. 109–432,".
- 13 **1520.** Page 1009, line 9: after that line insert:
- 14 "Section 2087h. 71.45 (1t) (i) of the statutes is created to read:
- 15 71.45 (**1t**) (i) Those issued under s. 231.03 (6), on or after the effective date of this paragraph .... [revisor inserts date], if the proceeds from the obligations that are issued are used by a health facility, as defined in s. 231.01 (5), to fund the acquisition of information technology hardware or software.".
- 19 **1521.** Page 1009, line 12: after "(1dx)," insert "(3h), (3i),".
- 20 **1522.** Page 1009, line 17: delete the material beginning with that line and ending with page 1011, line 7.
- 22 **1523.** Page 1011, line 7: after that line insert:
- 23 "Section 2094m. 71.47 (3h) of the statutes is created to read:

4

5

6

7

8

9

10

11

12

13

16

19

20

21

22

23

24

1	71.47 <b>(3h)</b>	WORKPLACE WELLNESS PROGRAM CREDIT.	(a)	Definitions.	In this
2	subsection:				

- 1. "Claimant" means a person who files a claim under this subsection.
- 2. "Health risk assessment" means a computer-based health-promotion tool consisting of a questionnaire; a biometric heath screening to measure vital health statistics, including blood pressure, cholesterol, glucose, weight, and height; a formula for estimating health risks; an advice database; and a means to generate reports.
- 3. "Workplace wellness program" means a health or fitness program, as defined by rule under s. 560.207 (4), and includes health risk assessments and includes the following programs or services:
  - a. Smoking cessation.
    - b. Weight management.
- c. Stress management.
- d. Worker injury prevention programs.
  - e. Health screenings.
- 17 f. Nutrition education.
- g. Health or fitness incentive programs.
  - (b) *Filing claims*. Subject to the limitations provided in this subsection and s. 560.207, for taxable years beginning after December 31, 2008, a claimant may claim as a credit against the taxes imposed under s. 71.43, up to the amount of those taxes, in each taxable year for 3 years, an amount that is equal to 30 percent of the amount that the claimant paid in the taxable year to provide a workplace wellness program to any of the claimant's employees who are employed in this state, not including any amount paid to acquire, construct, rehabilitate, remodel, or repair real property.

- (c) *Limitations*. 1. The maximum amount of the credits that may be claimed under this subsection and ss. 71.07 (3h) and 71.28 (3h) in any taxable year is \$2,500,000 for all claimants who employ 50 or fewer employees in the taxable year and \$2,500,000 for all claimants who employ more than 50 employees in the taxable year.
- 2. Partnerships, limited liability companies, and tax-option corporations may not claim the credit under this subsection, but the eligibility for, and the amount of, the credit are based on their payment of amounts under par. (b). A partnership, limited liability company, or tax-option corporation shall compute the amount of credit that each of its partners, members, or shareholders may claim and shall provide that information to each of them. Partners, members of limited liability companies, and shareholders of tax-option corporations may claim the credit in proportion to their ownership interests.
- (d) *Administration*. Section 71.28 (4) (e) to (h), as it applies to the credit under s. 71.28 (4), applies to the credit under this subsection.".
  - **1524.** Page 1011, line 7: after that line insert:
- "Section 2094h. 71.47 (3i) of the statutes is created to read:
- 71.47 (3i) BIODIESEL FUEL PRODUCTION CREDIT. (a) Definitions. In this subsection:
  - 1. "Biodiesel fuel" has the meaning given in s. 168.14 (2m) (a).
  - 2. "Claimant" means a person who is engaged in the business of producing biodiesel fuel in this state and who files a claim under this subsection.
  - (b) *Filing claims*. Subject to the limitations provided in this subsection, for taxable years beginning after December 31, 2007, and before January 1, 2011, for a

 $\mathbf{2}$ 

- claimant who produces at least 2,500,000 gallons of biodiesel fuel in this state in the taxable year, a claimant may claim as a credit against the tax imposed under s. 71.43, up to the amount of the tax, an amount that is equal to the number of gallons of biodiesel fuel produced by the claimant in this state in the taxable year multiplied by 10 cents.
- (c) *Limitations*. 1. The maximum amount of the credit that a claimant may claim under this subsection in a taxable year is \$1,000,000.
- 2. Partnerships, limited liability companies, and tax-option corporations may not claim the credit under this subsection, but the eligibility for, and the amount of, the credit are based on their biodiesel fuel production, as described under par. (b). A partnership, limited liability company, or tax-option corporation shall compute the amount of credit that each of its partners, members, or shareholders may claim and shall provide that information to each of them. Partners, members of limited liability companies, and shareholders of tax-option corporations may claim the credit in proportion to their ownership interests.
- (d) *Administration*. Section 71.28 (4) (e) to (h) as it applies to the credit under s. 71.28 (4), applies to the credit under this subsection.".
- **1525.** Page 1012, line 12: after "subsection" insert "and s. 560.208".
- **1526.** Page 1012, line 22: after that line insert:
  - "2m. a. The maximum amount of the credits that may be claimed under this subsection and ss. 71.07 (3p) and 71.28 (3p) in fiscal year 2007–08 is \$600,000, as allocated under s. 560.208.

3

7

8

9

10

11

- b. The maximum amount of the credits that may be claimed under this  $\mathbf{2}$ subsection and ss. 71.07 (3p) and 71.28 (3p) in fiscal year 2008–09, and in each fiscal year thereafter, is \$700,000, as allocated under s. 560.208.".
- **1527.** Page 1013, line 11: after "Administration." insert "1.". 4
- **1528.** Page 1013, line 11: delete "(e) to" and substitute "(e), (g), and". 5
- **1529.** Page 1013, line 12: after that line insert: 6
  - "2. If the allowable amount of the claim under par. (b) exceeds the tax otherwise due under s. 71.43 or no tax is due under s. 71.43, the amount of the claim not used to offset the tax due shall be certified by the department of revenue to the department of administration for payment by check, share draft, or other draft drawn from the appropriation account under s. 20.835 (2) (bn).".
- 12 **1530.** Page 1013, line 13: before that line insert:
- 13 "Section 2095d. 71.47 (3w) (title) of the statutes is amended to read:
- 14 71.47 (3w) (title) Enterprise Rural enterprise zone jobs credit.
- 15 **Section 2095e.** 71.47 (3w) (a) 1. of the statutes is amended to read:
- 16 71.47 (3w) (a) 1. "Base year" means the taxable year beginning during the 17 calendar year prior to the calendar year in which the <u>rural</u> enterprise zone in which 18 the claimant is located takes effect.
  - **Section 2095f.** 71.47 (3w) (a) 4. of the statutes is amended to read:
- 71.47 (3w) (a) 4. "Enterprise Rural enterprise zone" means a zone designated 20 21under s. 560.799.".
- **1531.** Page 1013, line 19: delete "-a- an" and substitute "a rural". 22
- **1532.** Page 1013, line 25: before "enterprise" insert "rural". 23
- **1533.** Page 1014, line 3: before "enterprise" insert "rural". 24

- 1 **1534.** Page 1014, line 14: on lines 14 and 16, before "enterprise" insert
- 2 "rural".
- 3 **1535.** Page 1015, line 10: delete "<u>a an</u>" and substitute "a <u>rural</u>".
- 4 **1536.** Page 1016, line 4: delete lines 4 to 20.
- 5 **1537.** Page 1017, line 20: delete "2008" and substitute "2009".
- 6 **1538.** Page 1020, line 7: delete "2007" and substitute "2008".
- 7 **1539.** Page 1021, line 7: after that line insert:
- 8 "Section 2116s. 71.49 (1) (cn) of the statutes is created to read:
- 9 71.49 (1) (cn) Biodiesel fuel production credit under s. 71.47 (3i).".
- 10 **1540.** Page 1021, line 8: delete lines 8 to 10.
- 11 **1541.** Page 1021, line 12: after that line insert:
- 12 **"Section 2118m.** 71.49 (1) (dt) of the statutes is created to read:
- 13 71.49 (1) (dt) Workplace wellness program credit under s. 71.47 (3h).".
- 14 **1542.** Page 1021, line 21: before "enterprise" insert "<u>rural</u>".
- 15 **1543.** Page 1021, line 21: after "(2m)," insert "dairy manufacturing facility
- investment credit under s. 71.47 (3p),".
- 17 **1544.** Page 1021, line 22: delete "71.28" and substitute "71.47".
- 18 **1545.** Page 1021, line 22: after that line insert:
- 19 "**Section 2121m.** 71.53 (2) (c) of the statutes is amended to read:
- 20 71.53 (2) (c) The claimant was under 18 65 years of age at the close of the year
- 21 to which the claim relates, except that this limitation does not apply if the claimant
- was at least 18 years of age at the close of the year to which the claim relates and is

- either married or has a dependent whom he or she claims under section 151 (c) of the
   Internal Revenue Code during the year to which the claim relates.
- **Section 2121n.** 71.53 (2) (d) of the statutes is amended to read:
  - 71.53 (2) (d) The claimant was claimed as a dependent for federal income tax purposes by another person during the year to which the claim relates but this limitation shall not apply if the claimant was 62 65 years of age or older at the close of the year to which the claim relates.".
- **1546.** Page 1022, line 1: delete that line and substitute:
- 9 "71.54 (1) (f) 2001 and thereafter to 2007. (intro.) The amount".
- **1547.** Page 1022, line 2: delete "and thereafter" and substitute: "and thereafter to 2007".
- **1548.** Page 1022, line 3: after that line insert:
- **"Section 2121se.** 71.54 (1) (g) of the statutes is created to read:
  - 71.54 (1) (g) 2008 and thereafter. The amount of any claim filed in 2008 and thereafter and based on property taxes accrued or rent constituting property taxes accrued during the previous year is limited as follows:
  - 1. If the household income was \$8,000 or less in the year to which the claim relates, the claim is limited to 80 percent of the property taxes accrued or rent constituting property taxes accrued or both in that year on the claimant's homestead.
  - 2. If the household income was more than \$8,000 in the year to which the claim relates, the claim is limited to 80 percent of the amount by which the property taxes accrued or rent constituting property taxes accrued or both in that year on the claimant's homestead exceeds 8.146 percent of the household income exceeding \$8,000.

- 3. No credit may be allowed if the household income of a claimant exceeds \$25,800.".
- **1549.** Page 1022, line 4: delete lines 4 to 6.
- **1550.** Page 1022, line 8: delete "2007" and substitute "2009".
- 1551. Page 1022, line 9: delete lines 9 to 11 and substitute "beginning after December 31, 2008, the dollar amount for the maximum household income under sub. (1) (g) 3. shall be increased each year by a".
- **1552.** Page 1022, line 13: delete "the previous" and substitute "that".
- **1553.** Page 1022, line 15: delete "2005" and substitute "2008".
- **1554.** Page 1022, line 23: delete lines 23 and 24 and substitute "(g) 2. such that as a claimant's income increases from the threshold income under sub. (1) (g) 2., to an amount that exceeds the maximum household income".
- **1555.** Page 1025, line 21: after that line insert:

- "Section 2135e. 71.78 (2) of the statutes is amended to read:
  - 71.78 (2) DISCLOSURE OF NET TAX. The department shall make available upon suitable forms prepared by the department information setting forth the net Wisconsin income tax, Wisconsin franchise tax, or Wisconsin gift tax reported as paid or payable in the returns filed by any individual or corporation, and any amount of delinquent taxes owed, as described in s. 73.03 (62), by any such individual or corporation, for any individual year upon request. When making available information setting forth the delinquent taxes owed by an individual or corporation, the information shall include interest, penalties, fees, and costs, which are unpaid for more than 90 days after all appeal rights have expired, except that such information may not be provided for any person who has reached an agreement or

compromise with the department, or the department of justice, under s. 71.92 and is in compliance with that agreement, regarding the payment of delinquent taxes, or the name of any person who is protected by a stay that is in effect under the Federal Bankruptcy Code. Before the request is granted, the person desiring to obtain the information shall prove his or her identity and shall be required to sign a statement setting forth the person's address and reason for making the request and indicating that the person understands the provisions of this section with respect to the divulgement, publication or dissemination of information obtained from returns as provided in sub. (1). The use of a fictitious name is a violation of this section. Within 24 hours after any information from any such tax return has been so obtained, the department shall mail to the person from whose return the information has been obtained a notification which shall give the name and address of the person obtaining the information and the reason assigned for requesting the information. The department shall collect from the person requesting the information a fee of \$4 for each return."

**1556.** Page 1038, line 16: after that line insert:

"Section 2139h. 71.83 (1) (ce) of the statutes is created to read:

71.83 (1) (ce) *Health savings accounts*. For taxable years beginning after December 31, 2007, any person who is liable for a penalty for federal income tax purposes under section 223 (f) (4) of the Internal Revenue Code is liable for a penalty equal to 33 percent of that penalty. The department of revenue shall assess, levy, and collect the penalty under this paragraph as it assesses, levies, and collects taxes under this chapter."

**1557.** Page 1039, line 15: delete lines 15 to 24.

1	<b>1558.</b> Page 1039, line 24: after that line insert:		
2	"Section 2142e. Subchapter XVI of chapter 71 [precedes 71.98] of the statutes		
3	is created to read:		
4	CHAPTER 71		
5	SUBCHAPTER XVI		
6	INTERNAL REVENUE CODE UPDATE		
7	71.98 Internal Revenue Code update. The following federal laws, to the		
8	extent that they apply to the Internal Revenue Code, apply to this chapter:		
9	(1) Health savings accounts. Section 233 of the Internal Revenue Code,		
10	relating to health savings accounts, subject to the following:		
11	(a) For taxable years beginning after December 31, 2007, and ending before		
12	January 1, 2009, an individual may claim under this chapter 50 percent of the tax		
13	benefits of an account that are allowable under federal law.		
14	(b) For taxable years beginning after December 31, 2008, an individual may		
15	claim under this chapter the tax benefits of an account to the same extent as is		
16	allowable under federal law.".		
17	<b>1559.</b> Page 1040, line 11: delete the material beginning with that line and		
18	ending with on page 1041, line 23.		
19	<b>1560.</b> Page 1041, line 24: delete the material beginning with that line and		
20	ending with page 1042, line 8.		
21	<b>1561.</b> Page 1042, line 9: delete lines 9 to 17.		
22	<b>1562.</b> Page 1042, line 18: delete the material beginning with that line and		
23	ending with page 1043, line 7.		
24	<b>1563.</b> Page 1043, line 8: delete lines 8 to 16.		

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

**1564.** Page 1044, line 1: delete the material beginning with that line and ending with page 1045, line 16.

**1565.** Page 1045, line 16: after that line insert:

**"Section 2153p.** 73.03 (62) of the statutes is amended to read:

73.03 (62) To prepare and maintain a list of all persons who owe delinquent taxes, including interest, penalties, fees, and costs, to the department, in excess of \$25,000 \$5,000, which are unpaid for more than 90 days after all appeal rights have expired, and; to post the names of persons from this list on the Internet at a site that is created and maintained by the department for this purpose; and to distribute the posted information to Internet search engines so the information is searchable. The Internet site shall list the name, address, type of tax due, and amount of tax due, including interest, penalties, fees, and costs for each person who has one of the delinquent taxpayer accounts, and the Internet site shall also contain a special page for the persons who have the 100 largest delinquent taxpayer accounts. Except as otherwise provided in this subsection, the department shall update the Internet site on a quarterly basis, and shall send the updates to the Internet search engines. The department may not post on the Internet or distribute to Internet search engines the name of any person who has reached an agreement or compromise with the department, or the department of justice, under s. 71.92 and is in compliance with that agreement, regarding the payment of delinquent taxes, or the name of any person who is protected by a stay that is in effect under the Federal Bankruptcy Code; the Internet posting and Internet search engines shall be updated each business day, as defined in s. 562.01 (3m), to comply with these prohibitions.".

**1566.** Page 1046, line 1: delete lines 1 to 25.

**1567.** Page 1047, line 1: delete lines 1 to 4.

**1568.** Page 1047, line 5: delete lines 5 to 12.

**1569.** Page 1047, line 12: after that line insert:

"Section 2160d. 74.35 (2m) of the statutes is amended to read:

74.35 (**2m**) EXCLUSIVE PROCEDURE. A claim that property is exempt, other than a claim that property is exempt under s. 70.11 (21) (a) or (27), may be made only in an action under this section. Such a claim may not be made by means of an action under s. 74.33 or an action for a declaratory judgment under s. 806.04.

**SECTION 2160f.** 74.35 (5) (d) of the statutes is amended to read:

74.35 (5) (d) No claim may be made under this section based on the contention that the tax was unlawful because the property is exempt from taxation under s. 70.11 (21) (a) or (27).

**SECTION 2160h.** 76.025 (1) of the statutes is amended to read:

76.025 (1) The property taxable under s. 76.13 shall include all franchises, and all real and personal property of the company used or employed in the operation of its business, excluding property that is exempt from the property tax under s. 70.11 (39) and (39m), such motor vehicles as are exempt under s. 70.112 (5) and treatment plant and pollution abatement equipment exempt under s. 70.11 (21) (a). The taxable property shall include all title and interest of the company referred to in such property as owner, lessee or otherwise, and in case any portion of the property is jointly used by 2 or more companies, the unit assessment shall include and cover a proportionate share of that portion of the property jointly used so that the assessments of the property of all companies having any rights, title or interest of

 $\mathbf{2}$ 

any kind or nature whatsoever in any such property jointly used shall, in the aggregate, include only one total full value of such property.".

**1570.** Page 1047, line 12: after that line insert:

"Section 2160m. 74.485 (6) of the statutes is amended to read:

74.485 (6) DISTRIBUTION. A county that collects a penalty under this section shall distribute 50% of the amount of the penalty to the taxation district in which the land related to the penalty is located. If the land related to the penalty is located in 2 or more taxation districts, the county shall distribute 50% of the amount of the penalty to the taxation districts in proportion to the equalized value of the land related to the penalty that is located in each taxation district. A taxation district shall distribute 50% of any amount it receives under this subsection to an adjoining taxation district, if the taxation district in which the land related to the penalty is located annexed the land related to the penalty from the adjoining taxation district in either of the 2 years preceding a distribution under this subsection. The county shall distribute the amounts provided under this subsection no later than 30 days after receiving a payment under sub. (5).".

**1571.** Page 1047, line 12: after that line insert:

**"Section 2160d.** 74.25 (1) (a) 6. of the statutes is amended to read:

74.25 (1) (a) 6. Pay to the county treasurer 20% of collections of occupational taxes on coal docks, 20% of collections of the taxes imposed under ss. 77.04 and 77.84 (2) (a) and (am), and all 20% of collections of payments for closed lands under s. 77.84 (2) (b) and (bm).

**Section 2160f.** 74.25 (1) (a) 8. of the statutes is amended to read:

1	74.25 (1) (a) 8. Retain for the taxation district all woodland tax law collection		
2	under s. $77.16$ and, $80\%$ of collections of the taxes imposed under ss. $77.04$ and $77.8$		
3	(2) (a) and (am), and 80% of collections of payments for closed lands under s. 77.8		
4	(2) (b) and (bm).		
5	<b>Section 2160r.</b> 74.30 (1) (f) of the statutes is amended to read:		
6	74.30 (1) (f) Pay to the county treasurer 20% of collections of occupational tax		
7	on coal docks, $20\%$ of collections of the taxes imposed under ss. $77.04$ and $77.84$ (		
8	(a) and (am), and all $\underline{20\%}$ of collections of payments for closed lands under s. 77.8		
9	(2) (b) and (bm).		
10	<b>Section 2160t.</b> 74.30 (1) (h) of the statutes is amended to read:		
11	74.30 (1) (h) Retain for the taxation district all woodland tax law collection		
12	under s. $77.16$ and, $80\%$ of collections of the taxes imposed under ss. $77.04$ and $77.8$		
13	(2) (a) and (am), and 80% of collections of payments for closed lands under s. 77.8		
14	(2) (b) and (bm).".		
15	<b>1572.</b> Page 1047, line 13: delete lines 13 to 20.		
16	1573. Page 1049, line 22: delete the material beginning with that line and		
17	ending with page 1051, line 12.		
18	<b>1574.</b> Page 1051, line 12: after that line insert:		
19	"Section 2166d. 76.81 of the statutes is amended to read:		
20	76.81 Imposition. There is imposed a tax on the real property of, and the		
21	tangible personal property of, every telephone company, excluding property that is		
22	exempt from the property tax under s. 70.11 (39) and (39m), motor vehicles that are		
23	exempt under s. 70.112 (5), property that is used less than 50% in the operation of		

a telephone company, as provided under s.  $70.112\ (4)\ (b)$ , and treatment plant and

pollution abatement equipment that is exempt under s. 70.11 (21) (a). Except as provided in s. 76.815, the rate for the tax imposed on each description of real property and on each item of tangible personal property is the net rate for the prior year for the tax under ch. 70 in the taxing jurisdictions where the description or item is located. The real and tangible personal property of a telephone company shall be assessed as provided under s. 70.112 (4) (b).".

**1575.** Page 1051, line 13: delete the material beginning with that line and ending with page 1052, line 3.

**1576.** Page 1052, line 4: delete the material beginning with that line and ending with page 1053, line 6, and substitute:

**"Section 2168d.** 77.22 (1) of the statutes is amended to read:

77.22 (1) There Except as provided in sub. (1m), there is imposed on the grantor of real estate a real estate transfer fee at the rate of 30 cents for each \$100 of value or fraction thereof on every conveyance not exempted or excluded under this subchapter. In regard to land contracts the value is the total principal amount that the buyer agrees to pay the seller for the real estate. This fee shall be collected by the register at the time the instrument of conveyance is submitted for recording. Except as provided in s. 77.255, at the time of submission the grantee or his or her duly authorized agent or other person acquiring an ownership interest under the instrument, or the clerk of court in the case of a foreclosure under s. 846.16 (1), shall execute a return, signed by both grantor and grantee, on the form prescribed under sub. (2). The register shall enter the fee paid on the face of the deed or other instrument of conveyance before recording, and, except as provided in s. 77.255, submission of a completed real estate transfer return and collection by the register

of the fee shall be prerequisites to acceptance of the conveyance for recording. The register shall have no duty to determine either the correct value of the real estate transferred or the validity of any exemption or exclusion claimed. If the transfer is not subject to a fee as provided in this subchapter, the reason for exemption shall be stated on the face of the conveyance to be recorded by reference to the proper subsection under s. 77.25.

**Section 2168e.** 77.22 (1m) of the statutes is created to read:

77.22 (1m) (a) For conveyances that are recorded during the period beginning on July 1, 2010, and ending June 30, 2011, the rate of the fee imposed under sub. (1) is 20 cents for each \$100 of value or fraction thereof on every conveyance not exempted or excluded under this subchapter.

(b) For conveyances that are recorded after June 30, 2011, the rate of the fee imposed under sub. (1) is 10 cents for each \$100 of value or fraction thereof on every conveyance not exempted or excluded under this subchapter.

**SECTION 2169d.** 77.24 of the statutes is renumbered 77.24 (1) and amended to read:

77.24 (1) Twenty Except as provided in subs. (2) and (3), 20 percent of all fees collected under this subchapter shall be retained by the county and the balance shall be transmitted to the state. Remittances shall be made monthly by the county treasurers to the department of revenue by the 15th day of the month following the close of the month in which the fee was collected. The remittance to the department shall be accompanied by the returns executed under s. 77.22.

**Section 2169e.** 77.24 (2) of the statutes is created to read:

77.24 (2) For conveyances that are recorded during the period beginning on July 1, 2010, and ending June 30, 2011, 40 percent of all fees collected under this

subchapter shall be retained by the county and the balance shall be transmitted to the state. Remittances shall be made monthly by the county treasurers to the department of revenue by the 15th day of the month following the close of the month in which the fee was collected. The remittance to the department shall be accompanied by the returns executed under s. 77.22.

**SECTION 2169f.** 77.24 (3) of the statutes is created to read:

- 7 77.24 (3) For conveyances that are recorded after June 30, 2011, all fees collected under this subchapter shall be retained by the county.".
  - **1577.** Page 1053, line 7: delete the material beginning with that line and ending with page 1054, line 4.
  - **1578.** Page 1054, line 5: delete lines 5 to 16.
- **1579.** Page 1054, line 17: delete the material beginning with that line and ending with page 1056, line 10.
- **1580.** Page 1056, line 15: delete the material beginning with that line and ending with page 1097, line 10.
  - **1581.** Page 1097, line 11: delete the material beginning with that line and ending with page 1098, line 3.
    - **1582.** Page 1098, line 4: delete the material beginning with that line and ending with page 1100, line 13.
    - **1583.** Page 1100, line 15: delete the material beginning with that line and ending with page 1101, line 8, and substitute:
    - "77.52 (2) (a) 11. The producing, fabricating, processing, printing or imprinting of tangible personal property for a consideration for consumers who furnish directly or indirectly the materials used in the producing, fabricating, processing, printing

or imprinting. This subdivision does not apply to the printing or imprinting of
tangible personal property which will be subsequently transported outside the state
for use outside the state by the consumer for advertising purposes that results in
printed material, catalogs, or envelopes that are exempt under s. 77.54 (25) or
(25m).".
1584. Page 1101, line 9: delete the material beginning with that line and
ending with page 1133, line 14.
<b>1585.</b> Page 1133, line 14: after that line insert:
"Section 2300d 77 52 (2r) of the statutes is created to read:

77.52 (2r) No part of the charge for services provided by a temporary help company, as defined in s. 108.02 (24m), is subject to tax under sub. (2), if the client for whom the services are provided controls the means of performing the services and is responsible for the satisfactory completion of the services."

**1586.** Page 1133, line 14: after that line insert:

"Section 2356d. 77.54 (9a) (i) of the statutes is created to read:

77.54 (9a) (i) A cemetery company or corporation described under section 501 (c) (13) of the Internal Revenue Code, if the tangible personal property or taxable services are used exclusively by the cemetery company or corporation for the purposes of the company or corporation.".

**1587.** Page 1133, line 21: delete the material beginning with that line and ending with page 1134, line 16, and substitute:

"Section 2383d. 77.54 (25) of the statutes is amended to read:

77.54 **(25)** The gross receipts from the sale of and the storage of printed material which is designed to advertise and promote the sale of merchandise, or to

advertise the services of individual business firms, which printed material is purchased and stored for the purpose of subsequently transporting it outside the state by the purchaser for use thereafter solely outside the state. This subsection does not apply to catalogs and the envelopes in which the catalogs are mailed.".

**1588.** Page 1134, line 18: delete "sales price" and substitute "gross receipts".

**1589.** Page 1134, line 22: delete the material beginning with that line and ending with page 1135, line 12, and substitute:

**"Section 2386b.** 77.54 (26) of the statutes is amended to read:

77.54 (26) The gross receipts from the sales of and the storage, use, or other consumption of tangible personal property which becomes a component part of an industrial waste treatment facility that is exempt under s. 70.11 (21) (a) or that would be exempt under s. 70.11 (21) (a) if the property were taxable under ch. 70, or tangible personal property which becomes a component part of a waste treatment facility of this state or any agency thereof, or any political subdivision of the state or agency thereof as provided in s. 40.02 (28). The exemption includes replacement parts therefor, and also applies to chemicals and supplies used or consumed in operating a waste treatment facility and to purchases of tangible personal property made by construction contractors who transfer such property to their customers in fulfillment of a real property construction activity. This exemption does not apply to tangible personal property installed in fulfillment of a written construction contract entered into, or a formal written bid made, prior to July 31, 1975.".

**1590.** Page 1135, line 13: delete the material beginning with that line and ending with page 1144, line 11.

**1591.** Page 1144, line 11: after that line insert:

23

24

**"Section 2391d.** 77.54 (30) (a) 1m. of the statutes is created to read: 1 2 77.54 (30) (a) 1m. Biomass, as defined in s. 196.378 (1) (ar), that is used for fuel 3 sold for residential use. 4 **Section 2410d.** 77.54 (47) (b) 2. of the statutes is amended to read: 5 77.54 (47) (b) 2. The shooting facility is a nonprofit organization that charges 6 for shooting at the facility, but is not required to pay the tax imposed under s. 77.52 7 on its gross receipts from such charges because the charges are for occasional sales. 8 as provided under sub. (7m), or because the charges satisfy the exemption under s. 9 77.52 (2) (a) 2. b.". **1592.** Page 1144, line 17: delete lines 17 to 22. 10 11 **1593.** Page 1144, line 22: after that line insert: 12 **"Section 2419c.** 77.54 (56) of the statutes is created to read: 13 77.54 (56) (a) The gross receipts from the sale of and the storage, use, or other 14 consumption of a product whose power source is wind energy, direct radiant energy 15 received from the sun, or gas generated from anaerobic digestion of animal manure 16 and other agricultural waste, if the product produces at least 200 watts of alternating 17 current or 600 British thermal units per day, except that the exemption under this 18 subsection does not apply to an uninterruptible power source that is designed 19 primarily for computers. 20 (b) Except for the sale of electricity or energy that is exempt from taxation 21under sub. (30), the gross receipts from the sale of and the storage, use, or other

consumption of electricity or energy produced by a product described under par. (a).".

**1594.** Page 1144, line 22: after that line insert:

**"Section 2.** 77.54 (55) of the statutes is created to read:

- 1 77.54 **(55)** The gross receipts from the sale of and the storage, use, or other consumption of all of the following:
  - (a) United States coins.
- 4 (b) United States currency.
- 5 (c) Any of the following that are made from gold, silver, platinum, or palladium, 6 or any combination of such metals:
- 7 1. Bars.

- 8 2. Ingots, not including the mold in which the metal is cast.
- 9 3. Coins.
- 10 (d) Commemorative medallions.".
- 11 **1595.** Page 1144, line 23: delete the material beginning with that line and ending with page 1148, line 12.
- 13 **1596.** Page 1148, line 13: delete lines 13 to 15.
- 14 **1597.** Page 1148, line 16: delete the material beginning with that line and ending with page 1153, line 13.
- 16 **1598.** Page 1154, line 3: delete the material beginning with that line and ending with page 1160, line 2.
- 18 **1599.** Page 1160, line 3: delete lines 3 to 6.
- 19 **1600.** Page 1160, line 7: delete the material beginning with that line and ending with page 1172, line 8.
- 21 **1601.** Page 1172, line 8: after that line insert:
- 22 "Section 2460d. 77.68 of the statutes is created to read:

 $\mathbf{2}$ 

- **77.68 Deposit into the transportation fund. (1)** No later than July 1, 2009, the department shall estimate the total amount of the taxes imposed under ss. 77.52 and 77.53 on the sale or use of motor vehicles, motor vehicle parts, and motor vehicle services that are taxable under s. 77.52 **(2) (a)** 10. that the department collected in fiscal year 2008–09.
- (2) Beginning on July 1, 2009, and on July 1 of each year thereafter, the department shall estimate the total amount of the taxes imposed under ss. 77.52 and 77.53 on the sale or use of motor vehicles, motor vehicle parts, and motor vehicle services that are taxable under s. 77.52 (2) (a) 10. that the department will collect in that fiscal year.
- (3) On July 1, 2009, and on each July 1 thereafter, the department shall transfer from s. 20.855 (4) (fd) to the transportation fund 50 percent of the amount calculated by subtracting the amount estimated under sub. (1) from the amount estimated under sub. (2), if subtracting the amount estimated under sub. (1) from the amount estimated under sub. (2) results in a positive number.".
- **1602.** Page 1173, line 17: delete the material beginning with that line and ending with page 1175, line 14, and substitute:
- "Section 2482m. 77.89 (2) (a) of the statutes is renumbered 77.89 (2) and amended to read:
- 77.89 (2) Each municipal treasurer shall pay 20% of each payment received under sub. (1) and under ss. 77.84 (2) (a) and, (am), (b), and (bm), 77.85, and 77.876 to the county treasurer and shall deposit the remainder in the municipal treasury. The payment to the county treasurer for money received before November 1 of any year shall be made on or before the November 15 after its receipt. For money received

8

9

10

13

14

15

16

17

18

- on or after November 1 of any year, the payment to the county treasurer shall be made on or before November 15 of the following year.
- 3 **Section 2482p.** 77.89 (2) (b) of the statutes is repealed.".
- 4 **1603.** Page 1175, line 24: after "(3g)," insert "(3h), (3i),".
- 5 **1604.** Page 1176, line 7: delete the material beginning with that line and ending with page 1179, line 12.
  - **1605.** Page 1179, line 20: delete the material beginning with that line and ending with page 1180, line 3.
    - **1606.** Page 1180, line 4: delete the material beginning with that line and ending with page 1182, line 24.
- 11 **1607.** Page 1182, line 25: delete the material beginning with that line and ending with page 1187, line 9, and substitute:
  - "Section **2506d.** 79.043 (5) of the statutes is amended to read:
  - 79.043 (5) Except as provided under s. 79.02 (3) (e), for the distribution beginning in 2005 and subsequent years ending in 2007, each county and municipality shall receive a payment under this section and s. 79.035 that is equal to the amount of the payment determined for the county or municipality under this section and s. 79.035 in 2004.
    - **Section 2507d.** 79.043 (6) of the statutes is created to read:
- 79.043 (6) (a) Except as provided in s. 79.02 (3) (e), in 2008 each county shall receive a payment under this section and s. 79.035 that is equal to the amount it received in 2007, minus 5 percent of the amount it received in 2007.

(b) Except as provided in s. 79.02 (3) (e) and par. (c), in 2008 each municipality
shall receive a payment under this section and s. 79.035 that is equal to the amount
it received in 2007.

(c) Except as provided in s. 79.02 (3) (e), in 2008 each municipality shall receive a payment under this section and s. 79.035 that is equal to the amount it received in 2007, minus 12.5 percent of the amount it received in 2007, if the municipality's 2007 population exceeded 20,000 and if its distribution under this section and s. 79.035 in 2007 exceeded an amount equal to the municipality's 2007 population multiplied by \$250.

**Section 2507e.** 79.043 (7) of the statutes is created to read:

79.043 (7) Except as provided under s. 79.02 (3) (e), for the distribution in 2009 and subsequent years, each county and municipality shall receive a payment under this section and s. 79.035 that is equal to the amount of the payment determined for the county or municipality under this section and s. 79.035 in 2008."

**1608.** Page 1187, line 9: after that line insert:

"Section 2505d. 79.04 (1) (intro.) of the statutes is amended to read:

79.04 (1) (intro.) Annually, except for production plants that begin operation after December 31, 2003, or begin operation as a repowered production plant after December 31, 2003, and except as provided in sub. (4m), the department of administration, upon certification by the department of revenue, shall distribute to a municipality having within its boundaries a production plant, general structure, or substation, used by a light, heat, or power company assessed under s. 76.28 (2) or 76.29 (2), except property described in s. 66.0813 unless the production plant or substation is owned or operated by a local governmental unit located outside of the

 $\mathbf{2}$ 

municipality, or by an electric cooperative assessed under ss. 76.07 and 76.48, respectively, or by a municipal electric company under s. 66.0825 the amount determined as follows:

**Section 2505e.** 79.04 (1) (b) 1. of the statutes is amended to read:

79.04 (1) (b) 1. Beginning with the distribution under this subsection in 1991, and ending with the distribution under this subsection in 2008, the amount determined under par. (a) to value property used by a light, heat or power company in a municipality may not be less than the amount determined to value the property for the distribution to the municipality under this subsection in 1990, subject to subds. 2., 3. and 4.

**Section 2505f.** 79.04 (2) (a) of the statutes is amended to read:

79.04 (2) (a) Annually, except for production plants that begin operation after December 31, 2003, or begin operation as a repowered production plant after December 31, 2003, and except as provided in sub. (4m), the department of administration, upon certification by the department of revenue, shall distribute from the shared revenue account or, for the distribution in 2003, from the appropriation under s. 20.835 (1) (t), 2003 stats., to any county having within its boundaries a production plant, general structure, or substation, used by a light, heat or power company assessed under s. 76.28 (2) or 76.29 (2), except property described in s. 66.0813 unless the production plant or substation is owned or operated by a local governmental unit that is located outside of the municipality in which the production plant or substation is located, or by an electric cooperative assessed under ss. 76.07 and 76.48, respectively, or by a municipal electric company under s. 66.0825 an amount determined by multiplying by 6 mills in the case of property in a town and by 3 mills in the case of property in a city or village the first \$125,000,000 of the

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

amount shown in the account, plus leased property, of each public utility except qualified wholesale electric companies, as defined in s. 76.28 (1) (gm), on December 31 of the preceding year for "production plant, exclusive of land," "general structures," and "substations," in the case of light, heat and power companies, electric cooperatives or municipal electric companies, for all property within the municipality in accordance with the system of accounts established by the public service commission or rural electrification administration, less depreciation thereon as determined by the department of revenue and less the value of treatment plant and pollution abatement equipment, as defined under s. 70.11 (21) (a), as determined by the department of revenue plus an amount from the shared revenue account or, for the distribution in 2003, from the appropriation under s. 20.835 (1) (t), 2003 stats., determined by multiplying by 6 mills in the case of property in a town, and 3 mills in the case of property in a city or village, of the total original cost of production plant, general structures, and substations less depreciation, land and approved waste treatment facilities of each qualified wholesale electric company, as defined in s. 76.28 (1) (gm), as reported to the department of revenue of all property within the municipality. The total of amounts, as depreciated, from the accounts of all public utilities for the same production plant is also limited to not more than \$125,000,000. The amount distributable to a county under this subsection and sub. (6) in any year shall not exceed \$100 times the population of the county.

**Section 2505g.** 79.04 (2) (am) 1. of the statutes is amended to read:

79.04 (2) (am) 1. Beginning with the distribution under this subsection in 1991, and ending with the distribution under this subsection in 2008, the amount determined under par. (a) to value property used by a light, heat or power company

in a county may not be less than the amount determined to value the property for the distribution to the county under this subsection in 1990, subject to subds. 2. and 3.

**SECTION 2505h.** 79.04 (4m) of the statutes is created to read:

79.04 (4m) Beginning with distributions in 2009, for production plants described under subs. (1) and (2), if in any year the payments to the municipality and county in which the production plant is located would be greater under subs. (6) and (7) (c) 1. based on the production plant's name-plate capacity than under sub. (1) or (2) based on the depreciated net book value of the production plant, the municipality and county shall receive payments under subs. (6) and (7) (c) 1., rather than under sub. (1) or (2), beginning in that year and in each year thereafter.

**Section 2505i.** 79.04 (6) (a) of the statutes is amended to read:

79.04 (6) (a) Annually, beginning in 2005, for production plants that begin operation after December 31, 2003, or begin operation as a repowered production plant after December 31, 2003, except as provided in sub. (4m), the department of administration, upon certification by the department of revenue, shall distribute payments from the public utility account, as determined under par. (b), to each municipality and county in which a production plant is located, if the production plant has a name-plate capacity of at least one megawatt and is used by a light, heat, or power company assessed under s. 76.28 (2) or 76.29 (2), except property described in s. 66.0813, unless the production plant is owned or operated by a local governmental unit located outside of the municipality; by a qualified wholesale electric company, as defined in s. 76.28 (1) (gm); by a wholesale merchant plant, as defined in s. 196.491 (1) (w); by an electric cooperative assessed under ss. 76.07 and 76.48, respectively; or by a municipal electric company under s. 66.0825.

**Section 2505j.** 79.04 (6) (c) 1. of the statutes is amended to read:

 $\mathbf{2}$ 

79.04 (6) (c) 1. If the production plant is located in a city or village, the city or village receives a payment equal to two-thirds of the amount determined under par. (b) and the county in which the city or village is located receives a payment equal to one-third of the amount determined under par. (b). If the production plant is located in a town, the town receives a payment equal to one-third of the amount determined under par. (b), and the county in which the town is located receives a payment equal to two-thirds of the amount determined under par. (b), except that with regard to a production plant that generates electricity from wind power the town where the plant is located receives a payment equal to two-thirds of the amount determined under par. (b) and the county where the plant is located receives a payment equal to one-third of the amount determined under par. (b). If a municipality is located in more than one county, the county in which the production plant is located shall receive the county portion of the payment."

**1609.** Page 1187, line 9: after that line insert:

"Section 2505d. 79.04 (1) (a) of the statutes is amended to read:

79.04 (1) (a) An amount from the shared revenue account or, for the distribution in 2003, from the appropriation under s. 20.835 (1) (t), 2003 stats., determined by multiplying by 3 mills in the case of a town, and 6 mills in the case of a city or village, the first \$125,000,000 of the amount shown in the account, plus leased property, of each public utility except qualified wholesale electric companies, as defined in s. 76.28 (1) (gm), on December 31 of the preceding year for "production plant, exclusive of land," "general structures," and "substations," in the case of light, heat and power companies, electric cooperatives or municipal electric companies, for all property within a municipality in accordance with the system of accounts

established by the public service commission or rural electrification administration, less depreciation thereon as determined by the department of revenue and less the value of treatment plant and pollution abatement equipment, as defined under s. 70.11 (21) (a), as determined by the department of revenue plus an amount from the shared revenue account or, for the distribution in 2003, from the appropriation under s. 20.835 (1) (t), 2003 stats., determined by multiplying by 3 mills in the case of a town, and 6 mills in the case of a city or village, of the first \$125,000,000 of the total original cost of production plant, general structures, and substations less depreciation, land and approved waste treatment facilities of each qualified wholesale electric company, as defined in s. 76.28 (1) (gm), as reported to the department of revenue of all property within the municipality. The total of amounts, as depreciated, from the accounts of all public utilities for the same production plant is also limited to not more than \$125,000,000. The amount distributable to a municipality under this subsection and sub. (6) in any year shall not exceed \$300 times the population of the municipality."

**1610.** Page 1187, line 10: delete the material beginning with that line and ending with page 1190, line 21, and substitute:

"Section 2522d. 79.14 of the statutes is amended to read:

**79.14 School levy tax credit.** The appropriation under s. 20.835 (3) (b) is \$319,305,000 in 1994, 1995, and 1996; \$469,305,000 beginning in 1997 and ending in 2006; and \$593,050,000 beginning in 2007 and ending in 2008; and \$793,050,000 in each year thereafter.".

**1611.** Page 1190, line 21: after that line insert:

"Section 2524g. 84.013 (3m) (d) of the statutes is created to read:

84.013 (3m) (d) The department shall begin construction of the major highway project enumerated under sub. (3) (ra) no later than July 1, 2009, and shall complete construction of this project no later than July 1, 2011.".

**1612.** Page 1190, line 21: after that line insert:

**"Section 2523p.** 84.013 (4) (a) of the statutes is amended to read:

84.013 (4) (a) Subject to s. ss. 13.489 (1m) and 84.06 (1r), in preparation for future major highway projects, the department may perform preliminary engineering and design work and studies for possible major highway projects not listed under sub. (3), but no major highway may be constructed unless the project is listed under sub. (3) or approved under sub. (6).".

**1613.** Page 1190, line 21: after that line insert:

**"Section 2523p.** 83.015 (2) (b) of the statutes is amended to read:

83.015 (2) (b) In any county with a highway commissioner appointed under s. 83.01 (1) (b) or (c), the county highway committee shall be only a policy–making body determining the broad outlines and principles governing administration and the county highway commissioner shall have the administrative powers and duties prescribed for the county highway committee under par. (a), sub. (3) (a) and ss. 27.065 (4) (b) and (13), 32.05 (1) (a), 82.08, 83.01 (6), 83.013, 83.018, 83.025 (1) and (3), 83.026, 83.035, 83.04, 83.05 (1), 83.07 to 83.09, 83.12, 83.14 (6), 83.17, 83.18, 83.42 (3) and (4), 84.01 (5), 84.06 (3), 84.07 (1) and (2), 84.09 (1), (3) (a) to (c) and (4), 84.10 (1), 86.04 (1) and (2), 86.07 (2) and (2m), 86.19 (3), 86.34 (1), 114.33 (5), 349.07 (2), 349.11 (4) and (10) and 349.15 (2). No statutory power, duty or function specified elsewhere for the county highway commissioner may be deemed impliedly repealed for the sole reason that reference to it has been omitted in this paragraph.".

**1614.** Page 1190, line 21: after that line insert:

"Section 2524p. 84.01 (21) of the statutes is amended to read:

84.01 (21) Motor vehicle weighing stations. The department, as a part of the improvement and maintenance of highways, may acquire, construct and maintain lands and facilities, including scales or weighing stations, for weighing, measuring or inspecting vehicles and loads operating on any public highway in the state. Lands necessary may be adjacent or contiguous to the highway and weighing station facilities may be constructed and maintained upon the traveled portion of the highway or any other part thereof. The department may not construct or locate a motor vehicle weighing facility in or adjacent to the village of Rockland in La Crosse County.".

**1615.** Page 1191, line 18: after that line insert:

"Section 2528g. 84.02 (5) (a) of the statutes is amended to read:

84.02 (5) (a) As often as it deems necessary, the department shall publish highway service maps showing the state trunk highway system and such other main highways and other features as may seem desirable. Such highway service maps shall be sold by the department at a price to be fixed by it, which shall be not less than cost. The department may permit the use of the base plates for other maps and publications in consideration of a fair fee for such use. The department shall make and publish or duplicate such highway service maps as are required for its use, and, in only one fiscal year of each fiscal biennium, shall publish folded highway maps of Wisconsin for free distribution to the public. The department shall ensure that the folded highway maps bear information regarding the requirements of s. 347.48 (4).".

**1616.** Page 1192, line 2: after that line insert:

"Section 2531c. 84.06 (1m) of the statutes is amended to read:

84.06 (1m) Plans. The Subject to sub. (1r), the department may prepare plans, estimates, and specifications and undertake and perform all surveys, investigations, and engineering work for any highway improvement within its jurisdiction. When provision has been made for the necessary funds for any such highway improvement and, if federal aid is to be utilized, when the project has been approved by the proper federal authorities, the department may proceed as provided in this section, with due regard to any applicable federal requirement or regulation.

**Section 2531e.** 84.06 (1r) of the statutes is created to read:

- 84.06 (1r) Value engineering for certain projects. (a) In this subsection, "value engineering" has the meaning given in 23 CFR 627.3.
- (b) The department shall employ value engineering for any highway improvement project under sub. (1m) for which the cost of construction, utilities, and rights-of-way is in excess of \$5,000,000, as adjusted under par. (g), or that otherwise meets criteria established by the department under par. (c), and shall assure that a value engineering study and analysis is performed on each such project. Any value engineering study and analysis related to engineering work performed by a consultant under contract with the department under s. 84.01 (13) may not be performed by the same consultant unless that consultant maintains separate and distinct organizational separation of its value engineering and design sections.
- (c) The department shall establish criteria for determining which projects, in addition to those having total project costs in excess of \$5,000,000, as adjusted under par. (g), on which the department will employ value engineering.

 $\mathbf{2}$ 

- (d) After review and for compelling reasons, the secretary may waive the requirement under par. (b) for any project. Any such waiver shall be in writing, state the reasons for the waiver, and apply only to a single project.
- (e) For each project for which the department performs a value engineering study and analysis under this subsection, the department shall include in the study and analysis an identification of the cost of all design elements for the project that are considered by the department to be context–sensitive design elements and the department's justification for any increased project costs resulting from these design elements.
- (f) Annually, the department shall submit a report to the governor, and to the appropriate standing committees of the legislature under s. 13.172 (3), on the department's employment of value engineering under par. (b), the criteria established by the department under par. (c), and all waivers made under par. (d). All project information included in any report required under this paragraph shall be reported on both a cumulative basis from the inception of the project and on an updated basis for the period since the department's last report under this paragraph. The report under this paragraph shall also include all of the following information:
  - 1. The number of value engineering studies conducted.
  - 2. The cost of conducting the studies.
  - 3. The estimated construction cost of the projects studied.
  - 4. The total number of study recommendations.
- 5. The total estimated savings that would result from all recommendations if approved and implemented.
  - 6. The number of recommendations approved.
  - 7. The total savings that resulted from the approved recommendations.

 $\mathbf{2}$ 

- 8. The cost of all context-sensitive design elements included in completed projects.
  - (g) Beginning on the first day of the 13th month beginning after the effective date of this paragraph .... [revisor inserts date], and annually thereafter, the department shall adjust the dollar amounts specified in pars. (b) and (c) in proportion to any change in the cost of construction, utilities, and rights-of-way since the effective date of this paragraph.".
    - **1617.** Page 1193, line 8: after that line insert:
      - "Section 2532h. 84.101 of the statutes is created to read:
  - 84.101 Ronald Reagan Memorial Highway. The department shall designate and mark the route of USH 14 from the Wisconsin-Illinois border to Madison as the "Ronald Reagan Memorial Highway" in recognition and appreciation of the public career of Ronald Reagan, who served 2 terms of office with distinction as the 40th president of the United States and who subsequently demonstrated grace and dignity in his struggle with Alzheimer's disease.".
    - **1618.** Page 1193, line 9: delete lines 9 to 14.
- **1619.** Page 1193, line 19: after that line insert:
  - "Section 2534p. 84.25 (4) of the statutes is renumbered 84.25 (4) (a) and amended to read:
    - 84.25 (4) (a) After Except as provided in par. (b), after the establishment of any controlled-access highway, no street or highway or private driveway, shall be opened into or connected with any controlled-access highway without the previous consent and approval of the department in writing, which shall be given only if the public

interest shall be served thereby and shall specify the terms and conditions on which such consent and approval is given.

**Section 2534r.** 84.25 (4) (b) of the statutes is created to read:

84.25 **(4)** (b) Notwithstanding any other provision of this chapter, if the department denies approval under par. (a), or fails to provide approval under par. (a) within 60 days after a request for approval is made to the department, related to any access point to any part of a controlled–access highway located within a municipality, that municipality may approve any access point specified in the request for approval made to the department. Any approval under this paragraph shall be in writing and shall specify the terms and conditions on which the approval is given.".

- **1620.** Page 1193, line 20: delete lines 20 to 25.
- **1621.** Page 1194, line 1: delete lines 1 to 12.
- **1622.** Page 1194, line 25: after "(6w)," insert "(6y),".
- **1623.** Page 1194, line 25: delete "and (b)" and substitute "and (b) to (c)".
- **1624.** Page 1195, line 4: delete the material beginning with "<u>, and from</u>" and ending with "section" on line 6.
- **1625.** Page 1195, line 19: delete "\$2,708,341,000" and substitute 18 "\$2,668,661,500".
- **1626.** Page 1195, line 25: delete the material beginning with "<u>, to make</u>" and ending with "<u>this section</u>," on page 1196, line 2.
- **1627.** Page 1196, line 3: after that line insert:
- 22 "Section **2540c.** 85.024 (2) of the statutes is amended to read:

85.024 (2) The department shall administer a bicycle and pedestrian facilities program to award grants of assistance to political subdivisions for the planning, development, or construction of bicycle and pedestrian facilities. For purposes of this subsection, "bicycle and pedestrian facilities" do not include sidewalks or street beautification measures. The department shall award from the appropriation under s. 20.395 (2) (nx) (ox) grants to political subdivisions under this section. A political subdivision that is awarded a grant under this section shall contribute matching funds equal to at least 25% 20 percent of the amount awarded under this section. Any improvement project for which a political subdivision receives a grant under this section shall be let by contract based on bids and the contract shall be awarded to the lowest competent and responsible bidder.".

**1628.** Page 1196, line 23: after that line insert:

"Section 2541r. 85.035 of the statutes is amended to read:

85.035 Reduction of department appropriations. Where Subject to ss. 16.50 (1) (c) and 85.62, where the secretary deems that economic conditions warrant, the secretary, in conjunction with submission of estimates under s. 16.50, may recommend to the secretary of administration that authorized department appropriations be reduced to reflect revenue deficiencies."

**1629.** Page 1196, line 24: delete that line.

**1630.** Page 1197, line 1: delete lines 1 to 5.

**1631.** Page 1199, line 19: delete lines 19 to 25.

**1632.** Page 1200, line 1: delete lines 1 to 7.

**1633.** Page 1200, line 7: after that line insert:

"Section 2550e. 85.243 of the statutes is repealed.".

 $\mathbf{2}$ 

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

**1634.** Page 1201, line 8: after that line insert:

**"Section 2550s.** 86.07 (2) of the statutes is amended to read:

86.07 (2) No Subject to sub. (2m), no person shall make any excavation or fill or install any culvert or make any other alteration in any highway or in any manner disturb any highway or bridge without a permit therefor from the highway authority maintaining the highway. Such permit shall contain the statement and be subject to the condition that the work shall be constructed subject to such rules and regulations as may be prescribed by said authority and be performed and completed to its satisfaction, and in the case of temporary alterations that the highway or bridge shall be restored to its former condition, and that the permittee shall be liable to the town or county or state, as the case may be, for all damages which occur during the progress of said work or as a result thereof. Nothing herein shall abridge the right of the department or the county board or its highway committee to make such additional rules, regulations and conditions not inconsistent herewith as may be deemed necessary and proper for the preservation of highways, or for the safety of the public, and to make the granting of any such permit conditional thereon. If Subject to sub. (2m), if any culvert is installed or any excavation or fill or any other alteration is made in violation of the provisions of this subsection, the highway may be restored to its former condition by the highway authority in charge of the maintenance thereof; and any person who violates this subsection shall be punished by a fine of not less than \$5 nor more than \$100, or by imprisonment not exceeding 6 months, or both.

**Section 2550t.** 86.07 (2m) of the statutes is created to read:

86.07 (2m) Notwithstanding s. 86.073, if the department denies an application for a permit under sub. (2), or fails to approve a permit under sub. (2) within 60 days after application for the permit is made to the department, related to any part of a state trunk highway located within a municipality, that municipality may issue a permit under sub. (2) approving and authorizing the work, activity, or alteration, with respect to this state trunk highway, that was the subject of the permit application made to the department, regardless of what authority maintains the highway or whether the highway has been designated a connecting highway under s. 86.32.".

**1635.** Page 1201, line 8: after that line insert:

"Section 2550u. 85.62 of the statutes is created to read:

85.62 Transportation fund balance and appropriation reductions. (1) During any fiscal biennium, if at any time the department determines that, at the close of the fiscal biennium, net appropriations and fund reserves from the transportation fund will exceed estimated transportation fund revenues by more than \$30,000,000, the department shall develop a plan to eliminate, by the close of the fiscal biennium, this projected deficit by reducing all state funds appropriations under s. 20.395, other than those for debt service and sum sufficient appropriations, from the transportation fund as equitably as reasonable. This plan may not include the reduction of any state funds appropriation if the reduction would violate a condition imposed by the federal government on the receipt of federal funds or if the reduction would violate the federal or state constitution. The secretary shall submit this plan to the joint committee on finance.

- (2) If the cochairpersons of the joint committee on finance do not notify the secretary within 14 working days after the date on which the secretary submits the plan under sub. (1) that the committee has scheduled a meeting for the purpose of reviewing the plan, the secretary shall implement the plan. If, within 14 working days after the date on which the secretary submits the plan, the cochairpersons of the joint committee on finance notify the secretary that the committee has scheduled a meeting for the purpose of reviewing the plan, the secretary may implement the plan only as approved, or modified and approved, by the committee.
- (3) Notwithstanding ss. 13.101 (6) and 20.001 (3) (c), if a plan is implemented as provided under sub. (2), the appropriation accounts shall be reduced as specified in the plan implemented under sub. (2).".
  - **1636.** Page 1201, line 9: delete lines 9 to 13.
- **1637.** Page 1203, line 20: delete the material beginning with that line and ending with page 1204, line 8.
  - **1638.** Page 1204, line 9: after that line insert:
  - "Section 2558d. 88.68 (4) of the statutes is amended to read:
- 88.68 (4) Whenever the cleaning out, deepening or reconstruction of a drain crossing a public highway requires the lowering of a culvert through such highway in order to provide effective drainage, the drainage board may proceed to lower such culvert only after obtaining a permit under s. 86.07 (2) or (2m). In lieu of issuing a permit, the authority in charge of maintenance of the highway may proceed to do the work itself. Except as provided in s. 86.075, the expenses involved in such lowering shall be borne by the drainage district, or as provided by mutual agreement between the highway authority and the drainage board.".

- **1639.** Page 1204, line 13: after that line insert:
- 2 "Section 2589f. 93.07 (25) of the statutes is repealed.".
- **1640.** Page 1204, line 14: delete lines 14 to 25.
- **1641.** Page 1205, line 1: delete lines 1 to 13.
- **1642.** Page 1205, line 13: after that line insert:
- **"Section 2592d.** 93.46 (2) (e) of the statutes is amended to read:
- 7 93.46 (2) (e) The department may not make a grant under this subsection or
- 8 s. 93.47 that exceeds 75 percent of project costs.
- **Section 2592f.** 93.47 of the statutes is repealed.".
- **1643.** Page 1206, line 19: substitute "\$13" for "\$14".
- **1644.** Page 1206, line 21: substitute "\$13" for "\$14".
- **1645.** Page 1206, line 23: substitute "41" for "44".
- **1646.** Page 1207, line 5: substitute "\$3.25" for "\$3.50".
- **1647.** Page 1207, line 8: substitute "\$110" for "\$120".
- **1648.** Page 1207, line 11: substitute "<u>0.72</u>" for "<u>0.75</u>".
- **1649.** Page 1207, line 14: substitute "\$26" for "\$28".
- **1650.** Page 1207, line 21: substitute "<u>\$36</u>" for "\$38".
- **1651.** Page 1208, line 2: substitute "\$13" for "\$14".
- **1652.** Page 1210, line 10: after that line insert:
- 20 "Section 2602h. 96.11 (3) of the statutes is created to read:
- 96.11 (3) (a) 1. Except as provided under par. (c), a corn producer who sells corn
- that is subject to the assessment levied under sub. (1) under the marketing order for

- corn shall pay an additional assessment of 0.4 cents per bushel, to be collected and remitted to the marketing board in the manner provided in the marketing order for the assessment under sub. (1).
- 2. If the marketing order for corn provides for rebates under s. 96.13 (2), a producer to whom subd. 1. applies may obtain a rebate of the assessment paid under subd. 1. in the manner provided in the marketing order for obtaining a rebate of the assessment levied under sub. (1).
- (b) The marketing board for corn shall use the moneys received under par. (a) for the purposes of the marketing order for corn.
- (c) 1. The department shall conduct a referendum on whether to terminate the assessment under par. (a) if after June 30, 2011, and before January 1, 2012, 10 percent of the producers who sell corn that is subject to the assessment levied under sub. (1) petition for a referendum. The assessment under par. (a) does not apply after June 30, 2012, if a referendum to terminate the assessment is approved by affected producers.
  - 2. If the marketing order for corn is terminated, pars. (a) and (b) do not apply. **Section 2602j.** 96.17 (1) of the statutes is amended to read:
- 96.17 (1) Any due and payable assessment levied under a marketing order or under s. 96.11 (3) (a) and every sum due under either a marketing order or agreement in a specified amount shall constitute a personal debt of every person so assessed or who is otherwise liable and the same sum shall be due and payable to the secretary or the marketing board according to the terms and conditions of the marketing order or agreement. In the event any person fails to pay the full amount of such assessment or such other sum on or before the due date, the secretary may add to such unpaid assessment or sum an amount not exceeding 10% of the amount due to defray the cost

 $\mathbf{2}$ 

of enforcing collection. In the event any person fails to pay any due and payable assessment or sum, the secretary may bring a civil action against such person for collection, together with the above specified 10%.".

**1653.** Page 1210, line 10: after that line insert:

"Section 2602p. 98.04 (2) of the statutes is renumbered 98.04 (2) (intro.) and amended to read:

98.04 (2) (intro.) A municipality that is required to establish a department of weights and measures under sub. (1) may contract, instead of establishing its own department, do any of the following:

(a) Contract with the department of agriculture, trade, and consumer protection to enforce the provisions of this chapter within the municipality's jurisdiction instead of establishing its own department if the department of agriculture, trade and consumer protection agrees to enter into such a contract. The department of agriculture, trade and consumer protection may charge the municipality fees sufficient to cover the department's costs under the contract. A municipality may recover an amount not to exceed the cost of these fees by assessing fees on the persons who receive services under the weights and measures program.

**Section 2602q.** 98.04(2)(b) of the statutes is created to read:

98.04 (2) (b) Contract with a private party who has been certified under sub.

(3) to enforce the provisions of this chapter within the municipality's jurisdiction. A municipality may recover an amount not to exceed the municipality's cost of contracting with a private party under this paragraph by assessing fees on the persons who receive services under the weights and measures program.

**Section 2602r.** 98.04 (3) of the statutes is created to read:

- 1 98.04 (3) The department of agriculture, trade and consumer protection shall 2 promulgate rules implementing a program for certifying a person as qualified to act 3 as a private contractor under sub. (2) (b).". **1654.** Page 1210, line 11: delete lines 11 to 16. 4 5 **1655.** Page 1210, line 16: after that line insert: 6 "Section 2607p. 100.55 of the statutes is created to read: 7 100.55 Motor vehicle fuel retailer contracts; prohibited provisions. A 8 contract between a retailer of motor vehicle fuel in this state and a motor vehicle fuel 9 wholesaler may not prohibit the retailer from, or penalize the retailer for, selling motor vehicle fuel containing ethanol or biodiesel.". 10 11 **1656.** Page 1210, line 17: delete lines 17 to 22. **1657.** Page 1210, line 22: after that line insert: 12 "Section 2608m. 101.02 (7m) of the statutes is repealed.". 13 **1658.** Page 1210, line 23: delete lines 23 to 25. 14 **1659.** Page 1211, line 1: delete lines 1 to 25. 15 **1660.** Page 1212, line 1: delete lines 1 to 13. 16 **1661.** Page 1212, line 19: after that line insert: 17 18 "Section 2614d. 101.124 of the statutes is repealed.". **1662.** Page 1212, line 19: after that line insert: 19 20 **"Section 2615c.** 101.14 (4m) (a) 1m. of the statutes is created to read:
- 21 101.14 (4m) (a) 1m. "Community-based residential facility" means a facility
  22 that meets the definition in s. 50.01 (1g) and that is licensed under s. 50.03 (1).
- 23 **Section 2615e.** 101.14 (4m) (a) 4. of the statutes is repealed.

1 **Section 2615g.** 101.14 (4m) (am) of the statutes is repealed.  $\mathbf{2}$ **Section 2615i.** 101.14 (4m) (b) (intro.) and 2. of the statutes are consolidated, 3 renumbered 101.14 (4m) (bg) and amended to read: 101.14 (4m) (bg) The department shall require an automatic fire sprinkler 4 5 system or 2-hour fire resistance in every multifamily dwelling that contains any of 6 the following: 2. More than 20 more than 12 dwelling units and that is served by 7 a public water system with adequate pressure for fire sprinklers. 8 **Section 2615L.** 101.14 (4m) (b) 1. of the statutes is repealed. 9 **Section 2615m.** 101.14 (4m) (b) 3. of the statutes is repealed. 10 **Section 2615p.** 101.14 (4m) (br) of the statutes is created to read: 11 101.14 (4m) (br) The department shall require 2-hour fire resistance in every 12 multifamily dwelling that contains more than 12 dwelling units and that is not 13 served by a public water system with adequate pressure for fire sprinklers. 14 **Section 2615r.** 101.14 (4m) (bv) of the statutes is created to read: 15 101.14 (4m) (by) The department may not require an automatic sprinkler system or 2-hour fire resistance in any multifamily dwelling that has 12 or fewer 16 17 dwelling units. 18 **Section 2615s.** 101.14 (4m) (c) of the statutes is repealed. 19 **Section 2615t.** 101.14 (4m) (d) of the statutes is repealed. 20 **Section 2615u.** 101.14 (4m) (e) of the statutes is repealed. 21**Section 2615v.** 101.14 (4m) (g) of the statutes is created to read: 22 101.14 (4m) (g) For every multifamily dwelling the initial construction of 23 which is begun on or after the effective date of this paragraph .... [revisor inserts 24date, all of the following shall apply:

1. Any exterior point at ground level on the multifamily dwelling may not be
farther than 250 feet from a driveable surface. The driveable surface may not be
paved unless the lack of pavement is impracticable.

2. Any interior point in the multifamily dwelling may not be farther than 100 feet from the nearest emergency exit from the multifamily dwelling.

**SECTION 2615w.** 101.14 (4m) (gm) of the statutes is created to read:

- 101.14 **(4m)** (gm) For every community-based residential facility the initial constriction of which is begun on or after the effective date of this paragraph .... [revisor inserts date], all of the following shall apply:
- 1. Any exterior point at ground level on the community-based residential facility may not be farther than 250 feet from a driveable surface. The driveable surface may not be paved unless the lack of pavement is impracticable.
- 2. Any interior point in the community-based residential facility may not be farther than 100 feet from the nearest emergency exit from the community-based residential facility.

**Section 2615x.** 101.14 (4m) (h) of the statutes is created to read:

101.14 (4m) (h) Any ordinance enacted by a political subdivision that relates to the use of automatic fire sprinkler systems or to the placement of fire resistance structures or materials shall conform to this section. Any ordinance enacted or authorized under s. 66.1019 (3) (b), 2005 stats., 101.02 (7m), 2005 stats., or 101.975 (3), 2005 stats., is no longer valid.

**Section 2615y.** 101.141 of the statutes is amended to read:

101.141 Record keeping of fires. The department shall maintain records of all fires occurring in this state. For each fire that involves a building and that results in one or more fatalities, the department shall include in the records the

24

June 1, 2008.".

cause of the fire, the age of the building, what the building was used for, and any other 1  $\mathbf{2}$ relevant information concerning the building, as determined by the department. 3 Such records shall be open to public inspection during normal business hours.". **1663.** Page 1215, line 3: delete lines 3 to 9. 4 5 **1664.** Page 1215, line 18: delete lines 18 to 20 and substitute "funds, or to pay 6 accrued or capitalized interest.". 7 **1665.** Page 1215, line 20: after that line insert: 8 "Section 2629k. 101.143 (9m) (j) of the statutes is created to read: 9 101.143 (9m) (j) In each fiscal year in which revenue obligations are 10 outstanding under this subsection, the department of administration and the department of commerce shall periodically determine whether the balance in the 11 12 petroleum inspection fund exceeds the amount necessary to make all of the 13 expenditures that are required to be made out of the fund, other than under this 14 paragraph, and, if so, the department of administration shall expend the excess to 15 retire revenue obligations issued under this subsection.". 16 **1666.** Page 1216, line 1: delete lines 1 to 6. **1667.** Page 1216, line 17: after that line insert: 17 18 "Section 2634d. 101.19 (1) (k) of the statutes, as created by 2005 Wisconsin 19 Act 456, is amended to read: 20 101.19 (1) (k) Administering subch. VII, except that the department may not 21 charge a fee for an emergency elevator mechanic's license under s. 101.985 (2) (c) or 22 a conveyance operation permit under s. 101.983 (2) for a platform lift, stairway chair

lift, or any other lift in a private residence. This paragraph applies beginning on

24

Act 456, is amended to read:

1	<b>1668.</b> Page 1218, line 6: after that line insert:
2	"Section 2639c. 101.975 (3) of the statutes is repealed.".
3	<b>1669.</b> Page 1218, line 6: after that line insert:
4	"Section 2640c. 101.981 (1) (c) of the statutes is renumbered 101.981 (1) (c)
5	(intro.) and amended to read:
6	101.981 (1) (c) (intro.) "Conveyance" means an elevator, an escalator, a
7	dumbwaiter, a belt manlift, a moving walkway, a platform lift, a personnel hoist, a
8	material hoist and a stairway chair lift, and any other similar device, such as an
9	automated people mover, used to elevate or move people or things, as provided in the
10	rules of the department. "Conveyance" does not include -a- any of the following:
11	1. A grain elevator -a.
12	2. A ski lift or towing device, or an.
13	3. An amusement or thrill ride.
14	<b>Section 2640f.</b> 101.981 (1) (c) 4. of the statutes is created to read:
15	101.981 (1) (c) 4. An elevator, a dumbwaiter, a platform lift, or a stairway chair
16	lift that is located in an individual residential dwelling unit or any other type of lift
17	that moves people and that is located in such a unit.
18	<b>Section 2640j.</b> 101.981 (1) (c) 5. of the statutes is created to read:
19	101.981 (1) (c) 5. A personnel hoist or a material hoist.
20	<b>Section 2640m.</b> 101.9815 of the statutes is created to read:
21	101.9815 Applicability. Sections 101.983, 101.984, 101.985, and 101.988
22	apply beginning on June 1, 2008.

**Section 2640p.** 101.983(2)(c) of the statutes, as created by 2005 Wisconsin

 $\mathbf{2}$ 

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

101.983 (2) (c) *Inspections*. The department may not issue or renew a permit under this subsection unless the department has received an inspection report for the conveyance issued by an elevator inspector licensed under s. 101.985 (3) indicating that the conveyance complies with this subchapter and any applicable rules promulgated under this subchapter. Upon request of the owner of a private residence containing a newly installed platform lift, stairway chair lift, or residential lift or of the new owner of a private residence containing a previously installed platform lift, stairway chair lift, or residential lift, the department shall inspect the lift or equipment for compliance with this subchapter and any applicable rules promulgated under this subchapter. This inspection by the department does not exempt the owner from the requirement to ensure that the department receives an inspection report from a licensed elevator inspector. Upon performing this inspection, the department shall give the owner notice of relevant conveyance safety requirements and shall instruct the owner as to the procedure for obtaining periodic inspections and renewing the permit under which the lift or equipment is operated. **Section 2640t.** 101.983 (2) (d) of the statutes, as created by 2005 Wisconsin

**SECTION 2640t.** 101.983 (2) (d) of the statutes, as created by 2005 Wisconsin Act 456, is amended to read:

101.983 (2) (d) Term and posting requirements. A permit issued under this subsection has a term of one year, except that a permit applicable to a platform lift, stairway chair lift, or residential lift in a private residence is valid until ownership of the private residence is transferred, at which time the new owner shall apply for renewal of the permit under par. (b). The owner of the building or residence in which a conveyance is located shall display the permit under par. (a) applicable to the conveyance on or in the conveyance or, if applicable, in the machinery room.".

 $\mathbf{2}$ 

**1670.** Page 1220, line 20: after that line insert:

"Section 2641t. 101.988 (3) of the statutes, as created by 2005 Wisconsin Act 456, is amended to read:

101.988 (3) Penalties. Any person who violates this subchapter or rules promulgated under this subchapter may be fined not more than \$1,500 or imprisoned for not more than 30 days or both, except that, notwithstanding s. 939.61 (1), the owner of a private residence in which a conveyance is located may not be fined or required to pay a forfeiture to this state as a result of any violation involving that conveyance.".

- **1671.** Page 1220, line 21: delete the material beginning with that line and ending with page 1221, line 3.
- **1672.** Page 1221, line 4: delete lines 4 to 6.
- **1673.** Page 1221, line 7: delete lines 7 to 16.
- **1674.** Page 1221, line 17: delete lines 17 to 22.
- **1675.** Page 1221, line 23: delete lines 23 and 24.
- **1676.** Page 1221, line 24: after that line insert:
- 17 "Section **2649h.** 103.49 (2m) (b) 1. of the statutes is amended to read:

103.49 (2m) (b) 1. The laborer, worker, mechanic, or truck driver is employed to go to the source of mineral aggregate such as sand, gravel, or stone that is to be immediately incorporated into the work, and not stockpiled or further transported by truck, pick up that mineral aggregate, and deliver that mineral aggregate to the site of a project that is subject to this section by depositing and immediately incorporating the material substantially in place into the work at the final location where the material is to be incorporated into the work, directly or through spreaders

from the transporting vehicle, without the need to pick up and move the material to that final location.

**SECTION 2649j.** 103.50 (2m) (b) 1. of the statutes is amended to read:

103.50 (2m) (b) 1. The laborer, worker, mechanic, or truck driver is employed to go to the source of mineral aggregate such as sand, gravel, or stone that is to be immediately incorporated into the work, and not stockpiled or further transported by truck, pick up that mineral aggregate, and deliver that mineral aggregate to the site of a project that is subject to this section by depositing and immediately incorporating the material substantially in place into the work at the final location where the material is to be incorporated into the work, directly or through spreaders from the transporting vehicle, without the need to pick up and move the material to that final location.".

- **1677.** Page 1221, line 25: delete that line.
- **1678.** Page 1222, line 1: delete lines 1 to 3.
- **1679.** Page 1223, line 12: after that line insert:
- **"Section 2651r.** 110.08 (5) of the statutes is created to read:

110.08 (5) (a) Unless an alternative plan has been approved under par. (b), after the effective date of this paragraph .... [revisor inserts date], the department shall maintain a local examining center in each municipality in which a local examining center was located on December 1, 2006. If the department closed any local examining center in a municipality between December 1, 2006, and the effective date of this paragraph .... [revisor inserts date], and the department maintains no other local examining center in that municipality on the effective date of this paragraph .... [revisor inserts date], the department shall, as soon as possible, open a local

examining center in that municipality. Any local examining center required to be opened under this paragraph may not be closed by the department.

(b) In lieu of maintaining or opening a local examining center in a municipality under par. (a), the department may submit to the joint committee on finance an alternative plan for providing services that would otherwise be provided at the local examining center in the municipality. If the cochairpersons of the joint committee on finance do not notify the department within 14 working days after the date of the department's submittal of the plan that the committee has scheduled a meeting for the purpose of reviewing the plan, the department may implement the plan as proposed. If, within 14 working days after the date of the department's submittal, the cochairpersons of the committee notify the department that the committee has scheduled a meeting for the purpose of reviewing the proposed plan, the department may implement the plan only upon approval of the committee."

**1680.** Page 1223, line 12: after that line insert:

"Section 2651u. 110.08 (1m) of the statutes is amended to read:

110.08 (1m) Each operator's license examiner shall receive informational training on the powers and duties of the department relating to organ donor information under s. 343.175 once every 2 years and, for operator's license examiners hired after January 1, 1997, prior to initial assignment to operator's license examining activities. The informational training under this subsection shall be developed by all organ procurement organizations, as defined in s. 343.01 (2) (dg) 340.01 (41k), in cooperation with the department."

**1681.** Page 1228, line 9: after that line insert:

**"Section 2661h.** 111.70 (1) (ag) of the statutes is created to read:

 $\mathbf{2}$ 

- 111.70 (1) (ag) 1. Except as provided in subds. 2. and 3., if the employer is a municipal employer, "available revenue" means the sum of the allowable property tax levy and payments received under ss. 86.30 and 86.32 and subch. I of ch.79.
- 2. If the employer is a school district, "available revenue" has the same meaning as "revenue" in s. 121.90 (1m).
- 3. If the employer is a technical college district, "available revenue" means the sum of the allowable property tax levy and payments received under s. 38.28.".

**1682.** Page 1228, line 9: after that line insert:

"Section 2663h. 111.70 (1) (dm) of the statutes is amended to read:

111.70 (1) (dm) "Economic issue" means salaries, overtime pay, sick leave, payments in lieu of sick leave usage, vacations, clothing allowances in excess of the actual cost of clothing, length-of-service credit, continuing education credit, shift premium pay, longevity pay, extra duty pay, performance bonuses, health insurance, life insurance, dental insurance, disability insurance, vision insurance, long-term care insurance, worker's compensation and unemployment insurance, social security benefits, vacation pay, holiday pay, lead worker pay, temporary assignment pay, retirement contributions, supplemental retirement benefits, severance or other separation pay, hazardous duty pay, certification or license payment, and limitations on layoffs that create a new or increased financial liability on the employer and contracting or subcontracting of work that would otherwise be performed by municipal employees in the collective bargaining unit with which there is a labor dispute."

**1683.** Page 1228, line 9: after that line insert:

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

"Section 2664d. 111.70 (1) (a) of the statutes is renumbered 111.70 (1) (ar) and amended to read:

111.70 (1) (ar) "Collective bargaining" means the performance of the mutual obligation of a municipal employer, through its officers and agents, and the representative of its municipal employees in a collective bargaining unit, to meet and confer at reasonable times, in good faith, with the intention of reaching an agreement, or to resolve questions arising under such an agreement, with respect to wages, hours and conditions of employment, and with respect to a requirement of the municipal employer for a municipal employee to perform law enforcement and fire fighting services under s. 61.66, except as provided in sub. (4) (m), (n), (o), (p), and (pr) and s. 40.81 (3) and except that a municipal employer shall not meet and confer with respect to any proposal to diminish or abridge the rights guaranteed to municipal employees under ch. 164. The duty to bargain, however, does not compel either party to agree to a proposal or require the making of a concession. Collective bargaining includes the reduction of any agreement reached to a written and signed document. The municipal employer shall not be required to bargain on subjects reserved to management and direction of the governmental unit except insofar as the manner of exercise of such functions affects the wages, hours and conditions of employment of the municipal employees in a collective bargaining unit. In creating this subchapter the legislature recognizes that the municipal employer must exercise its powers and responsibilities to act for the government and good order of the jurisdiction which it serves, its commercial benefit and the health, safety and welfare of the public to assure orderly operations and functions within its jurisdiction, subject to those rights secured to municipal employees by the constitutions of this state and of the United States and by this subchapter.".

**1684.** Page 1228, line 9: after that line insert:

**"Section 2661n.** 111.335 (1) (d) of the statutes is created to read:

111.335 (1) (d) 1. In this paragraph, "educational agency" means a school district, a cooperative educational service agency, a county children with disabilities education board, a state prison under s. 302.01, a juvenile correctional facility, as defined in s. 938.02 (10p), a secured residential care center for children and youth, as defined in s. 938.02 (15g), the Wisconsin Center for the Blind and Visually Impaired, the Wisconsin Educational Services Program for the Deaf and Hard of Hearing, the Mendota Mental Health Institute, the Winnebago Mental Health Institute, a state center for the developmentally disabled, a private school, a charter school, a private, nonprofit, nonsectarian agency under contract with a school board under s. 118.153 (3) (c), or a nonsectarian private school or agency under contract with the board of school directors in a 1st class city under s. 119.235 (1).

- 2. Notwithstanding s. 111.322, it is not employment discrimination because of conviction record for an educational agency to refuse to employ or to terminate from employment an individual who has been convicted of a felony and who has not been pardoned for that felony.".
- **1685.** Page 1228, line 9: after that line insert:
- 19 "Section 2661j. 111.31 (6) of the statutes is created to read:
  - 111.31 **(6)** The legislature finds that the prohibition against discrimination on the basis of arrest or conviction record under s. 111.335 is a matter of statewide concern, requiring uniform enforcement at the state, county, and municipal levels.
    - **SECTION 2661k.** 111.335 (1) (c) 1. of the statutes is amended to read:

111.335 (1) (c) 1. Has been convicted of any felony, misdemeanor or other
offense the circumstances of which substantially relate to the circumstances of the
particular job or licensed activity; or.

**Section 2661L.** 111.335 (1) (c) 3. of the statutes is created to read:

111.335 (1) (c) 3. Has been convicted of a sex offense, as defined in s. 301.45 (1d) (b), or found to have committed a violation of federal law or the law of another state that would be a sex offense, as defined in s. 301.45 (1d) (b), if committed in this state, and who has not been pardoned for that sex offense.

**Section 2661m.** 111.335 (1) (c) 4. of the statutes is created to read:

111.335 (1) (c) 4. Has been convicted of a violent offense, as defined in s. 301.048 (2) (bm) 1., or found to have committed a violation of federal law or the law of another state that would be a violent offense, as defined in s. 301.048 (2) (bm) 1., if committed in this state, and who has not been pardoned for that violent offense.

**Section 26610.** 111.335 (2) of the statutes is created to read:

111.335 (2) No county, city, village, or town may adopt any provision concerning employment discrimination based on arrest or conviction record that prohibits any activity that is allowed under this section.".

**1686.** Page 1228, line 9: after that line insert:

"Section 2661r. 111.35 (3) (a) (intro.) of the statutes is amended to read:

111.35 (3) (a) (intro.) Notwithstanding s. 111.322, it is not employment discrimination because of use of a lawful product off the employer's premises during nonworking hours for an employer, labor organization, employment agency, licensing agency, or other person to offer <u>financial incentives related to employee</u> health care benefits that are intended to discourage use of a lawful product. Those

financial incentives may include offering a policy or plan of life, health, or disability insurance coverage under which the type of coverage or the price of coverage for an individual who uses a lawful product off the employer's premises during nonworking hours differs from the type of coverage or the price of coverage provided for an individual who does not use that lawful product, if all of the following conditions apply:

**Section 2661s.** 111.35 (3) (b) (intro.) of the statutes is amended to read:

111.35 (3) (b) (intro.) Notwithstanding s. 111.322, it is not employment discrimination because of nonuse of a lawful product off the employer's premises during nonworking hours for an employer, labor organization, employment agency, licensing agency, or other person to offer financial incentives related to employee health care benefits that are intended to encourage use of a lawful product. Those financial incentives may include offering a policy or plan of life, health, or disability insurance coverage under which the type of coverage or the price of coverage for an individual who does not use a lawful product off the employer's premises during nonworking hours differs from the type of coverage or the price of coverage provided for an individual who uses that lawful product, if all of the following conditions apply:".

- **1687.** Page 1228, line 10: delete lines 10 to 17.
- **1688.** Page 1228, line 17: after that line insert:
- 21 "Section 2681m. 114.31 (3) (a) of the statutes is renumbered 114.31 (3).
- **Section 2681n.** 114.31 (3) (b) of the statutes is repealed.".
- **1689.** Page 1228, line 17: after that line insert:
- **SECTION 2670b.** 111.70 (4) (cm) 6. am. of the statutes is amended to read:

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

111.70 (4) (cm) 6. am. Upon receipt of a petition to initiate arbitration, the commission shall make an investigation, with or without a formal hearing, to determine whether arbitration should be commenced. If in determining whether an impasse exists the commission finds that the procedures set forth in this paragraph have not been complied with and such compliance would tend to result in a settlement, it may order such compliance before ordering arbitration. The validity of any arbitration award or collective bargaining agreement shall not be affected by failure to comply with such procedures. Prior to the close of the investigation each party shall submit in writing to the commission its single final offer containing its final proposals on all issues in dispute that are subject to interest arbitration under this subdivision or under subd. 5s. in collective bargaining units to which subd. 5s. applies. If, under state law, the municipal employer is subject to limitations on available revenue, a final offer may not require the expenditure for compensation and fringe benefits per employee in any year of the contract to be more than the amount expended in the previous year increased by the allowable percentage increase in available revenue. If the commission determines that a final offer requires greater expenditure, the commission shall return the offer to the party and the party shall revise it before submitting it again. If a party fails to submit a single, ultimate final offer, the commission shall close the investigation based on the last written position of the party. If the last written position requires greater expenditure than permitted under this subd. 6. am., the commission shall consider that the party failed to submit an offer. The municipal employer may not submit a qualified economic offer under subd. 5s. after the close of the investigation. Such final Final offers may include only mandatory subjects of bargaining, except that a permissive subject of bargaining may be included by a party if the other party does not object

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

and shall then be treated as a mandatory subject. No later than such time, the parties shall also submit to the commission a stipulation, in writing, with respect to all matters which are agreed upon for inclusion in the new or amended collective bargaining agreement. The commission, after receiving a report from its investigator and determining that arbitration should be commenced, shall issue an order requiring arbitration and immediately submit to the parties a list of 7 arbitrators. Upon receipt of such list, the parties shall alternately strike names until a single name is left, who shall be appointed as arbitrator. The petitioning party shall notify the commission in writing of the identity of the arbitrator selected. Upon receipt of such notice, the commission shall formally appoint the arbitrator and submit to him or her the final offers of the parties. The final offers shall be considered public documents and shall be available from the commission. In lieu of a single arbitrator and upon request of both parties, the commission shall appoint a tripartite arbitration panel consisting of one member selected by each of the parties and a neutral person designated by the commission who shall serve as a chairperson. An arbitration panel has the same powers and duties as provided in this section for any other appointed arbitrator, and all arbitration decisions by such panel shall be determined by majority vote. In lieu of selection of the arbitrator by the parties and upon request of both parties, the commission shall establish a procedure for randomly selecting names of arbitrators. Under the procedure, the commission shall submit a list of 7 arbitrators to the parties. Each party shall strike one name from the list. From the remaining 5 names, the commission shall randomly appoint an arbitrator. Unless both parties to an arbitration proceeding otherwise agree in writing, every individual whose name is submitted by the commission for appointment as an arbitrator shall be a resident of this state at the time of

submission and every individual who is designated as an arbitration panel chairperson shall be a resident of this state at the time of designation.

**SECTION 2670d.** 111.70 (4) (cm) 6. b. of the statutes is amended to read:

111.70 (4) (cm) 6. b. The arbitrator shall, within 10 days of his or her appointment, establish a date and place for the conduct of the arbitration hearing. Upon petition of at least 5 citizens of the jurisdiction served by the municipal employer, filed within 10 days after the date on which the arbitrator is appointed, the arbitrator shall hold a public hearing in the jurisdiction for the purpose of providing the opportunity to both parties to explain or present supporting arguments for their positions and to members of the public to offer their comments and suggestions. The final offers of the parties, as transmitted by the commission to the arbitrator, shall serve as the basis for continued negotiations, if any, between the parties with respect to the issues in dispute. At any time prior to the arbitration hearing, either party, with the consent of the other party, and subject to the expenditure limit in subd. 6.

**Section 2671b.** 111.70 (4) (cm) 7. of the statutes is amended to read:

111.70 (4) (cm) 7. 'Factor Factors given greatest weight.' In making any decision under the arbitration procedures authorized by this paragraph, the arbitrator or arbitration panel shall consider and shall give the greatest weight to the expenditure limit in subd. 6. am. and any state law or directive lawfully issued by a state legislative or administrative officer, body or agency which places limitations on expenditures that may be made or revenues that may be collected by a municipal employer. The arbitrator or arbitration panel shall give an accounting of the consideration of this factor these factors in the arbitrator's or panel's decision.".

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

**1690.** Page 1228, line 17: after that line insert:

**"Section 2667d.** 111.70 (4) (cm) 5s. of the statutes is amended to read:

111.70 (4) (cm) 5s. 'Issues subject to arbitration.' In a collective bargaining unit consisting of school district professional employees, the municipal employer or the labor organization may petition the commission to determine whether the municipal employer has submitted a qualified economic offer. The commission shall appoint an investigator for that purpose. If the investigator finds that the municipal employer has submitted a qualified economic offer, the investigator shall determine whether a deadlock exists between the parties with respect to all economic issues. If the municipal employer submits a qualified economic offer applicable to any period beginning on or after July 1, 1993, no economic issues are subject to interest arbitration under subd. 6. for that period, except that only the impact of contracting out or subcontracting work that would otherwise be performed by municipal employees in the collective bargaining unit is subject to interest arbitration under subd. 6. In such a collective bargaining unit, economic issues concerning the wages, hours or conditions of employment of the school district professional employees in the unit for any period prior to July 1, 1993, are subject to interest arbitration under subd. 6. for that period. In such a collective bargaining unit, noneconomic issues applicable to any period on or after July 1, 1993, are subject to interest arbitration after the parties have reached agreement and stipulate to agreement on all economic issues concerning the wages, hours or conditions of employment of the school district professional employees in the unit for that period. In such a collective bargaining unit, if the commission's investigator finds that the municipal employer has submitted a qualified economic offer and that a deadlock exists between the parties

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

with respect to all economic issues, the municipal employer may implement the qualified economic offer. On the 90th day prior to expiration of the period included within the qualified economic offer, if no agreement exists on that day, the parties are deemed to have stipulated to the inclusion in a new or revised collective bargaining agreement of all provisions of any predecessor collective bargaining agreement concerning economic issues, or of all provisions of any existing collective bargaining agreement concerning economic issues if the parties have reopened negotiations under an existing agreement, as modified by the terms of the qualified economic offer and as otherwise modified by the parties. In such a collective bargaining unit, on and after that 90th day, a municipal employer that refuses to bargain collectively with respect to the terms of that stipulation, applicable to the 90-day period prior to expiration of the period included within the qualified economic offer, does not violate sub. (3) (a) 4. Any such unilateral implementation after August 11, 1993, during the 90-day period prior to expiration of the period included within a qualified economic offer, operates as a full, final and complete settlement of all economic issues between the parties for the period included within the qualified economic offer. The failure of a labor organization to recognize the validity of such a lawful qualified economic offer does not affect the obligation of the municipal employer to submit economic issues to arbitration under subd. 6.

**Section 2667h.** 111.70 (4) (p) of the statutes is created to read:

111.70 (4) (p) Competitive contracting; all municipal employers. 1. A municipal employer's decision to enter into contracts with persons who are not employed by the municipal employer for the performance of services, as that decision relates to mandatory and permissive subjects of collective bargaining under this subchapter, is subject to this paragraph.

 $\mathbf{2}$ 

- 2. A municipal employer may solicit bids from persons who are not employed by the municipal employer to perform services that are currently being performed by its municipal employees only if the municipal employer does all of the following:
- a. Notifies the labor organization that is recognized or certified to represent the municipal employees who are currently performing the services that the municipal employer intends to solicit the bids.
- b. Conducts an internal cost study to determine the total costs incurred by the municipal employer in having its municipal employees perform the services. The costs determined by this study shall be denominated the "current internal cost." The study shall also determine the percentage of the current internal cost that is attributable to wages and benefits paid to the municipal employees who perform the services and who are represented by the labor organization. This percentage shall be denominated the "labor cost ratio" and shall be expressed as a decimal. The municipal employer shall keep the study confidential until after all bids solicited under subd. 3. have been received by the municipal employer.
- 3. After conducting the study under subd. 2. b., the municipal employer may solicit and receive bids from persons who are not employed by the municipal employer to perform any services that are currently being performed by its municipal employees. These bids shall be denominated the "preliminary external bids" and shall provide for services that are at least substantially similar to those currently being performed by the municipal employer's employees.
- 4. No later than 30 days after receiving the final bid under subd. 3., the municipal employer shall select the preliminary external bid that the municipal employer considers the most advantageous to the municipal employer. The sum of

the cost of this bid and the municipal employer's cost in administering any contract entered into pursuant to the bid shall be denominated the "selected external cost."

5. After determining the selected external cost under subd. 4., the municipal employer shall subtract that number from an amount equal to 90 percent of the current internal cost determined under subd. 2. b. and shall then multiply the result by the labor cost ratio determined under subd. 2. b. The product shall be denominated the "required labor savings." No later than 5 days after selecting the preliminary external bid under subd. 4., the municipal employer shall notify the labor organization that represents the municipal employees of the required labor savings and shall also provide to the labor organization a copy of the preliminary external bid selected under subd. 4.

6. a. If the required labor savings is an amount less than or equal to zero, the municipal employer is required to bargain collectively any decision to enter into contracts with persons who are not employed by the municipal employer for the performance of services for the municipal employer, and the impact of any such decision on the wages, hours, and conditions of employment of the municipal employees who would otherwise perform those services.

b. If the required labor savings is an amount greater than zero, the municipal employer is not required to bargain collectively any decision to enter into contracts with persons who are not employed by the municipal employer for the performance of services for the municipal employer, or the impact of any such decision on the wages, hours, and conditions of employment of the municipal employees who would otherwise perform those services, unless the labor organization that represents the municipal employees notifies the municipal employer, in writing, that the municipal employees agree to be subject to the nonbinding arbitration mechanism under subd.

- 7. The notification must be received by the municipal employer no later than 15 days after the municipal employer selects the preliminary external bid under subd. 4.
- 7. a. No later than 30 days after the municipal employer receives the notification under subd. 6. b., each party shall submit to an arbitrator a proposal to reduce the current internal cost determined under subd. 2. b. by an amount at least equal to the required labor savings determined under subd. 5. The reductions specified in the proposals shall come entirely from changes to the wages, hours, or conditions of employment of the municipal employees who are represented by the labor organization. The arbitrator shall be selected using the process under par. (cm) 6. am. The arbitrator may select any item from each proposal to reduce the current internal cost determined under subd. 2. b. by an amount at least equal to the required labor savings determined under subd. 5.
- b. If the labor organization rejects the arbitrator's proposal, the municipal employer is not required to bargain collectively the decision to contract with persons who are not employed by the municipal employer for the performance of the services for the municipal employer, or the impact of any such decision on the wages, hours, and conditions of employment of the municipal employees who would otherwise perform those services.
- c. If the municipal employer rejects the arbitrator's proposal, the municipal employer is required to bargain collectively the decision to contract with persons who are not employed by the municipal employer for the performance of the services for the municipal employer, and the impact of any such decision on the wages, hours, and conditions of employment of the municipal employees who would otherwise perform those services.

- d. If neither party rejects the arbitrator's proposal within 10 days after the arbitrator presents the proposal to the parties, the proposal shall be final and binding on both parties and shall be incorporated into a collective bargaining agreement.
- e. Any rejection of an arbitrator's proposal under subd. 7. b. or c. shall be made no later than 10 days after the arbitrator presents the proposal to the parties.
- 8. If the arbitrator's proposal is not rejected by either party and is incorporated into a collective bargaining agreement, the municipal employer may not solicit and receive bids from persons who are not employed by the municipal employer to perform the service covered by the arbitrator's proposal for a period of 3 years from the date that the arbitrator submits his or her proposal to the parties.".
  - **1691.** Page 1228, line 17: after that line insert:
- "Section **2666h.** 111.70 (4) (m) (title) of the statutes is amended to read:
- 14 111.70 (4) (m) (title) Prohibited subjects of bargaining; school districts.
- **Section 2666p.** 111.70 (4) (pr) of the statutes is created to read:
  - 111.70 (4) (pr) Prohibited subjects of bargaining; counties. In a county, the municipal employer is prohibited from bargaining collectively with respect to a sheriff's decision to have persons other than municipal employees provide meals to inmates, provided that the sheriff determines that the meals can be provided at a cost lower than if municipal employees provide the meals. The municipal employer is also prohibited from bargaining collectively the impact of that decision on the wages, hours, or conditions of employment of the municipal employees who perform those services."
    - **1692.** Page 1228, line 17: after that line insert:

1	"Section 2680d. 111.91 (2) (nm) of the statutes is created to read:
2	111.91 (2) (nm) The requirements related to continuing coverage for a
3	dependent student on a medical leave of absence under s. 632.895 (15).".
4	<b>1693.</b> Page 1228, line 17: after that line insert:
5	"Section 2665g. 114.09 (title) of the statutes is amended to read:
6	114.09 (title) Reckless Intoxicated and reckless flying; penalty.
7	<b>SECTION 2665h.</b> 114.09 (1) (a) of the statutes is renumbered 114.09 (1) (a)
8	(intro.) and amended to read:
9	114.09 (1) (a) (intro.) In this subsection, "drug":
10	1. "Drug" has the meaning specified in s. 450.01 (10).
11	<b>Section 2665j.</b> 114.09 (1) (a) 2. of the statutes is created to read:
12	114.09 (1) (a) 2. "Prohibited alcohol concentration" means an alcohol
13	concentration of $0.04$ or more if there is no passenger in the aircraft, more than $0.00$
14	if there is a passenger in the aircraft.
15	<b>Section 2665L.</b> 114.09 (1) (b) of the statutes is renumbered 114.09 (1) (b) 1.
16	and amended to read:
17	114.09 (1) (b) 1. No person may operate an aircraft in the air or on the ground
18	or water while under the influence of intoxicating liquor or controlled substances or
19	controlled substance analogs under ch. 961 or a combination thereof, under the
20	influence of any other drug to a degree which renders him or her incapable of safely
21	operating an aircraft, or under the combined influence of intoxicating liquor and any
22	other drug to a degree which renders him or her incapable of safely operating an
23	aircraft <del>, nor</del> .

2. No person may operate an aircraft in the air or on the ground or water in a
careless or reckless manner so as to endanger the life or property of another. In
determining whether the operation was careless or reckless the court shall consider $\frac{1}{2}$
the standards for safe operation of aircraft prescribed by federal statutes or
regulations governing aeronautics.

3. The court shall make a written report of all convictions, including bail or appearance money forfeitures, obtained under this section to the department, which shall send the report to the proper federal agency.

**SECTION 2665n.** 114.09 (1) (b) 1m. of the statutes is created to read:

114.09 (1) (b) 1m. No person may operate an aircraft in the air or on the ground if the person has a prohibited alcohol concentration.

**Section 2665r.** 114.09 (2) of the statutes is repealed and recreated to read:

114.09 (2) (a) Any person violating sub. (1) (b) 1. or 1m.:

- 1. Shall forfeit not less than \$150 nor more than \$300, except as provided in subs. 6. and 7.
- 2. Except as provided in subd. 6., shall be fined not less than \$350 nor more than \$1,100 and imprisoned for not less than 5 days nor more than 6 months if the number of convictions under ss. 940.09 (1) and 940.25 in the person's lifetime, plus the total number of suspensions, revocations, and other convictions counted under s. 343.307 (1) within a 10-year period, equals 2, except that suspensions, revocations, or convictions arising out of the same incident or occurrence shall be counted as one.
- 3. Except as provided in subds. 6. and 7., shall be fined not less than \$600 nor more than \$2,000 and imprisoned for not less than 30 days nor more than one year in the county jail if the number of convictions under ss. 940.09 (1) and 940.25 in the person's lifetime, plus the total number of suspensions, revocations, and other

convictions counted under s. 343.307 (1), equals 3, except that suspensions, revocations, or convictions arising out of the same incident or occurrence shall be counted as one.

- 4. Except as provided in subds. 6. and 7., shall be fined not less than \$600 nor more than \$2,000 and imprisoned for not less than 60 days nor more than one year in the county jail if the number of convictions under ss. 940.09 (1) and 940.25 in the person's lifetime, plus the total number of suspensions, revocations, and other convictions counted under s. 343.307 (1), equals 4, except that suspensions, revocations, or convictions arising out of the same incident or occurrence shall be counted as one.
- 5. Except as provided in subds. 6. and 7., is guilty of a Class H felony and shall be fined not less than \$600 and imprisoned for not less than 6 months if the number of convictions under ss. 940.09 (1) and 940.25 in the person's lifetime, plus the total number of suspensions, revocations, and other convictions counted under s. 343.307 (1), equals 5 or more, except that suspensions, revocations, or convictions arising out of the same incident or occurrence shall be counted as one.
- 6. If there was a minor passenger under 16 years of age in the aircraft at the time of the violation that gave rise to the conviction under sub. (1) (b) 1. or 1m., the applicable minimum and maximum forfeitures, fines, or imprisonment under subd. 1., 2., 3., 4., or 5. for the conviction are doubled. An offense under sub. (1) (b) 1. or 1m., that subjects a person to a penalty under subd. 3., 4., or 5. when there is a minor passenger under 16 years of age in the aircraft is a felony and the place of imprisonment shall be determined under s. 973.02.
- 7. a. If a person convicted had an alcohol concentration of 0.17 to 0.199, the applicable minimum and maximum fines under subd. 3. to 5. are doubled.

- b. If a person convicted had an alcohol concentration of 0.20 to 0.249, the applicable minimum and maximum fines under subd. 3. to 5. are tripled.
- c. If a person convicted had an alcohol concentration of 0.25 or above, the applicable minimum and maximum fines under subd. 3. to 5. are quadrupled.
- (b) In par. (a) 1. to 5., the time period shall be measured from the dates of the refusals or violations that resulted in the revocation or convictions. If a person has a suspension, revocation, or conviction for any offense under a local ordinance or a state statute of another state that would be counted under s. 343.307 (1), that suspension, revocation or conviction shall count as a prior suspension, revocation, or conviction under par. (a) 1. to 5.
- (bm) 1. Except as provided in subd. 1. a. or b., the court shall order the person violating sub. (1) (b) 1. or 1m. to submit to and comply with an assessment by an approved public treatment facility as defined in s. 51.45 (2) (c) for examination of the person's use of alcohol, controlled substances, or controlled substance analogs and development of an airman safety plan for the person. The court shall notify the person, the department, and the proper federal agency of the assessment order. The assessment order shall:
- a. If the person is a resident, refer the person to an approved public treatment facility in the county in which the person resides. The facility named in the order may provide for assessment of the person in another approved public treatment facility. The order shall provide that, if the person is temporarily residing in another state, the facility named in the order may refer the person to an appropriate treatment facility in that state for assessment and development of an airman safety plan for the person satisfying the requirements of that state.

 $\mathbf{2}$ 

- b. If the person is a nonresident, refer the person to an approved public treatment facility in this state. The order shall provide that the facility named in the order may refer the person to an appropriate treatment facility in the state in which the person resides for assessment and development of an airman safety plan for the person satisfying the requirements of that state.
- c. Require a person who is referred to a treatment facility in another state under subd. 1. a. or b. to furnish the department written verification of his or her compliance from the agency that administers the assessment and airman safety plan program. The person shall provide initial verification of compliance within 60 days after the date of his or her conviction. The requirement to furnish verification of compliance may be satisfied by receipt by the department of such verification from the agency that administers the assessment and airman safety plan program.
- 2. The department of health and family services shall establish standards for assessment procedures and the airman safety plan programs by rule. The department of health and family services shall establish by rule conflict of interest guidelines for providers.
- 3. Prior to developing a plan that specifies treatment, the facility shall make a finding that treatment is necessary and appropriate services are available. The facility shall submit a report of the assessment and the airman safety plan within 14 days to the county department under s. 51.42, the plan provider, the department of transportation, the appropriate federal agency, and the person, except that, upon request by the facility and the person, the county department may extend the period for assessment for not more than 20 additional workdays. The county department shall notify the department of transportation regarding any such extension.

- 4. The assessment report shall order compliance with an airman safety plan. The report shall inform the person of the fee provisions under s. 46.03 (18) (f). The safety plan may include a component that makes the person aware of the effect of his or her offense on a victim and a victim's family. The safety plan may include treatment for the person's misuse, abuse, or dependence on alcohol, controlled substances, or controlled substance analogs. If the plan requires inpatient treatment, the treatment shall not exceed 30 days. An airman safety plan under this paragraph shall include a termination date consistent with the plan that shall not extend beyond one year. The county department under s. 51.42 shall assure notification of the department of transportation and the person of the person's compliance or noncompliance with assessment and treatment.
  - (c) Any person violating sub. (1) (b) 2.:
- 1. May be required to forfeit not less than \$25 nor more than \$200, except as provided in subd. 2.
- 2. May be fined not less than \$50 nor more than \$500 or imprisoned for not more than one year in the county jail or both if the total of convictions under sub. (1) (b) 2. equals 2 or more in a 4-year period. The 4-year period shall be measured from the dates of the violations that resulted in the convictions.".
  - **1694.** Page 1228, line 17: after that line insert:
- 20 "Section 2677b. 111.70 (4) (cm) 8s. of the statutes is renumbered 111.70 (4) (cm) 8s. a.
  - **Section 2677c.** 111.70 (4) (cm) 8s. b. of the statutes is created to read:
- 23 111.70 (4) (cm) 8s. b. If a school district unilaterally changes its employees' 24 health care coverage plan provider under par. (n), any costs savings realized because

 $\mathbf{2}$ 

of the change may not be included when determining the fringe benefit savings under subd. 8s. a.

**SECTION 2679f.** 111.70 (4) (n) of the statutes is created to read:

111.70 (4) (n) Municipal employer-initiated change in health care coverage plan provider. Notwithstanding the terms of a collective bargaining agreement, a municipal employer may unilaterally change its employees' health care coverage plan provider without the consent of any affected employee in the collective bargaining unit if the benefits provided by the new health care coverage plan provider are substantially similar to those provided by the former health care coverage plan provider and if either the persons who provide health care coverage under the new plan are the same as under the former plan or cost savings will result from changing the health care coverage plan provider. Any such unilateral change in health care coverage plan provider is not a violation of a collective bargaining agreement or a prohibited practice under sub. (3) (a) and, for purposes of a qualified economic offer, satisfies the requirement to maintain fringe benefits under sub. (1) (nc).

**Section 2679g.** 111.70 (4) (o) of the statutes is created to read:

111.70 (4) (o) *Prohibited subject of collective bargaining*. A municipal employer is prohibited from bargaining collectively with respect to the employer's selection of a health care coverage plan if the municipal employer offers to enroll the employees in a health care coverage plan under s. 40.51 (7) or in a health care coverage plan that is substantially similar to a plan offered under s. 40.51 (7). The commission shall use the criteria in rules promulgated by the commissioner of insurance under s. 601.41 (12) to determine if health care coverage plans are substantially similar.".

1	<b>1695.</b> Page 1228, line 17: after that line insert:
2	"Section 2671gp. 111.91 (2) (pj) of the statutes is created to read:
3	111.91 (2) (pj) The number of paid sick leave days in any year specified under
4	s. 230.35 (2) (am).".
5	<b>1696.</b> Page 1228, line 17: after that line insert:
6	"Section 2669p. 111.91 (2) (ph) of the statutes is created to read:
7	111.91 (2) (ph) A decision to convert from a biweekly pay period to a monthly
8	pay period.".
9	<b>1697.</b> Page 1229, line 11: after that line insert:
10	<b>"Section 2684p.</b> 115.28 (44) of the statutes is created to read:
11	115.28 (44) Teacher Merit Pay. Distribute the amounts appropriated under s
12	20.255 (2) (am) to school districts to assist in paying the costs of teacher merit pay
13	programs. The department shall promulgate rules to implement and administer this
14	subsection.".
15	<b>1698.</b> Page 1229, line 11: after that line insert:
16	"Section 2683p. 115.28 (46m) of the statutes is created to read:
17	115.28 (46m) School safety grants. From the appropriation under s. 20.255
18	(2) (dn), annually award grants to reimburse school districts for costs allowable
19	under the federal Safe and Drug-Free Schools and Communities Act, 20 USC 7103
20	to 7165, but not paid under that act. If the appropriation under s. $20.255\ (2)\ (dn)$ in
21	any fiscal year is insufficient to pay all claims under this subsection, the funds shal
22	be prorated among the school districts entitled to the funds.".
23	<b>1699.</b> Page 1229, line 11: after that line insert:

**"Section 2684.** 115.28 (46) of the statutes is created to read:

 $\mathbf{2}$ 

- 115.28 (46) Grants for science, technology, engineering, and mathematics programs. From the appropriation under s. 20.255 (2) (fz), award grants to school districts to develop innovative instructional programs in science, technology, engineering and mathematics; support pupils who are typically under-represented in these subjects; and increase the academic achievement of pupils in those subjects.".
- **1700.** Page 1229, line 12: delete lines 12 to 22.
- **1701.** Page 1229, line 23: delete the material beginning with that line and ending with page 1230, line 4.
- **1702.** Page 1230, line 5: delete lines 5 to 23.
- **1703.** Page 1231, line 1: delete lines 1 to 11.
- **1704.** Page 1231, line 12: delete the material beginning with that line and ending with page 1232, line 10.
- **1705.** Page 1232, line 11: delete the material beginning with that line and ending with page 1233, line 16.
- **1706.** Page 1234, line 12: delete lines 12 to 25.
- **1707.** Page 1235, line 1: delete that line.
- **1708.** Page 1235, line 13: delete lines 13 to 25.
- **1709.** Page 1236, line 1: delete lines 1 to 25.
- **1710.** Page 1237, line 1: delete lines 1 to 5.
- **1711.** Page 1237, line 5: after that line insert:
- 22 "Section 2711d. 115.85 of the statutes is created to read:

- 115.85 Autism Scholarship Program. (1) In this section, "eligible autistic child" means a child with a disability to whom all the following apply:
- (a) The school district in which the child resides or that the child is attending has identified the child as autistic.
- (b) The school district in which the child resides or that the child is attending has in effect for the child an individualized education program.
- (c) The child was enrolled in public school in the school year prior to the school year in which a scholarship under this section is first sought for the child, or the child is eligible to enter public school in the school year in which a scholarship under this section is first sought for the child.
- (2) (a) Except as provided in par. (b), upon receipt of an application from the parent of an eligible autistic child, the department shall award a scholarship to the child. The scholarship may be used only to pay tuition for the child to attend a special education program that implements the child's individualized education program and that is operated by a school district other than the school district that the child is attending or in which the child resides, by another public entity, or by a private provider approved by the department under sub. (3). The amount of the scholarship shall be \$15,000 or the actual tuition charged by the special education program, whichever is less.
- (b) No more than 200 scholarships may be awarded under this section in any school year.
- (3) The department shall establish standards for the approval of private providers for the purpose of sub. (2). The department may not approve a private provider unless the person providing special education or related services holds the appropriate license issued by the department.

1	(4) The department shall promulgate rules to implement and administer this
2	section, including procedures and deadlines for scholarship applications, payment
3	schedules for scholarships, and standards for the approval of private providers under
4	sub. (3).
5	<b>Section 2711e.</b> $115.88 (10)$ of the statutes is created to read:
6	115.88 (10) Autism scholarship program. Notwithstanding subs. (1m) to (3),
7	(6), and $(8)$ and ss. $115.93$ and $118.255$ $(4)$ , annually the department shall allocate
8	\$3,000,000 from the appropriation under s. 20.255 (2) (b) for autism scholarships
9	under s. 115.85.
10	<b>Section 2711f.</b> 115.882 of the statutes is amended to read:
11	115.882 Payment of state aid. Funds appropriated under s. 20.255 (2) (b)
12	shall be used first for the purpose of s. $115.88(4)$ and then for the purpose of s. $115.88$
13	(10). Costs eligible for reimbursement from the appropriation under s. $20.255~(2)~(b)$
14	under ss. 115.88 (1m) to (3), (6) and (8), 115.93, and 118.255 (4) shall be reimbursed
15	at a rate set to distribute the full amount appropriated for reimbursement for the
16	costs, not to exceed 100%.".
17	<b>1712.</b> Page 1237, line 5: after that line insert:
18	"Section 2713m. 118.15 (1) (cv) of the statutes is created to read:
19	118.15 (1) (cv) Attendance at a special education program operated by a public
20	entity or private provider under a scholarship awarded under s. 115.85 may be
21	substituted for attendance at a public or private school.".
22	<b>1713.</b> Page 1237, line 5: after that line insert:

"Section 2719m. 118.35 (4) of the statutes is amended to read:

118.35 (4) From the appropriation under s. $20.255$ (2) (fy), the department shall
award grants to nonprofit organizations, cooperative educational service agencies,
and the school district operating under ch. 119 for the purpose of providing advanced
curriculum and assessments for gifted and talented middle school pupils.".
1714. Page 1237, line 6: delete the material beginning with that line and
ending with page 1239, line 4, and substitute:
"Section 2717L. 118.43 (3) (intro.) of the statutes is amended to read:
118.43 (3) Contract requirements. (intro.) Except as provided in pars. (am)
and (ar) and sub. (4m), an achievement guarantee contract shall require the school
board to do all of the following in each participating school:
<b>Section 2717p.</b> 118.43 (4m) of the statutes is created to read:
118.43 (4m) Exceptions. A school district participating in the program under
this section may disregard the requirement to reduce class size to 15 in grades 2 or
3, or both, in any school.
Section 2717t. 118.43 (6) (b) (intro.) of the statutes is amended to read:
118.43 (6) (b) (intro.) From the appropriations appropriation under s. 20.255
(2) (cu) and (cv), subject to par. (c), the department shall pay to each school district
that has entered into a contract with the department under this section an amount
determined as follows:

**SECTION 2717v.** 118.43 (6) (b) 9. of the statutes is amended to read:

118.43 **(6)** (b) 9. In the 2005–06 and 2006–07 school years, \$2,000 multiplied by the number of low-income pupils enrolled in grades eligible for funding in each school in the school district covered by renewals of contracts under sub. (2) (g); and in the 2007–08 school year and any subsequent school year, \$2,250 multiplied by the

17

18

19

20

21

- 1 number of low-income pupils enrolled in grades eligible for funding in each school 2 in the school district covered by renewals of contracts under sub. (2) (g). A grade is 3 eligible for funding only if the school board has reduced class size in that grade to <u>15.</u>". 4 **1715.** Page 1239, line 5: delete lines 5 to 25.
- 5
- **1716.** Page 1240, line 1: delete lines 1 to 20. 6
- **1717.** Page 1240, line 20: after that line insert: 7
- 8 "Section 2731d. 118.53 of the statutes is created to read:
- 118.53 Racine County Parental Choice Program. (1) In this section: 9
- 10 (a) "Membership" has the meaning given in s. 121.004 (5).
- 11 (b) "Poverty line" means the federal poverty line as defined under 42 USC 9902 (2).12
  - (c) "Resident school board" means the school board of a resident school district.
- 14 (d) "Resident school district" means the school district in which a pupil resides.
- 15 (e) "Summer average daily membership equivalent" has the meaning given in 16 s. 121.004 (8).
  - (f) "Summer choice average daily membership equivalent" means the summer average daily membership equivalent of pupils who were attending a private school under this section on the 2nd Friday of January of the school term immediately preceding that summer or whose applications have been accepted under sub. (3) for attendance at the private school in the school term immediately following that summer.
- 23 (g) "Teacher" means a person who has primary responsibility for the academic 24 instruction of pupils.

 $\mathbf{2}$ 

- (2) Beginning in the 2009–10 school year, any pupil in grades kindergarten to 12 who resides within Racine County may attend any private school located in the county if all of the following apply:
- (a) The private school notified the state superintendent of its intent to participate in the program under this section by February 1 of the previous school year. The notice shall specify the number of pupils participating in the program under this section for which the school has space.
  - (b) The private school complies with 42 USC 2000d.
- (c) The private school meets all health and safety laws or codes that apply to public schools.
- (d) All of the private school's teachers have graduated from high school or been granted a declaration of equivalency of high school graduation.
- (e) The private school achieves accreditation by the Wisconsin North Central Association, the Wisconsin Religious and Independent Schools Accreditation, the Independent Schools Association of the Central States, or any other organization recognized by the National Council for Private School Accreditation, by December 31 of the 3rd school year following the first school year in which the private school participates in the program under this section.
- (3) (a) The pupil or the pupil's parent or guardian shall submit an application, on a form provided by the state superintendent, to the participating private school that the pupil wishes to attend. The pupil or the pupil's parent or guardian shall submit with the application sufficient documentation, as determined by the state superintendent by rule, for the participating private school to verify family income and to determine payments to be made on behalf of the pupil under subs. (4) and (4m).

- (b) Within 60 days after receiving the application, the private school shall notify the applicant, in writing, whether the application has been accepted. The state superintendent shall ensure that the private school determines which pupils to accept on a random basis, except that the private school may give preference in accepting applications to siblings of pupils accepted on a random basis.
- (c) If the private school rejects an applicant because it has too few available spaces, the pupil may transfer his or her application to a participating private school that has space available.
- (d) The pupil or the pupil's parent or guardian shall submit proof of the pupil's enrollment in the private school to the state superintendent prior to the school term.
- (4) (a) Annually, on or before October 15, a private school participating in the program under this section shall file with the department a report stating its summer average daily membership equivalent and its summer choice average daily membership equivalent for the purpose of sub. (4m).
- (b) Upon receipt of proof of enrollment under sub. (3) (d), the state superintendent shall pay to the parent or guardian, from the appropriation under s. 20.255 (2) (fr), an amount equal to the lesser of the following:
- 1. The amount equal to the private school's operating and debt service cost per pupil that is related to educational programming, as determined by the department.
- 2. a. If the pupil is a member of a family that has a total family income that is no more than 1.75 times the poverty line, \$6,500 in the 2009–10 school year, and, in each subsequent school year, the amount paid per pupil under this subd. 2. a. in the previous school year multiplied by the sum of 1.0 plus the percentage change from the previous school year to the current school year in the total amount appropriated under s. 20.255 (2) (ac) expressed as a decimal, but not less than zero.

 $\mathbf{2}$ 

- b. If the pupil is a member of a family that has a total family income that is more than 1.75 times the poverty line but no more than 2.25 times the poverty line, \$5,000 in the 2009–10 school year, and, in each subsequent school year, the amount paid per pupil under this subd. 2. b. in the previous school year multiplied by the sum of 1.0 plus the percentage change from the previous school year to the current school year in the total amount appropriated under s. 20.255 (2) (ac) expressed as a decimal, but not less than zero.
- c. If the pupil is a member of a family that has a total family income that is more than 2.25 times the poverty line but no more than 2.50 times the poverty line, \$3,500 in the 2009–10 school year, and, in each subsequent school year, the amount paid per pupil under this subd. 2. c. in the previous school year multiplied by the sum of 1.0 plus the percentage change from the previous school year to the current school year in the total amount appropriated under s. 20.255 (2) (ac) expressed as a decimal, but not less than zero.
- d. If the pupil is a member of a family that has a total family income that is more than 2.50 times the poverty line but no more than 2.75 times the poverty line, \$2,000 in the 2009–10 school year, and, in each subsequent school year, the amount paid per pupil under this subd. 2. d. in the previous school year multiplied by the sum of 1.0 plus the percentage change from the previous school year to the current school year in the total amount appropriated under s. 20.255 (2) (ac) expressed as a decimal, but not less than zero.
- e. If the pupil is a member of a family that has a total family income that is more than 2.75 times the poverty line, \$1,000 in the 2009–10 school year, and, in each subsequent school year, the amount paid per pupil under this subd. 2. e. in the previous school year multiplied by the sum of 1.0 plus the percentage change from

the previous school year to the current school year in the total amount appropriated under s. 20.255 (2) (ac) expressed as a decimal, but not less than zero.

- (c) The state superintendent shall pay 25 percent of the total amount under par. (b) in September, 25 percent in November, 25 percent in February, and 25 percent in May. The state superintendent may include the entire amount under sub. (4m) in one of those installments or apportion the entire amount among one or more of those installments. The department shall send the check to the private school. The parent or guardian shall restrictively endorse the check for the use of the private school.
- (4m) In addition to the payment under sub. (4) the state superintendent shall pay to the parent or guardian of each pupil enrolled in a private school under this section, in the manner described in sub. (4) (c), an amount determined by multiplying 40 percent of the payment under sub. (4) by the quotient determined by dividing the summer choice average daily membership equivalent of the private school by the total number of pupils for whom payments are being made under sub. (4).
- (5) The state superintendent shall ensure that pupils and parents and guardians of pupils who reside in Racine County are informed annually of the private schools participating in the program under this section.
- (6) The resident school board shall provide transportation to pupils of the resident school district attending a private school under this section if required under s. 121.54 and may claim transportation aid under s. 121.58 for pupils so transported.
- (7) (a) Each private school participating in the program under this section shall meet at least one of the following standards:
- 1. At least 70 percent of the pupils in the program advance one grade level each year.

- 2. The private school's average attendance rate for the pupils in the program is at least 90 percent.
- 3. At least 80 percent of the pupils in the program demonstrate significant academic progress.
- 4. At least 70 percent of the families of pupils in the program meet parent involvement criteria established by the private school.
- (am) Each private school participating in the program under this section is subject to uniform financial accounting standards established by the department. Annually by September 1 following a school year in which a private school participated in the program under this section, the private school shall submit to the department all of the following:
- 1. An independent financial audit of the private school conducted by a certified public accountant, accompanied by the auditor's statement that the audit report is free of material misstatements and fairly presents pupil costs under sub. (4) (b) 1. The audit under this subdivision shall be limited in scope to those records that are necessary for the department to make payments under subs. (4) and (4m).
  - 2. Evidence of sound fiscal practices, as prescribed by the department by rule.
- (c) A private school may not require a pupil attending the private school under this section to participate in any religious activity if the pupil's parent or guardian submits to the pupil's teacher or the private school's principal a written request that the pupil be exempt from such activities.
- (d) By August 1 before the first school term of participation in the program, or by May 1 if the private school begins participating in the program during summer school, each private school participating in the program under this section shall submit to the department all of the following:

- 1. A copy of the school's current certificate of occupancy issued by the city, town, or village in which the school is located. If the private school moves to a new location, the private school shall submit a copy of the new certificate of occupancy issued by the city, town, or village in which the school is located to the department before the attendance of pupils at the new location and before the next succeeding date specified in s. 121.05 (1) (a). A temporary certificate of occupancy does not meet the requirement of this subdivision.
  - 2. Evidence of financial viability, as prescribed by the department by rule.
- 3. Proof that the private school's administrator has participated in a fiscal management training program approved by the department.
- (e) Annually, each private school participating in the program under this section shall administer a nationally normed standardized test in reading, mathematics, and science to pupils attending the school under the program in the 4th, 8th, and 10th grades. The private school may administer additional standardized tests to such pupils.
- (8) There is created a pupil assignment council composed of one representative from each private school participating in the program under this section. Annually by June 30, the council shall make recommendations to the participating private schools to achieve, to the extent possible, a balanced representation of pupils participating in the program under this section.
- (9) If any accrediting agency specified under sub. (2) (e) determines during the accrediting process that a private school does not meet all of the requirements under s. 118.165 (1), it shall report that failure to the department.
- (10) (a) The state superintendent may issue an order barring a private school from participating in the program under this section in the current school year if the

 $\mathbf{2}$ 

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

- state superintendent determines that the private school has done any of the following:
  - 1. Misrepresented information required under sub. (7) (d).
  - 2. Failed to provide the notice required under sub. (2) (a), or the information required under sub. (7) (am) or (d), by the date or within the period specified.
    - 3. Failed to refund to the state any overpayment made under sub. (4) (b) or (4m) by the date specified by department rule.
    - 4. Failed to meet at least one of the standards under sub. (7) (a) by the date specified by department rule.
    - (am) If the state superintendent determines that any of the following have occurred, he or she may issue an order barring the private school from participating in the program under this section in the following school year:
    - 1. The private school's application for accreditation has been denied by the accrediting organization.
    - 2. The private school has not achieved accreditation within the period allowed under sub. (2) (e).
    - (b) The state superintendent may issue an order immediately terminating a private school's participation in the program under this section if he or she determines that conditions at the private school present an imminent threat to the health or safety of pupils.
    - (c) Whenever the state superintendent issues an order under par. (a), (am), or (b), he or she shall immediately notify the parent or guardian of each pupil attending the private school under this section.

1	(d) The state superintendent may withhold payment from a parent or guardian
2	under subs. (4) and (4m) if the private school attended by the child of the parent or
3	guardian violates this section.
4	(11) The department shall promulgate rules to implement and administer this
5	section.".
6	<b>1718.</b> Page 1240, line 20: after that line insert:
7	"Section 2732L. 118.55 (7t) (a) of the statutes is amended to read:
8	118.55 (7t) (a) A school board may establish a written policy limiting the
9	number of credits for which the school board will pay under subs. (5) and (7r) (d) to
10	the equivalent of $18 \ \underline{9}$ postsecondary semester credits per pupil.".
11	<b>1719.</b> Page 1240, line 24: delete "115.445,".
12	1720. Page 1241, line 5: after that line insert:
13	"Section 2733m. 119.16 (11) of the statutes is created to read:
14	119.16 (11) Residency requirements. (a) In this subsection, "teacher" means
15	any person holding a license or permit issued by the state superintendent whose
16	employment by a school district requires that he or she hold that license or permit.
17	(b) The board may not require, as a condition of employment, that a teacher
18	reside within the district.".
19	1721. Page 1241, line 5: after that line insert:
20	"Section 2733m. 119.23 (2) (a) (intro.) of the statutes is amended to read:
21	119.23 (2) (a) (intro.) Subject to par. (b), any pupil in grades kindergarten to 12
22	who resides within the city may attend, at no charge, any grade from kindergarten
23	to 12 at a private school located in the city, and any pupil in grades 9 to 12 who resides

within the city may attend, at no charge, any grade from 9 to 12 at a private school located outside the city but within Milwaukee County, if all of the following apply:".

**1722.** Page 1241, line 6: delete lines 6 to 18.

**1723.** Page 1241, line 18: after that line insert:

**"Section 2734m.** 119.23 (7) (d) 1. of the statutes is amended to read:

119.23 (7) (d) 1. A copy of the school's current certificate of occupancy issued by the city, village, or town in which the private school is located. If the private school moves to a new location, the private school shall submit to the department a copy of the new certificate of occupancy issued by the city to the department, village, or town in which the private school is located before the attendance of pupils at the new location and before the next succeeding date specified in s. 121.05 (1) (a). A temporary certificate of occupancy does not meet the requirement of this subdivision."

**1724.** Page 1241, line 18: after that line insert:

**"Section 2734q.** 119.23 (7) (e) 1. of the statutes is amended to read:

119.23 (7) (e) 1. Annually, each private school participating in the program under this section shall administer a nationally normed standardized test in reading, mathematics, and science or an examination adopted or approved by the state superintendent under s. 118.30 (1) to pupils attending the school under the program in the 4th, 8th, and 10th grades. The private school may administer additional standardized tests to such pupils. Beginning in 2006 and annually thereafter until 2011, the private school shall provide the scores of all standardized tests that it administers to the School Choice Demonstration Project.".

**1725.** Page 1241, line 18: after that line insert:

"Section 2734d. 119.23 (2) (b) of the statutes is renumbered 119.23 (2) (b) 1. and amended to read:

119.23 (2) (b) 1. No In the 2007–08 school year, no more than 22,500 pupils, as counted under s. 121.004 (7), may attend private schools under this section. In the 2008–09 school year and in each school year thereafter, the total number of pupils, as counted under s. 121.004 (7), who may attend private schools under this section is equal to the number of pupils permitted to attend private schools under this section in the previous school year plus 2,250.

2. Whenever the state superintendent determines that the limit <u>under subd.</u>

1. is reached, he or she shall issue an order prohibiting the participating private schools from accepting additional pupils until he or she determines that the number of pupils attending private schools under this section has fallen below the limit.".

**1726.** Page 1241, line 18: after that line insert:

"Section 2735p. 120.12 (15) of the statutes is amended to read:

120.12 (15) School hours. Establish rules scheduling the hours of a normal school day. The school board may differentiate between the various elementary and high school grades in scheduling the school day. The equivalent of 180 such days, as defined in s. 115.01 (10), shall be held during the school term. This subsection shall not be construed to eliminate a school district's duty to bargain with the employee's collective bargaining representative over any calendaring proposal which is primarily related to wages, hours and conditions of employment.".

- **1727.** Page 1241, line 19: delete lines 19 to 25.
- **1728.** Page 1242, line 1: delete lines 1 to 14.
- **1729.** Page 1242, line 14: after that line insert:

23

<del>180</del>.

"Section 2737p. 120.13 (2) (g) of the statutes is amended to read: 1  $\mathbf{2}$ 120.13 (2) (g) Every self-insured plan under par. (b) shall comply with ss. 3 49.493 (3) (d), 631.89, 631.90, 631.93 (2), 632.746 (10) (a) 2. and (b) 2., 632.747 (3), 4 632.85, 632.853, 632.855, 632.87 (4) and (5), 632.895 (9) to (14) (15), 632.896, and 5 767.25 (4m) (d).". 6 **1730.** Page 1242, line 14: after that line insert: 7 **"Section 2738d.** 120.18 (1) (d) of the statutes is amended to read: 120.18 (1) (d) The number of school days taught, including holidays, and the 8 9 number of hours of direct pupil instruction provided in each school, by teachers 10 legally qualified to teach. 11 **Section 2738i.** 121.004 (5m) of the statutes is created to read: 12 121.004 (5m) MINIMUM REQUIRED HOURS OF INSTRUCTION. "Minimum required 13 hours of instruction" means the number of hours of direct pupil instruction required for the grade in which a pupil is enrolled as specified in s. 121.02 (1) (f). 14 15 **Section 2738L.** 121.004 (7) (c) 1. b. of the statutes is amended to read: 16 121.004 (7) (c) 1. b. A pupil enrolled in a 5-year-old kindergarten program 17 requiring full-day attendance for less than 5 days a week for an entire school year 18 term shall be counted as the result obtained by multiplying the number of hours in 19 each day in which the pupil is enrolled by the total number of days for which the pupil 20 is enrolled, and dividing the result by the product of the total number of hours of 21 attendance per day required of first grade pupils in the school district multiplied by

**Section 2738n.** 121.004 (7) (cm) of the statutes is amended to read:

1	121.004 (7) (cm) A pupil enrolled in a 4-year-old kindergarten program that
2	provides the minimum required number of hours of direct pupil instruction under s.
3	121.02 (1) (f) 2. shall be counted as 0.6 pupil if the program annually provides at least
4	87.5 additional hours of outreach activities.
5	<b>Section 2738p.</b> 121.006 (2) (a) of the statutes is renumbered 121.006 (2) (a)
6	(intro.) and amended to read:
7	121.006 (2) (a) (intro.) Hold school for at least 180 days each year the minimum
8	required hours of instruction, less any days of the following:
9	1. Hours during which the state superintendent determines that school is not
10	held or educational standards are not maintained as the result of a strike by school
11	district employees, the days to be computed in accordance with s. $115.01 (10)$ .
12	<b>Section 2738r.</b> 121.006 (2) (a) 2. of the statutes is created to read:
13	121.006 (2) (a) 2. Hours during which school is closed by order of a local health
14	officer, as defined in s. 250.01 (5).".
15	1731. Page 1242, line 14: after that line insert:
16	"Section 2739c. 121.004 (7) (f) of the statutes is renumbered 121.004 (7) (f) 1.
17	and amended to read:
18	121.004 (7) (f) 1. —A—In the 2007-08 and 2008-09 school years, a pupil who
19	transfers from one school district to another under s. 121.85 (3) (a) shall be counted
20	by the school district in which the pupil resides as 0.75 pupil or, if appropriate, as a
21	number equal to the result obtained by multiplying 0.75 by the appropriate fraction
22	under par. (c), (cm) or (d).
23	<b>SECTION 2739e.</b> 121.004 (7) (f) 2. of the statutes is created to read:

121.004 (7) (f) 2. Beginning with the 2009–10 school year and ending with the 2014–15 school year, a pupil who transfers from one school district to another under s. 121.85 (3) (a) shall be counted by the school district in which the pupil resides as 0.675 pupil or, if appropriate, as a number equal to the result obtained by multiplying 0.675 by the appropriate fraction under par. (c), (cm) or (d).".

**1732.** Page 1242, line 15: delete lines 15 to 21.

**1733.** Page 1242, line 21: after that line insert:

**"Section 2740m.** 121.05 (1) (a) 4. of the statutes is created to read:

121.05 **(1)** (a) 4. Pupils residing in the school district who are attending a special education program operated by another school district, another public entity, or a private provider as a result of receiving a scholarship under s. 115.85.".

**1734.** Page 1242, line 21: after that line insert:

**Section 2740e.** 121.02 (1) (f) 1. of the statutes is repealed.

**SECTION 2740g.** 121.02 (1) (f) 2. of the statutes is renumbered 121.02 (1) (f) and amended to read:

121.02 (1) (f) Annually, schedule at least 437 hours of direct pupil instruction in kindergarten, at least 1,050 hours of direct pupil instruction in grades 1 to 6 and at least 1,137 hours of direct pupil instruction in grades 7 to 12. Scheduled hours under this subdivision paragraph include recess and time for pupils to transfer between classes but do not include the lunch period. Scheduled hours under this paragraph may include no more than 35 hours on Saturdays. A school board operating a 4-year-old kindergarten program may use up to 87.5 of the scheduled hours for outreach activities.".

1735. Page 1242, line 21: after that line insert:

**"Section 2740g.** 121.08 (4) (a) 2. of the statutes is amended to read: 1 2 121.08 (4) (a) 2. Divide the sum under subd. 1. by the total amount of state aid 3 that all school districts are eligible to be paid from the appropriations appropriation 4 under s. 20.255 (2) (ac) and (r), calculated as if the reduction reductions under par. pars. (b) and (br) had not occurred. 5 6 **Section 2740s.** 121.08 (4) (a) 3. of the statutes is amended to read: 7 121.08 (4) (a) 3. Multiply the amount of state aid that the school district is 8 eligible to be paid from the appropriations appropriation under s. 20.255 (2) (ac) and 9 (r), calculated as if the reduction reductions under par. pars. (b) and (br) had not 10 occurred, by the quotient under subd. 2.". **1736.** Page 1242, line 21: after that line insert: 11 12 **"Section 2740f.** 121.02 (1) (k) 4. of the statutes is created to read: 13 121.02 (1) (k) 4. If the school board operates a 4-year-old kindergarten 14 program, use model learning standards developed by the department, instruct 15 pupils enrolled in the program based on the model standards, and maintain a 16 written, sequential curriculum plan for the program.". **1737.** Page 1242, line 22: delete lines 22 to 25. 17 18 **1738.** Page 1243, line 1: delete lines 1 to 15. **1739.** Page 1243, line 15: after that line insert: 19 20 "Section 2744h. 121.23 (2) (intro.) of the statutes is amended to read: 21 121.23 (2) (intro.) If a school district holds less than 180 days of school the 22 minimum required hours of instruction as the result of a strike by school district 23 employees, for the purposes of computing general aid, the state superintendent shall

compute the school district's primary and secondary ceiling costs per member in

1	accordance with the procedure specified in pars. (a) to (e). In making the calculation,
2	the state superintendent shall:".
3	<b>1740.</b> Page 1243, line 15: after that line insert:
4	"Section 2744d. 121.08 (4) (br) of the statutes is created to read:
5	121.08 (4) (br) The amount of state aid that a school district located in whole
6	or in part in Racine County is eligible to be paid from the appropriation under 20.255
7	(2) (ac) shall also be reduced by 45 percent of the sum of the amounts paid under s.
8	118.53 (4) and (4m) in the current school year for pupils who reside in the school
9	district and attend a private school under s. 118.53 in the current school year.
10	Section 2744f. 121.08 (4) (d) of the statutes is amended to read:
11	121.08 (4) (d) The state superintendent shall ensure that the total amount of
12	aid reduction under pars. (a) and, (b), and (br) lapses to the general fund.".
13	1741. Page 1243, line 16: delete the material beginning with that line and
14	ending with page 1244, line 4.
15	<b>1742.</b> Page 1244, line 5: delete lines 5 to 10.
16	<b>1743.</b> Page 1244, line 11: delete lines 11 to 14 and substitute:
17	"Section 2748d. 121.58 (2) (a) 1. of the statutes is amended to read:
18	121.58 (2) (a) 1. For each pupil so transported whose residence is at least 2
19	miles and not more than 5 miles from the school attended, $\$30\ \$41$ per school year
20	in the 2005–06 school year and \$35 per school year thereafter.
21	<b>Section 2748h.</b> 121.58 (2) (a) 2. of the statutes is amended to read:
22	121.58 (2) (a) 2. For each pupil so transported whose residence is more than
23	5 miles and not more than 8 miles from the school attended, \$45 <u>\$71</u> per school year

in the 2005–06 school year and \$55 per school year thereafter.

1 **Section 2748p.** 121.58 (2) (a) 3. of the statutes is amended to read: 2 121.58 (2) (a) 3. For each pupil so transported whose residence is more than 3 8 miles and not more than 12 miles from the school attended, \$82 \$151 per school year in the 2005 school year and \$110 per school year thereafter. 4 5 **Section 2748t.** 121.58 (2) (a) 4. of the statutes is amended to read: 6 121.58 (2) (a) 4. For each pupil so transported whose residence is more than 7 12 miles from the school attended, \$150 \$350 per school year in the 2005-06 school 8 year and \$180 per school year thereafter. 9 **Section 2748v.** 121.58 (2) (c) of the statutes is amended to read: 10 121.58 (2) (c) A school district which provides transportation to and from a 11 school under s. 121.54 (9) shall be paid state aid for such transportation at the rate 12 of \$12 per school year per pupil so transported in the 2005-06 school year and \$15 13 \$17 per school year per pupil so transported thereafter. Such state aid shall be 14 reduced proportionately in the case of a pupil transported for less than a full year 15 because of nonenrollment. State aid for such transportation shall not exceed the 16 actual cost thereof.". **1744.** Page 1244, line 15: delete lines 15 to 22. 17 18 **1745.** Page 1244, line 22: after that line insert: 19 "Section 2749m. 121.81 (2) (br) of the statutes is created to read: 20 121.81 (2) (br) Notwithstanding pars. (a) and (b), if a pupil awarded a 21 scholarship under s. 115.85 attends a school district other than his or her school 22 district of residence under that section, all of the following apply: 23 1. The application need not be accompanied by a written declaration regarding

establishing residence in the school district.

- 2. The school district may not waive or refund tuition for the pupil.
- 2 3. The pupil shall not be considered a resident pupil in computing general aid under subch. II.".
  - **1746.** Page 1244, line 22: after that line insert:
  - "Section 2749d. 121.85 (3m) of the statutes is created to read:
  - 121.85 (3m) LIMITATION. Beginning in the 2009–10 school year, a pupil may not transfer to another school district or another attendance area under sub. (3) if the pupil did not transfer to another school district or attendance area under sub. (3) in the previous school year.
    - **SECTION 2749h.** 121.85 (6) (a) 2. of the statutes is amended to read:
  - 121.85 **(6)** (a) 2. Multiply In the 2007–08 and 2008–09 school years multiply the number of transfer pupils, as counted for membership purposes under s. 121.004 (7), by 0.25; and in each school year thereafter, multiply the number of transfer pupils, as counted for membership purposes under s. 121.004 (7), by 0.225.
    - **Section 2749p.** 121.85 (6) (b) 2. of the statutes is amended to read:
  - 121.85 (6) (b) 2. In each school year, the school district of attendance of pupils transferring from one school district to another under sub. (3) (a) shall receive an amount equal to that produced by multiplying the number of pupils transferred into the school district under sub. (3) (a) in the previous school year by the amount produced by dividing the school district's net school cost by the sum of the membership, plus the number of pupils transferred into the school district of attendance in the previous school year under sub. (3) (a). This subdivision applies to aid paid in the 1995–96 school year only if the number of pupils transferring from one school district to another under sub. (3) (a) in the 1994–95 school year constitutes

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

less than 5% of the total membership of the school district of attendance, and beginning in the 2009–10 school year, multiplying the product by 0.9.

**Section 2749t.** 121.88 of the statutes is created to read:

**121.88 State aid sunset.** Except for state aid under s. 121.85 (6) (ar), no state aid may be distributed under this subchapter after the 2014–15 school year.".

**1747.** Page 1245, line 4: after that line insert:

**"Section 2751m.** 121.91 (3) (a) of the statutes is amended to read:

121.91 (3) (a) If a school board wishes to exceed the limit under sub. (2m) otherwise applicable to the school district in any school year, it shall promptly adopt a resolution supporting inclusion in the final school district budget of an amount equal to the proposed excess revenue. The resolution shall specify whether the proposed excess revenue is for a recurring or nonrecurring purpose, or, if the proposed excess revenue is for both recurring and nonrecurring purposes, the amount of the proposed excess revenue for each purpose. The resolution shall be filed as provided in s. 8.37. Within 10 days after adopting the resolution, the school board shall notify the department of the scheduled date of the referendum and submit a copy of the resolution to the department. The school board shall call a special referendum for the purpose of submitting the resolution to the electors of the school district for approval or rejection. In lieu of a special referendum, the school board may specify that the The referendum shall be held at the next succeeding spring primary or election or September primary or general election, if such election is to be held not sooner than 42 days after the filing of the resolution of the school board. The school district clerk shall certify the results of the referendum to the department within 10 days after the referendum is held.".

**1748.** Page 1245, line 4: after that line insert:

"Section 2751g. 121.91 (2m) (f) of the statutes is created to read:

121.91 (2m) (f) Annually, the commissioner of insurance shall determine whether a school board provides health care benefits to its employees that are substantially similar in quality to, and no more expensive per employee than, the state health plan under s. 40.51 (6). Notwithstanding par. (e), if the commissioner of insurance determines that a school board does not provide such health care benefits to its employees, the school district's revenue limit shall be determined under par. (e), except that \$200 shall be substituted for the result under par. (e) 2.".

- **1749.** Page 1246, line 3: delete the material beginning with that line and ending with page 1247, line 8.
- **1750.** Page 1247, line 9: delete lines 9 to 20.
- **1751.** Page 1247, line 22: delete ", (L) 4., and (m) and" and substitute ", and".
- **1752.** Page 1248, line 14: after that line insert:
- 15 "Section **2757t.** 125.02 (6g) of the statutes is created to read:
  - 125.02 (6g) "Full-service restaurant" means an establishment where meals are prepared, served, and sold to transients or the general public for consumption on the premises and in which the sale of alcohol beverages accounts for 50 percent or less of the establishment's gross receipts for the most recent alcohol beverage licensing year.".
    - **1753.** Page 1249, line 7: after that line insert:
- **"Section 2759h.** 125.51 (4) (v) 1. of the statutes is amended to read:
- 23 125.51 (4) (v) 1. A full-service restaurant that has a seating capacity of 300 or 24 more persons. Notwithstanding sub. (3) (a) and (b), a "Class B" license issued under

this subdivision authorizes the retail sale of intoxicating liquor only for consumption
on the premises where sold. If a "Class B" license issued under this subdivision is
surrendered to the issuing municipality, revoked, or not renewed, the municipality
may not reissue the license to any applicant other than a full-service restaurant. A
person that holds a "Class B" license, other than one issued under this subdivision,
that is surrendered, revoked, or not renewed may not apply for issuance of a "Class
B" license under this subdivision.".
1754. Page 1249, line 7: after that line insert:
"Section 2759hb. 126.06 (4) of the statutes is created to read:
126.06 (4) Sunset. Subsections (1) to (3) do not apply after December 31, 2009.
<b>Section 2759hc.</b> 126.14 (4) of the statutes is created to read:
126.14 (4) Sunset. Subsections (1) to (3) do not apply after December 31, 2009.
<b>Section 2759hd.</b> 126.15 (9) of the statutes is created to read:
126.15 (9) Sunset. Subsections (1) to (8) do not apply after December 31, 2009.
<b>Section 2759he.</b> 126.16 (8) (a) 1. of the statutes is amended to read:
126.16 (8) (a) 1. The grain dealer reports, for at least 2 consecutive years, no
more than \$500,000 in annual grain payments under s. 126.11 (9) (a) and, before
January 1, 2010, the grain dealer pays the quarterly fund assessment that would
have been required of the grain dealer if the grain dealer had been a contributing
grain dealer on the most recent quarterly installment date under s. $126.15(7)$ .
<b>Section 2759hf.</b> 126.16 (8) (a) 2. of the statutes is amended to read:
126.16 (8) (a) 2. The grain dealer's annual financial statement under s. 126.13
shows positive equity for at least 2 consecutive years and, before January 1, 2010,

the grain dealer pays the quarterly fund assessment that would have been required

1	of the grain dealer if the grain dealer had been a contributing grain dealer on the
2	most recent quarterly installment date under s. 126.15 (7).
3	<b>Section 2759hg.</b> 126.29 (4) of the statutes is created to read:
4	126.29 (4) Sunset. Subsections (1) to (3) do not apply after December 31, 2009.
5	<b>Section 2759hi.</b> 126.30 (8) of the statutes is created to read:
6	126.30 (8) Sunset. Subsections (1) to (7) do not apply after December 31, 2009.
7	<b>Section 2759hj.</b> 126.31 (8) (a) 1. of the statutes is amended to read:
8	126.31 (8) (a) 1. The grain warehouse keeper reports grain warehouse capacity
9	under s. 126.26 (2) (e) of less than 300,000 bushels for at least 2 consecutive license
10	years and, before January 1, 2010, the grain warehouse keeper pays the quarterly
11	fund assessment that would have been required of the grain warehouse keeper if the
12	grain warehouse keeper had been a contributing grain warehouse keeper on the most
13	recent quarterly installment date under s. 126.30 (6).
14	<b>Section 2759hk.</b> 126.31 (8) (a) 2. of the statutes is amended to read:
15	126.31 (8) (a) 2. The grain warehouse keeper's annual financial statement
16	under s. 126.28 shows positive equity for at least 2 consecutive years and, before
17	January 1, 2010, the grain warehouse keeper pays the quarterly fund assessment
18	that would have been required of the grain warehouse keeper if the grain warehouse
19	keeper had been a contributing grain warehouse keeper on the most recent quarterly
20	installment date under s. 126.30 (6).
21	<b>Section 2759hL.</b> 126.45 (5) of the statutes is created to read:
22	126.45 (5) Sunset. Subsections (1) to (4) do not apply after December 31, 2009.
23	<b>Section 2759hm.</b> 126.46 (8) of the statutes is created to read:
24	126.46 (8) Sunset. Subsections (1) to (7) do not apply after December 31, 2009.

**Section 2759hn.** 126.47 (7) (a) 1. of the statutes is amended to read:

126.47 (7) (a) 1. The milk contractor reports not more than \$1,500,000 in milk
payroll obligations under s. 126.41 (6) (a) for at least 2 consecutive years and, before
January 1, 2010, the milk contractor pays the quarterly fund assessment that would
have been required of the milk contractor if the milk contractor had been a
contributing milk contractor on the most recent quarterly installment date under $\mathbf{s}$ .
126.46 (6).
<b>Section 2759hp.</b> 126.47 (7) (a) 2. of the statutes is amended to read:
126.47 (7) (a) 2. The milk contractor's annual financial statement under s.
126.44 shows positive equity for at least 2 consecutive years, the financial
statements are audited or, if the milk contractor reports $\$6,000,000$ or less in annual
milk payroll obligations under s. 126.41 (6) (a), reviewed, and, before January 1,
2010, the milk contractor pays the quarterly fund assessment that would have been
required of the milk contractor if the milk contractor had been a contributing milk
contractor on the most recent quarterly installment date under s. $126.46$ (6).
<b>Section 2759hq.</b> 126.59 (4) of the statutes is created to read:
126.59 (4) Sunset. Subsections (1) to (3) do not apply after December 31, 2009.
Section 2759hr. 126.595 (3) of the statutes is created to read:
126.595 (3) Sunset. Subsections (1) and (2) do not apply after December 31,
2009.
Section 2759hs. 126.60 (8) of the statutes is created to read:
126.60 (8) Sunset. Subsections (1) to (7) do not apply after December 31, 2009.
<b>Section 2759ht.</b> 126.61 (7) (a) 1. of the statutes is amended to read:
126.61 (7) (a) 1. The vegetable contractor reports less than \$1,000,000 in
annual contract obligations under s. 126.56 (9) (a) for at least 2 consecutive years
and, before January 1, 2010, the vegetable contractor pays the quarterly fund

assessment that would have been required of the vegetable contractor if the vegetable contractor had been a contributing vegetable contractor on the most recent quarterly installment date under s. 126.60 (6).

**Section 2759hu.** 126.61 (7) (a) 2. of the statutes is amended to read:

126.61 (7) (a) 2. The vegetable contractor's annual financial statement under s. 126.58 shows positive equity for at least 2 consecutive years and, before January 1, 2010, the vegetable contractor pays the quarterly fund assessment that would have been required of the vegetable contractor if the vegetable contractor had been a contributing vegetable contractor on the most recent quarterly installment date under s. 126.60 (6).

**Section 2759hv.** 126.71 (1e) of the statutes is created to read:

126.71 (1e) Sunset. Subsection (1) does not apply after December 31, 2009.

**SECTION 2759hw.** 126.71 (2) (b) of the statutes is amended to read:

126.71 (2) (b) If Before January 1, 2010, if the contractor was a contributing contractor when the default occurred, the department shall use the security proceeds to reimburse the sources under s. 126.72 from which the department makes any claim payment under sub. (1). If the security amount exceeds the amount payable under sub. (1) from the sources under s. 126.72, the department shall use the remaining security proceeds to pay the balance of the allowed claims. If the security amount is not adequate to pay the full remaining balance, the department shall pay claimants on a prorated basis in proportion to their allowed claims.

**Section 2759hx.** 126.72 (4) of the statutes is created to read:

126.72 (4) SUNSET. Subsections (1) to (3) do not apply after December 31, 2009.

**SECTION 2759hy.** 126.81 (3) of the statutes is amended to read:

24

126.81 (3) Modify Before January 1, 2010, modify the fund assessments 1 2 provided under s. 126.15, 126.30, 126.46, or 126.60, as provided in s. 126.88. 3 **Section 2759hz.** 126.88 of the statutes is renumbered 126.88 (1). 4 **Section 2759ib.** 126.88 (2) of the statutes is created to read: 5 126.88 (2) SUNSET. Subsection (1) does not apply after December 31, 2009.". **1755.** Page 1249, line 8: delete lines 8 to 13. 6 7 **1756.** Page 1250, line 3: delete lines 3 to 25. **1757.** Page 1251, line 1: delete lines 1 to 24. 8 **1758.** Page 1252, line 1: delete lines 1 to 24. 9 **1759.** Page 1253, line 1: delete lines 1 and 2. 10 **1760.** Page 1253, line 3: delete the material beginning with that line and 11 12 ending with page 1255, line 10. 13 **1761.** Page 1255, line 10: after that line insert: 14 **"Section 2850n.** 145.01 (4m) of the statutes is renumbered 145.01 (4m) (intro.) 15 and amended to read: 16 145.01 (4m) Failing private sewage system. (intro.) "Failing private sewage 17 system" has the meaning specified under s. 145.245 (4). means a private sewage system that causes or results in any of the following conditions: 18 19 **Section 2850nm.** 145.01 (4m) (a) to (e) of the statutes are created to read: 145.01 (4m) (a) The discharge of sewage into surface water or groundwater. 20 21 (b) The introduction of sewage into zones of saturation that adversely affects 22 the operation of a private sewage system.

(c) The discharge of sewage to a drain tile or into zones of bedrock.

(d) The discharge of sewage to the surface of the ground.

(e) The failure to accept sewage discharges and back up of sewage into the structure served by the private sewage system.

**SECTION 2850p.** 145.20 (5) (a) of the statutes is amended to read:

145.20 (5) (a) The department shall establish a maintenance program to be administered by governmental units responsible for the regulation of private sewage systems. The department shall determine the private sewage systems to which the maintenance program applies. At a minimum the maintenance program is applicable to all new or replacement private sewage systems constructed in a governmental unit after the date on which the governmental unit adopts this program. The department may apply the maintenance program by rule to private sewage systems constructed in a governmental unit responsible for the regulation of private sewage systems on or before the date on which the governmental unit adopts the program. The department shall determine the private sewage systems to which the maintenance program applies in governmental units that de did not meet the conditions for eligibility under s. 145.245 (9), 2005 stats., except that the maintenance program in these governmental units does not commence until January 1, 2008.

**Section 2850r.** 145.245 of the statutes is repealed.".

**1762.** Page 1256, line 15: after that line insert:

"Section 2861m. 146.38 (2) of the statutes is amended to read:

146.38 (2) All persons, organizations, or evaluators reviewing or evaluating, whether from one or more entities, who review or evaluate the services of health care providers shall keep a record of their investigations, inquiries, proceedings and conclusions. No such record may be released to any person under s. 804.10 (4) or

otherwise except as provided in sub. (3). No such record may be used in any civil or criminal action for personal injuries against the health care provider or facility; however, information, documents or records presented during the review or evaluation may not be construed as immune from discovery under s. 804.10 (4) or use in any civil or criminal action merely because they were so presented. Any person who testifies during or participates in the review or evaluation may testify in any civil action as to matters within his or her knowledge, but may not testify as to information obtained through his or her participation in the review or evaluation, nor as to any conclusion of such review or evaluation.

**Section 2861p.** 146.38 (3) (d) of the statutes is amended to read:

146.38 (3) (d) In a report in statistical form that is filed with a regulatory agency, accrediting agency, or person that publicly reports quality and patient safety information. The report may identify any provider or facility to which the statistics relate;

- **SECTION 2861q.** 146.38 (3) (e) of the statutes is repealed.".
- **1763.** Page 1256, line 16: delete lines 16 to 24.
- **1764.** Page 1257, line 1: delete lines 1 to 25.
- **1765.** Page 1258, line 1: delete lines 1 to 11.
- **1766.** Page 1258, line 12: delete lines 12 to 25.
- **1767.** Page 1259, line 1: delete lines 1 to 6.
- **1768.** Page 1259, line 6: after that line insert:
- 22 "Section 2862c. 146.40 (4g) (a) 2. (intro.) of the statutes is amended to read:
- 23 146.40 **(4g)** (a) 2. (intro.) A listing of all individuals about whom the department is notified under sub. (4r) (a) or (am), for whom the department is

1	conducting an investigation under sub. (4r) (b), for whom the department makes
2	findings under sub. (4r) (b), and to whom any of the following applies:".
3	1769. Page 1259, line 9: delete lines 9 and 10 and substitute "appropriation
4	under s. 20.435 (5) (cf), the department shall provide \$37,500 annually to an entity
5	that satisfies".
6	<b>1770.</b> Page 1259, line 17: after that line insert:
7	"Section 2872d. 146.89 (1) (d) 2. of the statutes is amended to read:
8	146.89 (1) (d) 2. A private school, as defined in s. 115.001 (3r), that participates
9	in the Racine County Parental Choice Program under s. 118.53 or the Milwaukee
10	Parental Choice Program under s. 119.23.
11	<b>Section 2872g.</b> 146.89 (1) (g) 3. of the statutes is amended to read:
12	146.89 (1) (g) 3. A private school, as defined in s. 115.001 (3r), that participates
13	in the Racine County Parental Choice Program under s. 118.53 or the Milwaukee
14	Parental Choice Program under s. 119.23.".
15	<b>1771.</b> Page 1259, line 17: after that line insert:
16	<b>"Section 2870pc.</b> 146.83 (1) (b) of the statutes is amended to read:
17	146.83 (1) (b) Receive a copy of the patient's health care records upon payment
18	of fees, as established by rule under sub. (3m) (3g).
19	Section 2870pd. 146.83 (3g) of the statutes is created to read:
20	146.83 (3g) (a) Except as provided under par. (b), the maximum amount that
21	a health care provider may charge under sub. (1) (b) for duplicate patient health care
22	records is \$1 per page, plus applicable tax and the actual costs of postage or other
23	means of delivering the requested records to the requester.

(b) For years beginning after 2007, fees specified under par. (a) increase annually by a percentage equal to the percentage change between the U.S. consumer price index for all urban consumers, U.S. city average, for the month of August of the previous year and the U.S. consumer price index for all urban consumers, U.S. city average, for the month of August of the year before the previous year, as determined by the federal department of labor. The department of health and family services shall specify this increase annually on the department's Web site.

**Section 2870pe.** 146.83 (3m) (a) (intro.) of the statutes is amended to read:

146.83 (3m) (a) (intro.) The department shall, by rule, prescribe fees that are based on an approximation of actual costs. The fees, plus applicable tax, are the maximum amount that a health care provider may charge under sub. (1) (b) for duplicate patient health care records and under sub. (1) (c) for duplicate X-ray reports or the referral of X-rays to another health care provider of the patient's choice. The rule shall also permit the health care provider to charge for actual postage or other actual delivery costs. In determining the approximation of actual costs for the purposes of this subsection, the department may consider all of the following factors:

**SECTION 2870pf.** 146.83 (3m) (a) 2., 3. and 4. of the statutes are amended to read:

146.83 (3m) (a) 2. The varying cost of retrieval of records X-ray reports, based on the different media on which the records reports are maintained.

- 3. The cost of separating requested patient health care records X-ray reports from those that are not requested.
  - 4. The cost of duplicating requested patient health care records X-ray reports.".

**1772.** Page 1259, line 17: after that line insert:

**"Section 2872L.** 146.82 (2) (a) 22. of the statutes is created to read:

146.82 (2) (a) 22. By a person specified in subd. 21. to a correctional officer of the department of corrections who has custody of or is responsible for the supervision of a prisoner, to a person designated with custodial authority over a prisoner by a jailer, or to a law enforcement officer or other person who is responsible for transferring a prisoner to or from a prison or jail, if the patient health care record indicates that the prisoner has a communicable disease and disclosure of that information is necessary for the health and safety of the prisoner or of other prisoners, of the person to whom the information is disclosed, or of any other employee of the prison or jail."

- **1773.** Page 1259, line 20: delete that line.
- **1774.** Page 1260, line 2: after that line insert:

"Section 2875h. 146.997 (5m) of the statutes is created to read:

146.997 (5m) IMMUNITY FROM LIABILITY. (a) Except as provided in par. (c), any health care provider, health care facility, or employee of a health care provider or health care facility that in good faith reports or provides any information, the disclosure of which is not expressly prohibited by state law or rule or federal law or regulation or that in good faith initiates, participates in, or testifies in any action or proceeding, is immune from all civil and criminal liability that may result from any act or omission in reporting or providing that information, or in initiating, participating in, or testifying in that action or proceeding, as follows:

1. Reports to a person specified in sub. (2) (a) (intro.) any information relating to an allegation that a health care provider, health care facility, or employee of a

- 1 health care provider or health care facility has committed a violation specified in sub.
- 2 (2) (a) 1. or 2., has engaged in unprofessional conduct, or has acted negligently in
- 3 treating a patient.

- 2. Initiates, participates in, or testifies in any action or proceeding in which it is alleged that a health care provider, health care facility, or employee of a health care provider or health care facility has committed a violation specified in sub. (2) (a) 1. or 2., has engaged in unprofessional conduct, or has acted negligently in treating a patient.
- 3. Provides to any legislator or legislative committee any information relating to an allegation that a health care provider, health care facility, or employee of a health care provider or health care facility has committed a violation specified in sub.

  (2) (a) 1. or 2., has engaged in unprofessional conduct, or has acted negligently in treating a patient.
- 4. Provides to any prospective employer of an employee or former employee of a health care provider or health care facility any information relating to an allegation that the employee or former employee has committed a violation specified in sub. (2) (a) 1. or 2., has engaged in unprofessional conduct, or has acted negligently in treating a patient.
- (b) For purposes of par. (a), a health care provider, health care facility, or employee of a health care provider or health care facility is not acting in good faith if the health care provider, health care facility, or employee reports any information under par. (a) 1. that the health care provider, health care facility, or employee knows or should know is false or misleading; initiates, participates in, or testifies in any action or proceeding under par. (a) 2. based on information that the health care provider, health care facility, or employee knows or should know is false or

misleading; or provides any information under par. (a) 3. or 4. that the health care provider, health care facility, or employee knows or should know is false or misleading.

(c) Paragraph (a) does not apply to a health care provider, health care facility, or employee of a health care provider or health care facility that reports or provides any information or that initiates, participates in, or testifies in any action or proceeding concerning a violation specified in sub. (2) (a) 1. or 2., unprofessional conduct, or negligence in treating a patient by the health care provider, health care facility, or employee.

**Section 2875i.** 146.997 (5r) of the statutes is created to read:

146.997 (5r) Disclosure of patient information. Notwithstanding s. 146.82 (1), a health care provider, health care facility, or employee of a health care provider or health care facility reporting information as described in sub. (5m) (a) 1. to an agency specified in sub. (2) (a) (intro.) that has the authority to investigate and take action against a health care provider, health care facility, or employee of a health care provider or health care facility for the type of conduct reported may disclose to that agency the name of the patient who is the subject of, and a description of the event giving rise to, the alleged violation, unprofessional conduct, or negligent act. A person receiving identifying information about a patient under this subsection shall keep the information confidential and may not disclose that information except for the purpose of investigating and taking action on the alleged violation, unprofessional conduct, or negligent act, with the informed consent of the patient or of a person authorized by the patient, or as authorized by a court of record.".

**1775.** Page 1260, line 14: delete lines 14 to 16.

- 1 **1776.** Page 1262, line 3: delete lines 3 and 4.
- 2 **1777.** Page 1264, line 4: after that line insert:
- 3 "Section 2904c. 165.25 (12) (title) of the statutes is created to read:
- 4 165.25 (12) (title) JUVENILE JUSTICE AND DELINQUENCY PREVENTION.
- **Section 2904cm.** 165.25 (14) (title) of the statutes is created to read:
- 6 165.25 (14) (title) RECOMMEND LEGISLATION.
- 7 **Section 2904d.** 165.25 (15) (title) of the statutes is created to read:
- 8 165.25 (15) (title) Cooperation and assistance.".
- 9 **1778.** Page 1264, line 9: after that line insert:
- 10 "Section 2905r. 165.845 (title) and (1) (intro.) of the statutes are created to
- 11 read:
- 12 **165.845** (title) **Criminal offense data. (1)** (intro.) The department of justice
- shall:".
- 14 **1779.** Page 1264, line 9: after that line insert:
- 15 "Section **2905p.** 165.8285 (1m) of the statutes is created to read:
- 16 165.8285 (1m) For the purpose of determining if a person is violating s. 301.49
- 17 (3), the department of justice shall, through the transaction information for
- management of enforcement system, provide local law enforcement agencies with
- access to the information provided under s. 301.49 (2) by the department of
- 20 corrections.".
- 21 **1780.** Page 1264, line 10: delete lines 10 to 24.
- 22 **1781.** Page 1265, line 1: delete lines 1 to 25.
- 23 **1782.** Page 1266, line 1: delete lines 1 to 8.

and amended to read:

23

**1783.** Page 1267, line 6: after that line insert: 1 2 "Section 2909n. 165.95 (title) of the statutes is created to read: 3 165.95 (title) Deferred prosecution and treatment for alcohol and drug 4 offenders. 5 **Section 2909nm.** 165.986 (title) of the statutes is created to read: 6 165.986 (title) Law enforcement officer supplement grants. **Section 2909p.** 165.987 (title) of the statutes is created to read: 7 165.987 (title) Grants for recording custodial interrogations.". 8 **1784.** Page 1267, line 7: delete lines 7 to 25. 9 **1785.** Page 1267, line 25: after that line insert: 10 11 **"Section 2910g.** 166.03 (2) (a) 8. of the statutes is created to read: 166.03 (2) (a) 8. Administer expenditure of federal homeland security funds.". 12 13 **1786.** Page 1268, line 1: delete lines 1 to 25. **1787.** Page 1269, line 1: delete lines 1 to 5. 14 **1788.** Page 1269, line 20: delete the material beginning with that line and 15 16 ending with page 1271, line 4. **1789.** Page 1271, line 5: delete lines 5 to 8. 17 **1790.** Page 1271, line 8: after that line insert: 18 "Section 2913rc. 169.06 (4) (title) of the statutes is repealed and recreated to 19 20 read: 21 169.06 (4) (title) EXEMPTIONS. 22 **Section 2913rg.** 169.06 (4) of the statutes is renumbered 169.06 (4) (a) (intro.)

1 169.06 (4) (a) (intro.) Subsections (1) to (3m) do not apply to wild any of the following:

1. Wild animals that are released into the wild after being accidentally trapped or confined.

**Section 2913rm.** 169.06 (4) (a) 2. of the statutes is created to read:

169.06 (4) (a) 2. Persons who take or possess live native wild reptiles or amphibians while engaged in an activity at an annual community event sponsored by a municipality or sporting club if the event is conducted for charitable purposes and the department determines that the activity does not pose a significant risk to the health or viability of the relevant species where the event is held.

**Section 2913rr.** 169.12 (7) of the statutes is created to read:

169.12 (7) EXEMPTION FOR CERTAIN COMMUNITY EVENTS. Subsection (1) does not apply to persons who take or possess live native wild reptiles or amphibians while engaged in an activity at an annual community event sponsored by a municipality or sporting club if the event is conducted for charitable purposes and the department determines that the activity does not pose a significant risk to the health or viability of the relevant species where the event is held.".

- **1791.** Page 1271, line 9: delete lines 9 to 21.
- **1792.** Page 1271, line 22: delete lines 22 to 24.
- **1793.** Page 1272, line 1: delete lines 1 to 25.
- **1794.** Page 1273, line 1: delete lines 1 to 24.
- **1795.** Page 1274, line 6: delete lines 6 to 9.
- **1796.** Page 1274, line 9: after that line insert:
- **Section 2924c.** 185.981 (4t) of the statutes is amended to read:

185.981 **(4t)** A sickness care plan operated by a cooperative association is subject to ss. 252.14, 631.17, 631.89, 631.95, 632.72 (2), 632.745 to 632.749, 632.85, 632.853, 632.855, 632.87 (2m), (3), (4), and (5), 632.895 (10) to (14) (15), and 632.897 (10) and chs. 149 and 155.

**SECTION 2924f.** 185.983 (1) (intro.) of the statutes is amended to read:

185.983 (1) (intro.) Every such voluntary nonprofit sickness care plan shall be exempt from chs. 600 to 646, with the exception of ss. 601.04, 601.13, 601.31, 601.41, 601.42, 601.43, 601.44, 601.45, 611.67, 619.04, 628.34 (10), 631.17, 631.89, 631.93, 631.95, 632.72 (2), 632.745 to 632.749, 632.775, 632.79, 632.795, 632.85, 632.853, 632.855, 632.87 (2m), (3), (4), and (5), 632.895 (5) and (9) to (14) (15), 632.896, and 632.897 (10) and chs. 609, 630, 635, 645, and 646, but the sponsoring association shall:".

- **1797.** Page 1276, line 24: delete "sub. (5)" and substitute "sub. (5)".
- **1798.** Page 1277, line 7: delete "fund." and substitute "fund subs. (4) and (5).".
- **1799.** Page 1277, line 7: after that line insert:
- 16 "Section **2928mm.** 196.202 (4) of the statutes is created to read:
  - 196.202 (4) RULES. The commission may promulgate rules for designating a commercial mobile radio service provider, upon petitioning the commission, as an eligible telecommunications carrier for purposes of participation under the federal universal service fund.".
    - **1800.** Page 1277, line 7: after that line insert:
- "Section 2928mm. 196.218 (3) (a) 3. (intro.) of the statutes is amended to read:
  196.218 (3) (a) 3. (intro.) The commission shall designate the method by which
  the contributions under this paragraph shall be calculated and collected. The Except

as provided in par. (am), the method shall ensure that the contributions are sufficient 1  $\mathbf{2}$ to generate the following amounts:". 3 **1801.** Page 1277, line 14: after that line insert: "Section 2929em. 196.218 (3) (am) of the statutes is created to read: 4 5 196.218 (3) (am) Beginning in fiscal year 2007-08, the total amount of 6 contributions by all telecommunications providers in a fiscal year under par. (a) 3. 7 may not exceed \$32,038,400.". **1802.** Page 1278, line 13: delete lines 13 to 21. 8 9 **1803.** Page 1278, line 21: after that line insert: 10 **"Section 2931g.** 196.218 (5s) of the statutes is created to read: 196.218 (5s) Annual Audits. Annually the legislative audit bureau shall 11 12 conduct a performance audit of at least one of the programs that receives funding 13 from the universal service fund. The audit bureau shall distribute a copy of each 14 audit report under this subsection to the commission, to the cochairs of the joint 15 committee on finance, and to the cochairs of the joint audit committee. Audits under 16 this subsection are in addition to the audits required under sub. (2) (d).". **1804.** Page 1279, line 13: delete lines 13 to 20. 17 **1805.** Page 1279, line 20: after that line insert: 18 19 **"Section 2933rg.** 196.491 (3) (d) (intro.) of the statutes is amended to read: 196.491 (3) (d) (intro.) Except as provided under par. (e) and s. 196.493, the 20 21 commission shall approve an application filed under par. (a) 1. for a certificate of

public convenience and necessity only if the commission determines all of the

**Section 2933rr.** 196.493 of the statutes is repealed.".

22

23

following:

 $\mathbf{2}$ 

**1806.** Page 1279, line 20: after that line insert:

"Section 2933rm. 196.491 (3) (dm) of the statutes is amended to read:

196.491 (3) (dm) In making a determination required under par. (d), the commission may not consider a factual conclusion in a strategic energy assessment or the report required under 2007 Wisconsin Act .... (this act), section 9139 (2t) (f), unless the conclusion is independently corroborated in the hearing under par. (b).".

**1807.** Page 1279, line 20: after that line insert:

"Section 2933rb. 196.497 (3) of the statutes is renumbered 196.497 (3) (a) and amended to read:

196.497 (3) (a) The commission shall serve as an advocate on behalf of the citizens of this state before the federal department of energy and other federal agencies on matters related to the <u>centralized interim storage or long-term disposal</u> of <u>high-level</u> radioactive waste and transuranic waste, <u>including any license</u> application submitted to the nuclear regulatory commission for a license to operate a federal repository for the long-term disposal of high-level radioactive waste and transuranic waste.

**Section 2933rf.** 196.497 (3) (b) of the statutes is created to read:

196.497 (3) (b) The commission's advocacy under par. (a) may take the form of participation under s. 196.02 (12), a request to the attorney general to intervene under sub. (7), participation in the intervention or advocacy of other states or of an interstate organization, or any other form of advocacy the commission determines is appropriate.

**Section 2933rk.** 196.497 (6) of the statutes is amended to read:

196.497 (6) Monitor federal activity. The commission shall monitor activity
in congress and the federal government related to the <u>centralized interim storage or</u>
long-term disposal of high-level radioactive waste and transuranic waste. The
commission may advise the congressional delegation from this state of action which
is needed to protect the interests of the state.
Section 2933ro. 196.497 (14) (title) of the statutes is repealed and recreated
to read:
196.497 (14) (title) Cooperation.
<b>Section 2933rs.</b> 196.497 (14) of the statutes is renumbered 196.497 (14) (a).
Section 2933rw. 196.497 (14) (a) (title) of the statutes is created to read:

**Section 2933ry.** 196.497 (14) (b) of the statutes is created to read:

196.497 (14) (a) (title) State agencies.

196.497 (14) (b) Facility owners and operators. The owner or operator of a nuclear power plant shall assist the commission to the fullest extent possible and shall provide any data or other information required by the commission in fulfilling its duties under this section. In this paragraph, "nuclear power plant" means a nuclear-powered electric generating facility, and includes a facility that is no longer used to generate electricity.".

**1808.** Page 1279, line 21: delete lines 21 to 25.

**1809.** Page 1280, line 1: delete lines 1 to 25.

**1810.** Page 1281, line 1: delete lines 1 to 25.

**1811.** Page 1282, line 1: delete lines 1 to 25.

**1812.** Page 1283, line 1: delete lines 1 to 25.

**1813.** Page 1284, line 1: delete lines 1 to 8.

- **1814.** Page 1284, line 9: delete lines 9 to 15.
- **1815.** Page 1284, line 16: delete lines 16 to 25.
- **1816.** Page 1285, line 1: delete lines 1 to 18.
- **1817.** Page 1285, line 19: delete lines 19 to 25.
- **1818.** Page 1286, line 1: delete lines 1 to 25.
- **1819.** Page 1287, line 1: delete lines 1 to 25.
- **1820.** Page 1288, line 1: delete lines 1 to 25.
- **1821.** Page 1289, line 1: delete lines 1 to 25.
- **1822.** Page 1290, line 1: delete lines 1 to 24.
- **1823.** Page 1291, line 1: delete lines 1 to 25.
- **1824.** Page 1292, line 1: delete lines 1 to 25.
- **1825.** Page 1293, line 1: delete lines 1 to 25.
- **1826.** Page 1294, line 1: delete lines 1 to 25.
- **1827.** Page 1295, line 1: delete lines 1 to 25.
- **1828.** Page 1296, line 1: delete lines 1 to 25.
- **1829.** Page 1297, line 1: delete lines 1 to 25.
- **1830.** Page 1298, line 1: delete lines 1 to 25.
- **1831.** Page 1299, line 1: delete lines 1 to 6.
- **1832.** Page 1299, line 8: delete lines 8 to 10.
- **1833.** Page 1299, line 10: after that line insert:
- **"Section 2994b.** 227.01 (13) (kg) of the statutes is created to read:

227.01	<b>(13)</b>	(kg)	Establishes	programmatic	goals	for	grants	and	loans
administered	d by a	depar	tment or inde	ependent agenc	y unde	rs.	15.04 (1)	(n)."	•

- **1834.** Page 1299, line 18: delete lines 18 to 20.
- **1835.** Page 1299, line 20: after that line insert:
- 5 "Section **2996c.** 227.112 of the statutes is created to read:
  - **227.112 State funding of mandates.** (1) In this section:
  - (a) "Local governmental unit" has the meaning given in s. 19.42 (7u).
  - (b) "Mandate" has the meaning given in s. 13.59 (1) (b), and includes a provision in a rule placing a requirement on a local governmental unit. The term does not include any rule provision that relates to employment discrimination or the compensation, benefits, leave, collective bargaining rights, or conditions of employment of employees or retirees of a local governmental unit or prevailing wages under s. 66.0903.
  - (2) An agency may not promulgate a rule or take an action on or after the effective date of this subsection .... [revisor inserts date], that imposes a mandate, other than a mandate that has a minimal fiscal effect, and shall not, on or after the effective date of this subsection .... [revisor inserts date], take an action required by a law enacted on or after the effective date of this subsection .... [revisor inserts date], if the action would impose a mandate, other than a mandate that has a minimal fiscal effect, unless there is a sufficient amount in the appropriation account under s. 20.505 (1) (bm) or such other appropriations designated by law for such purpose for providing reimbursement under s. 16.59 to local governmental units for their approximate costs that are attributable to the mandate without jeopardizing reimbursement under s. 16.59 for other mandates.".

**1836.** Page 1307, line 8: after that line insert:

"Section 2997zk. 227.28 of the statutes is created to read:

**227.28 Periodic review of administrative rules.** (1) Every agency that has promulgated administrative rules shall determine which of those rules has been in effect for 6 years before the effective date of this subsection .... [revisor inserts date], without being modified or revised and shall submit a notice to the chief clerk of each house of the legislature regarding those rules.

- (2) When a rule, other than a rule described in sub. (1), has been in effect for 6 years without being modified or revised, the agency that promulgated the rule shall submit a notice to the chief clerk of each house of the legislature regarding the rule.
- (3) The notice under sub. (1) or (2) shall be submitted in triplicate and shall include the rule's identifying name and number, the date it was promulgated and last modified or revised, if appropriate, and a brief description of the subject matter of the rule. The presiding officer of each house of the legislature shall, within 10 working days following the day on which a notice is received, direct the appropriate chief clerk to refer the notice to one standing committee. Each chief clerk shall enter a statement in the journal of his or her house that the notice has been received and to which committee the rule has been referred.
- (4) Upon receipt of notice that a rule has been referred to a committee under sub. (3), the chairperson of the committee shall notify, in writing, each committee member of the referral. If the chairperson of the committee does not notify the head of the agency that the committee has scheduled a meeting for the purpose of reviewing the rule within 14 working days after the date of the notification, the rule shall continue in effect. If, within 14 working days after the date of the notification,

- the chairperson of the committee notifies the head of the agency that the committee has scheduled a meeting for the purpose of reviewing the rule, the committee may schedule a meeting to discuss the rule.
- (5) (a) A committee may be convened upon the call of its chairperson to review a rule. A committee may hold a public hearing to review a rule. Except as provided under par. (b), the committee review period for each committee extends for 30 days after referral under sub. (3). If the chairperson of a committee takes either of the following actions within the 30-day period, the committee review period for that committee is continued for 30 days from the date on which the first 30-day review period would have expired:
- 1. Requests in writing that the agency meet with the committee to review the rule.
- 2. Publishes or posts notice that the committee will hold a meeting or hearing to review the rule and immediately sends a copy of the notice to the agency.
- (b) If a committee in one house votes to object to and recommend suspension of a rule under sub. (6), the chairperson of the committee shall immediately notify the chairperson of the committee to which the rule was referred in the other house. Upon receipt of the notice, the review period for the committee in the other house immediately ceases and no further action on the rule may be taken under this subsection, but the committee in the other house may proceed under sub. (6) to vote to object and recommend suspension of the rule.
- (c) If a committee has not concluded its jurisdiction over a rule before the day specified under s. 13.02 (1) for the next legislature to convene, that jurisdiction immediately ceases and, within 10 working days after that date, the presiding officer of the appropriate house shall refer the rule to the appropriate standing committee

- as provided under sub. (3). The committee review period that was interrupted by the loss of jurisdiction under this paragraph continues for the committee to which the rule is referred under this paragraph beginning on the date of referral under this paragraph.
- (6) A committee, by a majority vote of a quorum of the committee during the review period under sub. (5), may object to a rule and recommend that the rule be suspended if a change in circumstances since enactment of the law upon which the rule is based makes the rule no longer necessary to accomplish the policy objective of that law.
- (7) When a standing committee objects to a rule and recommends that the rule be suspended, the committee shall, within 30 days of the date of the objection, meet and take executive action regarding the introduction, in the committee's house of the legislature, of a bill to support the objection and suspension of the rule. The committee shall introduce the bill within 5 working days after taking executive action in favor of introduction of the bill unless the bill cannot be introduced during this time period under the joint rules of the legislature.".
  - **1837.** Page 1307, line 9: delete lines 9 to 25.
- **1838.** Page 1308, line 1: delete lines 1 to 9.
- 1839. Page 1308, line 10: delete the material beginning with that line and ending with page 1309, line 21.
- **1840.** Page 1310, line 4: delete "235, or" and substitute "235, or".
- **1841.** Page 1310, line 8: delete lines 8 and 9.
- **1842.** Page 1310, line 10: delete lines 10 to 13.
- **1843.** Page 1310, line 14: delete lines 14 to 18 and substitute.

- 1 "Section 3010. 230.08 (2) (L) 6. of the statutes is repealed.
- **Section 3011.** 230.08 (2) (of) of the statutes is repealed.".
- **1844.** Page 1310, line 19: delete lines 19 to 22.
- **1845.** Page 1310, line 25: after that line insert:
- 5 "Section 3013s. 230.08 (2) (y) of the statutes is repealed.".
- **1846.** Page 1311, line 5: delete lines 5 to 25.
- **1847.** Page 1312, line 1: delete lines 1 to 8.
- **1848.** Page 1312, line 8: after that line insert:
  - "Section 3019d. 230.35 (2) of the statutes is renumbered 230.35 (2) (a) and amended to read:
    - 230.35 (2) (a) Leave Subject to pars. (am) and (b), leave of absence with pay owing to sickness and leave of absence without pay, other than annual leave and leave under s. 103.10, shall be regulated by rules of the director, except that unused sick leave shall accumulate from year to year. After July 1, 1973, employees appointed to career executive positions under the program established under s. 230.24 or positions designated in s. 19.42 (10) (L) or 20.923 (4), (7), (8), and (9) or authorized under s. 230.08 (2) (e) shall have any unused sick leave credits restored if they are reemployed in a career executive position or in a position under s. 19.42 (10) (L) or 20.923 (4), (7), (8), and (9) or authorized under s. 230.08 (2) (e), regardless of the duration of their absence. Restoration of unused sick leave credits if reemployment is to a position other than those specified above shall be in accordance with rules of the director.
      - **SECTION 3019h.** 230.35 (2) (b) of the statutes is created to read:

230.35 (2) (b) A person who holds a state office, as defined in s. 5.02 (23), and who accrues sick leave under this subsection may not receive sick leave while in state office during any term of office that begins after the effective date of this paragraph .... [revisor inserts date].".

**1849.** Page 1312, line 8: after that line insert:

**"Section 3020r.** 230.81 (2) of the statutes is amended to read:

230.81 (2) Nothing in this section prohibits an employee from disclosing information to an appropriate law enforcement agency, a state or federal district attorney in whose jurisdiction the crime is alleged to have occurred, a state or federal grand jury or a judge in a proceeding commenced under s. 968.26 (1), or disclosing information pursuant to any subpoena issued by any person authorized to issue subpoenas under s. 885.01. Any such disclosure of information is a lawful disclosure under this section and is protected under s. 230.83.".

**1850.** Page 1312, line 8: after that line insert:

"Section 3021gp. 230.35 (2) (am) of the statutes is created to read:

230.35 **(2)** (am) An employee, other than a protective occupation participant, as defined in s. 40.02 (48), who is not specified in par. (b) and who is initially employed by the state after the effective date of this paragraph .... [revisor inserts date], shall only receive 6 paid sick leave days in any year.".

**1851.** Page 1312, line 8: after that line insert:

"Section 3022k. 231.03 (6) (intro.) of the statutes is amended to read:

231.03 **(6)** (intro.) Subject to s. 231.08 (7) <u>and (8)</u>, issue bonds of the authority to do any of the following:

**Section 3022n.** 231.08 (1) of the statutes is amended to read:

231.08 (1) Subject to sub. subs. (7) and (8), the authority may from time to time issue bonds for any corporate purpose. All such bonds or other obligations of the authority issued under this chapter are declared to be negotiable for all purposes, notwithstanding their payment from a limited source and without regard to any other law. The authority shall employ the building commission as its financial consultant to assist and coordinate the issuance of bonds and notes of the authority.

**Section 3022r.** 231.08 (8) of the statutes is created to read:

231.08 (8) If the authority is requested to issue bonds for a purpose specified in s. 231.03 (6), the authority shall issues the bonds in the amounts requested if the authority determines that revenues associated with the project that is to be financed with the bond proceeds are sufficient to pay all debt service and related costs of the bonds.".

- **1852.** Page 1315, line 12: after that line insert:
- "Section **3024d.** 234.03 (2m) of the statutes is amended to read:
- 15 234.03 (**2m**) To issue notes and bonds in accordance with ss. 234.08, 234.40, 234.50, 234.60, 234.61, 234.626, 234.63, and 234.65, and 234.66.".
  - **1853.** Page 1315, line 23: delete the material beginning with that line and ending with page 1316, line 14, and substitute:
- 19 "**Section 3027f.** 234.165 (3) of the statutes is created to read:

234.165 (3) For the purpose of housing grants and loans under s. 560.9803 and housing grants under s. 560.9805, in fiscal year 2007–08 the authority shall transfer to the department of commerce \$2,000,000 of its actual surplus under this section and in fiscal year 2008–09 the authority shall transfer to the department of commerce \$2,000,000 of its actual surplus under this section.

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

1	SECTION 3028fm.	234.165 (3) of the statutes, as affected by 2007 Wisconsin Act
2	(this act), is repeale	d.".

**1854.** Page 1316, line 14: after that line insert:

"Section 3028k. Chapter 235 of the statutes is repealed.".

**1855.** Page 1316, line 14: after that line insert:

"Section 3028h. 234.165 (4) of the statutes is created to read:

234.165 (4) Notwithstanding sub. (2) (c), for the purpose of a grant to a pulp and paper mill under 2007 Wisconsin Act .... (this act), section 9103 (3w), in fiscal year 2007–08 the authority shall transfer to the department of agriculture, trade and consumer protection \$2,500,000 of its actual surplus under this section.".

**1856.** Page 1316, line 14: after that line insert:

"Section 3028k. 234.265 (2) of the statutes is amended to read:

234.265 (2) Records or portions of records consisting of personal or financial information provided by a person seeking a grant or loan under s. 234.04, 234.08, 234.49, 234.59, 234.61, 234.63, 234.65, 234.67, 234.83, 234.84, 234.90, 234.905, 234.907, or 234.91, seeking a loan under ss. 234.621 to 234.626, seeking financial assistance under s. 234.66, 2005 stats., seeking investment of funds under s. 234.03 (18m), or in which the authority has invested funds under s. 234.03 (18m), unless the person consents to disclosure of the information.

**Section 3028m.** 234.40 (4) of the statutes is amended to read:

234.40 (4) The limitations established in ss. 234.18, 234.50, 234.60, 234.61, 234.63, and 234.65, and 234.66 are not applicable to bonds issued under the authority of this section. The authority may not have outstanding at any one time

bonds for veterans housing loans in an aggregate principal amount exceeding \$61,945,000, excluding bonds being issued to refund outstanding bonds.

**SECTION 30280.** 234.50 (4) of the statutes is amended to read:

234.50 (4) The limitations established in ss. 234.18, 234.40, 234.60, 234.61, 234.63, and 234.65, and 234.66 are not applicable to bonds issued under the authority of this section. The authority may not have outstanding at any one time bonds for housing rehabilitation loans in an aggregate principal amount exceeding \$100,000,000, excluding bonds being issued to refund outstanding bonds. The authority shall consult with and coordinate the issuance of bonds with the building commission prior to the issuance of bonds.

**Section 3028q.** 234.60 (2) of the statutes is amended to read:

234.60 (2) The limitations in ss. 234.18, 234.40, 234.50, 234.61, 234.63, and 234.65, and 234.66 do not apply to bonds or notes issued under this section.

**Section 3028s.** 234.61 (1) of the statutes is amended to read:

234.61 (1) Upon the authorization of the department of health and family services, the authority may issue bonds or notes and make loans for the financing of housing projects which are residential facilities as defined in s. 46.28 (1) (d) and the development costs of those housing projects, if the department of health and family services has approved the residential facilities for financing under s. 46.28 (2). The limitations in ss. 234.18, 234.40, 234.50, 234.60, 234.63, and 234.65, and 234.66 do not apply to bonds or notes issued under this section. The definition of "nonprofit corporation" in s. 234.01 (9) does not apply to this section.

**Section 3028u.** 234.63 (2) (b) of the statutes is amended to read:

234.63 **(2)** (b) The limits in ss. 234.18 <del>(1)</del>, 234.40, 234.50, 234.60, 234.61, and 234.65, and 234.66 do not apply to bonds issued under par. (a).

- **Section 3028w.** 234.65 (1) (b) of the statutes is amended to read:
- 2 234.65 (1) (b) The limits in ss. 234.18, 234.40, 234.50, 234.60, and 234.61, and
- 3 234.66 do not apply to bonds or notes issued under this section.
- **Section 3028y.** 234.66 of the statutes is repealed.".
- **1857.** Page 1316, line 15: delete lines 15 to 24.
- **1858.** Page 1317, line 1: delete lines 1 to 25.
- **1859.** Page 1318, line 1: delete lines 1 to 3.
- **1860.** Page 1318, line 7: delete "\$2,569,900 \$2,969,900" and substitute "\$2,569,900".
- **61.** Page 1318, line 8: delete "and not more than \$3,569,900 in fiscal year 2008–09".
- **1862.** Page 1319, line 1: delete lines 1 to 5.

- **1863.** Page 1319, line 11: after that line insert:
- 14 "Section 3036s. 252.15 (5) (a) 21. of the statutes is created to read:
  - 252.15 (5) (a) 21. By a person specified in subd. 20. to a correctional officer of the department of corrections who has custody of or is responsible for the supervision of a prisoner, to a person designated with custodial authority over a prisoner by a jailer, or to a law enforcement officer or other person who is responsible for transferring a prisoner to or from a prison or jail, if the test results indicate that the prisoner has tested positive for the presence of HIV, as defined in s. 252.01 (1m), antigen or nonantigenic products of HIV, or an antibody to HIV and disclosure of that information is necessary for the health and safety of the prisoner or of other

- prisoners, of the person to whom the information is disclosed, or of any other employee of the prison or jail.".
- **1864.** Page 1320, line 5: delete lines 5 to 12.
- **1865.** Page 1320, line 17: delete lines 17 to 25.
- **1866.** Page 1321, line 1: delete lines 1 to 25.
- **1867.** Page 1322, line 1: delete lines 1 to 6.
- **1868.** Page 1322, line 6: after that line insert:
- 8 "Section 3055j. 253.07 (5) of the statutes is created to read:
  - 253.07 (5) Abstinence education. For each fiscal year, the department shall apply to the federal secretary of health and human services for an abstinence education, as defined in 42 USC 710 (b) (2), allotment under 42 USC 710 (a). If the secretary grants the allotment, the department shall accept the allotment and use the allotment in accordance with 42 USC 710 and all federal regulations and guidelines applicable to the allotment.".
- **1869.** Page 1322, line 6: after that line insert:
- **"Section 3055e.** 253.07 (1) (a) 3. of the statutes is repealed.
- **Section 3055f.** 253.07 (1) (b) 3. of the statutes is repealed.
- **Section 3055g.** 253.07 (2) (b) of the statutes is amended to read:
  - 253.07 (2) (b) The department shall allocate state and federal family planning funds under its control <u>under this section</u> in a manner which will promote the development and maintenance of an integrated system of community health services. It shall maximize the use of existing community family planning services by encouraging local contractual arrangements. The department may provide family planning funds under this section to a county or other governmental body.

 $\mathbf{2}$ 

other than a county or other governmental body that provides counseling services with respect to the termination of pregnancy or that provides abortion services. The department may not provide family planning funds under this section to a private entity.

**Section 3055h.** 253.07 (2) (c) of the statutes is amended to read:

253.07 (2) (c) The department shall coordinate the delivery of family planning services by allocating family planning funds <u>under this section</u> in a manner which maximizes coordination between <u>the agencies counties and other governmental</u> bodies that receive funds under this section.

**Section 3055i.** 253.07 (4) (b) of the statutes is amended to read:

253.07 (4) (b) For each fiscal year, \$67,500 to subsidize the provision by family planning agencies counties or other governmental bodies under this section of papanicolaou tests to individuals with low income. In this paragraph, "low income" means adjusted gross income that is less than 200% of the poverty line established under 42 USC 9902 (2).

**Section 3055i.** 253.07 (4) (c) of the statutes is amended to read:

253.07 (4) (c) For each fiscal year, \$54,000 to subsidize the provision by family planning agencies counties or other governmental bodies under this section of follow-up cancer screening.

**Section 3055k.** 253.07 (4) (d) of the statutes is amended to read:

253.07 (4) (d) For each fiscal year, \$31,500 as grants to applying family planning agencies counties or other governmental bodies under this section for employment in communities of licensed registered nurses, licensed practical nurses, certified nurse-midwives or licensed physician assistants who are members of a racial minority.

 $\mathbf{2}$ 

**Section 3055L.** 253.07 (6) of the statutes is created to read:

253.07 **(6)** LIMITATIONS. No county or other governmental body that receives funds under this section may contract with those funds to provide family planning services with another county, governmental body, or private entity that provides counseling services with respect to the termination of pregnancy or provides abortion services.".

- **1870.** Page 1322, line 7: delete the material beginning with that line and ending with page 1323, line 16.
- **1871.** Page 1323, line 17: delete lines 17 to 25.
- **1872.** Page 1324, line 1: delete lines 1 to 25.
- **1873.** Page 1325, line 1: delete lines 1 to 25.
- **1874.** Page 1326, line 1: delete lines 1 and 2.
- **1875.** Page 1326, line 3: delete lines 3 to 19.
- **1876.** Page 1326, line 24: delete the material beginning with that line and ending with page 1327, line 2.
- **1877.** Page 1344, line 11: after that line insert:
- 17 "Section **3072p.** 281.57 (7) (c) 1. of the statutes is amended to read:
  - 281.57 (7) (c) 1. Metropolitan sewerage districts that serve 1st class cities are limited in each fiscal year to receiving total grant awards not to exceed 33% of the sum of the amounts in the schedule for that fiscal year for the appropriation under s. 20.143 (3) (de) and the amount authorized under sub. (10) for that fiscal year plus the unencumbered balance at the end of the preceding fiscal year for the amount

- 1 authorized under sub. (10). This subdivision is not applicable to grant awards
- 2 provided during fiscal years 1985–86, 1986–87, 1988–89 and 1989–90.
- 3 **Section 3073k.** 281.59 (1m) (c) of the statutes is repealed.".
- 1878. Page 1344, line 23: delete the material beginning with that line and ending with page 1345, line 7.
- 6 **1879.** Page 1345, line 11: delete the material beginning with "<u>, and</u>" and ending with "<u>subsection</u>," on line 13.
- 8 **1880.** Page 1353, line 14: delete lines 14 to 22.
- 9 **1881.** Page 1354, line 1: delete lines 1 to 3.
- 10 **1882.** Page 1354, line 4: delete lines 4 to 13.
- 11 **1883.** Page 1354, line 14: delete lines 14 to 25.
- 12 **1884.** Page 1355, line 1: delete lines 1 to 13.
- 13 **1885.** Page 1356, line 9: delete lines 9 to 25.
- 14 **1886.** Page 1357, line 1: delete lines 1 to 17.

- 15 **1887.** Page 1357, line 17: after that line insert:
- 16 "Section 3099p. 301.03 (3k) of the statutes is created to read:
- 301.03 (3k) (a) Provide health care to prisoners. The department shall solicit proposals from entities to provide health care to prisoners under contract with the department. The department shall contract with an entity to provide health care to prisoners if all of the following are true:
- 21 1. The entity is an accredited correctional health care provider.
  - 2. Each facility at which the entity provides health care services is accredited.

- 3. The department would realize cost savings by contracting with an entity to provide health care services to inmates.
  - (b) The department shall provide a copy of its request-for-proposal for health care services to the joint committee on finance and the appropriate standing committee for each house of the legislature at the time it is issued. When the department accepts a bid, or when it rejects all submitted bids, the department shall notify the joint committee on finance and the appropriate standing committee for each house of the legislature and shall submit a complete copy of each submitted bid to those committees. If the department does not accept any submitted bid, the department shall report its reasons for rejecting each submitted bid to the joint committee on finance and the appropriate standing committee for each house of the legislature.".
- **1888.** Page 1357, line 24: delete that line.
- **1889.** Page 1358, line 1: delete lines 1 to 18.
- **1890.** Page 1359, line 3: delete lines 3 to 6.
- **1891.** Page 1359, line 7: delete lines 7 to 14.
- **1892.** Page 1359, line 15: delete the material beginning with that line and ending with page 1360, line 2.
- **1893.** Page 1360, line 3: delete lines 3 to 10.
- **1894.** Page 1362, line 13: delete lines 13 to 17.
- **1895.** Page 1362, line 20: delete "and, (ko), and (r)" and substitute "and (ko)".
- 1896. Page 1362, line 25: delete the material beginning with that line and ending with page 1363, line 4, and substitute:

**Section 3117d.** 301.26 (7) (a) of the statutes is amended to read:

301.26 (7) (a) For community youth and family aids under this section, amounts not to exceed \$44,145,100 for the last 6 months of 2005 2007, \$88,290,200 for 2006 2008, and \$44,145,100 for the first 6 months of 2007 2009."

**1897.** Page 1363, line 9: after that line insert:

**"Section 3118m.** 301.26 (7) (b) 2. of the statutes is amended to read:

301.26 (7) (b) 2. Each county's proportion of the total Part I juvenile arrests reported statewide under the uniform crime reporting system of the office department of justice assistance during the most recent 3-year period for which that information is available.".

**1898.** Page 1363, line 10: delete lines 10 to 15.

**1899.** Page 1363, line 24: delete the material beginning with that line and ending with page 1364, line 5.

**1900.** Page 1365, line 2: after that line insert:

"Section 3124m. 301.263 (3) of the statutes is amended to read:

301.263 (3) The department shall distribute 33% of the amounts distributed under sub. (1) based on each county's proportion of the violent Part I juvenile arrests reported statewide under the uniform crime reporting system of the office department of justice assistance in the department of administration, during the most recent 2-year period for which that information is available. The department shall distribute 33% of the amounts distributed under sub. (1) based on each county's proportion of the number of juveniles statewide who are placed in a juvenile correctional facility or a secured residential care center for children and youth during the most recent 2-year period for which that information is available. The

 $\mathbf{2}$ 

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

department shall distribute 34% of the amounts distributed under sub. (1) based on each county's proportion of the total Part I juvenile arrests reported statewide under the uniform crime reporting system of the office department of justice assistance, during the most recent 2-year period for which that information is available.".

**1901.** Page 1365, line 3: delete the material beginning with that line and ending with page 1366, line 17, and substitute:

"Section 3128d. 301.265 (3) of the statutes is amended to read:

301.265 (3) From the appropriations under s. 20.410 (3) (d) and (kj), the department shall allocate \$150,000 in each fiscal year to enter into a contract with an organization to provide services in Racine County, \$150,000 in each fiscal year to enter into a contract with an organization to provide services in Kenosha County, \$150,000 in each fiscal year to enter into a contract with an organization that is located in ward 1 in the city of Racine to provide services in Racine County, and \$150,000 in each fiscal year to enter into a contract with an organization to provide services in Brown County, and from the appropriation under s. 20.410 (3) (kj), the department shall allocate \$100,000 in each fiscal year to enter into a contract with an organization, for the diversion of youths from gang activities into productive activities, including placement in appropriate educational, recreational, and employment programs, and for alcohol or other drug abuse education and treatment services for participants in that organization's youth diversion program. organization that is located in ward 1 in the city of Racine shall have a recreational facility, shall offer programs to divert youths from gang activities, may not be affiliated with any national or state association, and may not have entered into a contract under s. 301.265 (3), 1995 stats. Notwithstanding s. 16.75 and except for

the contract that must be awarded to an organization that is be located in the city of Racine, the department may enter into —a—any contract under this subsection without soliciting bids or proposals and without accepting the lowest responsible bid or offer.".

**1902.** Page 1366, line 17: after that line insert:

"Section 3128g. 301.28 (2) (c) of the statutes is created to read:

301.28 (2) (c) The department may not require a correctional officer, who is certified under s. 146.50 (6g) to use an automated external defibrillator, to attend a training program on first aid, on the use of cardiopulmonary resuscitation, or on the use of an automated external defibrillator.".

**1903.** Page 1366, line 18: delete the material beginning with that line and ending with page 1367, line 2.

**1904.** Page 1367, line 2: after that line insert:

"Section 3128r. 301.30 of the statutes is amended to read:

301.30 Inmate wages, allowances and release payments. The department may pay a wage or an allowance and a release payment to inmates at its institutions, except that the department may not pay an inmate for participating in an education or program assignment and may not pay an inmate who is involuntarily unassigned. An inmate is involuntarily unassigned if he or she is willing and able to participate in a work, education, or program assignment, but no assignment is available to him or her. The department shall prescribe the amounts of pay and such hours, health and other conditions in connection with employment as are reasonable.".

**1905.** Page 1367, line 2: after that line insert:

1 "Section 3128p. 301.289 of the statutes is created to read:

- **301.289 Unit supervisor positions.** The department may not employ a unit supervisor or a person having comparable duties to supervise correctional institution security staff unless the person directly reports to the institution's security director."
- **1906.** Page 1367, line 3: delete lines 3 to 25.
- **1907.** Page 1368, line 6: delete "\$50" and substitute "\$50 \$100".
- **1908.** Page 1368, line 21: delete lines 21 and 22.
- **1909.** Page 1371, line 16: delete "has received" and substitute "receives".
- **1910.** Page 1374, line 20: after that line insert:
- "Section 3166n. 301.49 of the statutes is created to read:
  - 301.49 Vehicle registration plates for certain sex offenders. (1) Who is covered. A person who is subject to the global positioning system tracking requirement under s. 301.48 (2) or (2m) is subject to the vehicle registration plate requirement and prohibition under sub. (3) until the person is released from the global positioning system tracking requirement under s. 301.48 (2) or (2m).
  - (2) Information access. The department shall provide to the department of justice information sufficient to identify the persons who are covered under sub. (1) for the purpose of s. 165.8285 (1m).
  - (3) Vehicle registration plate requirements. (a) A person covered under sub. (1) who applies to the department of transportation for registration of a motor vehicle under ch. 341 shall identify himself or herself in the application as a person to whom special registration plates shall be issued under s. 341.14 (6y).

- (b) If a person registered in his or her name any motor vehicle under ch. 341 prior to becoming covered under sub. (1), the person shall, immediately after becoming covered under sub. (1), apply to the department of transportation for replacement registration plates under s. 341.16 (1) (c) for each such vehicle, identifying himself or herself in the application for replacement registration plates as a person to whom special registration plates shall be issued under s. 341.14 (6y).
- (c) A person covered under sub. (1) may not operate on a highway, as defined in s. 340.01 (22), a motor vehicle required to be registered under ch. 341 unless it displays registration plates issued under s. 341.14 (6y).
- (4) Penalties. (a) Whoever intentionally violates sub. (3) (a) or (b) is guilty of a Class G felony.
  - (b) Whoever intentionally violates sub. (3) (c) is guilty of a Class H felony.
- (5) NOTIFICATION. (a) The department shall notify a person under sub. (1) of his or her need to comply with the vehicle registration plate requirement and prohibition under sub. (3).
- (b) After notifying a person under par. (a) of the need to comply with this section, the person who is providing the notification shall require the person who is being notified to read and sign a form stating that he or she has been informed of the requirements of this section.
- (c) It is not a defense to liability under sub. (4) (a) or (b) that the person subject to sub. (3) was not required to read and sign a form under par. (b), was not provided with a form to read and sign under par. (b), or failed or refused to read or sign a form under par. (b). It is not a defense to liability under sub. (4) (a) or (b) that the person subject to sub. (3) did not receive notice under this subsection from the department.".

- **1911.** Page 1374, line 21: delete lines 21 to 24.
- **1912.** Page 1375, line 1: delete lines 1 and 2.
- **1913.** Page 1375, line 2: after that line insert:
- 4 "Section 3177m. 302.12 (1) of the statutes is amended to read:
  - 302.12 (1) The department may provide by rule for the payment of money to inmates, except that the department may not pay an inmate for participating in an education or program assignment and may not pay an inmate who is involuntarily unassigned. An inmate is involuntarily unassigned if he or she is willing and able to participate in a work, education, or program assignment, but no assignment is available to him or her. The rate may vary for different prisoners in accordance with the pecuniary value of the work performed, willingness, and good behavior. The payment of money to inmates working in the prison industries shall be governed by s. 303.01 (4).".
  - **1914.** Page 1375, line 3: delete lines 3 to 20.
- **1915.** Page 1375, line 21: delete the material beginning with that line and ending with page 1376, line 15 and substitute:
  - "Section 3180d. 302.386 (1) of the statutes is amended to read:
    - 302.386 (1) Except as provided in sub. (5), liability for medical and dental services furnished to residents housed in prisons identified in s. 302.01, in a juvenile correctional facility, or in a secured residential care center for children and youth, or to forensic patients in state institutions for those services that are not provided by employees of the department or under contract pursuant to s. 301.03 (3k) shall be limited to the amounts payable under ss. 49.43 to 49.47, except s. 49.468, for similar services. The department may waive any such limit if it determines that needed

services cannot be obtained for the applicable amount. No provider of services may bill the resident or patient for the cost of services exceeding the amount of the liability under this subsection.".

**1916.** Page 1376, line 15: after that line insert:

**"Section 3180m.** 302.43 of the statutes is amended to read:

302.43 Good time. Every inmate of a county jail is eligible to earn good time in the amount of one-fourth of his or her term for good behavior if sentenced to at least 4 days, but fractions of a day shall be ignored. An inmate shall be given credit for time served prior to sentencing under s. 973.155, including good time under s. 973.155 (4). An inmate who violates any law or any regulation of the jail, or neglects or refuses to perform any duty lawfully required of him or her, may be deprived by the sheriff of good time under this section, except that the sheriff shall not deprive the inmate of more than 2 days good time for any one offense without the approval of the court. An inmate who files an action or special proceeding, including a petition for a common law writ of certiorari, to which s. 807.15 applies shall be deprived of the number of days of good time specified in the court order prepared under s. 807.15 (3). This section does not apply to a person who is confined in the county jail in connection with his or her participation in a substance abuse treatment program that meets the requirements of s. 16.964 (12) (c) 165.95 (3), as determined by the office department of justice assistance under s. 16.964 (12) (j) 165.95 (9)."

**1917.** Page 1376, line 15: after that line insert:

"Section 3190m. 340.01 (3) (dg) of the statutes is created to read:

340.01 (3) (dg) Privately owned motor vehicles being used by an organ procurement organization, or by any person under an agreement with an organ

procurement organization, to transport organs for human transplantation or to transport medical personnel for the purpose of performing human organ harvesting or transplantation immediately after the transportation.

**Section 3190p.** 340.01 (3) (dh) of the statutes is created to read:

340.01 (3) (dh) Privately owned motor vehicles being operated in the course of a business and being used, in response to an emergency call from a treating physician or his or her designee declaring the transportation to be an emergency, to transport medical devices or equipment to a hospital or ambulatory surgery center, or to pick up medical devices or equipment for immediate transportation to a hospital or ambulatory surgery center, if the medical devices or equipment are to be used for human implantation or for urgent medical treatment immediately after the transportation."

## **1918.** Page 1376, line 15: after that line insert:

"Section 3180h. 302.388(2)(g) of the statutes is created to read:

302.388 (2) (g) If a prisoner's health summary form or complete medical file indicates that the prisoner has a communicable disease and if disclosure of that information is necessary for the health and safety of the prisoner or of other prisoners, of a correctional officer who has custody of or is responsible for the supervision of the prisoner, of a person designated with custodial authority over the prisoner by a jailer, of any other employee of the prison or jail, or of a law enforcement officer or other person who is responsible for transferring the prisoner to or from a prison or jail, receiving institution intake staff shall disclose that information to the persons specified in par. (f) 1. to 4. and to that correctional officer, person with custodial authority, law enforcement officer, or other person.

23

24

1	<b>Section 3180j.</b> 302.388 (3) of the statutes is renumbered 302.388 (3) (a).
2	SECTION 3180L. 302.388 (3) (b) of the statutes is created to read:
3	302.388 (3) (b) If a prisoner's treatment summary indicates that the prisoner
4	has a communicable disease and if disclosure of that information is necessary for the
5	health and safety of the prisoner or of other prisoners, of a correctional officer who
6	has custody of or is responsible for the supervision of the prisoner, of a person
7	designated with custodial authority over the prisoner by a jailer, of any other
8	employee of the prison or jail, or of a law enforcement officer or other person who is
9	responsible for transferring the prisoner to or from a prison or jail, the department
10	or jailer shall disclose that information to the persons to whom a treatment summary
11	may be made available under par. (a) and to that correctional officer, person with
12	custodial authority, law enforcement officer, or other person.".
13	1919. Page 1376, line 16: delete the material beginning with that line and
14	ending with page 1393, line 3, and substitute:
15	"Section 3204bg. 341.01 (2) (c) of the statutes is created to read:
16	341.01 (2) (c) "Sex offender" means a person who is covered under s. $301.49$ (1).
17	<b>Section 3204br.</b> 341.08 (1m) of the statutes is amended to read:
18	341.08 (1m) The forms for application for original registration and for renewal
19	of registration under sub. (1) shall include the information required under s. 85.103
20	(2) and, the applicant's birth date, and whether the applicant is a sex offender.
21	<b>Section 3205ec.</b> 341.14 (6y) of the statutes is created to read:

341.14 (6y) Upon application by a person who is a sex offender, the department shall issue and deliver to the person plates of a special design, in lieu of the plates that ordinarily would be issued for the vehicle, and shall renew the plates. The plates

 $\mathbf{2}$ 

shall have a chartreuse-colored background. The department shall design the plates so as to readily apprise law enforcement officers of the fact that the vehicle is owned by a sex offender. A fee of \$30, in addition to the prescribed registration fee, shall be charged for the issuance or renewal of such plates. If at any time the owner of the vehicle is no longer a sex offender, the owner may dispose of these special plates in a manner prescribed by the department and shall obtain replacement plates for the vehicle.

**Section 3205eg.** 341.145 (1) (g) of the statutes is created to read:

341.145 (1) (g) A registration plate of the same color and design as provided in s. 341.14 (6y) for a vehicle registered by a sex offender, which displays a registration number composed of letters or numbers, or both, not exceeding 6 positions and not less than one position, requested by the applicant.

**SECTION 3205ev.** 341.145 (1g) (f) of the statutes is created to read:

341.145 (1g) (f) The department may issue personalized registration plates under sub. (1) (g) to a person who is required to obtain special plates issued under s. 341.14 (6y).

**Section 3205he.** 341.16 (1) (c) of the statutes is created to read:

341.16 (1) (c) Upon receipt of an application by a sex offender required under s. 301.49 (3) (b) to be issued special registration plates under s. 341.14 (6y), along with a fee of \$15 for each plate, the department shall issue replacement plates under s. 341.14 (6y). Upon receipt of these replacement registration plates, the applicant shall remove and destroy all registration plates replaced.

**SECTION 3205hv.** 341.16 (4) of the statutes is amended to read:

1	341.16 (4) Any person issued replacement plates who fails to destroy the
2	original plates as required by sub. $\underline{(1)(c)}$ , $\underline{(2)}$ , or $\underline{(3)}$ may be required to forfeit not more
3	than \$200.".
4	<b>1920.</b> Page 1393, line 3: after that line insert:
5	"Section 3205dz. $341.14~(6r)~(f)~19m$ . of the statutes is amended to read:
6	341.14 (6r) (f) 19m. Persons who have had an immediate family member die
7	in combat while serving in the U.S. armed forces or in forces incorporated as part of
8	the U.S. armed forces if the immediate family member meets any of the eligibility
9	criteria specified in 10 USC 1126 (a) (1) to (3) for a gold star lapel button.".
10	<b>1921.</b> Page 1393, line 13: after that line insert:
11	"Section 3209bd. 341.25 (2) (cm) to (q) of the statutes are amended to read:
12	$341.25$ (2) (cm) Not more than $10,000 \dots 119.50 \underline{137.00}$
13	(d) Not more than 12,000
14	(e) Not more than 16,000
15	(f) Not more than $20,000 \dots 274.00 \underline{315.00}$
16	(g) Not more than $26,000 \dots 365.50 \underline{420.00}$
17	(h) Not more than 32,000
18	(i) Not more than 38,000 <u>593.50</u> <u>683.00</u>
19	(j) Not more than 44,000
20	(k) Not more than 50,000
21	(km) Not more than 54,000
22	(L) Not more than 56,000
23	(m) Not more than 62,000
24	(n) Not more than 68,000

- 3 (q) Not more than  $80,000 \dots \frac{1,969.50}{2,265.00}$ ".
- 1922. Page 1393, line 14: delete the material beginning with that line and ending with page 1404, line 9.
- 6 **1923.** Page 1404, line 10: delete lines 10 to 25.
- 7 **1924.** Page 1405, line 1: delete lines 1 to 7.
- 8 **1925.** Page 1405, line 8: delete the material beginning with that line and ending with page 1406, line 9.
- 10 **1926.** Page 1406, line 10: delete lines 10 to 23.
- 11 **1927.** Page 1406, line 23: after that line insert:
- 12 "Section 3215m. 342.14 (1) of the statutes is amended to read:
- 13 342.14 (1) For filing an application for the first certificate of title, \$28.50 \$40.50, by the owner of the vehicle.".
- 15 **1928.** Page 1407, line 4: after that line insert:
- **"Section 3216n.** 342.14 (3) of the statutes is amended to read:
- 17 342.14 (3) For a certificate of title after a transfer, \$28.50 \$40.50, by the owner of the vehicle.".
- 19 **1929.** Page 1407, line 5: delete lines 5 to 10.
- 20 **1930.** Page 1407, line 11: delete the material beginning with that line and ending with page 1408, line 18.
- 22 **1931.** Page 1408, line 21: after that line insert:
- 23 "Section 3220c. 343.01 (2) (dg) of the statutes is renumbered 340.01 (41k).".

- **1932.** Page 1412, line 16: delete lines 16 to 25.
- **1933.** Page 1413, line 1: delete lines 1 and 2.
- **1934.** Page 1414, line 11: delete "children and families" and substitute "workforce development".
- **1935.** Page 1429, line 11: delete lines 11 to 22.
- **1936.** Page 1430, line 14: after that line insert:
- 7 "Section 3315k. 343.307 (1) (intro.) of the statutes is amended to read:
- 8 343.307 (1) (intro.) The court shall count the following to determine the length 9 of a revocation under s. 343.30 (1q) (b) and to determine the penalty under s. 114.09 10 (2) and 346.65 (2):
- **Section 3315s.** 343.307 (1) (g) of the statutes is created to read:
- 12 343.307 (1) (g) Convictions for violations under s. 114.09 (1) (b) 1. or 1m.".
- **1937.** Page 1431, line 1: delete lines 1 to 10.
- **1938.** Page 1439, line 16: delete "workforce development children and families" and substitute "workforce development".
- **1939.** Page 1440, line 17: delete the material beginning with that line and ending with page 1441, line 13.
- **1940.** Page 1441, line 14: delete lines 14 to 24.
- **1941.** Page 1442, line 1: delete lines 1 to 24.
- **1942.** Page 1443, line 1: delete lines 1 and 2.
- **1943.** Page 1444, line 22: delete the material beginning with that line and ending with page 1445, line 13.
- **1944.** Page 1445, line 25: after that line insert:

**"Section 3427c.** 346.03 (1) of the statutes is amended to read:

346.03 (1) The operator of an authorized emergency vehicle, when responding to an emergency call or when in the pursuit of an actual or suspected violator of the law er, when responding to but not upon returning from a fire alarm, when transporting an organ for human transplantation, or when transporting medical personnel for the purpose of performing human organ harvesting or transplantation immediately after the transportation, may exercise the privileges set forth in this section, but subject to the conditions stated in subs. (2) to (5) (5m).

**Section 3427e.** 346.03 (5m) of the statutes is created to read:

346.03 (5m) The privileges granted under this section apply to the operator of an authorized emergency vehicle under s. 340.01 (3) (dg) or (dh) only if the operator has successfully completed a safety and training course in emergency vehicle operation that is taken at a technical college under ch. 38 or that is approved by the department and only if the vehicle being operated is plainly marked, in a manner prescribed by the department, to identify it as an authorized emergency vehicle under s. 340.01 (3) (dg) or (dh).".

**1945.** Page 1446, line 1: before that line insert:

**"Section 3428m.** 346.49 (1) (a) of the statutes is amended to read:

346.49 (1) (a) Unless otherwise provided in par. (b) or (c), any person violating s. 346.46 (1), (2m) or (4) or 346.47 may be required to forfeit not less than \$20 nor more than \$40 for the first offense and not less than \$50 nor more than \$100 for the 2nd or subsequent conviction within a year.

**Section 3428n.** 346.49 (1) (am) of the statutes is created to read:

346.49 (1) (am) Unless otherwise provide in par. (b) or (c), any person violating s. 346.46 (2m) may be required to forfeit not less than \$30 nor more than \$300.

**Section 3428p.** 346.49 (1) (c) of the statutes is amended to read:

346.49 (1) (c) If an operator of a vehicle violates s. 346.46 (1) where persons engaged in work in a highway maintenance or construction area or in a utility work area are at risk from traffic, any applicable minimum and maximum forfeiture specified in par. (a) or (am) for the violation shall be doubled.".

**1946.** Page 1446, line 1: delete the material beginning with that line and ending with page 1450, line 7.

**1947.** Page 1450, line 7: after that line insert:

**"Section 3432c.** 347.25 (1) of the statutes is amended to read:

347.25 (1) Except as provided in subs. (1m) (a), (1r), and (1s), an authorized emergency vehicle may be equipped with one or more flashing, oscillating, or rotating red lights, except that ambulances, fire department equipment, and privately owned motor vehicles under s. 340.01 (3) (d), (dg), or (dm) being used by personnel of a full-time or part-time fire department or, by members of a volunteer fire department or rescue squad, or by an organ procurement organization or any person under an agreement with an organ procurement organization, and privately owned motor vehicles under s. 340.01 (3) (dh) being used to transport or pick up medical devices or equipment, may be equipped with red or red and white lights, and shall be so equipped when the operator thereof is exercising the privileges granted by s. 346.03. The lights shall be so designed and mounted as to be plainly visible and understandable from a distance of 500 feet both during normal sunlight and during hours of darkness. No operator of an authorized emergency vehicle may use the

warning lights except when responding to an emergency call or when in pursuit of an actual or suspected violator of the law, when responding to but not upon returning from a fire alarm, when transporting an organ for human transplantation, when transporting medical personnel for the purpose of performing human organ harvesting or transplantation immediately after the transportation, or when necessarily parked in a position which is likely to be hazardous to traffic.

**Section 3432e.** 347.38 (4) of the statutes is amended to read:

347.38 (4) An authorized emergency vehicle shall be equipped with a siren, but such siren shall not be used except when such vehicle is operated in response to an emergency call or in the immediate pursuit of an actual or suspected violator of the law, when responding to but not upon returning from a fire alarm, when transporting an organ for human transplantation, or when transporting medical personnel for the purpose of performing human organ harvesting or transplantation immediately after the transportation, in which events the driver of such vehicle shall sound the siren when reasonably necessary to warn pedestrians and other drivers.".

- **1948.** Page 1452, line 4: delete lines 4 to 17.
- **1949.** Page 1452, line 17: after that line insert:
- 18 "Section 3435x. 349.132 of the statutes is created to read:
  - 349.132 Authority to immobilize, remove, impound, and dispose of motor vehicles for nonmoving traffic violations. (1) In this section:
    - (a) "Habitual parking violator" means a person who has received, more than 28 days previously, 3 or more citations for nonmoving traffic violations that remain unpaid and for which the person has not scheduled an appearance in court in response to the citations.

- (b) "Immobilization device" has the meaning given in s. 341.65 (1) (a).
- 2 (c) "Nonmoving traffic violation" has the meaning given in s. 345.28 (1) (c).
  - (d) "Owner" has the meaning given in s. 341.65 (1) (am).
  - (e) "Parking enforcer" means a traffic officer or any other person who enforces nonmoving traffic violations and who is employed by a municipality or county.
  - (2) The governing body of any municipality or county may by ordinance provide for the immobilization or removal, impoundment, and disposal of vehicles owned by habitual parking violators as provided in this section. Any ordinance under this section shall do all of the following:
  - (a) Limit application of the ordinance to those motor vehicles for which all of the following apply:
  - 1. The municipality or county has cited the owner of the motor vehicle for 3 or more nonmoving traffic violations that, at the time of the vehicle's immobilization or removal, occurred more than 28 days previously and for which the owner has neither paid the forfeiture for each of these violations nor scheduled an appearance in court in response to each of these citations.
  - 2. a. The municipality or county has mailed to the last-known address of the owner at least one notice that specifies, for each citation counted under subd. 1., the date on which the citation was issued, the license number of the vehicle involved, the place where the citation may be paid, the amount of the forfeiture, and the means by which the citation may be contested.
  - b. The notice under subd. 2. a. shall also inform the owner that any motor vehicle owned by him or her may be immobilized with an immobilization device or removed and impounded if, within 28 days after the owner has received 3 or more citations and at the time the vehicle is immobilized or removed and impounded, the

- owner has not either paid the forfeiture for each violation that occurred more than 28 days previously or scheduled an appearance in court in response to each citation issued more than 28 days previously for which the forfeiture has not been paid.
- c. The notice under this subdivision may be combined with any other notice provided by the municipality or county to the owner.
- (b) Authorize any parking enforcer who discovers any motor vehicle to which par. (a) applies that is legally or illegally parked on any portion of the street, highway, or publicly owned or leased parking facility within the corporate limits of the municipality or county to cause the motor vehicle to be immobilized with an immobilization device or removed to a suitable place of impoundment or both. Upon immobilization or removal of the motor vehicle, the parking enforcer shall follow the notification procedure specified in s. 341.65 (2) (b).
- (c) Specify whether the municipality or county may contract with a 3rd party for the performance of services related to immobilization or removal of motor vehicles, which services shall be rendered only at the request of a parking enforcer.
- (d) Provide for a reasonable removal fee, if any, that will be charged to remove an immobilization device placed on a vehicle under this section.
- (e) Provide for the recovery of reasonable towing or storage charges associated with the removal or impoundment of a vehicle, and of reasonable charges associated with disposal of a vehicle, under this section.
- (f) Require that, if the motor vehicle is immobilized, the parking enforcer or a 3rd party contractor place in a highly visible location and a reasonably secure manner on the vehicle, at the time of immobilization, a written notice that does all of the following:

- 1. Warns any driver of the vehicle that the immobilization device has been placed on the vehicle.
- 2. Either provides all of the information specified in par. (a) 2. a. or provides a telephone number at which an individual is available to provide such information 24 hours a day.
- 3. States the amount of the removal fee under par. (d), if any, that is in addition to any amount required to be paid as specified in the notice under par. (a) 2. a.
- (g) If the motor vehicle is immobilized in a time-limited, legal parking space, prohibit the municipality or county from issuing, after the vehicle's immobilization, any citation for a nonmoving traffic violation for the vehicle within the first 4 hours after the vehicle is immobilized and during any hours in which the municipal court or clerk's office of the circuit court that would be contacted to arrange an appearance for purposes of sub. (3) (b) is not open for regular business.
- (h) If the motor vehicle is immobilized, require the municipality or county, or a 3rd party contractor, to remove, or provide sufficient information to allow the vehicle owner to remove, the immobilization device without undue delay, not to exceed 3 hours, after receiving notice that the person has satisfied the requirements for release of the motor vehicle under sub. (3) (b).
- (3) (a) Any motor vehicle immobilized or impounded as provided in sub. (2) shall remain immobilized or impounded until lawfully claimed or disposed of as provided in this subsection and sub. (5).
- (b) The owner of a motor vehicle that is immobilized under sub. (2) may secure release of the motor vehicle by paying any removal fee specified in sub. (2) (d) and either paying all forfeitures specified in each notice under sub. (2) (a) 2. a. for citations counted under sub. (2) (a) 1. or scheduling an appearance in court in

response to all citations counted under sub. (2) (a) 1. for which the forfeitures have not been paid.

- (c) The owner of a motor vehicle that is removed and impounded under sub. (2) may secure release of the motor vehicle by paying any charges specified in sub. (2) (e) and either paying all forfeitures specified in the notice under sub. (2) (a) 2. a. for citations counted under sub. (2) (a) 1. or scheduling an appearance in court in response to all citations counted under sub. (2) (a) 1. for which the forfeitures have not been paid.
- (d) If an owner secures release of a motor vehicle under par. (b) or (c) by scheduling an appearance in court and thereafter fails to appear or fails to comply with any court order with respect to any citation counted under sub. (2) (a) 1. for which the forfeiture has not been fully paid, including failure to satisfy in full any court-ordered payment plan or other agreement approved by the court, the court may order a law enforcement officer, or an authorized employee or contractor of the municipality or county, to immobilize the motor vehicle involved in the nonmoving traffic violations or the municipality or county may cause the motor vehicle to be immobilized or removed and impounded as provided under sub. (2). If the court orders the motor vehicle immobilized, upon compliance with the court order, the court shall order a law enforcement officer, or an authorized employee or contractor of the municipality or county, to remove the immobilization device.
- (e) Notwithstanding par. (a), if any motor vehicle immobilized or impounded is an unregistered motor vehicle for purposes of s. 341.65 or an abandoned motor vehicle for purposes of s. 342.40, the municipality or county may take any action authorized under s. 341.65 or 342.40. Any vehicle immobilized under this section for

longer than the period specified in s. 342.40 (1m) shall be considered abandoned for purposes of s. 342.40.

- (4) The owner of any motor vehicle immobilized or removed and impounded as provided under this section is responsible for all charges associated with immobilizing, removing, impounding, and disposing of the motor vehicle, as provided under sub. (2) (d) and (e). Charges not recovered from the sale of the motor vehicle may be recovered in a civil action by the municipality or county against the owner.
- (5) The procedures and provisions of s. 341.65 (2) (f) to (h) shall apply with respect to the impoundment and disposal of motor vehicles authorized to be removed, impounded, and disposed of under this section to the same extent as these provisions apply to the impoundment and disposal of unregistered motor vehicles that are removed under authority of s. 341.65, except that reclamation of the motor vehicle by the owner requires compliance with sub. (3) rather than s. 341.65 (2) (e). The provisions of s. 349.13 (5) (b) shall apply with respect to vehicles removed or stored under this section to the same extent as these provisions apply with respect to vehicles removed or stored under authority of s. 349.13.
- (6) Any ordinance enacted under this section permitting immobilization of a motor vehicle may prohibit any person from removing, disconnecting, tampering with, or otherwise circumventing the operation of an immobilization device installed under this section except upon release of the motor vehicle to the owner or to make necessary repairs to a malfunctioning immobilization device.
- (7) Section 349.137 does not apply to the use of motor vehicle immobilization devices under this section.".
  - **1950.** Page 1452, line 18: delete lines 18 to 25.

- 1 **1951.** Page 1453, line 1: delete lines 1 to 3.
- 2 **1952.** Page 1454, line 24: delete that line.
- 3 **1953.** Page 1455, line 1: delete lines 1 to 20.
- 4 **1954.** Page 1456, line 5: delete lines 5 to 25.
- 5 **1955.** Page 1457, line 1: delete lines 1 to 20.
- 6 **1956.** Page 1458, line 8: delete lines 8 to 11.
- 7 **1957.** Page 1475, line 13: delete lines 13 to 24.
- 8 **1958.** Page 1476, line 1: delete lines 1 to 24.
- 9 **1959.** Page 1476, line 25: delete the material beginning with that line and ending with page 1477, line 9.
- 11 **1960.** Page 1477, line 9: after that line insert:
- **"Section 3563p.** 560.01 (2) (a) (title) of the statutes is amended to read:
- 13 560.01 **(2)** (a) (title) State economic policy <u>and comprehensive economic</u>
  14 development report.
- **Section 3563q.** 560.01 (2) (a) of the statutes is renumbered 560.01 (2) (a) 1.
- **Section 3563r.** 560.01 (2) (a) 2. of the statutes is created to read:
- 560.01 (2) (a) 2. Annually, no later than October 1, the department shall submit to the chief clerk of each house of the legislature for distribution to the legislature under s. 13.172 (2) a comprehensive report assessing economic development programs administered by the department. The report shall include all of the following information:
- a. A summary of the programs administered by the department.

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

- b. Quantifiable performance measures directly related to the purpose of the
   program.
  - c. A comparison of expected and actual program outcomes.
  - d. A summary of program costs and benefits.
    - e. The number of grants and loans awarded under each program.
    - f. The amount of each grant and loan.
    - g. The recipient of each grant or loan.
    - h. The total amount of grants and loans received by each recipient.
    - i. Any recommended changes to an economic development program.

**Section 3563rm.** 560.03 (15) (intro.) of the statutes is amended to read:

560.03 (15) (intro.) Annually, submit Include in the report required under s. 560.01 (2) (a) 2. a summary of the employment impact estimates required under s. 560.034 (2) and a report to the chief clerk of each house of the legislature, for distribution to the legislature under s. 13.172 (2), analyzing the use of industrial revenue bond financing under s. 66.1103 and accurately reporting the benefits of that use, including the effect on employment in this state including, but not limited to:

**Section 3563rp.** 560.03 (26) of the statutes is amended to read:

560.03 (26) In cooperation with the department of financial institutions and the board of regents of the University of Wisconsin System, annually conduct and publish include in the report required under s. 560.01 (2) (a) 2. the results of a study of Wisconsin businesses to determine new business formation trends and identify obstacles faced by new Wisconsin businesses and areas where changes in governmental policy may satisfy the needs of new Wisconsin businesses. As part of the study, the department of commerce shall conduct a survey of Wisconsin businesses."

**1961.** Page 1477, line 15: after that line insert: 1 2 "Section 3563u. 560.038 of the statutes is repealed. 3 **Section 3563w.** 560.039 of the statutes is repealed.". **1962.** Page 1477, line 20: after that line insert: 4 5 "Section 3564d. 560.07 (9) of the statutes is amended to read: 6 560.07 (9) On or before July 1, 1985, and every July 1 thereafter, submit to the 7 chief clerk of each house of the legislature, for distribution to the appropriate 8 standing committees under s. 13.172 (3), a Include in the report stating required 9 under s. 560.01 (2) (a) 2. information regarding the net jobs gain due to the funds 10 provided to Forward Wisconsin, Inc., under s. 20.143 (1) (bm). 11 **Section 3564f.** 560.08 (2) (m) of the statutes is amended to read: 12 560.08 (2) (m) No later than January 1 of each odd-numbered year, submit to 13 the investment board a Include in the report describing required under s. 560.01 (2) 14 (a) 2. a description of the types of investments in businesses in this state which will 15 have the greatest likelihood of enhancing economic development in this state. The 16 department shall submit to the investment board a copy of the information required 17 under this paragraph. 18 **Section 3564h.** 560.081 (2) (i) of the statutes is amended to read: 19 560.081 (2) (i) As part of the report required under s. <del>15.04 (1) (d)</del> 560.01 (2) 20 (a) 2., include a report on the effects of the state main street program. 21 **Section 3564j.** 560.10 of the statutes is repealed. 22 **Section 3564L.** 560.11 (1) (b) of the statutes is amended to read: 23 560.11 (1) (b) Periodically Include in the report to the department of natural 24 resources and to the administrator of the federal environmental protection agency

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

22

23

required under s. 560.01 (2) (a) 2. information concerning the compliance of the state small business stationary source technical and environmental compliance assistance program with the federal paperwork reduction act, 44 USC 3501 to 3520, the federal regulatory flexibility act, 5 USC 601 to 612, and the federal equal access to justice act, 5 USC 504. The department shall submit to the department of natural resources and to the administrator of the federal environmental protection agency a copy of the information required under this paragraph."

**1963.** Page 1477, line 22: delete the material beginning with that line and ending with page 1478, line 18.

**1964.** Page 1478, line 22: after that line insert:

**"Section 3565j.** 560.13 (7) of the statutes is amended to read:

560.13 (7) On or before December 31, 1998, and annually thereafter, the <u>The</u> department shall submit a <u>include in the</u> report on <u>required under s. 560.01 (2) (a)</u>

2. an evaluation of the effectiveness of the program under this section to the legislature under s. 13.172 (2) and. The department shall provide to the governor and the department of administration a copy of the evaluation required under this subsection."

**1965.** Page 1478, line 23: after that line insert:

"Section 3566q. 560.137 of the statutes is repealed.".

**1966.** Page 1479, line 2: after that line insert:

21 "Section 3568g. 560.14 (3m) of the statutes is repealed.".

**1967.** Page 1479, line 5: after that line insert:

**"Section 3571d.** 560.155 (4) of the statutes is amended to read:

560.155 (4) No later than January 31, 2002, the The department shall submit
to the legislature under s. 13.172 (2) a report on include in the report required under
s. 560.01 (2) (a) 2. an evaluation of the operation and effectiveness of the grant
program under this section.".
<b>1968.</b> Page 1479, line 13: after that line insert:
"Section 3574g. 560.17 (5r) of the statutes is repealed.
<b>Section 3574i.</b> 560.17 (6m) (a) of the statutes is renumbered 560.17 (6m) and
amended to read:
560.17 (6m) Except as provided in par. (b), in order to To receive a grant or loan
under this section a person or business shall contribute cash, from a source other
than the state, in an amount that equals at least $25\%$ of the total cost of the project.
SECTION 3574k. 560.17 (6m) (b) of the statutes is repealed.
<b>Section 3574m.</b> 560.17 (7) (a) of the statutes is amended to read:
560.17 (7) (a) Except as provided in par. (am), the department shall designate
staff to evaluate applications for grants or loans and assist the board under this
section. The board shall act on an application for a grant or loan at its next regularly
scheduled meeting after the department determines that the application is complete,
except that the board shall act on an application for a loan under sub. (5r) and advise
the applicant of its decision within 45 days after the department determines that the
application is complete.".
<b>1969.</b> Page 1480, line 5: after that line insert:
"Section 3577k. $560.205(1)(g)$ of the statutes is amended to read:
560.205 (1) (g) It is not engaged in real estate development, insurance,

banking, lending, lobbying, political consulting, professional services provided by

 $\mathbf{2}$ 

attorneys, accountants, business consultants, physicians, or health care consultants, wholesale or retail trade, leisure, hospitality, transportation, or construction, except construction of power production plants that derive energy from a renewable resource, as defined in s. 196.378 (1) (h).".

**1970.** Page 1480, line 5: after that line insert:

"Section 3577q. 560.205 (3) (c) (intro.) of the statutes is amended to read:

560.205 (3) (c) *Annual report*. (intro.) Annually, no later than September 15, the The department shall submit a include in the report to the chief clerk of each house of the legislature for distribution to the legislature under s. 13.172 (2), listing required under s. 560.01 (2) (a) 2. all of the following information:".

**1971.** Page 1480, line 21: after that line insert:

**"Section 3578d.** 560.207 of the statutes is created to read:

**560.207** Workplace wellness programs. (1) The department shall implement a program to certify workplace wellness programs under ss. 71.07 (3h), 71.28 (3h), and 71.47 (3h).

(2) If the department certifies a workplace wellness program under sub. (1), the department shall determine the amount of workplace wellness program credits to allocate to the business providing the workplace wellness program. The total amount of workplace wellness program credits allocated to businesses in any year may not exceed \$5,000,000. In any year, the department may not allocate more than \$2,500,000 in credits to businesses with more than 50 employees, and may not allocate more than \$2,500,000 in credits to businesses with 50 or fewer employees.

(3) The department shall inform the department of revenue of every business
whose workplace wellness program is certified under sub. (1) and the amount of
credits allocated to the business.
(4) The department, in consultation with the department of revenue and the
department of health and family services, shall promulgate rules to administer this
section.".
<b>1972.</b> Page 1480, line 21: after that line insert:
"Section 3578d. 560.2055 of the statutes is created to read:
560.2055 Grants for venture center. The department may not make grants
to establish, start up, operate, or maintain a corporation for the purpose of raising
capital to promote investments in emerging industries in this state unless the
corporation satisfies the following requirements:
(1) The corporation is established as a nongovernmental organization with a
mission to facilitate access to venture capital for Wisconsin-based businesses.
(2) The board of directors of the corporation includes all of the following:
(a) High-level business executives.
(b) The presidents or chief operating officers of business associations.
(c) Partners, officers, or shareholders of institutional investment businesses.
(d) Executives from technology transfer organizations.
(3) The corporation collaborates with governmental and academic economic
development organizations working for the improvement of Wisconsin's technology
business sectors.".
<b>1973.</b> Page 1480, line 21: after that line insert:
business sectors.".

"Section 3578h. 560.208 of the statutes is created to read:

- **560.208** Dairy manufacturing facility investment credit. (1) The department of commerce shall implement a program to certify taxpayers as eligible for the dairy manufacturing facility investment credit under ss. 71.07 (3p), 71.28 (3p), and 71.47 (3p).
- (2) If the department of commerce certifies a taxpayer under sub. (1), the department of commerce shall determine the amount of credits to allocate to that taxpayer. The total amount of dairy manufacturing facility investment credits allocated to taxpayers in fiscal year 2007–08 may not exceed \$600,000 and the total amount of dairy manufacturing facility investment credits allocated to taxpayers in fiscal year 2008–09, and in each fiscal year thereafter, may not exceed \$700,000.
- (3) The department of commerce shall inform the department of revenue of every taxpayer certified under sub. (1) and the amount of credits allocated to the taxpayer.
- (4) The department of commerce, in consultation with the department of revenue, shall promulgate rules to administer this section.".

## **1974.** Page 1484, line 1: before that line insert:

"Section 3581zd. 560.42(5) of the statutes is amended to read:

560.42 (5) Report. Beginning in 2003 and biennially thereafter, the Annually, the center shall prepare a report describing its activities under this section since the period covered in the previous report. The department shall submit the report with the report required under s. 560.55. The report may include recommendations for the legislature, governor, public records board, and regulatory agencies on simplifying the process of applying for permits, of reviewing and making determinations on permit applications, and of issuing permits, and shall include

 $\mathbf{2}$ 

information on the number of requests for assistance, the types of assistance provided, and the center's success in resolving conflicts in permit application and review processes. The department shall include the information required under this subsection in the report required under s. 560.01 (2) (a) 2.

**Section 3581zf.** 560.52 (1) (intro.) of the statutes is amended to read:

560.52 (1) INVENTORY. (intro.) The department shall prepare and periodically, revise, and include in the report required under s. 560.01 (2) (a) 2. an inventory of existing entrepreneurial assistance programs offered in this state. In preparing the inventory, the department may do any of the following:

**Section 3581zh.** 560.55 of the statutes is amended to read:

of each odd-numbered year thereafter, the The department shall submit to include in the governor and to the chief clerk of each house of the legislature, for distribution to the legislature under s. 13.172 (2), a report describing required under s. 560.01 (2) (a) 2. a description of the department's activities and the result of the department's activities under s. 560.54 since the period covered in the previous report submitted under this section. The department shall combine this report with the report required under s. 560.42 (5) and may combine this report with other reports published by the department, including the report under s. 15.04 (1) (d) provide to the governor a copy of the information required under this section. The report may include recommendations for legislative proposals to change the entrepreneurial assistance programs and intermediary assistance programs."

**1975.** Page 1487, line 15: after that line insert:

**"Section 3621d.** 560.607 (2) of the statutes is repealed.".

1	<b>1976.</b> Page 1489, line 10: delete lines 10 to 25.
2	<b>1977.</b> Page 1489, line 25: after that line insert:
3	"Section 3636d. 560.799 (title) of the statutes is amended to read:
4	560.799 (title) Enterprise Rural enterprise zone.
5	<b>Section 3636e.</b> 560.799 (1) (bm) 1. of the statutes is amended to read:
6	560.799 (1) (bm) 1. A business' employees in a economic rural enterprise zone.
7	Section 3636f. 560.799 (3) (a) of the statutes is amended to read:
8	560.799 (3) (a) The department may designate not more than 10 rural
9	enterprise zones. The department may designate an area as a <u>rural</u> enterprise zone
10	if the area does not exceed 50 acres and if the area does not include any part of a city
11	of the 1st class or a city with a population greater than 200,000.
12	Section 3636g. 560.799 (5) (a) of the statutes is amended to read:
13	560.799 (5) (a) A business that begins operations in a <u>rural</u> enterprise zone.
14	<b>Section 3636h.</b> 560.799 (5) (b) of the statutes is amended to read:
15	560.799 (5) (b) A business that relocates to a <u>rural</u> enterprise zone from outside
16	this state, if the business offers compensation and benefits to its employees working
17	in the zone for the same type of work that are at least as favorable as those offered
18	to its employees working outside the zone, as determined by the department.
19	<b>Section 3636i.</b> 560.799 (5) (c) (intro.) of the statutes is amended to read:
20	560.799 (5) (c) (intro.) A business that expands operations in a <u>rural</u> enterprise
21	zone, but only if any of the following apply:
22	<b>Section 3636j.</b> 560.799 (5) (c) 1. b. of the statutes is amended to read:
23	560.799 (5) (c) 1. b. The business offers compensation and benefits for the same
24	type of work to its employees working in the <u>rural</u> enterprise zone that are at least

1	as favorable as those offered to its employees working in this state but outside the
2	zone, as determined by the department.
3	<b>Section 3636k.</b> 560.799 (5) (c) 2. (intro.) of the statutes is amended to read:
4	560.799 (5) (c) 2. (intro.) The business makes a capital investment in property
5	located in the <u>rural</u> enterprise zone and all of the following apply:
6	<b>Section 3636L.</b> 560.799 (6) (b) 2. of the statutes is amended to read:
7	560.799 (6) (b) 2. Leaves the <u>rural</u> enterprise zone to conduct substantially the
8	same business outside of the <u>rural</u> enterprise zone.
9	<b>Section 3636m.</b> 560.799 (6) (b) 3. of the statutes is amended to read:
10	560.799 (6) (b) 3. Ceases operations in the <u>rural</u> enterprise zone and does not
11	renew operation of the business or a similar business in the <u>rural</u> enterprise zone
12	within 12 months.".
13	1978. Page 1490, line 7: after that line insert:
14	"Section 3638b. 560.80 (3m) of the statutes is repealed.
15	<b>Section 3638d.</b> 560.80 (5) of the statutes is amended to read:
16	560.80 (5) "Eligible recipient" means a person who is eligible to receive a grant
17	$under\ s.\ 560.82\ (5)\ (a)\ or\ 560.837\ or\ a\ grant\ or\ loan\ under\ s.\ 560.83\ (5)\ (a)\ or\ (b).$
18	SECTION 3638f. 560.80 (5m) of the statutes is repealed.
19	<b>Section 3638h.</b> 560.80 (11) of the statutes is amended to read:
20	560.80 (11) "Project" means a development project, an early planning project,
21	a finance project, an education and training project or a revolving fund project.
22	Section 3638j. 560.81 (4) of the statutes is repealed.
23	SECTION 3638L. 560.837 of the statutes is repealed.
24	<b>Section 3638n.</b> 560.84 (1) (f) of the statutes is amended to read:

 $\mathbf{2}$ 

560.84 (1) (f) That the project meets all criteria set forth in s. 560.82, or 560.83, whichever is appropriate.

**SECTION 3638p.** 560.84 (1) (j) of the statutes is amended to read:

560.84 (1) (j) If a development project, finance project, or education and training project, that funds from the grant or loan will not be used to refinance existing debt.

**Section 3638r.** 560.85 (3) (b) of the statutes is amended to read:

560.85 (3) (b) Develop procedures, with the approval of the board, to evaluate applications, monitor project performance, and audit grants and loans awarded for development projects under s. 560.83, projects under s. 560.835, 2001 stats., and finance projects and education and training projects under s. 560.837, 2005 stats.

**Section 3638t.** 560.905 (2) of the statutes is amended to read:

560.905 (2) The standing committees of the senate and assembly with jurisdiction over science and technology shall advise the department concerning the administration of this subchapter. The department shall consider the advice of the committees in carrying out the functions under this subchapter. Annually, the The department shall submit a include in the report on required under s. 560.01 (2) (a) 2. information regarding the department's activities to the chief clerk of each house of the legislature for distribution under s. 13.172 (3) under this subchapter and shall request that the information required under this subsection be distributed to the standing committees with jurisdiction over science and technology.

**Section 3638v.** 560.91 of the statutes is repealed.

**Section 3638x.** 560.915 (2) and (3) of the statutes are repealed.

**SECTION 3638z.** 560.915 (4) (a) of the statutes is renumbered 560.915 (4) (a) (intro.) and amended to read:

560.915 (4	(a) (intro.) Prepare a biennial <u>Include in the</u> report <u>required under</u>
s. 560.01 (2) (a	) 2. information on the disposition and repayment of moneys spent
under <del>sub. (2).</del>	s. 560.915 (2), 2005 stats., and request that the information required
under this para	agraph be distributed to the standing committees with jurisdiction
over economic	development, until the year in or after which the earlier of the
following occur	<u>s:</u>
Section 3	<b>3638zb.</b> 560.915 (4) (b) (intro.) of the statutes is repealed.
Section 3	<b>3638zd.</b> 560.915 (4) (b) 1. of the statutes is renumbered 560.915 (4)
(a) 1. and amer	nded to read:
560.915 (4	4) (a) 1. All financial assistance provided by the department under
sub. (2) (b) s. 50	60.915 (2) (b), 2005 stats., has been repaid.
Section 3	<b>3638zf.</b> 560.915 (4) (b) 2. of the statutes is renumbered 560.915 (4) (a)
2. and amende	d to read:
560.915 (4	(a) 2. The secretary certifies in the report that the department will
not provide mo	ney to any fund under sub. (2) (b) s. 560.915 (2) (b), 2005 stats.
Section 3	<b>3638zh.</b> 560.915 (4) (c) of the statutes is amended to read:
560.915 (4	(c) Deposit all money received in repayment of financial assistance
under <del>sub. (2) (</del>	(b) s. 560.915 (2) (b), 2005 stats., in the general fund.
Section 3	<b>3638zj.</b> 560.925 of the statutes is repealed.
Section 3	<b>3638zL.</b> 560.935 of the statutes is repealed.".
<b>1979.</b> F	Page 1490, line 8: delete lines 8 to 25.
<b>1980.</b> F	Page 1491, line 1: delete lines 1 to 25.
<b>1981.</b> F	Page 1492, line 1: delete lines 1 to 7.
	Page 1493, line 13: delete lines 13 to 25.

- 1 **1983.** Page 1494, line 1: delete lines 1 to 25.
- 2 **1984.** Page 1495, line 1: delete lines 1 to 7.
- 3 **1985.** Page 1495, line 13: after that line insert:
- 4 "Section 3652j. 601.41 (12) of the statutes is created to read:
- 5 601.41 (12) Substantially similar health care coverage plan. The commissioner shall promulgate rules that set out a standardized summary of benefits provided under health care coverage plans, including plans offered under s. 40.51 (7), for use in determining whether a health care coverage plan is substantially similar to a plan offered under s. 40.51 (7).".
- 1986. Page 1495, line 17: delete "developing" and substitute "approving".
- 11 **1987.** Page 1497, line 21: after that line insert:
- 12 "Section **3660b.** 609.76 of the statutes is created to read:
- 609.76 Coverage of student on medical leave. Limited service health organizations, preferred provider plans, and defined network plans are subject to s. 632.895 (15).".
- 16 **1988.** Page 1497, line 22: delete lines 22 to 24.
- 17 **1989.** Page 1498, line 1: delete lines 1 to 25.
- 18 **1990.** Page 1499, line 1: delete lines 1 to 11.
- 19 **1991.** Page 1500, line 7: delete lines 7 to 24.
- 20 **1992.** Page 1501, line 1: delete lines 1 to 25.
- 21 **1993.** Page 1502, line 1: delete lines 1 to 25.
- 22 **1994.** Page 1503, line 1: delete lines 1 to 24.
- 23 **1995.** Page 1504, line 1: delete lines 1 and 2.

- **1996.** Page 1504, line 3: delete lines 3 and 4.
- **1997.** Page 1504, line 5: delete lines 5 to 8.
- **1998.** Page 1504, line 8: after that line insert:
- 4 "Section 3687w. 632.895 (15) of the statutes is created to read:
  - 632.895 (15) COVERAGE OF STUDENT ON MEDICAL LEAVE. (a) Subject to pars. (b) and (c), every disability insurance policy, and every self-insured health plan of the state or a county, city, town, village, or school district, that provides coverage for a person as a dependent of the insured because the person is a full-time student shall continue to provide dependent coverage for the person if, due to a medically necessary leave of absence, he or she ceases to be a full-time student.
  - (b) A policy or plan is not required to continue coverage under par. (a) unless the person submits documentation and certification of the medical necessity of the leave of absence from the person's attending physician. The date on which the person submits the documentation and certification shall be the date on which the coverage continuation under par. (a) begins.
  - (c) A policy or plan is required to continue coverage under par. (a) only until any of the following occurs:
  - 1. The person advises the policy or plan that he or she does not intend to return to school full time.
    - 2. The person becomes employed full time.
    - 3. The person obtains other health care coverage.
- 4. The person marries and is eligible for coverage under his or her spouse's health care coverage.

 $\mathbf{2}$ 

	5. The person reaches the age at which coverage as a dependent who is a
fu	all-time student would otherwise end under the terms and conditions of the policy
or	r plan.

- 6. Coverage of the insured through whom the person has dependent coverage under the policy or plan is discontinued or not renewed.
- 7. One year has elapsed since the person's coverage continuation under par. (a) began and the person has not returned to school full time.".
- **1999.** Page 1504, line 9: delete lines 9 to 24.
- **2000.** Page 1505, line 1: delete lines 1 to 25.
- **2001.** Page 1506, line 1: delete lines 1 to 8.
- **2002.** Page 1508, line 8: after that line insert:
- **"Section 3701s.** 655.002 (1) (e) of the statutes is amended to read:
- 13 655.002 (1) (e) A Regardless of where organized, a corporation organized and
  14 operated in this state for the primary purpose of providing the medical services of
  15 physicians or nurse anesthetists.".
  - **2003.** Page 1508, line 23: after that line insert:
- **"Section 3702c.** 655.27 (6) of the statutes is amended to read:
  - 655.27 (6) Purpose and integrity of fund. The fund is established to curb the rising costs of health care by financing part of the liability incurred by health care providers as a result of medical malpractice claims and to ensure that proper claims are satisfied. The fund, including any net worth of the fund, is held in irrevocable trust for the sole benefit of health care providers participating in the fund and proper claimants. Moneys in the fund may not be used for any other purpose of the state. No executive budget bill or bills introduced under s. 16.47 (1m) may use moneys in

- the fund for, or transfer moneys from the fund to be used for, any purpose not specified in this chapter or in s. 20.145 (2) as a use for moneys in the fund.".
- **2004.** Page 1508, line 24: delete that line.
- **2005.** Page 1509, line 1: delete lines 1 to 7.
- **2006.** Page 1509, line 18: delete lines 18 to 25.
- **2007.** Page 1510, line 1: delete lines 1 to 21.
- **2008.** Page 1510, line 21: after that line insert:
- 8 "Section 3706d. 757.02 (5) of the statutes is renumbered 757.02 (5) (a) and amended to read:
  - 757.02 (5) (a) Except for retired judges appointed under s. 753.075, and except as provided in par. (b), each supreme court justice, court of appeals judge and circuit court judge included under ch. 40 shall accrue sick leave at the rate established under s. 230.35 (2) for the purpose of credits under s. 40.05 (4) (b) and for premium payment determinations under s. 40.05 (4) and (5).
- **Section 3706h.** 757.02 (5) (b) of the statutes is created to read:
  - 757.02 (5) (b) No supreme court justice, court of appeals judge, or circuit court judge may receive sick leave as a supreme court justice, court of appeals judge, or circuit court judge included under ch. 40 beginning on the date that the next supreme court justice, court of appeals judge, or circuit court judge assumes office after the effective date of this paragraph .... [revisor inserts date].".
    - **2009.** Page 1510, line 21: after that line insert:
- **"Section 3707b.** 753.06 (6) (e) of the statutes is amended to read:
- 23 753.06 (6) (e) Juneau County. The circuit has one branch. Commencing August 1, 2008, the circuit has 2 branches.".

- **2010.** Page 1511, line 24: delete that line.
- **2011.** Page 1512, line 1: delete lines 1 to 5.
- 2012. Page 1512, line 6: delete the material beginning with that line and ending with page 1514, line 7.
- **2013.** Page 1514, line 8: delete the material beginning with that line and ending with page 1515, line 3.
- **2014.** Page 1515, line 4: delete the material beginning with that line and ending with page 1516, line 9.
- **2015.** Page 1516, line 10: delete lines 10 to 24.
- **2016.** Page 1517, line 1: delete lines 1 to 17.
- **2017.** Page 1518, line 7: delete lines 7 to 21.
- **2018.** Page 1519, line 21: delete lines 21 to 25.
- **2019.** Page 1520, line 1: delete lines 1 to 12.
- **2020.** Page 1521, line 5: delete lines 5 to 25.
- **2021.** Page 1522, line 1: delete lines 1 to 11.
- **2022.** Page 1522, line 19: delete lines 19 to 24.
- **2023.** Page 1523, line 1: delete lines 1 to 25.
- **2024.** Page 1524, line 1: delete lines 1 to 25.
- **2025.** Page 1525, line 1: delete lines 1 to 25.
- **2026.** Page 1526, line 1: delete lines 1 to 7.
- **2027.** Page 1526, line 20: delete lines 20 to 25.
- **2028.** Page 1527, line 1: delete lines 1 and 2.

**2029.** Page 1527, line 2: after that line insert:

"Section 3749m. 778.30 (1) (intro.) of the statutes is amended to read:

778.30 (1) (intro.) In addition to the procedures under s. 23.795 or 345.47 or under this chapter for the collection of forfeitures, costs, assessments, surcharges, municipal debt reduced to judgement, or restitution payments if a defendant fails to pay the forfeiture, costs, assessment, surcharge, municipal debt reduced to judgement, or restitution payment within the period specified by the circuit court, the court may do any of the following:

**Section 3750g.** 800.09 (1) (intro.) of the statutes is amended to read:

800.09 (1) JUDGMENT. (intro.) If a municipal court finds a defendant guilty, it may render judgment by ordering restitution under s. 800.093 and payment of a forfeiture, plus costs, fees, and surcharges imposed under ch. 814. The court shall apply any payment received on a judgment that includes restitution to first satisfy any payment of restitution ordered, then to pay the forfeiture, costs, fees, and surcharges. If the judgment is not paid, the court may proceed under par. (a), (b), or (c), or (d) or any combination of those paragraphs, as follows:

**Section 3750j.** 800.09(1)(d) of the statutes is created to read:

800.09 (1) (d) The court may issue an order assigning not more than 25 percent of the defendant's commissions, earnings, salaries, wages, pension benefits, benefits under ch. 102, and other money due or to be due in the future, including lottery prizes, to the municipal court for payment of the unpaid restitution, forfeiture, costs, fees, or surcharges.".

**2030.** Page 1529, line 8: delete lines 8 to 17.

2031.	Page 1529, line 18:	delete the	material	beginning	with the	at line	and
ending with p	page 1530, line 2.						

**2032.** Page 1530, line 2: after that line insert:

"Section 3759b. 812.30 (7m) of the statutes is created to read:

812.30 (7m) "Emancipated minor" has the meaning given in s. 48.375 (2) (e).

**SECTION 3759c.** 812.30 (8) of the statutes is amended to read:

812.30 (8) "Household income" means the disposable earnings of the debtor and dependents during any month in which the garnishment is in effect, plus unearned income received by the debtor and dependents in that month, less any of the debtor's earnings assigned by court order under ch. 767. If the debtor is a minor, other than an emancipated minor, "household income" means the disposable earnings of the debtor, of the parent that the debtor resides with for at least 50 percent of the month, and of the parent's dependents during any month in which the garnishment is in effect, plus unearned income received by the debtor, the parent, and the parent's dependents in that month, less any of the debtor's and the parent's earnings assigned by court order under ch. 767.

**Section 3759f.** 812.34 (2) (b) 3. of the statutes is created to read:

812.34 (2) (b) 3. The debtor is a minor, other than an emancipated minor, and the judgment that the creditor is seeking to satisfy is not a judgment for unpaid restitution, costs, a forfeiture, or a surcharge entered under s. 895.035 (2m) (a) or (b) by the court assigned to exercise jurisdiction under chs. 48 and 938.

**Section 3759h.** 812.35 (5) of the statutes is amended to read:

812.35 (5) Upon Except as provided in sub. (7), after being served, the garnishee shall determine whether the garnishee may become obligated to the

debtor for earnings earned within pay periods beginning within 13 weeks after the date of service. If it is unlikely that the garnishee will become so obligated, the garnishee shall send a statement of that fact to the creditor by the end of the 7th business day after receiving the earnings garnishment form under sub. (3). The creditor shall send a copy of this statement to the court within 7 business days after receipt of the statement.

**Section 3759j.** 812.35 (6) of the statutes is amended to read:

812.35 (6) If Except as provided in sub. (7), if the garnishee may become obligated to the debtor for earnings earned within pay periods beginning within 13 weeks after the date of service, but one or more earnings garnishments against the debtor have already been served on the garnishee and not terminated, the garnishee shall retain the earnings garnishment form and place the garnishment into effect the pay period after the last of any prior earnings garnishments terminates. The garnishee shall notify the debtor of the amount of the garnishment and shall notify the creditor of the amount owed on the pending garnishments by the end of the 7th business day after receipt of the garnishment form under sub. (3). If, before the earnings garnishment takes effect, the garnishee determines that it is unlikely that the garnishee will continue to be obligated to the debtor for earnings, the garnishee shall notify the creditor and court under sub. (5) within 7 business days after making that determination.

**Section 3759k.** 812.35 (7) of the statutes is created to read:

812.35 (7) (a) If the debtor is a minor, other than an emancipated minor, the garnishment shall continue until the amount that the creditor is seeking in the garnishment is paid in full. The garnishee shall determine whether the garnishee may become obligated to the debtor for earnings earned after the date of service. If

 $\mathbf{2}$ 

it is unlikely that the garnishee will become so obligated, the garnishee shall send a statement of that fact to the creditor by the end of the 7th business day after receiving the earnings garnishment form under sub. (3). The creditor shall send a copy of this statement to the court within 7 business days after receipt of the statement.

(b) If the garnishee may become obligated to the debtor for earnings earned within pay periods beginning after the date of service, but one or more earnings garnishments against the debtor have already been served on the garnishee and not terminated, the garnishee shall retain the earnings garnishment form and place the garnishment into effect the pay period after the last of any prior earnings garnishments terminates. The garnishee shall notify the debtor of the amount of the garnishment and shall notify the creditor of the amount owed on the pending garnishments by the end of the 7th business day after receipt of the garnishment form under sub. (3). If before the earnings garnishment takes effect the garnishee determines that it is unlikely that the garnishee will continue to be obligated to the debtor for earnings, the garnishee shall notify the creditor and court under par. (a) within 7 business days after making that determination.

**Section 3759m.** 812.40 of the statutes is amended to read:

812.40 Stipulated extension. At any time while an earnings garnishment is in effect, the debtor and creditor, if the debtor is an adult or an emancipated minor, may stipulate in writing to an extension of the earnings garnishment for additional pay periods. The extension may commence on the first day after the earnings garnishment ends and shall end within 13 weeks after the last day of the last pay period affected by the earnings garnishment. The garnishee shall be bound by the extension if a copy of the stipulation is delivered or mailed to the garnishee, together

with an additional garnishee fee under s.	812.33, before the last day of the last pay		
period affected by the earnings garnishment or any prior stipulated extension of th			
earnings garnishment. A stipulated extension is void and the garnishee f			
refunded if, prior to the last day of the last pay period affected by the earning			
garnishment, the garnishee is served under s. 812.35 (3) by a creditor seeking			
satisfy a different judgment against the debtor.			
Section 3759q. 812.44 (3) (form) of the statutes is amended to read:			
812.44 <b>(3)</b> (form)			
STATE OF WISCONSIN			
CIRCUIT COURT: County			
A.B., Creditor			
vs.	File or Reference Number		
C.D., Debtor EARNI			
and GARNI			
E.F., Garnishee			
THE STATE OF WISCONSIN, To the	e garnishee:		
The creditor has been awarded a coun	rt judgment that has not been paid. As a		
result, the creditor claims that a total of \$	is owed by the debtor, as follows:		
A. Unpaid balance on judgment	\$		
B. Unpaid postjudgment interest	\$		
C. Costs of this earnings garnishmer	nt		
(estimated)	\$		
TOTAL	\$		

The creditor believes that you will owe the debtor for earnings within the next 13 weeks, or if the debtor is an unemancipated minor, within any time. If the \$15 fee is tendered with these papers, you are directed by the court to do the following:

# DETERMINE WHETHER YOU WILL

## OWE THE DEBTOR EARNINGS

- 1. Determine if you are likely to owe the debtor for earnings in pay periods beginning within the next 13 weeks, or if the debtor is an unemancipated minor, within any time.
- 2. If you are not likely to owe the debtor for earnings in pay periods beginning within the next 13 weeks, or if the debtor is an unemancipated minor, within any time, send a statement stating that fact to the creditor by the end of the 7th business day after receiving the earnings garnishment forms.

## IF THE DEBTOR SENDS YOU AN ANSWER

- 3. Whenever you receive a debtor's answer form from the debtor, mail a copy of the answer form to the creditor by the end of the 3rd business day after receipt of that form. Include the date you received the answer form on the copy sent to the creditor.
- 4. If the debtor's answer form claims a complete exemption or defense, do not withhold or pay to the creditor any part of the debtor's earnings under this garnishment unless you receive an order of the court directing you to do so.

# MULTIPLE EARNINGS GARNISHMENTS

5. If the debtor's earnings are already being garnisheed when you receive this earnings garnishment, place this earnings garnishment into effect the pay period after the last of any prior earnings garnishments terminates. Notify the debtor of the amount of the garnishment and notify the creditor of the amount owed on the

pending garnishments by the end of the 7th business day after you receive these forms. If there are no prior pending earnings garnishments against the debtor's earnings, place this earnings garnishment into effect the pay period after you receive it.

# EARNINGS GARNISHMENTS LAST 13 WEEKS,

# EXCEPT FOR PUBLIC EMPLOYEES

6. The garnishment of the earnings of employees of the state of Wisconsin and its political subdivisions and of unemancipated minors remain in effect until the judgment is satisfied. The garnishment of earnings of other employees will affect the debtor's earnings for all pay periods beginning within 13 weeks after you receive it, unless the debtor's earnings are already being garnisheed. If this earnings garnishment is delayed under paragraph 5, above, it will affect the debtor's earnings for all pay periods beginning within 13 weeks after the first day of the pay period that you put this earnings garnishment into effect. If the amount claimed by the creditor is fully paid before the end of the 13 weeks, this earnings garnishment will terminate at that point.

# PAYING THE CREDITOR

7. Between 5 and 10 business days after each payday of a pay period affected by this earnings garnishment, pay the creditor 20% of the debtor's disposable earnings for that pay period. Payment is complete upon mailing.

## EFFECT OF COURT-ORDERED

## ASSIGNMENTS FOR SUPPORT

8. If the debtor has assigned his or her earnings for support by court order, those support payments take priority over this earnings garnishment. If 25% or more of the debtor's disposable earnings is assigned for support by court order, do not pay any

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

part of the debtor's earnings to the creditor. Instead, send the creditor a statement of that fact by the end of the 7th business day after you receive these forms. If less than 25% of the debtor's earnings is assigned for support by court order, the amount the creditor must be paid is reduced so that the total of earnings assigned and garnisheed does not exceed 25% of the debtor's disposable earnings. **EXTENSIONS** 9. The debtor and creditor may agree in writing to extend this earnings garnishment for additional pay periods beginning within 13 weeks after this earnings garnishment would otherwise terminate if the debtor is an adult or an emancipated minor. If you receive a written extension stipulation, and an additional garnishee fee for each extension, you must honor it unless a different garnishment against this debtor's earnings is served upon you before the extension takes effect. In that case, the extension is void and you must return the extension fee to the party who paid it to you. **Section 3759r.** 812.44 (4) (form) of the statutes is amended to read: 812.44 **(4)** (form) STATE OF WISCONSIN CIRCUIT COURT:.... County A.B., Creditor File or Reference Number.... vs. C.D., Debtor **EXEMPTION NOTICE** EARNINGS GARNISHMENT and E.F., Garnishee

23

24

To the debtor: 1 2 The creditor was awarded a judgment against you or your spouse by.... (County 3 Circuit or Federal District) Court on the.... day of..... (year) That judgment not having been fully paid, the creditor has now filed a garnishment proceeding against 4 5 your earnings from the garnishee. This means that the creditor is seeking to take 6 some of your earnings to satisfy part or all of the judgment against you or your 7 spouse. 8 The total amount of the creditor's claim is as follows: 9 Unpaid balance on judgment \$.... 10 Unpaid postjudgment interest \$.... 11 Costs: \$.... a. Garnishment filing fee 12 13 b. Garnishee fee \$.... \$.... c. Service of process (estimate) 14 15 TOTAL \$.... 16 By law, you are entitled to an exemption of not less than 80% of your disposable 17 earnings. Your "disposable earnings" are those remaining after social security and 18 federal and state income taxes are withheld. 19 Your earnings are completely exempt from garnishment if: 20 1. Your household income is below the federal poverty level. See the enclosed 21schedules and worksheet to determine if you qualify for this exemption.

2. You receive aid to families with dependent children, relief funded by a relief

block grant under ch. 49, relief provided by counties under section s. 59.53 (21) of the

Wisconsin Statutes, medical assistance, supplemental security income, food stamps,

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

or veterans benefits based on need under 38 USC 501 to 562 or section 45.351 (1) of the Wisconsin Statutes, or have received these benefits within the past 6 months.

- 3. At least 25% of your disposable earnings are assigned by court order for support.
- 4. You are a minor, other than an emancipated minor, and the judgment that the creditor is seeking to satisfy is not a judgment for unpaid restitution, costs, a forfeiture, or a surcharge entered under section 895.035 (2m) (a) or (b) of the Wisconsin Statutes by the court assigned to exercise jurisdiction under chapters 48 and 938 of the Wisconsin Statutes.

If the garnishment of 20% of your disposable income would result in the income of your household being below the poverty line, the garnishment is limited to the amount of your household's income in excess of the poverty line.

If you qualify for a complete exemption or for a limit on the amount of the garnishment to the amount that your household's income exceeds the poverty line, you must give or mail a copy of the enclosed debtor's answer form to the garnishee in order to receive that increased exemption.

If your circumstances change while the garnishment is in effect, you may file a new answer at any time.

If you do not qualify for a complete exemption, but you will not be able to acquire the necessities of life for yourself and your dependents if your earnings are reduced by this earnings garnishment, you may ask the court in which this earnings garnishment was filed to increase your exemption or grant you other relief.

IF YOU NEED ASSISTANCE

CONSULT AN ATTORNEY

To the garnishee:

If you have earnings that are being garnisheed that are exempt or subject to a
defense, the sooner you file your answer or seek relief from the court, the sooner such
relief can be provided. This If you are an adult or an emancipated minor, this
earnings garnishment affects your earnings in pay periods beginning within 13
weeks after it was served on the garnishee. You may agree in writing with the
creditor to extend it for additional 13-week periods until the debt is paid. <u>If you are</u>
an unemancipated minor, this earnings garnishment affects your earnings after it
was served on the garnishee and until the amount that the creditor is seeking in the
garnishment is paid in full.
PENALTIES
If you wrongly claim an exemption or defense in bad faith, or if the creditor
wrongly objects to your claim in bad faith, the court may order the person who acted
in bad faith to pay court costs, actual damages and reasonable attorney fees.
<b>Section 3759s.</b> 812.44 (5) (form) of the statutes is amended to read:
812.44 <b>(5)</b> (form)
STATE OF WISCONSIN
CIRCUIT COURT: County
A.B., Creditor
vs. File or Reference Number
C.D., Debtor EARNINGS GARNISHMENT
and DEBTOR'S ANSWER
E.F., Garnishee

1	My earnings are COMPLETELY EXEMPT from earnings garnishment because:
2	1. The judgment has been paid or is void.
3	2. (5) (form) paragraph 2. I receive, am eligible for, or have within 6 months
4	received, aid to families with dependent children, relief funded by a relief block grant
5	under ch. 49, relief provided by counties under section 59.53 (21) of the Wisconsin
6	Statutes, medical assistance, supplemental security income, food stamps, or
7	veterans benefits based on need under $38~\mathrm{USC}~501$ to $562$ or section $45.351~(1)$ of the
8	Wisconsin Statutes.
9	$\dots$ 3. At least 25% of my disposable earnings are assigned for support by court order.
10	4. My household income is less than the poverty line, or this garnishment would
11	cause that to happen.
12	4m. I am not an emancipated minor and the judgment that the creditor is seeking
13	to satisfy is not a judgment for unpaid restitution, costs, a forfeiture, or a surcharge
14	entered under section 895.035 (2m) (a) or (b) of the Wisconsin Statutes by the court
15	assigned to exercise jurisdiction under chapters 48 and 938 of the Wisconsin
16	Statutes.
17	5. I have another defense to this earnings garnishment (explain briefly).
18	
19	
20	I understand that if I claim a complete exemption or defense in bad faith, I may
21	be held liable to the creditor for actual damages, costs and reasonable attorney fees.
22	DATE Signature of Debtor
23	${\rm Address} \;$
24	Telephone Number
25	Date Received by Garnishee".

- **2033.** Page 1530, line 3: delete lines 3 to 8.
- **2034.** Page 1530, line 9: delete lines 9 to 15.
- **2035.** Page 1530, line 16: delete lines 16 to 20.
- **2036.** Page 1530, line 21: delete lines 21 to 24.
- **2037.** Page 1531, line 1: delete lines 1 to 24.
- **2038.** Page 1531, line 24: after that line insert:

"Section 3767m. 823.02 of the statutes is renumbered 823.02 (1) and amended to read:

823.02 (1) —An Except as provided in sub. (2), an action to enjoin a public nuisance may be commenced and prosecuted in the name of the state, either by the attorney general on information obtained by the department of justice, or upon the relation of a private individual, sewerage commission created under ss. 200.01 to 200.15 or a county, having first obtained leave therefor from the court. An action to enjoin a public nuisance may be commenced and prosecuted by a city, village, town or a metropolitan sewerage district created under ss. 200.21 to 200.65 in the name of the municipality or metropolitan sewerage district, and it is not necessary to obtain leave from the court to commence or prosecute the action. The same rule as to liability for costs shall govern as in other actions brought by the state. No stay of any order or judgment enjoining or abating, in any action under this section, may be had unless the appeal is taken within 5 days after notice of entry of the judgment or order or service of the injunction. Upon appeal and stay, the return to the court of appeals or supreme court shall be made immediately.

**Section 3767n.** 823.02 (2) of the statutes is created to read:

- 823.02 (2) The department of justice may not bring or join in an action under this section if the activity, use, or practice alleged to be a nuisance is not in violation of any state statute, rule, permit, or approval or local ordinance or regulation.".
- **2039.** Page 1532, line 1: delete lines 1 to 9.
- **2040.** Page 1532, line 10: delete lines 10 to 18.
  - **2041.** Page 1532, line 19: delete the material beginning with that line and ending with page 1533, line 4.
    - **2042.** Page 1533, line 4: after that line insert:
  - "Section 3374q. 893.575 of the statutes is created to read:
  - 893.575 Actions concerning illegal drug use. (1) Except as otherwise provided in this section, an action under s. 895.10 shall be commenced within 2 years after the cause of action accrues or be barred. A cause of action under s. 895.10 accrues when a person who may recover has reason to know of the harm from illegal drug use that is the basis for the cause of action and has reason to know the illegal drug use is the cause of the harm.
  - (2) For a plaintiff, the time limit under this section is tolled while the individual potential plaintiff is incapacitated by the use of an illegal drug to the extent the individual cannot reasonably be expected to seek recovery under s. 895.10. For a defendant, the time limit under this section is tolled until 6 months after the individual potential defendant is convicted of a criminal drug offense.
  - (3) The time limit under this section for an action under s. 895.10 based on participation in the illegal drug market that occurred prior to the effective date of this subsection .... [revisor inserts date], does not begin to run until the effective date of this subsection .... [revisor inserts date].".

24

1	<b>2043.</b> Page 1533, line 7: after that line insert:
2	"Section 3775m. 895.08 of the statutes is created to read:
3	895.08 Short title. This section and ss. 895.09 to 895.13 shall be known as the
4	"Drug Dealer Liability Law."
5	SECTION 3775n. 895.09 of the statutes is created to read:
6	<b>895.09 Definitions.</b> In ss. 895.08 to 895.13:
7	(1) "Illegal drug" means a controlled substance or controlled substance analog
8	whose distribution is a violation of s. 961.41.
9	(2) "Illegal drug market" means the support system of illegal drug-related
10	operations, from production to retail sales, through which an illegal drug reaches the
11	user.
12	(3) "Individual drug user" means the individual whose illegal drug use is the
13	basis of an action brought under s. 895.10.
14	(4) "Level 1 offense" means possession of 7 grams or more, but less than 113
15	grams, or distribution of less than 28 grams of a specified illegal drug other than
16	marijuana, or possession of 454 grams or more, but less than 1.8 kilograms, of
17	marijuana, or possession of 25 plants or more, but less than 50 plants, containing
18	tetrahydrocannabinols, or distribution of less than 454 grams of marijuana.
19	(5) "Level 2 offense" means possession of 113 grams or more, but less than 227
20	grams, or distribution of 28 grams or more, but less than 56 grams, of a specified
21	illegal drug other than marijuana, or possession of 1.8 kilograms or more, but less
22	than 3.6 kilograms of marijuana, or possession of 50 plants or more, but less than 75

plants, containing tetrahydrocannabinols, or distribution of more than 454 grams,

but less than 2.3 kilograms, of marijuana.

 $\mathbf{2}$ 

- (6) "Level 3 offense" means possession of 227 grams or more, but less than 454 grams, or distribution of 56 grams or more, but less than 113 grams, of a specified illegal drug other than marijuana, or possession of 3.6 kilograms or more, but less than 7.3 kilograms of marijuana, or possession of 75 plants or more, but less than 100 plants, containing tetrahydrocannabinols, or distribution of more than 2.3 kilograms, but less than 4.5 kilograms, of marijuana.
- (7) "Level 4 offense" means possession of 454 grams or more or distribution of 113 grams or more of a specified illegal drug other than marijuana, or possession of 7.3 kilograms or more of marijuana, or possession of 100 plants or more containing tetrahydrocannabinols, or distribution of 4.5 kilograms or more of marijuana.
  - (8) "Marijuana" has the meaning given in s. 961.01 (14).
- (9) "Participate in the illegal drug market" means to distribute, possess with an intent to distribute, commit an act intended to facilitate the marketing or distribution of, or agree to distribute, possess with an intent to distribute, or commit an act intended to facilitate the marketing and distribution of an illegal drug. "Participate in the illegal drug market" does not include the purchase or receipt of an illegal drug for personal use only.
- (10) "Period of illegal drug use" means, in relation to the individual drug user, the time of the individual's first use of an illegal drug to the accrual of the cause of action under s. 895.10. The period of illegal drug use is presumed to commence 2 years before the cause of action accrues unless the defendant proves otherwise by clear and convincing evidence.
- (11) "Place of illegal drug activity" means, in relation to the individual drug user and unless the defendant proves otherwise by clear and convincing evidence, each assembly district in which a claim is made that the individual possesses or uses

an illegal drug or in which the individual resides, attends school, or is employed during the period of the individual's illegal drug use.

(12) "Place of participation" means, in relation to a defendant in an action brought under s. 895.10, each assembly district in which the person participates in the illegal drug market or in which the person resides, attends school, or is employed during the period of the person's participation in the illegal drug market.

**Section 3775p.** 895.10 of the statutes is created to read:

- 895.10 Liability for participation in the illegal drug market. (1) A person who knowingly participates in the illegal drug market within this state is liable for civil damages as provided in this section. A person may recover damages under this section for injury resulting from an individual's use of an illegal drug.
- (2) A law enforcement officer or agency, the state, or a person acting at the direction of a law enforcement officer or agency or the state, is not liable for participating in the illegal drug market, if the participation is in furtherance of an official investigation.
- (3) One or more of the following persons may bring an action for damages caused by an individual's use of an illegal drug:
- (a) A parent, legal guardian, child, spouse, or sibling of the individual drug user.
  - (b) An individual who was exposed to an illegal drug in utero.
  - (c) An employer of the individual drug user.
- (d) A medical facility, insurer, governmental agency, employer, or other entity that funds a drug treatment program or employee assistance program for the individual drug user or that otherwise expended money on behalf of the individual drug user.

- (e) A person injured as a result of the willful, reckless, or negligent actions of an individual drug user.
- (4) A person entitled to bring an action under this section may seek damages from a person who knowingly distributed, or knowingly participated in the chain of distribution of, an illegal drug that was used by the individual drug user.
- (5) A person entitled to bring an action under this section may recover all of the following:
- (a) Economic damages, including the cost of treatment and rehabilitation, medical expenses, loss of economic or educational potential, loss of productivity, absenteeism, support expenses, accidents or injury, and any other pecuniary loss proximately caused by the illegal drug use.
- (b) Noneconomic damages, including physical and emotional pain, suffering, physical impairment, emotional distress, mental anguish, disfigurement, loss of enjoyment, loss of companionship, services, and consortium, and other nonpecuniary losses proximately caused by an individual's use of an illegal drug.
  - (c) Exemplary damages.
  - (d) Reasonable attorney fees.
  - (e) Costs of the suit, including reasonable expenses for expert testimony.
- (6) (a) An individual drug user may not bring an action for damages under this section caused by the use of an illegal drug, except as provided in this subsection. An individual drug user may bring an action for damages caused by the use of an illegal drug only if all of the following conditions are met:
- 1. The individual personally discloses to law enforcement authorities, more than 6 months before filing the action, all of the information known to the individual regarding all of that individual's sources of illegal drugs.

5

6

7

8

9

10

11

12

13

14

15

17

18

19

20

21

22

23

24

1	2. The individual has not used an illegal drug within the 6 months before filing
2	the action.
3	3. The individual continues to remain free of the use of an illegal drug

- (b) A person entitled to bring an action under this subsection may seek damages only from a person who distributed, or is in the chain of distribution of, an illegal drug that was actually used by the individual drug user.
- (c) A person entitled to bring an action under this subsection may recover only the following damages:
- 1. Economic damages, including, but not limited to, the cost of treatment, rehabilitation, and medical expenses, loss of economic or educational potential, loss of productivity, absenteeism, accidents or injury, and any other pecuniary loss proximately caused by the person's illegal drug use.
  - 2. Reasonable attorney fees.

throughout the pendency of the action.

- 3. Costs of the suit, including reasonable expenses for expert testimony.
- **Section 3775q.** 895.11 of the statutes is created to read:
  - 895.11 Third-party cases and target communities. (1) A 3rd party may not pay damages awarded under s. 895.10, or provide a defense or money for a defense, on behalf of an insured under a contract of insurance or indemnification.
  - (2) A person whose participation in the illegal drug market constitutes the following level of offense shall be considered to have the following illegal drug market target community:
  - (a) For a level 1 offense, all assembly districts that comprise the person's place of participation.

- (b) For a level 2 offense, the target community described in par. (a) plus all assembly districts with a border contiguous to that target community.
- (c) For a level 3 offense, the target community described in par. (b) plus all assembly districts with a border contiguous to that target community.
  - (d) For a level 4 offense, the state.

**Section 3775r.** 895.12 of the statutes is created to read:

# 895.12 Joinder, responsibility, contribution, and standard of proof. (1) JOINDER OF PARTIES. (a) Two or more persons may join in one action under s. 895.10 as plaintiffs if their respective actions have at least one place of illegal drug activity in common and if any portion of the period of illegal drug use related to one plaintiff's damages overlaps with the period of illegal drug use related to every other plaintiff's damages.

- (b) Two or more persons may be joined in one action under s. 895.10 as defendants if those persons are liable to at least one plaintiff.
- (c) A plaintiff need not be interested in obtaining and a defendant need not be interested in defending against all the relief demanded. Judgment may be given for one or more plaintiffs according to their respective rights to relief and against one or more defendants according to their respective liabilities.
- (2) Comparative responsibility. (a) Section 895.045 applies to an action under s. 895.10.
- (b) The burden of proving the comparative negligence of the plaintiff is on the defendant, which shall be shown by clear and convincing evidence.
- (c) Comparative negligence may not be attributed to a plaintiff who is not an individual drug user.

 $\mathbf{2}$ 

- (3) Contribution among and recovery from multiple defendants. A person subject to liability under s. 895.10 has a right of action for contribution against another person subject to liability under that section. Contribution may be enforced either in the original action or by a separate action brought for that purpose. A plaintiff may seek recovery in accordance with s. 895.10 and existing law against a person whom a defendant has asserted a right of contribution.
- (4) STANDARD OF PROOF: EFFECT OF CRIMINAL DRUG CONVICTION. (a) Proof of participation in the illegal drug market in an action brought under s. 895.10 shall be shown by clear and convincing evidence. Except as otherwise provided in this section and ss. 895.09, 895.10, 895.11, and 895.13, other elements of the cause of action shall be shown by a preponderance of the evidence.
- (b) A person against whom recovery is sought who has a criminal conviction under ch. 961 or 21 USC 801 to 971, is estopped from denying participation in the illegal drug market. Such a conviction is also prima facie evidence of the person's participation in the illegal drug market during the 2 years preceding the date of an act giving rise to a conviction.
- (c) The absence of a criminal conviction described under par. (b) of a person against whom recovery is sought does not bar an action against that person.

**Section 3775s.** 895.13 of the statutes is created to read:

895.13 Attachment, execution, and stay. (1) Except as provided in sub. (3), a plaintiff under s. 895.10 may request an ex parte prejudgment attachment order from the court against all assets of a defendant sufficient to satisfy a potential award. If attachment is instituted, a defendant is entitled to an immediate court hearing. The court may lift the attachment if the defendant demonstrates that the assets will

 $\mathbf{2}$ 

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

21

- be available for a potential award or if the defendant posts a bond sufficient to cover a potential award.
- (2) A person against whom a judgment has been rendered under s. 895.10 is not eligible to exempt any property, of whatever kind, from process to levy or process to execute on the judgment.
- (3) Any assets sought to satisfy a judgment in an action under s. 895.10 that are named in a forfeiture action or that have been seized for forfeiture by any state or federal agency may not be used to satisfy a judgment unless and until the assets have been released following the conclusion of the forfeiture action or released by the agency that seized the assets.
- (4) The district attorney may represent the state or a political subdivision of the state in an action brought under s. 895.10.
- (5) On motion by a governmental agency involved in a drug investigation or prosecution, an action brought under s. 895.10 shall be stayed until the completion of the criminal investigation or prosecution that gave rise to the motion for a stay of the action.".
  - **2044.** Page 1533, line 7: after that line insert:
- "Section 3775g. 895.046 of the statutes is created to read:
- 19 **895.046** Remedies against manufacturers, distributors, sellers, and 20 promoters of products. (1) Definitions. In this section:
  - (a) "Claimant" means a person seeking damages or other relief for injury or harm to a person or property caused by or arising from a product.

- (b) "Relevant production period" means the time period during which the specific product that allegedly caused the claimant's injury or harm was manufactured, distributed, sold, or promoted.
- (2) APPLICABILITY. This section applies to all actions in which a claimant alleges that the manufacturer, distributor, seller, or promoter of a product is liable for an injury or harm to a person or property, including actions based on allegations that the design, manufacture, distribution, sale, or promotion of, or instructions or warnings about, a product caused or contributed to a personal injury or harm to a person or property, a private nuisance, or a public nuisance, and to all related or independent claims, including unjust enrichment, restitution, or indemnification.
- (3) Remedy with specific product identification. Except as provided in sub. (4), the manufacturer, distributor, seller, or promoter of a product may be held liable in an action under sub. (2) only if the claimant proves, in addition to any other elements required to prove his or her claim, that the manufacturer, distributor, seller, or promoter of a product manufactured, distributed, sold, or promoted the specific product alleged to have caused the claimant's injury or harm.
- (4) REMEDY WITHOUT SPECIFIC PRODUCT IDENTIFICATION. Subject to sub. (5), if a claimant cannot meet the burden of proof under sub. (3), the manufacturer, distributor, seller, or promoter of a product may be held liable for an action under sub. (2) only if the claimant proves all of the following:
- (a) That no other lawful process exists for the claimant to seek redress from another person for the injury or harm.
- (b) That the claimant has suffered an injury or harm that can be caused only by a product chemically identical to the specific product that allegedly caused the claimant's injury or harm.

- (c) That the manufacturer, distributor, seller, or promoter of a product manufactured, distributed, sold, or promoted a product that meets all of the following criteria:
- 1. Is chemically identical to the specific product that allegedly caused the claimant's injury or harm.
- 2. Was manufactured, distributed, sold, or promoted in this state during the time period in which the specific product that allegedly caused the claimant's injury or harm was manufactured, distributed, sold, or promoted.
- (dm) That the action names, as defendants, those manufacturers of a product who collectively, during the relevant production period, manufactured at least 80 percent of all products sold in this state that are chemically identical to the specific product that allegedly caused the claimant's injury or harm.
- (5) LIMITATION ON LIABILITY. No manufacturer, distributor, seller, or promoter of a product is liable under sub. (4) if more than 25 years have passed between the date that the manufacturer, distributor, seller, or promoter of a product last manufactured, distributed, sold, or promoted a product chemically identical to the specific product that allegedly caused the claimant's injury and the date that the claimant's cause of action accrued.
- (6) APPORTIONMENT OF LIABILITY. If more than one manufacturer, distributor, seller, or promoter of a product is found liable for the claimant's injury or harm under subs. (4) and (5), the court shall apportion liability among those manufacturers, distributors, sellers, and promoters, but that liability shall be several and not joint.".
  - **2045.** Page 1533, line 8: delete lines 8 to 25.
  - **2046.** Page 1534, line 1: delete lines 1 to 8.

 $\mathbf{2}$ 

**2047.** Page 1534, line 17: after that line insert:

"Section 3778r. 904.16 of the statutes is created to read:

**904.16 Health care reports.** Any reports that a state or federal regulatory agency requires a health care provider, as defined in s. 146.81 (1), to give or disclose to that state or federal regulatory agency and other information obtained by the state or federal regulatory agency regarding a health care provider may not be used as evidence in a civil or criminal action brought against the health care provider, except that those reports and information may be used as evidence in any administrative proceeding conducted by the state regulatory agency.".

**2048.** Page 1534, line 17: after that line insert:

"Section 3778q. 895.52 (6) (a) (intro.) of the statutes is amended to read:

895.52 **(6)** (a) (intro.) The private property owner collects money, goods or services in payment for the use of the owner's property for the recreational activity during which the death or injury occurs, and the aggregate value of all payments received by the owner for the use of the owner's property for recreational activities during the year in which the death or injury occurs exceeds \$2,000 \$10,000. The following do not constitute payment to a private property owner for the use of his or her property for a recreational activity:".

- **2049.** Page 1534, line 18: delete lines 18 to 24.
- **2050.** Page 1535, line 1: delete the material beginning with that line and ending with page 1536, line 7.
- **2051.** Page 1536, line 7: after that line insert:
- 23 "Section 3779e. 908.03 (6m) (d) of the statutes is renumbered 908.03 (6m) (d) 3. and amended to read:

908.03 (6m) (d) 3. After December 31, 2002 the effective date of this paragraph ..... [revisor inserts date], the department of health and family services shall, by rule, prescribe uniform fees for X-ray copies that are based on an approximation of actual costs. The fees, plus applicable tax, and are the maximum amount, plus applicable tax and the actual costs of postage or other means of delivering the requested records to the requester, that a health care provider may charge for certified duplicate patient health care records. The rule shall also allow the health care provider to charge for actual postage or other actual delivery costs duplicate X-ray copies.

4. For duplicate patient health care records and duplicate X-ray reports or the referral of X-rays to another health care provider that are requested before commencement of an action, s. 146.83 (1) (b) and (c), (3g), and (3m) applies.

**Section 3779f.** 908.03 (6m) (d) 1. and 2. of the statutes are created to read:

908.03 (6m) (d) 1. Except as provided in subd. 2., the maximum amount that a health care provider may charge for certified duplicate patient health care records is \$1 per page, plus applicable tax and the actual costs of postage or other means of delivering the requested records to the requester.

2. For years beginning after 2007, fees specified under subd. 1. increase annually by a percentage equal to the percentage change between the U.S. consumer price index for all urban consumers, U.S. city average, for the month of August of the previous year and the U.S. consumer price index for all urban consumers, U.S. city average, for the month of August of the year before the previous year, as determined by the federal department of labor. The department of health and family services shall specify this increase annually on the department's Web site."

**2052.** Page 1537, line 22: delete lines 22 to 25.

- **2053.** Page 1538, line 1: delete lines 1 to 25.
- **2054.** Page 1539, line 1: delete lines 1 to 13.
- **2055.** Page 1539, line 18: delete lines 18 to 25.
- **2056.** Page 1540, line 1: delete lines 1 to 14.
- **2057.** Page 1541, line 22: delete lines 22 to 25.
- **2058.** Page 1542, line 1: delete lines 1 to 7.
- **2059.** Page 1544, line 23: delete lines 23 to 25.
- **2060.** Page 1545, line 1: delete lines 1 to 25.
- **2061.** Page 1546, line 1: delete lines 1 to 25.
- **2062.** Page 1547, line 1: delete lines 1 to 24.
- **2063.** Page 1548, line 1: delete lines 1 to 12.
- **2064.** Page 1548, line 23: delete lines 23 to 25.
- **2065.** Page 1549, line 1: delete lines 1 to 4.
- **2066.** Page 1549, line 18: delete lines 18 to 25.
- **2067.** Page 1550, line 1: delete lines 1 to 25.
- **2068.** Page 1551, line 1: delete lines 1 to 11.
- **2069.** Page 1551, line 11: after that line insert:
- 18 "Section 3834br. 939.48 (1m) of the statutes is created to read:
- 939.48 (1m) (a) If an actor intentionally used force that was intended or likely to cause death or great bodily harm, the court shall presume that the actor reasonably believed that the force was necessary to prevent imminent death or great

- bodily harm to himself or herself if the actor makes such a claim under sub. (1) and any of the following applies:
  - 1. The person against whom the force was used was in the process of unlawfully and forcibly entering the actor's residence, the actor was present in the residence, and the actor knew or had reason to believe that an unlawful and forcible entry was occurring.
  - 2. The person against whom the force was used was in the actor's residence after unlawfully and forcibly entering it, the actor was present in the residence, and the actor knew or had reason to believe that the person had unlawfully and forcibly entered the residence.
  - (b) The presumption described in par. (a) does not apply if any of the following applies:
  - 1. The actor was engaged in an unlawful activity or was using his or her residence to further an unlawful activity at the time.
  - 2. The person against whom the force was used was a peace officer who entered or attempted to enter the actor's residence in the performance of his or her official duties. This subdivision applies only if at least one of the following applies:
  - a. The officer identified himself or herself to the actor before the force described in par. (a) was used by the actor.
- b. The actor knew or reasonably should have known that the person entering or attempting to enter his or her residence was a peace officer.".
  - **2070.** Page 1551, line 12: delete lines 12 to 16.
  - **2071.** Page 1551, line 16: after that line insert:
  - "Section 3834t. 948.10 (1) of the statutes is amended to read:

- 948.10 (1) Whoever, for purposes of sexual arousal or sexual gratification, 1  $\mathbf{2}$ causes a child to expose genitals or pubic area or exposes genitals or pubic area to a 3 child is guilty of a Class A misdemeanor I felony.". **2072.** Page 1551, line 17: delete lines 17 to 25. 4 **2073.** Page 1552, line 1: delete lines 1 to 5. 5 **2074.** Page 1552, line 6: delete lines 6 to 10. 6 **2075.** Page 1563, line 16: substitute "20.505 (6) (ku) 20.455 (2) (kx)" for 7 "20.505 (6) (ku)". 8 9 **2076.** Page 1563, line 16: after that line insert: 10 "Section 3866m. 961.472 (5) (b) of the statutes is amended to read: 961.472 (5) (b) The person is participating in a substance abuse treatment 11 12 program that meets the requirements of s. 16.964 (12) (c) 165.95 (3), as determined 13 by the office department of justice assistance under s. 16.964 (12) (i) 165.95 (9).". 2077. Page 1563, line 17: delete the material beginning with that line and 14 15 ending with page 1564, line 19. 16 **2078.** Page 1564, line 19: after that line insert:
- 17 **"Section 3870e.** 967.11 (1) of the statutes is amended to read:
- 967.11 (1) In this section, "approved substance abuse treatment program" 18 19 means a substance abuse treatment program that meets the requirements of s. 16.964 (12) (c) 165.95 (3), as determined by the office department of justice assistance 20 21under s. 16.964 (12) (i) 165.95 (9).".
- 22 **2079.** Page 1564, line 19: after that line insert:

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

"Section 3870d. 968.26 of the statutes is renumbered 968.26 (1) and amended to read:

968.26 (1) If Except as provided in sub. (2), if a person complains to a judge that he or she has reason to believe that a crime has been committed within his or her jurisdiction, the judge shall examine the complainant under oath and any witnesses produced by him or her and may, and at the request of the district attorney shall, subpoena and examine other witnesses to ascertain whether a crime has been committed and by whom committed. The extent to which the judge may proceed in the examination is within the judge's discretion. The examination may be adjourned and may be secret. Any witness examined under this section may have counsel present at the examination but the counsel shall not be allowed to examine his or her client, cross-examine other witnesses or argue before the judge. If it appears probable from the testimony given that a crime has been committed and who committed it, the complaint may be reduced to writing and signed and verified; and thereupon a warrant shall issue for the arrest of the accused. Subject to s. 971.23, if the proceeding is secret, the record of the proceeding and the testimony taken shall not be open to inspection by anyone except the district attorney unless it is used by the prosecution at the preliminary hearing or the trial of the accused and then only to the extent that it is so used. A court, on the motion of a district attorney, may compel a person to testify or produce evidence under s. 972.08 (1). The person is immune from prosecution as provided in s. 972.08 (1), subject to the restrictions under s. 972.085.

**Section 3870f.** 968.26 (2) of the statutes is created to read:

968.26 (2) A prisoner, as defined in s. 801.02 (7) (a) 2., who has reason to believe that a crime has been committed may complain only to the district attorney of the

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

county in which the alleged crime occurred. The district attorney may conduct an investigation and may file a complaint under sub. (1).

**SECTION 3870h.** 970.02 (1) (c) of the statutes is amended to read:

970.02 (1) (c) That the defendant is entitled to a preliminary examination if charged with a felony in any complaint, including a complaint issued under s. 968.26 (1), or when the defendant has been returned to this state for prosecution through extradition proceedings under ch. 976, or any indictment, unless waived in writing or in open court, or unless the defendant is a corporation or limited liability company.

**Section 3870n.** 971.02 (1) of the statutes is amended to read:

971.02 (1) If the defendant is charged with a felony in any complaint, including a complaint issued under s. 968.26 (1), or when the defendant has been returned to this state for prosecution through extradition proceedings under ch. 976, or any indictment, no information or indictment shall be filed until the defendant has had a preliminary examination, unless the defendant waives such examination in writing or in open court or unless the defendant is a corporation or limited liability company. The omission of the preliminary examination shall not invalidate any information unless the defendant moves to dismiss prior to the entry of a plea.".

- **2080.** Page 1564, line 20: delete the material beginning with that line and ending with page 1566, line 17.
  - **2081.** Page 1567, line 21: after that line insert:
- 21 "Section 3875s. 971.23 (1) (a) of the statutes is amended to read:
- 971.23 (1) (a) Any written or recorded statement concerning the alleged crime made by the defendant, including the testimony of the defendant in a secret

proceeding under s. 968.26 (1) or before a grand jury, and the names of witnesses to the defendant's written statements.".

**2082.** Page 1568, line 5: after that line insert:

**"Section 3876h.** 972.08 (1) (a) of the statutes is amended to read:

972.08 (1) (a) Whenever any person refuses to testify or to produce books, papers or documents when required to do so before any grand jury, in a proceeding under s. 968.26 (1) or at a preliminary examination, criminal hearing or trial for the reason that the testimony or evidence required of him or her may tend to incriminate him or her or subject him or her to a forfeiture or penalty, the person may nevertheless be compelled to testify or produce the evidence by order of the court on motion of the district attorney. No person who testifies or produces evidence in obedience to the command of the court in that case may be liable to any forfeiture or penalty for or on account of testifying or producing evidence, but no person may be exempted from prosecution and punishment for perjury or false swearing committed in so testifying.

**Section 3876g.** 972.085 of the statutes is amended to read:

**972.085** Immunity; use standard. Immunity from criminal or forfeiture prosecution under ss. 13.35, 17.16 (7), 77.61 (12), 93.17, 111.07 (2) (b), 128.16, 133.15, 139.20, 139.39 (5), 195.048, 196.48, 551.56 (3), 553.55 (3), 601.62 (5), 767.87 (4), 885.15, 885.24, 885.25 (2), 891.39 (2), 968.26 (1), 972.08 (1) and 979.07 (1) and ch. 769, provides immunity only from the use of the compelled testimony or evidence in subsequent criminal or forfeiture proceedings, as well as immunity from the use of evidence derived from that compelled testimony or evidence.".

**2083.** Page 1568, line 11: after that line insert:

"Section 3879g. 973.042 (5) of the statutes is amended to read: 1  $\mathbf{2}$ 973.042 (5) The secretary of administration shall credit part A of the surcharge 3 to the appropriation account under s. 20.410 (1) (gi). The secretary of administration 4 shall credit part B of the surcharge to the appropriation account under s. 20.455 (2) 5 (gj). The secretary of administration shall credit part C of the surcharge to the 6 appropriation account under s. 20.505 (6) 20.455 (5) (gj). 7 **Section 3879m.** 973.043 (3) of the statutes is amended to read: 8 973.043 (3) All moneys collected from drug offender diversion surcharges shall 9 be credited to the appropriation account under s. 20.505 (6) (ku) 20.455 (2) (kx) and 10 used for the purpose of making grants to counties under s. 16.964 (12) 165.95.". **2084.** Page 1568, line 12: delete the material beginning with that line and 11 12 ending with page 1569, line 17. **2085.** Page 1569, line 17: after that line insert: 13 14 "Section 3884L. 973.047 (1f) of the statutes is amended to read: 15 973.047 (1f) If a court imposes a sentence or places a person on probation for 16 a felony conviction or for a conviction for a violation of s. 940.225 (3m), or 944.20, or 17 948.10, the court shall require the person to provide a biological specimen to the state 18 crime laboratories for deoxyribonucleic acid analysis.". **2086.** Page 1569, line 18: delete lines 18 to 20. 19 **2087.** Page 1570, line 1: delete lines 1 to 4. 20 **2088.** Page 1570, line 12: after that line insert: 21 22 **"Section 3887m.** 973.155 (1m) of the statutes is amended to read: 23 973.155 (1m) A convicted offender shall be given credit toward the service of 24 his or her sentence for all days spent in custody as part of a substance abuse

- treatment program that meets the requirements of s. 16.964 (12) (c) 165.95 (3), as
- determined by the office department of justice assistance under s. 16.964 (12) (i)
- 3 <u>165.95 (9)</u> for any offense arising out of the course of conduct that led to the person's
- 4 placement in that program.".
- 5 **2089.** Page 1570, line 23: delete "paid only" and substitute "paid only".
- 6 **2090.** Page 1570, line 24: delete "to the general fund" and substitute "to the
- 7 general fund".
- 8 **2091.** Page 1570, line 24: delete "credited to the appropriation account under
- 9 <u>s. 20.455 (5) (hh)</u>".
- 10 **2092.** Page 1570, line 25: delete "949, the" and substitute "949, the".
- 11 **2093.** Page 1571, line 1: delete "general fund shall receive" and substitute
- 12 "general fund shall receive".
- 13 **2094.** Page 1571, line 2: delete "shall be credited to the appropriation account
- 14 under s. 20.455 (5) (hh)".
- **2095.** Page 1571, line 4: delete lines 4 to 17 and substitute.
- 16 "Section 3893. 973.30 of the statutes is repealed.".
- 17 **2096.** Page 1571, line 18: delete the material beginning with that line and
- ending with page 1573, line 13.
- 19 **2097.** Page 1573, line 14: delete lines 14 to 23.
- 20 **2098.** Page 1573, line 24: delete the material beginning with that line and
- ending with page 1576, line 18.
- 22 **2099.** Page 1576, line 18: after that line insert:
- 23 "Section **3926n.** 978.01 (2) (b) of the statutes is amended to read:

- 978.01 **(2)** (b) A district attorney serves on a part-time basis if his or her prosecutorial unit consists of Buffalo, Florence, Pepin, <u>or</u> Trempealeau <del>or Vernon</del> county.".
- **2100.** Page 1576, line 19: delete lines 19 to 22.
- **2101.** Page 1577, line 12: delete lines 12 to 17.
- **2102.** Page 1577, line 17: after that line insert:
- **"Section 3932d.** 985.02 (2) (a) of the statutes is amended to read:
- 985.02 (2) (a) The notice must be posted in at least 3 public places likely to give notice to persons affected. In this paragraph, the Internet is considered a public place."
- **2103.** Page 1577, line 18: delete lines 18 to 21.
- **2104.** Page 1577, line 22: delete lines 22 to 24.
- **2105.** Page 1577, line 24: after that line insert:
- 14 "Section 3934b. 2001 Wisconsin Act 16, section 9107 (13r) is repealed.".
- **2106.** Page 1577, line 25: delete that line.
- **2107.** Page 1578, line 3: delete lines 3 to 19 and substitute:
  - "[2005 Wisconsin Act 25] Section 9101 (4) (b) the secretary If the department of administration determines that it is in the best interest of the state to offer one or more parcels of state-owned property for sale, the secretary of administration shall submit a report to the secretary of the building commission containing an inventory of his or her recommendations to offer specified state properties the property for sale under in accordance with section 16.848 of the statutes, as created by this act, if the property is eligible for sale under that section and this subsection. The report shall

contain a description of the property and the reasons the	refor. A prop	erty may be		
included in the inventory for the recommendation. The se	ecretary of ad	ministration		
may recommend that a parcel of property be offered for sale	e with or with	out approval		
of the state agency having jurisdiction of the property. If	f, <u>during the</u>	<u>period</u> on or		
before June 30, 2007, or the period beginning on the effective date of this paragraph				
and ending on June 30, 2009, the building commission votes to approve the sale of				
any property included in the inventory a report und	der this par	agraph, the		
department of administration may offer the property for	sale under se	ection 16.848		
of the statutes, as created by this act.".				
<b>2108.</b> Page 1578, line 22: after that line insert:				
"Section 3936e. 2005 Wisconsin Act 25, section 910	05 (1) (h) 1. is	amended to		
read:				
[2005 Wisconsin Act 25] Section 9105 (1) (h)				
1. Projects financed by general fund supported				
borrowing:				
Madison — Purchase of space at University				
Square project	\$	39,850,000		
(Total project all funding sources \$56,850,000)				
— Sterling Hall renovation	37,500,000	20,000,000		
(Total project all funding sources \$39,500,000)	, ,			
Milwaukee — Golda Meir Library remodeling —		0.500.000		
Phase I		3,508,000		

(Total project all funding sources \$4,908,000)

1	— Columbia St. Mary's Columbia cam-	
2	pus medical facilities acquisition	
3	and remodeling	56,530,000
4	(Total project all funding sources	
5	\$112,120,000)	
6	Platteville — Tri-state initiative facilities	20,000,000
7	(Total project all funding sources \$50,615,000)	
8	Stevens Point — Waste management laboratory	1,789,000
9	Stout — Jarvis science wing addition and	
10	remodeling	40,637,000
11	Superior — Jim Dan Hill Library renovation	4,500,000
12	System — Classroom renovation/instructional	
13	technology	7,000,000
14	— Utility Improvements — $3$ campuses	21,008,000
15	(Total project all funding sources \$28,600,000)	
16	Whitewater — College of Business and Economics	
17	building	35,549,000
18	(Total project all funding sources	
19	\$41,039,000)".	
20	<b>2109.</b> Page 1579, line 9: after that line insert:	
21	"Section 3936p. 2005 Wisconsin Act 25, section 9105 (11) is	amended to read:
22	[2005 Wisconsin Act 25] Section 9105 (11) Sterling Hall.	Notwithstanding
23	section 18.04 (1) and (2) of the statutes, of the public debt a	uthorized for the

 $\mathbf{2}$ 

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

renovation of Sterling Hall, as enumerated in subsection (1) (h) 1., \$20,000,000 \$2,500,000 in public debt may not be contracted until after June 30, 2007.".

**2110.** Page 1579, line 13: delete the material beginning with "and" and ending with "2009," on line 14.

**2111.** Page 1579, line 25: after that line insert:

"Section 3937j. 2005 Wisconsin Act 456, section 7 (2) is amended to read:

[2005 Wisconsin Act 456] Section 7 (2) EMERGENCY RULES. Using the procedure under section 227.24 of the statutes, the department of commerce shall may promulgate as emergency rules the rules required under subchapter VII of chapter 101 of the statutes, as created by this act, and under section 101.19 (1) (k) of the statutes, as created by this act. Notwithstanding section 227.24 (1) (c) and (2) of the statutes, the emergency rules promulgated under this subsection may remain in effect until the date on which the permanent rules required under subchapter VII of chapter 101 of the statutes, as created by this act, and under section 101.19 (1) (k) of the statutes, as created by this act, take effect. Notwithstanding section 227.24 (1) (a) and (3) of the statutes, the department of commerce is not required to provide evidence that promulgating rules under this subsection as emergency rules is necessary for the preservation of the public peace, health, safety, or welfare and is not required to provide a finding of emergency for the rules promulgated under this subsection. The department of commerce shall promulgate the rules required under this subsection no later than the first day of the 9th month beginning after the effective date of this subsection.

**Section 3937m.** 2005 Wisconsin Act 456, section 7 (3) is amended to read:

[2005 Wisconsin Act 456] Section 7 (3) Effect on collective bargaining agreement that is in effect on the effective date of this subsection June 1, 2008, and that contains provisions that are inconsistent with subchapter VII of chapter 101 of the statutes, as created by this act, then, notwithstanding subchapter VII of chapter 101 of the statutes, as created by this act, the person may perform its obligations, and exercise its rights, under those provisions of the collective bargaining agreement until the collective bargaining agreement expires or is extended, modified, or renewed, whichever occurs first.

**SECTION 3937p.** 2005 Wisconsin Act 456, section 7 (5) is amended to read:

[2005 Wisconsin Act 456] Section 7 (5) Grace Period for Obtaining Permit to Operate existing conveyances. Notwithstanding section 101.983 (2) of the statutes, as created by this act, the owner or lessee of any conveyance, as defined in section 101.981 (1) (c) of the statutes, as created by this act, in operation on the effective date of this subsection June 1, 2008, shall obtain the permit required under section 101.983 (2) of the statutes, as created by this act, no later than the first day of the 6th month beginning after the effective date of this subsection June 1, 2008. This subsection does not apply to any person required to obtain a permit under rules of the department of commerce that are in effect before the effective date of this subsection."

- **2112.** Page 1581, line 1: delete the material beginning with that line and ending with page 1582, line 10.
- **2113.** Page 1582, line 11: delete lines 11 to 25.
- **2114.** Page 1583, line 1: delete lines 1 to 5.

- **2115.** Page 1583, line 15: delete lines 15 to 19.
  - **2116.** Page 1584, line 4: after that line insert:
  - "(10r) Transfer of office of justice assistance.
  - (a) Assets and liabilities. On the effective date of this paragraph, the assets and liabilities of the office of justice assistance, except those that are primarily related to administering federal homeland security moneys, shall become the assets and liabilities of the department of justice. On the effective date of this paragraph, the assets and liabilities of the office of justice assistance that are primarily related to administering federal homeland security moneys shall become the assets and liabilities of the department of military affairs.
  - (b) 1. On the effective date of this subsection, the authorized positions for the office of justice assistance funded from the appropriation under section 20.505 (6) (a), 2005 stats., are decreased by 2.42 GPR positions.
  - 2. On the effective date of this subsection, the authorized positions for the office of justice assistance funded from the appropriation under section 20.505 (6) (k), 2005 stats., are decreased by 1.40 PR positions.
  - 3. On the effective date of this subsection, the authorized positions for the office of justice assistance funded from the appropriation under section 20.505 (6) (m), 2005 stats., are decreased by 15.18 FED positions.
  - 4. On the effective date of this subsection, the incumbents holding the positions under subds. 1. to 3., except the executive director of the office of justice assistance and an individual holding a senior position, are transferred to the department of justice.

- (c) *Employee status*. Employees transferred under paragraph (b) shall have all the rights and the same status under subchapter V of chapter 111 and chapter 230 of the statutes in the department of justice or the department of military affairs, whichever is applicable, that they enjoyed in the office of justice assistance immediately before the transfer. Notwithstanding section 230.28 (4) of the statutes, no employee so transferred who has attained permanent status in class is required to serve a probationary period.
- (d) Tangible personal property. On the effective date of this paragraph, all tangible personal property, including records, of the office of justice assistance, except property that is primarily related to administering federal homeland security moneys, shall be transferred to the department of justice. On the effective date of this paragraph, all tangible personal property, including records, of the office of justice assistance that is primarily related to administering federal homeland security moneys shall be transferred to the department of military affairs.

#### (e) Contracts.

- 1. All contracts entered into by the office of justice assistance in effect on the effective date of this subdivision, except contracts that are primarily related to administering federal homeland security moneys, remain in effect and are transferred to the department of justice. The department of justice shall carry out any such contractual obligations unless modified or rescinded by the department of justice to the extent allowed under the contract.
- 2. All contracts entered into by the office of justice assistance in effect on the effective date of this subdivision that are primarily related to administering federal homeland security moneys remain in effect and are transferred to the department of military affairs. The department of military affairs shall carry out any such

contractual obligations unless modified or rescinded by the department of military affairs to the extent allowed under the contract.

- on the effective date of this paragraph, except matters that are primarily related to administering federal homeland security moneys, is transferred to the department of justice, and all materials submitted to or actions taken by the office of justice assistance with respect to the pending matter are considered as having been submitted to or taken by the department of justice. Any matter pending with the office of justice assistance on the effective date of this paragraph that is primarily related to administering federal homeland security moneys is transferred to the department of military affairs, and all materials submitted to or actions taken by the office of justice assistance with respect to the pending matter are considered as having been submitted to or taken by the department of military affairs.
  - (g) Rules and orders.
- 1. All rules promulgated for the office of justice assistance, except rules that are primarily related to administering federal homeland security moneys, and that are in effect on the effective date of this paragraph remain in effect until their specified expiration dates or until amended or repealed by the department of justice. All orders issued by the office of justice assistance, except orders that are primarily related to administering of federal homeland security moneys, and that are in effect on the effective date of this paragraph remain in effect until their specified expiration dates or until modified or rescinded by the department of justice.
- 2. All rules promulgated for the office of justice assistance that are primarily related to administering federal homeland security moneys and that are in effect on the effective date of this paragraph remain in effect until their specified expiration

dates or until amended or repealed by the department of military affairs. All orders issued by the office of justice assistance that are primarily related to administering federal homeland security moneys and that are in effect on the effective date of this paragraph remain in effect until their specified expiration dates or until modified or rescinded by the department of military affairs.".

### **2117.** Page 1584, line 4: after that line insert:

"(8k) Madison Child Care facilities and services. The department of administration shall not expend more than \$284,600 in fiscal year 2007–08, not more than \$179,900 in fiscal year 2008–09, and not more than \$89,400 in fiscal year 2009–10 from the appropriation under section 20.505 (5) (ka) of the statutes, as affected by this act, for the purpose of subsidizing child care facilities and services in the city of Madison under section 16.841 of the statutes, as affected by this act.".

## **2118.** Page 1584, line 4: after that line insert:

- "(9f) Position decreases.
- (a) The authorized FTE positions for the department of administration, funded from the appropriation under section 20.505 (1) (im) of the statutes, are decreased in each year of the 2007–09 fiscal biennium by 1.0 FTE positions, for the purpose of not creating an office of energy independence.
- (b) The authorized FTE positions for the department of administration, funded from the appropriation under section 20.505 (1) (mb) of the statutes, are decreased in each year of the 2007–09 fiscal biennium by 6.0 FTE positions, for the purpose of not creating an office of energy independence.
- (c) The authorized FTE positions for the department of administration, funded from the appropriation under section 20.505 (3) (g) of the statutes, are decreased in

 $\mathbf{2}$ 

each year of the 2007-09 fiscal bienn	ium by 2.0 FTE positions, for the purpose of no
creating an office of energy independ	lence.".

## **2119.** Page 1584, line 19: after that line insert:

- "(1L) Position adjustments.
- (a) The authorized FTE positions for the department of agriculture, trade and consumer protection are increased by 1.5 PR positions to be funded from the appropriation under section 20.115 (1) (gf) of the statutes.
- (b) The authorized FTE positions for the department of agriculture, trade and consumer protection are increased by 1.0 FED position to be funded from the appropriation under section 20.115 (1) (m) of the statutes.
- (c) The authorized FTE positions for the department of agriculture, trade and consumer protection are decreased by 0.05 PR position funded from the appropriation under section 20.115 (7) (ja) of the statutes.
- (d) The authorized FTE positions for the department of agriculture, trade and consumer protection are decreased by 0.5 PR position funded from the appropriation under section 20.115 (8) (jm) of the statutes.
- (e) The authorized FTE positions for the department of agriculture, trade and consumer protection are increased by 0.5 PR position to be funded from the appropriation under section 20.115 (8) (kL) of the statutes.
- (f) The authorized FTE positions for the department of agriculture, trade and consumer protection are decreased by 3.0 FED positions funded from the appropriation under section 20.115 (8) (pz) of the statutes.".

# **2120.** Page 1584, line 19: after that line insert:

"(1x) Study of incentives for dairy cooperatives. The department of agriculture, trade and consumer protection shall study methods of providing incentives to cooperatives that manufacture or process dairy products to modernize and expand their facilities. The department shall report the results of the study to the senate and assembly committees with jurisdiction over agriculture no later than June 30, 2008."

## **2121.** Page 1585, line 2: after that line insert:

"(3p) Report on food safety programs. The department of agriculture, trade and consumer protection and the department of health and family services, jointly, shall prepare a plan to consolidate their food safety programs into one food safety program administered by the department of agriculture, trade and consumer protection. The departments shall submit a report on the plan, including a description of the efficiencies and the amount of savings that would result from the plan, to the senate and assembly committees with jurisdiction over agriculture no later than June 30, 2008."

## **2122.** Page 1585, line 2: after that line insert:

"(5i) AGRICULTURAL PRODUCER SECURITY PROGRAM. No later than June 30, 2008, the department of agriculture, trade and consumer protection shall submit to the senate and assembly committees with jurisdiction over agriculture a proposal for new provisions to ensure the financial security of agricultural producers. The department shall consult with representatives of the grain, dairy, and vegetable industries in developing its proposal and shall seek to provide a more efficient and less costly method of protecting producers in case of default by a contractor.".

# **2123.** Page 1585, line 2: after that line insert:

"(3i) Nutrient management consultants. No later than June 30, 2008, the department of agriculture, trade and consumer protection shall submit a plan to the senate and assembly committees with jurisdiction over agriculture to increase access by farmers to individuals who are trained to assist in the preparation of plans for agricultural nutrient management and who do not work for a governmental agency in order to maximize the number of cropped acres covered by nutrient management plans."

## **2124.** Page 1585, line 2: after that line insert:

- "(3w) Grant for Pulp and paper mill. In the 2007-09 fiscal biennium, the department of agriculture, trade and consumer protection shall award a grant from the appropriations under section 20.115 (4) (g) and (u) of the statutes, as created by this act, to the first person who operates a pulp and paper mill in this state without the use of natural gas or coal, if all of the following apply:
- (a) The person submits a plan to the department specifying the proposed use of the grant and the secretary of natural resources approves the plan.
- (b) The department enters into a written agreement with the person that specifies the conditions for the use of the grant, including reporting and auditing requirements.
- (c) The person agrees in writing to submit to the department, within 6 months after spending the grant proceeds, a report detailing how the grant proceeds were spent.".

## **2125.** Page 1585, line 2: after that line insert:

"(4u) Grants for soybean crushing facilities. During the 2007-09 fiscal biennium, the department of agriculture, trade and consumer protection shall make

5

6

7

8

9

10

11

12

13

15

16

17

18

19

20

- grants from the appropriation under section 20.115 (4) (qm) of the statutes, as created by this act, for the construction of soybean crushing facilities with the capacity to process more than 20,000,000 bushels of soybeans per year.".
  - **2126.** Page 1585, line 2: after that line insert:
  - "(5j) Privacy protection position authorizations.
  - (a) The authorized FTE positions for the department of agriculture, trade and consumer protection are decreased by 1.5 PR positions funded from the appropriation under section 20.115 (8) (ks) of the statutes for purposes related to privacy protection.
  - (b) The authorized FTE positions for the department of agriculture, trade and consumer protection are increased by 3.0 FED positions funded from the appropriation under section 20.115 (8) (pz) of the statutes for purposes related to privacy protection.".
- 14 **2127.** Page 1585, line 4: delete lines 4 to 13.
  - **2128.** Page 1585, line 13: after that line insert:
    - "(1p) Authorized Positions. The authorized FTE positions for the arts board, funded from the appropriation under section 20.215 (1) (ka) of the statutes, are decreased by 1.0 PR position associated with the administration of the fine arts in state buildings program.".
    - **2129.** Page 1592, line 15: delete lines 15 and 16 and adjust the appropriate totals accordingly.
- 22 **2130.** Page 1592, line 21: delete lines 21 and 22 and adjust the appropriate totals accordingly.

15

1	2131.	Page 1593, line 2: delete lines 2 and 3 and adjust the appropriate totals
2	accordingly.	
3	2132.	Page 1594, line 4: delete lines 4 and 5 and adjust the appropriate totals

- **2132.** Page 1594, line 4: delete lines 4 and 5 and adjust the appropriate totals accordingly.
- 5 **2133.** Page 1595, line 4: delete lines 4 and 5 and adjust the appropriate totals accordingly.
- 7 **2134.** Page 1595, line 12: delete lines 12 and 13 and adjust the appropriate totals accordingly.
- 9 **2135.** Page 1595, line 17: delete lines 17 to 20 and adjust the appropriate totals accordingly.
- 11 **2136.** Page 1597, line 18: delete "7,000,000" and substitute "4,242,000" and adjust the appropriate totals accordingly.
- 2137. Page 1597, line 20: delete "90,000,000" and substitute "54,546,000"
  and adjust the appropriate totals accordingly.
  - **2138.** Page 1597, line 22: delete "10,000,000" and substitute "6,061,000" and adjust the appropriate totals accordingly.
- 17 **2139.** Page 1598, line 2: delete "5,000,000" and substitute "3,030,000" and adjust the appropriate totals accordingly.
- 2140. Page 1598, line 4: delete "3,000,000" and substitute "1,818,000" and adjust the appropriate totals accordingly.
- 21 **2141.** Page 1598, line 6: delete "5,000,000" and substitute "3,030,000" and adjust the appropriate totals accordingly.

 $\mathbf{2}$ 

- **2142.** Page 1598, line 8: delete "45,000,000" and substitute "27,273,000" and adjust the appropriate totals accordingly.
  - **2143.** Page 1603, line 24: after that line insert:
- "(9p) 2001–03 STATE BUILDING PROGRAM DELETIONS. In 2001 Wisconsin Act 16, section 9107 (1) (p), under projects financed by general fund supported borrowing, the 2001–03 state building program project identified as Discovery Place museum Racine is deleted and the appropriate totals are decreased accordingly.".
  - **2144.** Page 1603, line 24: after that line insert:
  - "(9t) 2005–07 State Building Program deletions. In 2005 Wisconsin Act 25, section 9105 (1) (h) 3., under projects financed by program revenue supported borrowing for the University of Wisconsin–Platteville, the 2005–07 state building program project identified as Purchase and remodeling of buildings at 300 W. Highway 151 and 825 Chestnut Street (housing) is deleted and the appropriate totals are decreased accordingly."
    - **2145.** Page 1603, line 24: after that line insert:
- "(9q) Adjustment of totals. in the 2005-07 Authorized State Building Program, the appropriate totals are adjusted to reflect the changes made by Section 3936e of this act.".
  - **2146.** Page 1604, line 2: after that line insert:
- "(1d) PREVENTION OF CHILD MALTREATMENT. The authorized FTE positions for the child abuse and neglect prevention board, funded from the appropriation under section 20.433 (1) (q) of the statutes, are increased by 1.0 SEG position for the purpose of supporting activities for the prevention of maltreatment of children."
- **2147.** Page 1604, line 3: after that line insert:

- "(3g) CIRCUIT COURT BRANCH IN JUNEAU COUNTY. The initial election for circuit judge for branch 2 of the circuit court for Juneau County shall be at the spring election of 2008 for terms commencing August 1, 2008, and ending July 31, 2014.
  - (3h) CIRCUIT COURT BRANCH IN JUNEAU COUNTY.
- (a) The authorized FTE positions for the circuit courts are increased by 1.0 GPR circuit judge position on June 30, 2008, to be funded from the appropriation under section 20.625 (1) (a) of the statutes, to provide an additional circuit court judge for the circuit court branch created by section 753.06 (6) (e) of the statutes, as affected by this act.
- (b) The authorized FTE positions for the circuit courts are increased by 1.0 GPR court reporter position on June 30, 2008, to be funded from the appropriation under section 20.625 (1) (a) of the statutes, to provide one court reporter for the circuit court branch created by section 753.06 (6) (e) of the statutes, as affected by this act.".

## **2148.** Page 1604, line 14: after that line insert:

- "(2u) Report on residual contamination. The department of commerce, with the assistance of the department of natural resources, shall prepare a report concerning the petroleum storage remedial action program under section 101.143 of the statutes. The department shall submit the report to the legislature in the manner provided in section 13.172 (2) of the statutes and to the governor no later than October 1, 2008. The department shall include all of the following information in the report:
- (a) The number of sites closed under the program with residual contamination since the beginning of the program.

- (b) The amount of residual contamination and the conditions at each site described in paragraph (a) at the time that the department of commerce or the department of natural resources made a determination that no further remedial action was required at the site.
- (c) An estimate of the cost of testing all sites described in paragraph (a) to determine the extent to which the residual contamination has changed since the site was closed.
- (d) A description of the extent to which insurance available for petroleum storage tanks differs from the coverage and benefits provided under the program and of how those differences affect the ability of owners or operators of underground petroleum product storage tanks to operate their businesses.
- (e) An analysis of the feasibility of developing a public and private risk financing program for underground petroleum product storage tanks to pay the costs of cleaning up future petroleum releases at sites closed under the petroleum storage remedial action program under section 101.143 of the statutes.".
  - **2149.** Page 1604, line 15: delete lines 15 to 23.
- **2150.** Page 1604, line 24: before that line insert:
- 18 "(4f) Position decreases.
  - (a) Economic and community development; general program operations. The authorized FTE positions for the department of commerce, funded from the appropriation under section 20.143 (1) (a) of the statutes, as affected by this act, are decreased by 2.0 FTE positions, for the purpose of decreasing positions under this section.

- (b) Economic and community development; American Indian economic liaison and gaming grants specialist and program marketing. The authorized FTE positions for the department of commerce, funded from the appropriation under section 20.143 (1) (kg) of the statutes, as affected by this act, are decreased by 1.0 FTE position, for the purpose of decreasing positions under this section.
- (c) Housing assistance; general program operations. The authorized FTE positions for the department of commerce, funded from the appropriation under section 20.143 (2) (a) of the statutes, as affected by this act, are decreased by 1.0 FTE position, for the purpose of decreasing positions under this section.
- (d) Executive and administrative services; general program operations. The authorized FTE positions for the department of commerce, funded from the appropriation under section 20.143 (4) (a) of the statutes, as affected by this act, are decreased by 1.0 FTE position, for the purpose of decreasing positions under this section.
- (5f) Loans for pulp and paper mill. Notwithstanding section 560.61 of the statutes, as affected by this act, the department of commerce shall make 2 loans, each in an amount not to exceed \$1,000,000, in the 2007–09 fiscal biennium from the appropriation account under section 20.143 (1) (ie) of the statutes, as affected by this act, to the first person who operates a pulp and paper mill in this state without the use of natural gas or coal. The department shall enter into an agreement with the recipient of the loan under this subsection that specifies the uses for the loan proceeds and reporting and auditing requirements.".
- **2151.** Page 1604, line 25: delete the material beginning with that line and ending with page 1606, line 19.

10

11

12

13

14

15

16

17

18

19

- 1 **2152.** Page 1606, line 20: delete lines 20 to 25.
- 2 **2153.** Page 1607, line 1: delete lines 1 to 5.
- 3 **2154.** Page 1607, line 6: delete lines 6 to 12.
- **2155.** Page 1607, line 17: after "unit" insert ", and includes an evaluation of the need to expand North Hall to provide more inmate housing, program space, and a servery".
  - **2156.** Page 1607, line 17: delete "and" and substitute ",".
- 8 **2157.** Page 1607, line 17: after that line insert:
- 9 "(3p) Youth diversion grant reductions.
  - (a) Notwithstanding the dollar amount specified under section 301.265 (1) of the statutes, the department of corrections shall reduce the amount of money allocated under section 301.265 (1) of the statutes by \$10,000 annually.
  - (b) Notwithstanding the dollar amounts specified under section 301.265 (3) of the statutes, the department of corrections shall reduce the amount of money allocated for each of the 4 contracts that are funded with moneys from the appropriation account under section 20.410 (3) (d) of the statutes by \$3,000 annually and shall reduce the amount of money allocated for the contract that is funded only with moneys from the appropriation account under section 20.410 (3) (kj) of the statutes by \$3,100 annually.".
    - **2158.** Page 1607, line 17: after that line insert:
- "(4L) Administrative costs. The department of corrections may not decrease funding under section 20.410 (1) (a) of the statutes, as affected by this act, for operating correctional facilities and contracting for housing inmates, the sex

 $\mathbf{2}$ 

5

9

13

14

15

16

17

18

19

20

21

22

23

- offender registration program, global positioning system tracking for child sex offenders, or probation, parole, and extended supervision services.".
- 2159. Page 1607, line 21: delete lines 21 and 22 and substitute "accounts under section 20.455 (2) (kp) and (n) of".
  - **2160.** Page 1607, line 23: after "statutes," insert "as affected by this act,".
- **2161.** Page 1608, line 5: delete lines 5 and 6 and substitute "accounts under section 20.455 (2) (kp) and (n) of".
- 8 **2162.** Page 1608, line 7: after "statutes," insert "as affected by this act,".
  - **2163.** Page 1610, line 16: delete "as affected by this act,".
- 10 **2164.** Page 1610, line 20: after that line insert:
- 11 "(3u) Payment of required employee contributions under the Wisconsin Retirement System.
  - (a) The definitions in section 20.001 of the statutes are applicable in this subsection.
  - (b) The secretary of administration shall determine for each state agency the amount that the agency is not required to spend during the period that begins on September 1, 2007, and ends on June 30, 2009, as a result of subsection (3w), section 40.03 (1) (e) of the statutes, as affected by this act, and section 40.05 (1) (b) 3. of the statutes, as created by this act, and from each appropriation from which the moneys would have been expended during that period, other than appropriations of federal revenues.
  - (c) From each sum certain appropriation of general purpose revenue identified in paragraph (b), the secretary of administration shall lapse to the general fund the amount specified in paragraph (b) that would otherwise have been expended from

 $\mathbf{2}$ 

each of the appropriations. After the secretary makes the lapse, each of the sum certain appropriations is decreased by the amount specified in paragraph (b) for that appropriation.

- (d) For each sum sufficient appropriation of general purpose revenue identified in paragraph (b), the expenditure estimate for the appropriation during the 2007–09 fiscal biennium is reestimated to subtract the amount specified in paragraph (b) for that appropriation.
- (e) Each sum certain program revenues or program revenues-service appropriations is decreased by the amount specified in paragraph (b) for that appropriation.
- (f) From each appropriation of segregated fund revenues or segregated fund revenues-service identified in paragraph (b), the secretary of administration shall lapse to the underlying fund the amount specified in paragraph (b) that would otherwise have been expended from each of the appropriations. After the secretary makes the lapse, each of the sum certain segregated revenues or segregated revenues-service appropriations is decreased by the amount specified in paragraph (b) for that appropriation and the expenditure estimate for each of the appropriations that is not a sum certain appropriation is reestimated to subtract the amount specified in paragraph (b) for that appropriation.
- (3w) Required employee contribution rates under the Wisconsin Retirement System for 2007. Notwithstanding the actions of the employee trust funds board in establishing employee contribution rates under section 40.05 (1) (a) 1. to 4. of the statutes for 2007, if any category of participating employees specified in section 40.05 (1) (a) 1. to 4. of the statutes has an employee contribution rate that is less than 5

 $\mathbf{2}$ 

percent for 2007, the rate shall become 5 percent beginning on September 1, 2007, and shall remain at that rate for the remainder of 2007.".

## **2165.** Page 1610, line 20: after that line insert:

- "(3f) Payment of Health Insurance Premiums.
- (a) The definitions in section 20.001 of the statutes are applicable in this subsection.
- (b) The secretary of administration shall determine for each state agency the amount that the agency is not required to spend during the period that begins on January 1, 2008, and ends on June 30, 2009, as a result of section 40.05 (4) (a) 4. of the statutes, as created by this act, and from each appropriation from which the moneys would have been expended during that period, other than appropriations of federal revenues.
- (c) From each sum certain appropriation of general purpose revenue identified in paragraph (b), the secretary of administration shall lapse to the general fund the amount specified in paragraph (b) that would otherwise have been expended from each of the appropriations. After the secretary makes the lapse, each of the sum certain appropriations is decreased by the amount specified in paragraph (b) for that appropriation.
- (d) For each sum sufficient appropriation of general purpose revenue identified in paragraph (b), the expenditure estimate for the appropriation during the 2007–09 fiscal biennium is reestimated to subtract the amount specified in paragraph (b) for that appropriation.

 $\mathbf{2}$ 

- (e) Each sum certain program revenues or program revenues-service appropriations is decreased by the amount specified in paragraph (b) for that appropriation.
- (f) From each appropriation of segregated fund revenues or segregated fund revenues-service identified in paragraph (b), the secretary of administration shall lapse to the underlying fund the amount specified in paragraph (b) that would otherwise have been expended from each of the appropriations. After the secretary makes the lapse, each of the sum certain segregated revenues or segregated revenues-service appropriations is decreased by the amount specified in paragraph (b) for that appropriation and the expenditure estimate for each of the appropriations that is not a sum certain appropriation is reestimated to subtract the amount specified in paragraph (b) for that appropriation.".
- **2166.** Page 1612, line 12: delete lines 12 to 23.
- **2167.** Page 1612, line 24: delete that line.
- **2168.** Page 1613, line 1: delete lines 1 to 25.
- **2169.** Page 1614, line 1: delete lines 1 to 25.
- **2170.** Page 1615, line 1: delete lines 1 to 25.
- **2171.** Page 1616, line 1: delete lines 1 to 24.
- **2172.** Page 1617, line 1: delete lines 1 to 8.
- **2174.** Page 1619, line 5: delete lines 5 to 18.

- 2 2175. Page 1620, line 25: delete the material beginning with that line and ending with page 1621, line 12.
  - **2176.** Page 1621, line 13: delete lines 13 to 19.
- **2177.** Page 1621, line 19: after that line insert:
  - "(9f) State centers for the developmentally disabled. The authorized FTE positions for the department of health and family services are increased by 6.64 PR positions on July 1, 2007, to be funded from the appropriation account under section 20.435 (2) (gk) of the statutes, for the purpose of performing services at the state centers for the developmentally disabled."
    - **2178.** Page 1621, line 19: after that line insert:
  - "(10i) DIVISION ADMINISTRATOR POSITION DECREASE. The authorized FTE positions for the department of health and family services, funded from the appropriation account under section 20.435 (2) (a) of the statutes, are decreased on July 1, 2007, by 1.0 GPR position for the function of division administrator.".
    - **2179.** Page 1621, line 19: after that line insert:
  - "(10j) CLIENT ASSISTANCE FOR REEMPLOYMENT AND ECONOMIC SUPPORT SYSTEM POSITIONS. The authorized FTE positions for the department of health and family services are increased by 10.5 FED positions on July 1, 2007, to be funded from the appropriation under section 20.435 (4) (n) of the statutes, for the purpose of supporting the client assistance for reemployment and economic support system.".
    - **2180.** Page 1621, line 22: delete lines 22 to 24.
- **2181.** Page 1622, line 1: delete lines 1 to 12.
  - **2182.** Page 1623, line 13: after that line insert:

- "(1p) Federal Byrne Justice Assistance grant funding for Multijurisdictional enforcement groups. For the 2007-08 and 2008-09 fiscal years, the department of justice shall allocate to multijurisdictional enforcement groups 44 percent of the federal Byrne Justice Assistance Grant awards appropriated under section 20.455 (2) (n) of the statutes.
  - (1q) Office of Justice Assistance Transfer.
- (a) The authorized FTE positions for the department of justice are increased by 1.40 PR positions to be funded from the appropriation under section 20.455 (2) (ky) of the statutes, as affected by this act, for the purpose of transferring duties of the office of justice assistance to the department of justice.
- (b) The authorized FTE positions for the department of justice are increased by 2.27 GPR positions to be funded from the appropriation under section 20.455 (2) (a) of the statutes for the purpose of transferring duties of the office of justice assistance to the department of justice.
- (c) The authorized FTE positions for the department of justice funded from the appropriation under section 20.455 (2) (m) of the statutes, as affected by this act, are increased by 13.58 FED positions for fiscal year 2007–08 and 11.58 FED positions for fiscal year 2008–09 for the purpose of transferring duties of the office of justice assistance to the department of justice.".

## **2183.** Page 1623, line 13: after that line insert:

"(1h) DISTRICT ATTORNEY POSITION; CHIPPEWA COUNTY. From the appropriation account under section 20.455 (2) (n) of the statutes, the department of justice shall expend \$16,700 in fiscal year 2007–08 and \$16,700 in fiscal year 2008–09 to provide .25 assistant district attorney position in Chippewa County."

**2184.** Page 1623, line 13: after that line insert:

"(1f) DISTRICT ATTORNEY POSITION; ST. CROIX COUNTY. From the appropriation account under section 20.455 (2) (n) of the statutes, the department of justice shall expend \$32,400 in fiscal year 2007–08 and \$64,800 in fiscal year 2008–09 to provide 1.0 assistant district attorney position in St. Croix County.".

**2185.** Page 1623, line 21: after that line insert:

- "(1j) Audit of 4-Year-old kindergarten programs. The legislative audit bureau shall conduct an audit of school district 4-year-old kindergarten programs. The audit shall address whether revenue limit authority and state aid for the programs is greater than the costs of the programs. The bureau shall file its report as described in section 13.94 of the statutes by January 15, 2009."
- **2186.** Page 1624, line 20: delete lines 20 to 25.
- **2187.** Page 1625, line 1: delete lines 1 to 6.
- **2188.** Page 1625, line 6: after that line insert:
  - "(3i) Audit of technical college programs. The legislative audit bureau shall conduct a performance evaluation audit of all technical college system programs that award grants for the purpose of recruiting or retaining students. The legislative audit bureau shall file its report as described in section 13.94 of the statutes by January 15, 2009."
    - **2189.** Page 1625, line 6: after that line insert:
  - "(5q) Autism scholarship program audit of the Autism Scholarship Program under section 115.85 of the statutes, as created by this act, based on the data

from the 2008–09 fiscal year. If the bureau performs the audit, it shall file its report as described in section 13.94 (1) (b) of the statutes by December 31, 2009.".

#### **2190.** Page 1625, line 6: after that line insert:

"(4c) Audit of higher educational aids board under subchapter III of chapter 39 of the statutes, as affected by this act. The legislative audit bureau shall file its report as described in section 13.94 (1) (b) of the statutes by January 15, 2009."

#### **2191.** Page 1625, line 6: after that line insert:

"(3w) Applicability of legislative rules of proceeding related to referral of bills and amendments to the joint survey committee on retirement systems. Section 13.50 (6) (a) of the statutes shall not apply to the actions of the legislature in enacting this act.".

## **2192.** Page 1625, line 6: after that line insert:

- "(4v) Legislative appropriation lapses and reestimates.
- (a) In this subsection, "state operations" means all purposes except aids to individuals and organizations and local assistance.
- (b) The cochairpersons of the joint committee on legislative organization shall take actions during the 2007–09 fiscal biennium to ensure that from general purpose revenue appropriations for state operations to the legislature under section 20.765 of the statutes, as affected by this act, an amount equal to a total of \$3,561,000 in fiscal year 2007–08 and a total of \$2,744,600 in fiscal year 2008–09 are lapsed from sum certain appropriation accounts or are subtracted from the expenditure estimates for any other types of appropriations, or both.".

**2193.** Page 1625, line 6: after that line insert:

"(5c) Audit of department of commerce educational loan forgiveness programs. The legislative audit bureau shall conduct a performance evaluation audit of all educational loan forgiveness programs administered by the department of commerce under subchapter I of chapter 560 of the statutes, as affected by this act. The legislative audit bureau shall file its report as described in section 13.94 (1) (b) of the statutes by January 15, 2009."

- **2194.** Page 1625, line 13: delete lines 13 to 17.
- **2195.** Page 1625, line 18: delete lines 18 to 22.
  - **2196.** Page 1627, line 15: after that line insert:

"(4u) Emissions inspection program position. The authorized FTE positions for the department of natural resources, funded from the appropriation under section 20.370 (2) (cf) of the statutes, are increased in each year of the 2007–09 fiscal biennium by 0.75 FTE position, for the purpose of the vehicle emission inspection and maintenance program. No department of natural resources positions related to the vehicle emission inspection and maintenance program may be funded from any appropriation from the segregated transportation fund during the 2007–09 fiscal biennium.".

# **2197.** Page 1627, line 15: after that line insert:

"(4d) National Wild Turkey Federation. The department of natural resources shall provide \$50,000 in fiscal year 2007–08 and \$50,000 in fiscal year 2008–09 from the appropriation under section 20.370 (1) (ht) of the statutes to the Wisconsin chapter of the National Wild Turkey Federation to be used fund the turkey habitat

restoration and management activities of a National Wild Turkey Federation biologist.".

## **2198.** Page 1627, line 15: after that line insert:

"(4p) Stewardship audit. The joint legislative audit committee is requested to, and may, direct the legislative audit bureau to perform a performance evaluation audit and financial audit of the Warren Knowles–Gaylord Nelson stewardship 2000 program under section 23.0917 of the statutes. If the committee directs the legislative audit bureau to perform an audit under this subsection, the bureau shall file its report as described in section 13.94 (1) (b) of the statutes before July 1, 2009."

## **2199.** Page 1627, line 15: after that line insert:

"(4j) Stewardship program report. The department of natural resources shall evaluate the Warren Knowles-Gaylord Nelson stewardship 2000 program and prepare a report that specifies goals for the stewardship program, including criteria for determining when the stewardship program should be concluded. The department of natural resources shall submit the report no later than January 1, 2009, to the joint committee on finance, to the legislature for distribution to the appropriate standing committees, in the manner provided under section 13.172 (3) of the statutes, to the speaker of the assembly, in the manner provided under section 13.172 (3) of the statutes, and to the senate majority leader."

# **2200.** Page 1627, line 15: after that line insert:

"(7x) Environmental review of transportation projects positions. The authorized FTE positions for the department of natural resources, funded from payments under section 86.31 of the statutes, as affected by this act, from the

- appropriation under section 20.395 (2) (fr) of the statutes, are decreased by 3.0 FTE positions, for the purpose of environmental review of local transportation projects.".
- **2201.** Page 1627, line 18: delete lines 18 to 25.
- **2202.** Page 1628, line 1: delete lines 1 to 10.
- **2203.** Page 1628, line 11: delete that line and substitute:
- 6 "(3k) School district efficiency incentive grants.".
- **2204.** Page 1628, line 25: after that line insert:
  - "(3L) SECOND CHANCE PARTNERSHIP. From the appropriation under section 20.255 (2) (cf) of the statutes, as affected by this act, the department of public instruction shall pay \$190,000 in each of the 2007–08 and 2008–09 fiscal years to the Second Chance Partnership, a nonprofit corporation, to create a pilot work-based learning program in which children at risk participate in apprenticeships while earning high school diplomas."
    - **2205.** Page 1629, line 7: delete lines 7 to 18.
- **2206.** Page 1629, line 18: after that line insert:
  - "(6q) Rules. Using the procedure under section 227.24 of the statutes, the department of public instruction shall promulgate the rules required under section 115.85 (4) of the statutes, as created by this act, for the period before the effective date of the permanent rules promulgated under that section, but not to exceed the period authorized under section 227.24 (1) (c) and (2) of the statutes. Notwithstanding section 227.24 (1) (a), (2) (b), and (3) of the statutes, the department is not required to provide evidence that promulgating a rule under this subsection as an emergency rule is necessary for the preservation of the public peace, health, safety, or welfare

and is not required to provide to a finding of emergency for a rule promulgated under this subsection.".

## **2207.** Page 1629, line 18: after that line insert:

- "(7f) Open Enrollment Program; extension of application period. Notwithstanding section 118.51 (3) of the statutes, the following timeline applies to the parent of a pupil who resides in the Marinette School District, who would have been eligible to attend Porterfield Elementary School in the 2007–08 school year if the Marinette school board had not voted to close the school, and who wishes to attend a public school in a nonresident school district in the 2007–08 school year under section 118.51 of the statutes:
- (a) The parent shall submit an application to the school board of the nonresident school district within 5 days after the effective date of this paragraph. Upon receipt of the application, the nonresident school board shall send a copy to the school board of the Marinette School District and the department of public instruction.
- (b) Not before the 6th day after the effective date of this paragraph, but by the12th day after the effective date of this paragraph:
- 1. The nonresident school board shall notify the applicant whether it has accepted the pupil and, if it has accepted the pupil, the specific school or program that the pupil may attend in the 2007–08 school year.
- 2. If the Marinette school board denies a pupil's enrollment in a nonresident school district, it shall notify the applicant and the nonresident school board.

- (c) Within 7 days after receiving notice of acceptance under paragraph (b) 1., the pupil's parent shall notify the nonresident school board of the pupil's intent to attend school in that school district in the 2007–08 school year.
- (d) By September 1, 2007, each nonresident school board that has accepted a pupil under this subsection for attendance in the 2007–08 school year shall report the name of the pupil to the Marinette school board.".

## **2208.** Page 1629, line 18: after that line insert:

"(8f) Audit of Student Achievement Guarantee in Education Program. The legislative audit bureau is requested to perform a financial and performance evaluation audit of the Student Achievement Guarantee in Education Program under section 118.43 of the statutes, as affected by this act. If the legislative audit bureau performs the audit, it shall file its report as described under section 13.94 (1) (b) of the statutes by January 15, 2009."

## **2209.** Page 1629, line 18: after that line insert:

"(9f) Audit of community service levies. The legislative audit bureau shall perform an audit of community education, training, recreational, cultural, or athletic programs and services established and maintained under section 120.13 (19) of the statutes through school district levies. The legislative audit bureau shall file its report as described under section 13.94 (1) (b) of the statutes by January 15, 2009.".

## **2210.** Page 1630, line 2: after that line insert:

- "(2t) Electric supply investigation.
- (a) In this subsection, "commission" means the public service commission.
- (b) The commission shall undertake an investigation of alternative strategies that the state could pursue to meet the expected demand for electricity in the state

 $\mathbf{2}$ 

- in 2033, which is the final year in which the operating licenses of the nuclear power plants in Wisconsin are set or expected to expire. The study shall be designed to provide guidance to the commission, the legislature, and the electric power industry in formulating policy and conducting planning regarding the future sources of electric power that will be available in this state, including intermediate and longer-term strategies for achieving alternative outcomes identified in the investigation.
- (c) The commission shall base the investigation under paragraph (b) on the development of scenarios that project in the forecast period the total demand for electricity and the proportions of electricity that are expected to be generated by different types of fuels. The commission shall do all of the following in designing and preparing these scenarios:
- 1. Incorporate appropriate factors, including the continuation of a reliable and adequate supply of electricity, the expected retirement of existing electric generating facilities, and construction of new electric generating facilities, the importation of electricity into the state, and expected improvements in the electric transmission system.
- 2. Include scenarios that include electric generating technologies that use coal as a fuel and produce no or minimal emissions of carbon dioxide, that use nuclear fuel in advanced light water reactors, and that use both types of technologies.
- 3. Base the scenarios on the assumed implementation of a range of energy efficiency programs, ranging from continuation of the current programs to implementation of programs that produce the maximum energy savings foreseen by the commission, and of a range of regulatory programs that result in an effective cost

- of air emissions of carbon dioxide ranging from \$0 per ton to the highest amount per ton foreseen by the commission.
  - 4. Identify the expected economic costs and environmental impacts of each scenario using established methodologies that facilitate comparisons of the scenarios.
  - (d) The commission shall appoint an advisory committee under section 15.04 (1) (c) of the statutes to assist in the investigation and report required under this subsection.
  - (e) The Energy Institute of the University of Wisconsin–Madison shall assist the commission in conducting the investigation under this subsection.
  - (f) The commission shall prepare a report on the results of its investigation under this subsection and shall submit the report to the legislature in the manner provided under section 13.172 (3) of the statutes, no later than the first day of the 19th month beginning after the effective date of this paragraph.".
    - **2211.** Page 1630, line 2: after that line insert:
  - "(2u) RAILROAD SAFETY ANALYST POSITION. The authorized FTE positions for the office of the commissioner of railroads, funded from the appropriation under section 20.155 (2) (g) of the statutes, is decreased by 1.0 PR railroad safety analyst position.".
  - **2212.** Page 1630, line 17: delete lines 17 to 23.
- **2213.** Page 1631, line 7: delete lines 7 to 14.
- **2214.** Page 1632, line 2: after that line insert:
- 22 "(2t) Waste treatment facility exemptions.
  - (a) Notwithstanding any other provision of chapter 70 of the statutes, property tax assessments under section 70.11 (21) of the statutes, as affected by this act, as

of January 1, 2007, supersede any other property tax assessments under section 70.11 (21), 2005 stats., for property tax assessments as of January 1, 2007, that are made prior to the effective date of this subsection. Notwithstanding sections 70.47 (7) and 70.995 (8) of the statutes, an objection to a property tax assessment under section 70.11 (21) of the statutes, as affected by this act, for property tax assessments as of January 1, 2007, may be filed no later than 60 days after the effective date of this subsection or no later than the time allowed under sections 70.47 (7) and 70.995 (8) of the statutes, whichever is later.

(b) Section 77.54 (26) of the statutes, as affected by this act, does not apply to tangible personal property purchased in fulfillment of a contract to construct, repair, or improve a waste treatment facility, if the contract is entered into, or a formal bid is made, prior to the effective date of this subsection and the tangible personal property is affixed to and made a structural part of the waste treatment facility.".

# **2215.** Page 1632, line 2: after that line insert:

"(3i) Department of Revenue, audit bureau review of sales and use tax court case. The legislative audit bureau, in consultation with the department of revenue, shall review and analyze the state supreme court's decision in the case of *Department of Revenue v. River City Refuse Removal, Inc.*, 2007 WI 27, which was decided on March 8, 2007. The review and analysis shall focus on the imposition of the sales and use tax on transfers of goods and services between affiliated businesses. The audit bureau and the department of revenue shall estimate the potential losses of state and local sales and use tax revenues as a result of the court decision, and shall present options to the legislature for the mitigation of any potential revenue losses. On or before October 1, 2007, the audit bureau shall submit a written report of its findings,

- conclusions, and recommendations to the chief clerk of each house of the legislature,
- for distribution to the legislature under section 13.172 (2) of the statutes.".
- **2216.** Page 1632, line 9: delete lines 9 to 14.
- **2217.** Page 1632, line 21: delete lines 21 to 25.
- **2218.** Page 1633, line 1: delete lines 1 to 4.
- **2219.** Page 1633, line 19: after that line insert:
- 7 "(3f) License plates for sex offenders.

- (a) In this subsection, "sex offender" has the meaning given in section 341.01(2) (c) of the statutes, as created by this act.
  - (b) Not later than 60 days after the effective date of this paragraph, a person who, as of the effective date of this paragraph, is a sex offender and who has any motor vehicle registered in his or her name under chapter 341 of the statutes shall apply to the department of transportation for replacement registration plates for each such vehicle, identifying himself or herself in the application as a person to whom special registration plates shall be issued under section 341.14 (6y) of the statutes, as created by this act. Upon receipt of such an application, along with a fee of \$15 for each plate, the department of transportation shall issue to the applicant replacement registration plates under section 341.14 (6y) of the statutes, as created by this act. Upon receipt of these replacement registration plates, the applicant shall remove and destroy all registration plates replaced."
    - **2220.** Page 1634, line 3: after that line insert:
- 22 "(4d) Study and report regarding truck size and weight limits.
  - (a) The department of transportation shall contract for a study of Wisconsin's truck size and weight limit laws, to identify changes in those laws that would have

- a net benefit to Wisconsin's economy, when considering the costs of protecting highway infrastructure and safety, and the benefits that would result from reducing the cost of truck transportation.
- (b) The consultant that undertakes the study under this subsection shall review those vehicle configurations, changes in seasonal restrictions, and other policy issues that were found to have a net benefit in the cost-benefit analysis in the Minnesota truck size and weight project final report that was issued in June 2006.
- (c) The department of transportation shall appoint an advisory committee under section 227.13 of the statutes to assist in the review and report required under this subsection. The advisory committee shall include representation from the department of commerce and local governmental units, trucking companies, industries and small businesses that depend on truck transport, enforcement agencies, and other groups and individuals that are interested in and knowledgeable about truck size and weight limits. All advisory committee members may present written commentary on or dissenting views from the report and the department of transportation shall incorporate that commentary and any dissents into the final report.
- (d) The consultant that undertakes the study under this subsection shall prepare a report on the results of its study. The department of transportation shall submit the report to the legislature in the manner provided under section 13.172 (3) of the statutes, no later than January 1, 2009.".

# **2221.** Page 1634, line 13: after that line insert:

"(5f) Engineering Positions. The authorized FTE positions for the department of transportation, funded from the appropriation under section 20.395 (3) (cr) of the

- statutes, are decreased by 25.0 FTE positions, for the purpose of preliminary work related to the reconstruction of the Zoo interchange in Milwaukee County.".
- 2222. Page 1634, line 15: delete "The" and substitute "Subject to federal approval as provided in paragraph (e), the".
- **2223.** Page 1635, line 13: delete "any grant awarded" and substitute "the department of transportation shall award the grant".
- **2224.** Page 1635, line 13: delete "(a)" and substitute "(a),".
- **2225.** Page 1635, line 14: delete "approved shall be paid" and substitute "approved,".
- **2226.** Page 1635, line 17: delete lines 17 to 21.
- **2227.** Page 1636, line 6: after that line insert:

- 12 "(9i) VILLAGE OF ROTHSCHILD BUSINESS USH 51 WIDENING PROJECT.
  - (a) In the 2007-09 fiscal biennium, from the appropriation under section 20.395 (2) (iq) of the statutes, the department of transportation shall award a grant under section 84.185 of the statutes, as affected by this act, in the amount of \$200,000, to the village of Rothschild in Marathon County to widen the business route of USH 51 in or near the village of Rothschild from 2 lanes to 4 lanes and for related improvements if the village of Rothschild applies for the grant and contributes funds for the project that total at least \$200,000. The provisions of section 84.185 of the statutes, as affected by this act, relating to the awarding of grants, the amount of grants, and the eligibility requirements for grants do not apply to grants awarded under this paragraph.
  - (b) In the 2007-09 fiscal biennium, from the appropriation under section 20.395 (3) (cq) of the statutes, as affected by this act, the department of

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

- transportation shall provide \$238,300 to the village of Rothschild in Marathon

  County for the project described in paragraph (a).".
  - **2228.** Page 1636, line 6: after that line insert:
- "(12t) RECONSTRUCTION PROJECT IN CITY OF COLBY. The department of transportation shall commence, during the 2008–09 fiscal year, a reconstruction project on that portion of STH 13 that is Division Street in the city of Colby in Marathon County."
  - **2229.** Page 1636, line 6: after that line insert:
  - "(8n) OREGON BYPASS PROJECT. The department of transportation shall, in the 2007-09 fiscal biennium, complete the Oregon bypass project on USH 14 in Dane County, which project includes the reconstruction of 2 existing lanes and the construction of 2 new lanes from CTH "MM" to STH 138.".
    - **2230.** Page 1636, line 6: after that line insert:
  - "(9z) Improvement project in the village of Rib Lake. Notwithstanding limitations on the amount and use of aids provided under section 86.31 of the statutes, as affected by this act, or on eligibility requirements for receiving aids under section 86.31 of the statutes, as affected by this act, in the 2007–09 fiscal biennium, from the appropriation under section 20.395 (2) (ft) of the statutes, as affected by this act, the department of transportation shall award a grant under section 86.31 (3r) of the statutes, as affected by this act, in the amount of \$5,750, to the village of Rib Lake in Taylor County for improvements to McComb Avenue in the village of Rib Lake."
    - **2231.** Page 1636, line 6: after that line insert:

"(9u) Kenosha-Racine-Milwaukee commuter rail extension project. The department of transportation may submit a request to the joint committee on finance to supplement the appropriation account under section 20.395 (2) (ct) of the statutes by up to \$800,000 in the 2007–09 fiscal biennium from the appropriation account under section 20.865 (4) (u) of the statutes, for preliminary engineering for the Kenosha-Racine-Milwaukee commuter rail extension project. The committee may supplement the appropriation account under section 20.395 (2) (ct) of the statutes by up to \$800,000 in the 2007–09 fiscal biennium from the appropriation account under section 20.865 (4) (u) of the statutes, for this purpose, only if the legislature has enacted legislation, signed by the governor, establishing a financing mechanism sufficient to pay all costs, including capital and operating costs but excluding any federal share of costs, related to this commuter rail extension project and commuter rail service resulting from this extension. Notwithstanding section 13.101 (3) of the statutes, the committee is not required to find that an emergency exists prior to making the supplementation under this subsection."

### **2232.** Page 1636, line 6: after that line insert:

"(9y) Stillwater Bridge project consultant. During the 2007-09 fiscal biennium, the department of transportation shall enter into a contract with a financial consultant to work on aspects of the financing of the construction of the Stillwater Bridge across the St. Croix River between the town of Houlton in St. Croix County and the city of Stillwater, Minnesota, if federal funds are provided to this state for this purpose."

**2233.** Page 1637, line 1: delete lines 1 to 5.

**2234.** Page 1637, line 7: delete lines 7 to 11.

 $\mathbf{2}$ 

**2235.** Page 1637, line 16: delete the material beginning with that line and ending with page 1638, line 5.

#### **2236.** Page 1638, line 16: after that line insert:

"(3u) Law school funding. Notwithstanding section 16.42 (1) (e) of the statutes, in submitting information under section 16.42 of the statutes for the purposes of the 2009–11 biennial budget bill, the Board of Regents of the University of Wisconsin System shall submit a dollar amount for the appropriation under section 20.285 (1) (a) of the statutes as though the amount appropriated to the board in fiscal year 2008–09 under section 20.285 (1) (a) of the statutes were reduced by \$2,000,000, for the purpose of reducing general purpose revenue funding for the law school of the University of Wisconsin."

#### **2237.** Page 1638, line 16: after that line insert:

"(3k) High-demand faculty funding. Of the amounts appropriated to the Board of Regents of the University of Wisconsin System under section 20.285 (1) (a) of the statutes, the board shall allocate \$2,000,000 in fiscal year 2007-08 and \$4,000,000 in fiscal year 2008-09 for the purpose of supporting supplemental salary increases for faculty whose services are in high demand by other higher educational institutions. Of the amounts allocated under this subsection, the board may expend no more than 60 percent for faculty at the University of Wisconsin-Madison or the University of Wisconsin-Milwaukee. This subsection does not affect the board's authority to increase faculty salaries under section 36.09 (1) (j) of the statutes to recognize competitive factors."

**2238.** Page 1639, line 7: after that line insert:

- "(3u) Council on Veterans programs. Notwithstanding section 15.497 (2)

  (intro.), the term of the initial representative of the Wisconsin Council of the Military

  Officers Association of America and The Retired Enlisted Association on the council

  on veterans programs shall expire on July 1, 2008."
- **2239.** Page 1639, line 7: after that line insert:
  - "(3g) Planning and study of cemetery in Outagamie County. From the appropriation under section 20.485 (2) (tm) of the statutes, the department of veterans affairs shall expend an amount not to exceed \$35,000 during fiscal year 2007–08 for a capital planning and feasibility study of a new state veterans cemetery in Outagamie County."
- **2240.** Page 1639, line 9: delete lines 9 to 25.
- **2241.** Page 1640, line 1: delete lines 1 to 25.
- **2242.** Page 1641, line 1: delete lines 1 to 25.
- **2243.** Page 1642, line 1: delete lines 1 to 18.
- **2244.** Page 1642, line 19: delete lines 19 to 23.
- **2245.** Page 1642, line 24: delete lines 24 and 25.
- **2246.** Page 1643, line 1: delete lines 1 to 6.
- **2247.** Page 1643, line 8: delete lines 8 to 21.
- **2248.** Page 1643, line 22: delete the material beginning with that line and ending with page 1644, line 14.
- **2249.** Page 1644, line 23: after that line insert:
- 22 "(1f) Transfer of office of Justice Assistance.

- (a) The unencumbered balance in the appropriation account under section 20.505 (6) (i) of the statutes is transferred to the appropriation account under section 20.455 (3) (g) of the statutes.
  - (b) The unencumbered balance in the appropriation account under section 20.505 (6) (km) of the statutes is transferred to the appropriation account under section 20.410 (3) (ky) of the statutes.
  - (c) The unencumbered balance in the appropriation account under section 20.505 (6) (m) of the statutes is transferred to the appropriation account under section 20.455 (2) (m) of the statutes.
- (d) The unencumbered balance in the appropriation account under section 20.505 (6) (p) of the statutes is transferred to the appropriation account under section 20.455 (2) (n) of the statutes.".
- **2250.** Page 1647, line 19: delete lines 19 to 21.
- **2251.** Page 1652, line 6: delete lines 6 to 8.
- 2252. Page 1652, line 21: delete "\$3,833,000" and substitute "\$2,920,600".
- **2253.** Page 1652, line 22: delete "\$1,917,200" and substitute "\$982,100".
- **2254.** Page 1653, line 6: delete lines 6 to 13.
- **2255.** Page 1654, line 16: after that line insert:
- "(2u) Motor vehicle fuel retailer contracts. The treatment of section 100.55
  of the statutes, as created by this act, first applies to contracts entered into, renewed,
  or extended on the effective date of this subsection.".
- **2256.** Page 1654, line 17: after that line insert:

"(1p) Fine arts in state buildings program. The treatment of sections 13.48
(10) (a), $20.215$ (1) (k) and (ka) and $44.57$ of the statutes first applies with respect to
state buildings the design of which is approved by the building commission, or if not
subject to building commission approval, the design of which is approved by the
agency contracting for the design, on the effective date of this subsection.".
<b>2257.</b> Page 1654, line 22: delete lines 22 to 24.
<b>2258.</b> Page 1654, line 24: after that line insert:

- "(3d) Collection of fines and forfeitures. The treatment of section 59.25 (3) (j) of the statutes first applies to fines and forfeitures that are uncollected on the effective date of this subsection.".
  - **2259.** Page 1654, line 24: after that line insert:
- "(4c) Remedies against manufacturers. The treatment of section 895.046 of the statutes first applies to actions commenced on the effective date of this subsection.".
- **2260.** Page 1655, line 23: after that line insert:
- "(1p) Exposure to Child. The treatment of section 948.10 (1) of the statutes first applies to acts committed on the effective date of this subsection.".
  - **2261.** Page 1656, line 5: after that line insert:
- "(1w) John Doe Proceedings. The treatment of section 968.26 (2) of the statutes first applies to complaints made on the effective date of this subsection.".
  - **2262.** Page 1656, line 8: after that line insert:
- "(1d) Payment of employee required contributions. The renumbering and amendment of section 40.05 (1) (b) of the statutes and the creation of section 40.05 (1) (b) 4. of the statutes first apply to participating employees in the Wisconsin Retirement System who are covered by a collective bargaining agreement under

subchapter IV of chapter 111 of the statutes that contains provisions inconsistent with this act on the day on which the agreement expires or is extended, modified, or renewed, whichever occurs first.".

### **2263.** Page 1656, line 10: after that line insert:

"(1p) Final Offer Limits. The amendment of section 111.70 (4) (cm) 6. am. and b. and 7. of the statutes, and the creation of section 111.70 (1) (ag) of the statutes first apply to petitions for arbitration that relate to collective bargaining agreements that cover periods beginning on or after July 1, 2008, and that are filed under section 111.70 (4) (cm) 6. of the statutes, as affected by this act, on the effective date of this subsection.".

### **2264.** Page 1656, line 10: after that line insert:

"(1d) Competitive contracting. The treatment of section 111.70 (1) (a) (as it relates to competitive contracting) and (dm) and (4) (cm) 5s. and (p) of the statutes first applies to collective bargaining agreements under subchapter IV of chapter 111 of the statutes for which a notice of commencement of contract negotiations has been filed by either party on the effective date of this subsection."

# **2265.** Page 1656, line 10: after that line insert:

"(1L) Collective Bargaining related to provision of food services in country Jails. The treatment of section 111.70 (1) (a) (as it relates to the provision of food services in country jails) and (4) (m) (title) and (pr) of the statutes first applies to collective bargaining agreements entered into, extended, modified, or renewed, whichever occurs first, on the effective date of this subsection."

# **2266.** Page 1656, line 10: after that line insert:

 $\mathbf{2}$ 

"(2i) Municipal employer-initiated change in health care coverage. The treatment of section 111.70 (1) (a) (as it relates to health care) and (4) (n) and (o) of the statutes first applies to collective bargaining agreements entered into, extended, modified, or renewed, whichever occurs first, on the effective date of this subsection."

#### **2267.** Page 1656, line 15: after that line insert:

"(1q) The treatment of sections 5.37 (1), 5.64 (1) (ar) 1. a. and 1m. and (b), 5.91 (2), 7.50 (2) (a), and 10.02 (3) (b) 1. and (c) of the statutes first applies with respect to voting at the 2008 general election.".

#### **2268.** Page 1656, line 15: after that line insert:

"(2n) IDENTIFICATION REQUIREMENTS FOR VOTING. The treatment of sections 5.35 (6) (a) 4a., 6.29 (1), 6.33 (1), 6.345, 6.36 (2) (d), 6.79 (2) (a) and (d) and (7), 6.82 (1) (a), 6.86 (1) (ac) and (ar) and (3) (a) 1., 6.869, 6.87 (3) (d), 6.875 (6) (c) 1., 6.88 (3) (a), 6.965, 6.97 (title), (1), (2), and (4), 7.08 (1) (c) and (8) (title), 7.52 (3) (a), and 10.02 (3) (form) (a) of the statutes, the renumbering of section 6.79 (3) of the statutes, the renumbering and amendment of sections 6.87 (4) and 6.97 (3) of the statutes, the amendment of section 6.79 (3) (title) of the statutes, and the creation of sections 6.79 (3) (b), 6.87 (4) (b), and 6.97 (3) (a) and (c) of the statutes first apply with respect to voting at the 2009 spring primary election."

# **2269.** Page 1656, line 16: after that line insert:

"(1f) Indian gaming. The treatment of section 14.037 of the statutes first applies to concurrences issued by the governor on the effective date of this subsection.".

# **2270.** Page 1658, line 16: after that line insert:

 $\mathbf{2}$ 

"(10f) Patient Health Care Record Fees. The treatment of sections 146.83 (3g) and (3m) (a) (intro.) of the statutes, the renumbering and amendment of section 908.03 (6m) (d) of the statutes, and the creation of section 908.03 (6m) (d) 1. and 2. of the statutes first apply to requests for patient health care records and certified patient health care records made on the effective date of this subsection.".

#### **2271.** Page 1658, line 16: after that line insert:

"(10w) Immunity from Liability for, and disclosure of patient information by, Health care providers, facilities, and employees. The treatment of section 146.997 (5m) and (5r) of the statutes first applies to a health care provider, health care facility, or employee of a health care provider or health care facility that reports or provides information, or that initiates, participates in, or testifies in any action or proceeding, as described in section 146.997 (5m) (a) of the statutes, as created by this act, or that discloses patient information, as described in section 146.997 (5r) of the statutes, as created by this act, on the effective date of this subsection."

### **2272.** Page 1658, line 16: after that line insert:

"(9d) Care management organization provider payments. The treatment of section 46.284 (4) (m) of the statutes first applies to care management organizations that enter into or renew a contract with the department of health and family services to serve as a care management organization on the effective date of this subsection.".

- **2273.** Page 1658, line 18: delete lines 18 to 20.
- **2274.** Page 1658, line 21: delete lines 21 to 24.
- **2275.** Page 1658, line 24: after that line insert:

"(3x) MINNESOTA-WISCONSIN STUDENT RECIPROCITY AGREEMENT. The treatment of section 39.47 (1) and (3) of the statutes first applies to reimbursement owed under

- the Minnesota-Wisconsin student reciprocity agreement for the 2007-08 academic year.".
  - **2276.** Page 1659, line 3: after that line insert:
  - "(2g) COVERAGE OF STUDENT ON MEDICAL LEAVE. The treatment of sections 40.51 (8) and (8m), 66.0137 (4), 111.91 (2) (nm), 120.13 (2) (g), 185.981 (4t), 185.983 (1) (intro.), 609.76, and 632.895 (15) of the statutes first applies to all of the following:
  - (a) Except as provided in paragraphs (b) and (c), disability insurance policies that are issued or renewed, and self-insured governmental or school district health plans that are established, extended, modified, or renewed, on the effective date of this paragraph.
  - (b) Disability insurance policies covering employees who are affected by a collective bargaining agreement containing provisions inconsistent with this act that are issued or renewed on the earlier of the following:
    - 1. The day on which the collective bargaining agreement expires.
  - 2. The day on which the collective bargaining agreement is extended, modified, or renewed.
  - (c) Self-insured governmental or school district health plans covering employees who are affected by a collective bargaining agreement containing provisions inconsistent with this act that are established, extended, modified, or renewed on the earlier of the following:
    - 1. The day on which the collective bargaining agreement expires.
  - 2. The day on which the collective bargaining agreement is extended, modified, or renewed.".
    - **2277.** Page 1659, line 21: after that line insert:

- "(2t) Educational benefits. The treatment of section 21.49 (2m) of the statutes first applies to applications for tuition grants for an academic term that begins after the effective date of this subsection."
- 4 **2278.** Page 1659, line 23: delete lines 23 and 24.
- 5 **2279.** Page 1659, line 24: after that line insert:
- "(1g) Managed forest land payments. The treatment of sections 74.25 (1) (a)
  6. and 8. and 74.30 (1) (f) and (h) of the statutes, the repeal of section 77.89 (2) (b) of
  the statutes, and the renumbering and amendment of section 77.89 (2) (a) of the
  statutes first apply to payments by landowners after July 1, 2008.".
- 10 **2280.** Page 1660, line 1: delete lines 1 to 11.
- 11 **2281.** Page 1660, line 13: delete lines 13 and 14.
- 12 **2282.** Page 1660, line 17: after that line insert:
- "(2d) Revenue limit; health care benefits. The treatment of section 121.91 (2m) (f) of the statutes first applies to the calculation of a school district's revenue limit for the 2007-08 school year."
- 16 **2283.** Page 1660, line 18: delete lines 18 to 21.
- 17 **2284.** Page 1660, line 22: delete lines 22 to 24.
- 18 **2285.** Page 1661, line 1: delete lines 1 to 3.
- 19 **2286.** Page 1661, line 4: delete lines 4 to 7.
- 20 **2287.** Page 1661, line 7: after that line insert:
- "(9i) YOUTH OPTIONS PROGRAM. The treatment of section 118.55 (7t) (a) of the statutes first applies to the semester beginning after the effective date of this subsection."

**2288.** Page 1661, line 7: after that line insert:

2 "(10f) Transportation aid rates. The treatment of section 121.58 (2) (a) 1., 2.,

3., and 4. and (c) of the statutes first applies to transportation aid paid in the 2007-08

school year.".

1

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

**2289.** Page 1661, line 7: after that line insert:

"(9x) MILWAUKEE PARENTAL CHOICE PROGRAM ELIGIBILITY. The treatment of section 119.23 (2) (a) (intro.) and (7) (d) 1. of the statutes first applies to participation in the Milwaukee Parental Choice Program in the 2008–09 school year.".

**2290.** Page 1661, line 7: after that line insert:

"(9f) SCHOOL DISTRICT REFERENDA. The treatment of sections 67.05 (6a) (a) 2. a. and 121.91 (3) (a) of the statutes first applies to a resolution adopted under section 67.05 (6a) (a) of the statutes, or filed under 121.91 (3) (a) of the statutes, on the effective date of this subsection."

**2291.** Page 1661, line 7: after that line insert:

"(8k) RESIDENCY. The creation of section 119.16 (11) of the statutes first applies to teachers covered by a collective bargaining agreement that is in effect on the effective date of this subsection upon the expiration, extension, renewal, or modification of the agreement."

**2292.** Page 1661, line 21: delete lines 21 to 23.

**2293.** Page 1661, line 23: after that line insert:

"(3x) EDVEST; OUT OF STATE 529 PLANS. The treatment of section 71.05 (6) (b) 28. h., (with respect to the extent that it relates to a qualified tuition plan under 26 USC 529 (b) (1)), 31., and 32. (intro.) of the statutes first applies to taxable years beginning on January 1, 2008."

1	<b>2294.</b> Page 1661, line 23: after that line insert:
2	"(3c) Veterans and surviving spouses property tax credit. The treatment of
3	section 71.07 (6e) (a) 2. a., b., and c., 3. (intro.), b., and d., and 3e. of the statutes first
4	applies to taxable years beginning on January 1, 2009.".
5	<b>2295.</b> Page 1662, line 11: after that line insert:
6	"(6w) Homestead tax credit changes. The treatment of section 71.53 (2) (c) and
7	(d) of the statutes first applies to taxable years beginning on January 1, 2007.".
8	<b>2296.</b> Page 1662, line 11: after that line insert:
9	"(6g) Information technology bonds. The treatment of sections 71.05 (1) (c)
10	8., 71.26 (1m) (i), and 71.45 (1t) (i) of the statutes first applies to taxable years
11	beginning on January 1, 2008.".
12	<b>2297.</b> Page 1663, line 3: delete lines 3 to 7.
13	<b>2298.</b> Page 1663, line 7: after that line insert:
14	"(10i) Restaurant kitchen equipment. The treatment of section 70.11 (27m) of
15	the statutes first applies to the property tax assessments as of January 1, 2008.".
16	<b>2299.</b> Page 1663, line 7: after that line insert:
17	"(10t) Waste treatment facility. The renumbering and amendment of section
18	$70.11\ (21)\ (a)$ of the statutes, the amendment of sections $74.35\ (2m)$ and $74.35\ (5)\ (d)$
19	of the statutes, and the creation of 70.11 (21) (ab) of the statutes first apply
20	retroactively to the property tax assessments as of January 1, 2007.".
21	<b>2300.</b> Page 1663, line 8: delete lines 8 to 14.

10

11

12

- 2 2301. Page 1663, line 16: after "statutes" insert "(with respect to the extent that subdivision 28. h. refers to "mandatory student fees" and changes "claimant" to "owner of the account")".
- 4 **2302.** Page 1663, line 21: delete lines 21 to 23.
- 2303. Page 1663, line 24: delete the material beginning with that line and
  ending with page 1664, line 2.
- 7 **2304.** Page 1664, line 3: delete lines 3 to 5.
- 8 **2305.** Page 1664, line 8: after that line insert:
  - "(1w) State employee paid sick leave. The treatment of section 111.91 (2) (pj) of the statutes, the renumbering and amendment of section 230.35 (2) of the statutes, and the creation of section 230.35 (2) (am) of the statutes first apply to employees who are covered by collective bargaining agreements on the day on which the agreements are entered into, extended, modified, or renewed, whichever occurs first."
- **2306.** Page 1664, line 12: delete "and 3." and substitute ", 2., 2m., and 3.".
- 15 **2307.** Page 1664, line 15: delete lines 15 to 17.
- 16 **2308.** Page 1664, line 20: after that line insert:
- "(4u) FEE FOR ENGLISH AS A 2ND LANGUAGE COURSES. The treatment of section 38.24 (1m) (b) and (e) of the statutes first applies to students enrolled in the 2008–09 school year.".
- 20 **2309.** Page 1665, line 3: delete lines 3 to 5.
- 21 **2310.** Page 1665, line 17: after that line insert:

24

whichever occurs first.".

1	"(5x) Vehicle title fee increase. The treatment of section 342.14 (1) and (3) $$
2	of the statutes first applies to motor vehicle title applications submitted to the
3	department of transportation on the effective date of this subsection.".
4	2311. Page 1665, line 18: delete the material beginning with that line and
5	ending with page 1666, line 8.
6	2312. Page 1666, line 8: after that line insert:
7	"(8c) Liability for disobeying school crossing guards. The treatment of
8	section 346.49 (1) (a), (am), and (c) of the statutes first applies to violations of section
9	346.46 (2m) of the statutes occurring on the effective date of this subsection.".
10	2313. Page 1666, line 8: after that line insert:
11	"(11f) Value engineering. The treatment of sections $84.013$ (4) (a) and $84.06$
12	(1m) and (1r) of the statutes first applies to highway improvement projects for which
13	engineering work is commenced on the effective date of this subsection.".
14	<b>2314.</b> Page 1666, line 8: after that line insert:
15	"(7j) Organ transport vehicles. The treatment of sections $110.08\ (1m),340.01$
16	$(3) (dg) \ and \ (dh), \ 343.01 \ (2) \ (dg), \ 346.03 \ (1) \ and \ (5m), \ 347.25 \ (1), \ and \ 347.38 \ (4) \ of \ the$
17	statutes first applies to vehicles operated on the effective date of this subsection.".
18	2315. Page 1666, line 8: after that line insert:
19	"(7d) Prevailing wage on highway projects. The treatment of section $103.50$
20	(2m) (b) 1. of the statutes first applies to a laborer, worker, mechanic, or truck driver
21	who is affected by a collective bargaining agreement that contains provisions that
22	are inconsistent with section $103.50 \ (2m) \ (b) \ 1.$ of the statutes, as affected by this act,
23	on the day on which the agreement expires or is extended, modified, or renewed,

9

10

11

12

13

14

15

16

17

18

19

- **2316.** Page 1666, line 20: delete "and 3." and substitute ", 2., 2m., and 3.".
- 2 **2317.** Page 1666, line 20: delete "(b) and (bm) of the statutes first applies" and substitute "(bg) and (bm) of the statutes, the renumbering and amendment of section 36.27 (3p) (b) of the statutes, and the creation of section 36.27 (3p) (b) 2. of the statutes first apply".
- 6 **2318.** Page 1666, line 22: delete lines 22 to 24.
- 7 **2319.** Page 1667, line 1: delete lines 1 and 2.
  - **2320.** Page 1667, line 8: after that line insert:
    - "(3d) Prevailing wage on state and municipal projects. The treatment of sections 66.0903 (4) (b) 1. and 103.49 (2m) (b) 1. of the statutes first applies to a laborer, worker, mechanic, or truck driver who is affected by a collective bargaining agreement that contains provisions that are inconsistent with section 66.0903 (4) (b) 1. or 103.49 (2m) (b) 1. of the statutes, as affected by this act, on the day on which the agreement expires or is extended, modified, or renewed, whichever occurs first."
      - **2321.** Page 1667, line 10: delete lines 10 to 12.
      - **2322.** Page 1667, line 12: after that line insert:
      - "(3d) INCOME ASSIGNMENT. The treatment of sections 778.30 (1) (intro.) and 800.09 (1) (intro.) and (d) of the statutes first applies to judgments entered on the effective date of this subsection.".
        - **2323.** Page 1667, line 12: after that line insert:
- 21 "(3q) Brownfields revolving loans; local governments. The treatment of section 67.12 (12) (a) of the statutes, the renumbering and amendment of section

- 67.12 (1) (b) of the statutes, and the creation of section 67.12 (1) (b) 2. of the statutes, 1  $\mathbf{2}$ first apply to municipal debt that is issued on the effective date of this subsection.". 3 **2324.** Page 1667, line 12: after that line insert: 4 (2v)DOCUMENTARY PROOF OF CITIZENSHIP OR IMMIGRATION STATUS. The 5 renumbering and amendment of section 49.84 (5) of the statutes and the creation of 6 section 49.84 (5) (a) and (c) of the statutes first apply to applications for assistance 7 programs under chapter 49 of the statutes that are submitted on the effective date of this subsection.". 8 9 **2325.** Page 1667, line 12: after that line insert: 10 "(3i) DIRECT LEGISLATION. The treatment of section 9.20 (4) and (9) of the 11 statutes first applies to a petition that is filed on the effective date of this 12 subsection.". 13 **2326.** Page 1667, line 12: after that line insert: 14 "(3f) MONTHLY PAY PERIOD. The treatment of sections 16.415 (4), 16.53 (1) (d) 1., 15 36.11 (48m), and 111.91 (2) (ph) of the statutes first applies to employees who are 16 covered by collective bargaining agreements on the day on which the agreements are 17 entered into, extended, modified, or renewed, whichever occurs first.". **2327.** Page 1667, line 16: after that line insert: 18 19 "(1i) Madison Child Care facilities and Services. The treatment of sections 20 16.841, 20.505 (5) (ka) (by Section 534s), and 20.865 (2) (am), (gm), and (qm) of the 21statutes takes effect on July 1, 2010.". **2328.** Page 1667, line 16: after that line insert:
- 22

24

"(1k) Repeal of Sentencing Commission. The treatment of section 973.30 of the statutes takes effect retroactively on July 1, 2007.".

- 1 **2329.** Page 1667, line 23: delete lines 23 and 24.
- 2 **2330.** Page 1668, line 4: delete lines 4 and 5.
- 3 **2331.** Page 1668, line 5: after that line insert:
- "(3d) Collection of fines and forfeitures. The treatment of section 59.25 (3)
   (j) of the statutes and Section 9307 (3d) of this act take effect on the first day of the
- 6 4th month beginning after publication.".

14

15

16

17

18

19

20

- 7 **2332.** Page 1668, line 5: after that line insert:
- 8 "(4c) Remedies against manufacturers. The treatment of section 895.046 of the 9 statutes and Section 9307 (4c) of this act take effect on first day of the 2nd month 10 beginning after publication.".
- 11 **2333.** Page 1668, line 9: delete lines 9 to 11.
- 12 **2334.** Page 1668, line 11: after that line insert:
  - "(4f) Private sewage system replacement or rehabilitation program. The treatment of sections 20.143 (3) (de), 20.320 (3), 25.43 (3), 67.12 (12) (a), 145.20 (5) (a), 145.245, and 281.59 (1m) (c) of the statutes, the renumbering and amendment of section 145.01 (4m) of the statutes, and the creation of section 145.01 (4m) (a) to (e) of the statutes take effect on July 1, 2008."
    - **2335.** Page 1668, line 11: after that line insert:
  - "(3i) Fire safety requirements; community-based residential facilities. The treatment of section 101.14 (4m) (a) 1m. and (gm) of the statutes takes effect on the first day of the 7th month beginning after publication.
- 22 (3j) Fire safety requirements. The treatment of sections 50.035 (2) (d), 66.1019 (3) (a) and (b), 101.02 (7m), 101.14 (4m) (a) 4., (am), (b) (intro.), 1., 2., and 3., (br), (bv),

8

9

10

11

12

16

17

18

19

20

21

22

- 1 (c), (d), (e), (g), and (h) of the statutes takes effect on the first day of the 25th month 2 beginning after publication.".
  - **2336.** Page 1668, line 15: after that line insert:
- 4 "(1f) LICENSE PLATES FOR SEX OFFENDERS. The treatment of section 301.49 of the 5 statutes takes effect on the first day of the 6th month beginning after publication.".
- 6 **2337.** Page 1669, line 2: delete lines 2 and 3.
- 7 **2338.** Page 1669, line 6: after that line insert:
  - "(3v) PAYMENT OF REQUIRED EMPLOYEE CONTRIBUTIONS UNDER THE WISCONSIN RETIREMENT SYSTEM. The treatment of section 40.03 (1) (e), the renumbering and amendment of section 40.05 (1) (b) of the statutes, and the creation of section 40.05 (1) (b) 1. and 3. of the statutes take effect on September 1, 2007."
    - **2339.** Page 1669, line 14: after that line insert:
- "(2p) Ballot printing in first class cities. The treatment of sections 5.66 (1) and (1m) and 7.15 (2) (b) and (c) of the statutes takes effect on the first day of the 2nd month beginning after the effective date of this subsection."
  - **2340.** Page 1669, line 14: after that line insert:
  - "(2n) IDENTIFICATION REQUIREMENTS FOR VOTING. The treatment of sections 5.35 (6) (a) 4a., 6.29 (1), 6.33 (1), 6.345, 6.36 (2) (d), 6.79 (2) (a) and (d) and (7), 6.82 (1) (a), 6.86 (1) (ac) and (ar) and (3) (a) 1., 6.869, 6.87 (3) (d), 6.875 (6) (c) 1., 6.88 (3) (a), 6.965, 6.97 (title), (1), (2), and (4), 7.08 (1) (c) and (8) (title), 7.52 (3) (a), and 10.02 (3) (form) (a) of the statutes, the renumbering of section 6.79 (3) of the statutes, the renumbering and amendment of sections 6.87 (4) and 6.97 (3) of the statutes, the amendment of section 6.79 (3) (title) of the statutes, and the creation of sections 6.79

- 1 (3) (b), 6.87 (4) (b), and 6.97 (3) (a) and (c) of the statutes, take effect on January 1, 2009.".
- 3 **2341.** Page 1669, line 14: after that line insert:
- "(2u) Use of deductions for payments to labor organizations. The treatment of section 11.387 of the statutes takes effect on the first day of the 2nd month beginning after publication."
- 7 **2342.** Page 1669, line 20: delete "2008," and substitute "2009,".
- 8 **2343.** Page 1670, line 10: delete the material beginning with that line and ending with page 1671, line 2.
- 10 **2344.** Page 1671, line 3: delete lines 3 to 6.
- 11 **2345.** Page 1671, line 7: delete lines 7 and 8.
- 12 **2346.** Page 1671, line 9: delete lines 9 and 10.
- 13 **2347.** Page 1671, line 11: delete lines 11 and 12.
- 14 **2348.** Page 1671, line 13: delete lines 13 to 15.
- 15 **2349.** Page 1671, line 15: after that line insert:
- "(9w) VITAL RECORDS FEES. The treatment of sections 69.22 (1) (a) (by SECTION 1918h), 69.22 (1) (b) (by SECTION 1918j), 69.22 (1) (c) (by SECTION 1918L), 69.22 (1) (d)
  (by SECTION 1918n), and 69.22 (1m) (by SECTION 1918q) of the statutes and the repeal of section 69.22 (1p) of the statutes take effect on July 1, 2010."
- 20 **2350.** Page 1671, line 15: after that line insert:
- $^{\circ}$  (9i) Family care level of care need. The treatment of section 46.286 (1) (a)
- 22 1. of the statutes takes effect on January 1, 2008.".

- 2351. Page 1671, line 22: delete lines 22 to 25 and substitute "234.165 (3) of
   the statutes take effect on June 30, 2009.".
- 3 **2352.** Page 1672, line 1: after that line insert:
- "(2f) COVERAGE OF STUDENT ON MEDICAL LEAVE. The treatment of sections 40.51

  (8) and (8m), 66.0137 (4), 111.91 (2) (nm), 120.13 (2) (g), 185.981 (4t), 185.983 (1)

  (intro.), 609.76, and 632.895 (15) of the statutes and Section 9325 (2f) of this act take

  effect on the first day of the 7th month beginning after publication."
- 8 **2353.** Page 1673, line 3: substitute "(az)" for "(ad)".
- 9 **2354.** Page 1673, line 5: after that line insert:
- "(2f) Managed forest land payments. The treatment of sections 74.25 (1) (a)
  6. and 8. and 74.30 (1) (f) and (h) of the statutes, the repeal of section 77.89 (2) (b) of
  the statutes, and the renumbering and amendment of section 77.89 (2) (a) of the
  statutes take effect on July 1, 2008."
- **2355.** Page 1673, line 7: delete lines 7 to 15.
- 15 **2356.** Page 1673, line 17: delete lines 17 and 18.
- 16 **2357.** Page 1673, line 18: after that line insert:
- "(2k) Residency. The creation of section 119.16 (11) of the statutes takes effect on July 1, 2009.".
- 19 **2358.** Page 1674, line 9: after that line insert:
- 20 "(2d) Products powered by alternate resources. The treatment of section 77.54 (56) of the statutes takes effect on July 1, 2008.".
- 22 **2359.** Page 1674, line 10: delete lines 10 to 15.
- 23 **2360.** Page 1674, line 15: after that line insert:

1	"(3t) Waste treatment facility; sales tax exemption. The treatment of section
2	77.54~(26) (by Section 2386b) of the statutes takes effect on the first day of the 2nd
3	month beginning after publication.".
4	<b>2361.</b> Page 1674, line 15: after that line insert:
5	"(3j) Delinquent taxpayer Internet posting. The treatment of section 73.03
6	(62) of the statutes takes effect on the first day of the 3rd month beginning after
7	publication.".
8	<b>2362.</b> Page 1674, line 15: after that line insert:
9	"(3u) Temporary help agency. The treatment of section 77.52 (2r) of the
10	statutes takes effect on July 1, 2009.".
11	<b>2363.</b> Page 1674, line 15: after that line insert:
12	"(3v) Coins, currency, commemorative medallions. The treatment of section
13	77.54 (55) of the statutes takes effect on July 1, 2008.".
14	<b>2364.</b> Page 1674, line 15: after that line insert:
15	"(3f) Cemetery sales and use tax exemption. The treatment of section 77.54
16	(9a) (i) of the statutes takes effect on July 1, 2008.".
17	2365. Page 1674, line 16: delete the material beginning with that line and
18	ending with page 1676, line 22.
19	<b>2366.</b> Page 1676, line 22: after that line insert:
20	"(4f) Biomass used for fuel. The treatment of section $77.54\ (30)\ (a)\ 1m.$ of the
21	statutes takes effect on the first day of the 2nd month beginning after publication.".

**2367.** Page 1676, line 25: after that line insert:

- 1 "(5f) CLAY PIGEONS. The treatment of section 77.54 (47) (b) 2. of the statutes (by
- 2 Section 2410d) takes effect on July 1, 2007.".
- 3 **2368.** Page 1677, line 1: delete lines 1 to 4.
- 4 **2369.** Page 1677, line 4: after that line insert:
- 5 "(6i) Sales tax revenue; transportation fund. The treatment of sections
- 6 20.855 (4) (fd) of the statutes takes effect on July 1, 2009.".
- 7 **2370.** Page 1677, line 5: delete lines 5 to 9.
- 8 **2371.** Page 1677, line 10: delete lines 10 and 11.
- 9 **2372.** Page 1677, line 11: after that line insert:
- 10 "(9t) Waste treatment facility. The renumbering and amendment of section
- 11 70.11 (21) (a) of the statutes, the amendment of sections 74.35 (2m) and 74.35 (5) (d)
- of the statutes, and the creation of 70.11 (21) (ab) of the statutes take effect
- retroactively on January 1, 2007.".
- **2373.** Page 1677, line 12: after "(1j)" insert ", 77.52 (2) (a) 11.,".
- 15 **2374.** Page 1677, line 13: delete the material beginning with "and the" and
- ending with "take" on line 14 and substitute "takes".
- 17 **2375.** Page 1677, line 15: delete lines 15 and 16.
- 18 **2376.** Page 1677, line 17: delete lines 17 and 18.
- 19 **2377.** Page 1678, line 4: after "(f)," insert "(2j),".
- 20 **2378.** Page 1678, line 13: after "(a)" insert "and (b)".
- 21 **2379.** Page 1678, line 23: delete lines 23 to 25.
- 22 **2380.** Page 1679, line 8: after "(c)" insert ", (cm), (d), (e), (f), (g), (h), (i), (j),
- 23 (k), (km), (L), (m), (n), (o), (p), and (q)".

- 1 **2381.** Page 1679, line 8: delete "and".
- 2 **2382.** Page 1679, line 9: after that line insert:
- "(6f) LICENSE PLATES FOR SEX OFFENDERS. The treatment of sections 25.40 (1) (a)
  3., 84.59 (2) (b) (as it relates to license plates for sex offenders), 341.01 (2) (c), 341.08
  (1m), 341.14 (6y), 341.145 (1) (g) and (1g) (f), and 341.16 (1) (c) and (4) of the statutes
  and Section 9148 (3f) of this act take effect on the first day of the 6th month
- 7 beginning after publication.".
- 8 **2383.** Page 1679, line 13: delete "statute" and substitute "statutes".
- 9 **2384.** Page 1679, line 13: after that line insert:
- "(7x) Vehicle title fee increase. The treatment of section 342.14 (1) and (3) of the statutes and Section 9348 (5x) of this act take effect on October 1, 2007, or on the first day of the 3rd month beginning after publication, whichever is later."
- 13 **2385.** Page 1679, line 14: delete lines 14 to 16.
- **2386.** Page 1679, line 17: delete lines 17 and 18.
- 15 **2387.** Page 1679, line 20: after that line insert:
- "(11f) Value engineering. The treatment of sections 84.013 (4) (a) and 84.06 (1m) and (1r) of the statutes and Section 9348 (11f) of this act take effect on the first day of the 3rd month beginning after publication.".
- 19 **2388.** Page 1679, line 20: after that line insert:
- "(11i) GOLD STAR LICENSE PLATES. The treatment of section 341.14 (6r) (f) 19m.
  of the statutes takes effect on the first day of the 2nd month beginning after publication.".
- 23 **2389.** Page 1680, line 5: delete lines 5 and 6.

- 1 **2390.** Page 1680, line 7: after that line insert:
- 2 "(1t) Council on Veterans programs. The treatment of section 15.497 (2) (a)
- 3 of the statutes and the creation of section 15.497 (2) (e) and (f) of the statutes take
- 4 effect on first day of the 2nd month beginning after publication.".
- 5 **2391.** Page 1680, line 12: delete lines 12 and 13.
- 6 **2392.** Page 1680, line 23: after that line insert:
- 7 "(4h) Child care subsidies.
- 8 (a) Income eligibility. The treatment of section 49.155 (1m) (c) 1. (intro.), 1c.,
- 9 1m., 2., and 3. of the statutes takes effect on October 1, 2007, or on the day after
- 10 publication, whichever is later.
- 11 (b) Waiting list. The treatment of section 49.155 (1m) (intro.), (2), and (3) (a)
- of the statutes takes effect on October 1, 2007, or on the day after publication,
- whichever is later.".
- **2393.** Page 1680, line 24: delete lines 24 and 25.
- 15 **2394.** Page 1681, line 2: delete lines 2 to 25.
- 16 **2395.** Page 1682, line 1: delete lines 1 to 25.
- **2396.** Page 1683, line 1: delete lines 1 to 25.
- 18 **2397.** Page 1684, line 1: delete lines 1 to 25.
- 19 **2398.** Page 1685, line 1: delete lines 1 to 25.
- 20 **2399.** Page 1686, line 1: delete lines 1 to 25.
- 21 **2400.** Page 1687, line 1: delete lines 1 to 25.
- 22 **2401.** Page 1688, line 1: delete lines 1 to 13.
- 23 **2402.** Page 1688, line 13: after that line insert:

2

3

4

5

7

8

9

	"(2y)	DOCUMENTARY	PROOF	OF	CITIZENSHIP	OR	IMMIGRATION	STATUS.	The
renu	mberin	g and amendm	ent of s	secti	ion 49.84 (5)	of t	the statutes,	the creat	ion of
section 49.84 $(5)$ $(a)$ and $(c)$ of the statutes, and Section 9355 $(2y)$ of this act take effect									
on th	e first o	day of the 7th n	nonth b	egir	nning after p	ubli	cation.".		

- **2403.** Page 1688, line 14: delete lines 14 and 15.
- 6 **2404.** Page 1688, line 15: after that line insert:

"(5d) INCOME ASSIGNMENT. The treatment of sections 778.30 (1) (intro.) and 800.09 (1) (intro.) and (d) of the statutes and Section 9355 (3d) of this act take effect on the first day of the 4th month beginning after publication.".

10 (END)